TKK Dissertations 109 Espoo 2008

CMOS RADIO FREQUENCY CIRCUITS FOR SHORT-RANGE DIRECT-CONVERSION RECEIVERS

Doctoral Dissertation

Jouni Kaukovuori



Helsinki University of Technology Faculty of Electronics, Communications and Automation Department of Micro and Nanosciences TKK Dissertations 109 Espoo 2008

CMOS RADIO FREQUENCY CIRCUITS FOR SHORT-RANGE DIRECT-CONVERSION RECEIVERS

Doctoral Dissertation

Jouni Kaukovuori

Dissertation for the degree of Doctor of Science in Technology to be presented with due permission of the Faculty of Electronics, Communications and Automation for public examination and debate in Auditorium S4 at Helsinki University of Technology (Espoo, Finland) on the 9th of May, 2008, at 12 noon.

Helsinki University of Technology Faculty of Electronics, Communications and Automation Department of Micro and Nanosciences

Teknillinen korkeakoulu Elektroniikan, tietoliikenteen ja automaation tiedekunta Mikro- ja nanotekniikan laitos Distribution: Helsinki University of Technology Faculty of Electronics, Communications and Automation Department of Micro and Nanosciences P.O. Box 3000 FI - 02015 TKK FINLAND Tel. +358-9-451 2271 Fax +358-9-451 2269 E-mail: jouni.kaukovuori@st.com

© 2008 Jouni Kaukovuori

ISBN 978-951-22-9264-6 ISBN 978-951-22-9265-3 (PDF) ISSN 1795-2239 ISSN 1795-4584 (PDF) URL: http://lib.tkk.fi/Diss/2008/isbn9789512292653/

TKK-DISS-2442

Multiprint Oy Espoo 2008

Abstract

The research described in this thesis is focused on the design and implementation of radio frequency (RF) circuits for direct-conversion receivers. The main interest is in RF front-end circuits, which contain low-noise amplifiers, downconversion mixers, and quadrature local oscillator signal generation circuits. Three RF front-end circuits were fabricated in a short-channel CMOS process and experimental results are presented.

A low-noise amplifier (LNA) is typically the first amplifying block in the receiver. A large number of LNAs have been reported in the literature. In this thesis, wideband LNA structures are of particular interest. The most common and relevant LNA topologies are analyzed in detail in the frequency domain and theoretical limitations are found. New LNA structures are presented and a comparison to the ones found in the literature is made. In this work, LNAs are implemented with downconversion mixers as RF front-ends. The designed mixers are based on the commonly used Gilbert cell. Different mixer implementation alternatives are presented and the design of the interface between the LNA and the downconversion mixer is discussed.

In this work, the quadrature local oscillator signal is generated either by using frequency dividers or polyphase filters (PPF). Different possibilities for implementing frequency dividers are briefly described. Polyphase filters were already introduced by the 1970s and integrated circuit (IC) realizations to generate quadrature signals have been published since the mid-1990s. Although several publications where the performance of the PPFs has been studied either by theoretical calculations or simulations can be found in the literature, none of them covers all the relevant design parameters. In this thesis, the theory behind the PPFs is developed such that all the relevant design parameters needed in the practical circuit design have been calculated and presented with closed-form equations whenever possible. Although the main focus was on two-and three-stage PPFs, which are the most common ones encountered in practical ICs, the presented calculation methods can be extended to analyze the performance of multistage PPFs as well.

The main application targets of the circuits presented in this thesis are the short-range wireless sensor system and ultrawideband (UWB). Sensors are capable of monitoring temperature, pressure, humidity, or acceleration, for example. The amount of transferred data is typically small and therefore a modest bit rate, less than 1 Mbps, is adequate. The sensor system applied in this thesis operates at 2.4-GHz ISM band (Industrial, Scientific, and Medical). Since the sensors must be able to operate independently for several years, extremely low power consumption is required. In sensor radios, the receiver current consumption is dominated by the blocks and elements operating at the RF. Therefore, the target was to develop circuits that can offer satisfactory performance with a current consumption level that is small compared to other receivers targeted for common cellular systems.

On the other hand, there is a growing need for applications that can offer an extremely high data rate. UWB is one example of such a system. At the moment, it can offer data rates of up to 480 Mbps. There is a frequency spectrum allocated for UWB systems between 3.1 and 10.6 GHz. The UWB band is further divided into several narrower band groups (BG), each occupying a bandwidth of approximately 1.6 GHz. In this work, a direct-conversion RF frontend is designed for a dual-band UWB receiver, which operates in band groups BG1 and BG3, i.e. at 3.1 - 4.8 GHz and 6.3 - 7.9 GHz frequency areas, respectively. Clearly, an extremely wide bandwidth combined with a high operational frequency poses challenges for circuit design. The operational bandwidths and the interfaces between the circuit blocks need to be optimized to cover the wanted frequency areas. In addition, the wideband functionality should be achieved without using a number of on-chip inductors in order to minimize the die area, and yet the power consumption should be kept as small as possible.

The characteristics of the two main target applications are quite different from each other with regard to power consumption, bandwidth, and operational frequency requirements. A common factor for both is their short, i.e. less than 10 meters, range. Although the circuits presented in this thesis are targeted on the two main applications mentioned above, they can be utilized in other kind of wireless communication systems as well. The performance of three experimental circuits was verified with measurements and the results are presented in this work. Two of them have been a part of a whole receiver including baseband amplifiers and filters and analog-to-digital converters. Experimental circuits were fabricated in a 0.13-µm CMOS process. In addition, this thesis includes design examples where new circuit ideas and implementation possibilities are introduced by using 0.13-µm and 65-nm CMOS processes. Furthermore, part of the theory presented in this thesis is validated with design examples in which actual IC component models are used.

Tiivistelmä

Tässä väitöskirjassa esitetty tutkimus keskittyy suoramuunnosvastaanottimen radiotaajuudella (radio frequency, RF) toimivien piirien suunnitteluun ja toteuttamiseen. Työ keskittyy vähäkohinaiseen vahvistimeen (low-noise amplifier, LNA), alassekoittajaan ja kvadratuurisen paikallisoskillaattorisignaalin tuottavaan piiriin. Työssä toteutettiin kolme RF-etupäätä erittäin kapean viivanleveyden CMOS-prosessilla, ja niiden kokeelliset tulokset esitetään.

Vähäkohinainen vahvistin on yleensä ensimmäinen vahvistava lohko vastaanottimessa. Useita erilaisia vähäkohinaisia vahvistimia on esitetty kirjallisuudessa. Tämän työn kohteena ovat eritoten laajakaistaiset LNA-rakenteet. Tässä työssä analysoidaan taajuustasossa yleisimmät ja oleellisimmat LNA-topologiat. Lisäksi uusia LNA-rakenteita on esitetty tässä työssä ja niitä on verrattu muihin kirjallisuudessa esitettyihin piireihin. Tässä työssä LNA:t on toteutettu yhdessä alassekoittimen kanssa muodostaen RF-etupään. Työssä suunnitellut alassekoittimet perustuvat yleisesti käytettyyn Gilbertin soluun. Erilaisia sekoittajan suunnitteluvaihtoehtoja ja LNA:n ja alassekoittimen välisen rajapinnan toteutustapoja on esitetty.

Tässä työssä kvadratuurinen paikallisoskillaattorisignaali on muodostettu joko käyttämällä taajuusjakajia tai monivaihesuodattimia. Erilaisia taajuusjakajia ja niiden toteutustapoja käsitellään yleisellä tasolla. Monivaihesuodatinta, joka on alunperin kehitetty jo 1970-luvulla, on käytetty integroiduissa piireissä kvadratuurisignaalin tuottamiseen 1990-luvun puolivälistä lähtien. Kirjallisuudesta löytyy lukuisia artikkeleita, joissa monivaihesuodattimen toimintaa on käsitelty teoreettisesti laskien ja simuloinnein. Kuitenkaan kaikkia sen suunnitteluparametreja ei tähän mennessä ole käsitelty. Tässä työssä monivaihesuodattimen teoriaa on kehitetty edelleen siten, että käytännön piirisuunnittelussa tarvittavat oleelliset parametrit on analysoitu ja suunnitteluyhtälöt on esitetty suljetussa muodossa aina kuin mahdollista. Vaikka työssä on keskitytty yleisimpiin eli kaksi- ja kolmiasteisiin monivaihesuodattimiin, on työssä esitetty menetelmät, joilla laskentaa voidaan jatkaa aina useampiasteisiin suodattimiin asti.

Työssä esiteltyjen piirien pääkohteina ovat lyhyen kantaman sensoriradio ja erittäin laajakaistainen järjestelmä (ultrawideband, UWB). Sensoreilla voidaan tarkkailla esimerkiksi ympäristön lämpötilaa, kosteutta, painetta tai kiihtyvyyttä. Siirrettävän tiedon määrä on tyypillisesti vähäistä, jolloin pieni tiedonsiirtonopeus, alle 1 megabitti sekunnissa, on välttävä. Tämän työn kohteena oleva sensoriradiojärjestelmä toimii kapealla kaistalla 2,4 gigahertsin ISM-taajuusalueella (Industrial, Scientific, and Medical). Koska sensorien tavoitteena on toimia itsenäisesti ilman pariston vaihtoa useita vuosia, täytyy niiden kuluttaman virran olla erittäin vähäistä. Sensoriradiossa vastaanottimen tehonkulutuksen kannalta määräävässä asemassa ovat radiotaajuudella toimivat piirit. Tavoitteena oli tutkia ja kehittää piirirakenteita, joilla päästään tyydyttävään suorituskykyyn tehonkulutuksella, joka on vähäinen verrattuna muiden tavallisten langattomien tiedonsiirtojärjestelmien radiovastaanottimiin.

Toisaalta viime aikoina on kasvanut tarvetta myös järjestelmille, jotka kykenevät tarjoamaan erittäin korkean tiedonsiirtonopeuden. UWB on esimerkki tällaisesta järjestelmästä. Tällä hetkellä se tarjoaa tiedonsiirtonopeuksia aina 480 megabittiin sekunnissa. UWB:lle on varattu taajuusalueita 3,1 ja 10,6 gigahertsin taajuuksien välillä. Kyseinen kaista on edelleen jaettu pienempiin taajuusryhmiin (band group, BG), joiden kaistanleveys on noin 1,6 gigahertsiä. Tässä työssä on toteutettu RF-etupää radiovastaanottimeen, joka pystyy toimimaan BG1:llä ja BG3:lla eli taajuusalueilla 3,1 – 4,7 GHz ja 6,3 – 7,9 GHz. Erittäin suuri kaistanleveys

yhdistettynä korkeaan toimintataajuuteen tekee radiotaajuuspiirien suunnittelusta haasteellista. Piirirakenteiden toimintakaistat ja piirien väliset rajapinnat tulee optimoida riittävän laajoiksi käyttämättä kuitenkaan liian montaa piille integroitua kelaa piirin pinta-alan minimoimiseksi, ja lisäksi piirit tulisi toteuttaa mahdollisimman alhaisella tehonkulutuksella.

Työssä esiteltyjen piirien kaksi pääkohdetta ovat hyvin erityyppisiä, mitä tulee tehonkulutus-, kaistanleveys- ja toimintataajuusvaatimuksiin. Yhteistä molemmille on lyhyt, alle 10 metrin kantama. Vaikka tässä työssä esitellyt piirit onkin kohdennettu kahteen pääsovelluskohteeseen, voidaan esitettyjä piirejä käyttää myös muiden tiedonsiirtojärjestelmien piirien suunnitteluun. Tässä työssä esitetään mittaustuloksineen yhteensä kolme kokeellista piiriä yllämainittuihin järjestelmiin. Kaksi ensimmäistä kokeellista piiriä muodostaa kokonaisen radiovastaanottimen yhdessä analogisten kantataajuusosien ja analogia-digitaali-muuntimien kanssa. Esitetyt kokeelliset piirit on toteutettu käyttäen 0,13 μ m:n viivanleveyden CMOS-tekniikkaa. Näiden lisäksi työ pitää sisällään piirisuunnitteluesimerkkejä, joissa esitetään ideoita ja mahdollisuuksia käyttäen 0,13 μ m:n ja 65 nm:n viivanleveyden omaavia CMOS-tekniikoita. Lisäksi piirisuunnitteluesimerkein havainnollistetaan työssä esitetyn teorian paikkansapitävyyttä käyttämällä oikeita komponenttimalleja.

Preface

Experience is that marvelous thing that enables you to recognize a mistake when you make it again.

Franklin P. Jones

The work for this thesis was carried out in the Electronic Circuit Design Laboratory of Helsinki University of Technology (SMARAD2) between 2002 and 2007. The work presented in thesis is part of research projects funded in part by Nokia, Nokia Research Center, and the Finnish Funding Agency for Technology and Innovation (TEKES). During the years 2003 – 2007, I was privileged to be a postgraduate student of the Graduate School in Electronics, Telecommunications and Automation (GETA), which partially funded my studies. In addition, this thesis has been supported by the following foundations, which I would like to acknowledge: the Foundation of Technology (TES), Emil Aaltonen Foundation, HPY Research Foundation, Nokia Foundation, Walter Ahlström Foundation, the Finnish Society of Electronics Engineers (EIS), and Ulla Tuominen Foundation.

I would like to thank my supervisor, Prof. Jussi Ryynänen, for his competent and inspiring guidance and discussions concerning technical and non-technical matters. I would also like to thank Prof. Kari Halonen for the opportunity to work in his laboratory on interesting research topics and for the opportunity to write this thesis. In addition, I wish to thank D.Sc. Jarkko Jussila for his expert advice and valuable help with numerous problems. I also warmly thank Prof. Pietro Andreani and Dr. Domine Leenaerts for reviewing my thesis and for their valuable comments and suggestions.

My special thanks goes to Jere Järvinen and Mikko Hotti, with whom I've worked on the same projects and shared the same humorous (?) and relaxed working room for several years. Mikko Kaltiokallio is highly acknowledged for reading and commenting on parts of this thesis, keeping my ECCTD'07 presentation, and for other helpful assistance. Kari Stadius deserves huge thanks for his advices on one of the most tedious publications I've ever been involved with. The good company of other persons working in the DUST project, Prof. Saska Lindfors, Ville Saari, Tapio Rapinoja, and Olli Viitala, is also acknowledged. Tero Tikka deserves thanks for reading and commenting on parts of this thesis. I also thank all the staff at the whole laboratory for creating comfortable working atmosphere.

I'm thankful to my Mom for all her support and for financial help during the early stages of my studies. My parents-in-law deserve a mention for pleasant free-time activities, good food, and for child-care assistance.

Last, but certainly not least, I would like to express my warmest gratitude to my lovely wife Annukka, son Lauri, and daughter Veera. You are, to me, the dearest things in life.

Jouni Kaukovuori Vantaa, February 2008

Contents

A	bstract		i	
Т	iivistelmä		iii	
P	Preface v			
С	ontents		vi	
c				
S	ymbols		XII	
A	bbreviatio	ns	xix	
1	Intr	oduction	1	
	1.1 Rese	arch contribution and publications	1	
	1.2 Orga	nization of the Thesis	4	
2	Ove	rview of RF IC design for wireless radio systems	5	
	2.1 Rece	iver architectures	5	
	2.1.1	Direct-conversion architecture	5	
	2.1.2	Superheterodyne receiver	6	
	2.1.3	Low-IF architecture	7	
	2.1.4	Wideband-IF	8	
	2.1.5	RF front-end in different receiver topologies	8	
	2.2 Defin	nitions and general receiver design parameters	9	
	2.2.1	Input matching and stability	9	
	2.2.2	Voltage gain and insertion gain	9	
	2.2.3	Linearity	10	
	2.2.3	.1 Compression and desensitization	11	
	2.2.3	2 Intermodulation	12	
	2.2.4	Sensitivity and noise	13	
	2.2.4	1 Noise figure	13	
	2.2.4	2 Noise factor definitions for mixers	15	
	2.3 MOS	S transistor small-signal model and noise sources	15	
	2.4 Over	view of short-range radio systems	17	
	2.4.1	Wireless sensor systems	17	
	2.4.1	1 General requirements	18	

	2.4.1	.2 Overview of low-end extension for Bluetooth (BT LEE)	
	2.4.2	Ultrawideband	19
	2.4.2	.1 Target applications	
	2.4.2	.2 Multisystem challenges	
3	Low	z-noise amplifier design	26
	3.1 Gene	eral LNA design aspects	
	3.2 Narr	owband LNAs	27
	3.2.1	Inductively degenerated common-source LNA	
	3.2.2	Input matching	27
	3.2.2	Gain and effective transconductance	
	3.2.2		
	3.2.2	Linearity	
	3.2.2	.4 Cascode stage	
	3.2.3	Common-gate topology	
	3.2.3	.1 Input matching and voltage gain	
	3.2.3	.2 Noise of a CG stage	
	3.2.3	G_m -boosted CG stage	
	3.3 Wide	eband LNA topologies	
	3.3.1	IDCS LNA	
	3.3.1	.1 Input matching bandwidth	
	3.3.1	.2 Output signal current of an IDCS stage	
	3.3.1	.3 Noise figure	
	3.3.1	.4 Conclusion and remarks	
	3.3.2	IDCS LNA with RC-feedback	
	3.3.3	IDCS with additional signal path	
	3.3.3	.1 Comparison of basic IDCS LNA and its alternatives	
	3.3.4	LC ladder matching network	
	3.3.5	CS with shunt-resistor feedback	
	3.3.6	CS with common-drain feedback	53
	3.3.7	Wideband common-gate LNA	
	3.3.7	.1 Input network Q-value	
	3.3.7	.2 Input matching bandwidth	
	3.3.7	.3 Output signal current of CG stage	61
	3.3.7	.4 Noise figure	

	3.3.7	5 Conclusion and remarks	
	3.3.8	Reactive feedback	
	3.3.9	Comparison of wideband LNA topologies	
	3.4 Effec	et of bonding pad	
	3.5 Load	design	
	3.5.1	Narrowband loads	
	3.5.2	Wideband load structures	
	3.5.3	Load design for two-stage wideband amplifiers	
	3.5.3	1 Cascaded shunt-peak loads	
	3.5.3	2 Cross-connected shunt-peak load	80
4	Dow	nconversion mixer design	89
	4.1 Gene	ral mixer design aspects	
	4.2 Passi	ve mixers	
	4.3 Activ	e downconversion mixers	
	4.3.1	Interface to LNA	
	4.3.2	Mixer input stage	
	4.3.3	Switch quad	
	4.3.3	1 Conversion gain	
	4.3.3	2 Noise and linearity	
	4.3.4	Load design and interface to the baseband	
	4.4 Activ	ve mixer design techniques and alternatives	
	4.4.1	Quadrature mixers	
	4.4.2	Folded mixers	
	4.4.3	Mixers with current boosting	
	4.4.4	Merged LNA and mixer	
5	Qua	drature signal generation	109
	5.1 Gene	eral quadrature signal generation aspects	
	5.2 Statio	c frequency dividers	
	5.2.1	D-latch structures	
	5.3 Dyna	mic frequency dividers	
	5.4 First-	-order RC-CR networks	
	5.5 Passi	ve polyphase filters	
	5.5.1	Introduction to the analysis of polyphase filters	
	5.5.2	About PPF topologies, terminology, and notation	

	5.	5.3	PPF fre	equency responses and balances	117
	5.	5.4	Image-	reject ratio	118
		5.5.4	.1 E	Equal RC poles	. 119
		5.5.4	.2 7	Γwo-stage PPF with unequal RC poles	. 121
		5.5.4	.3 1	Three-stage PPF with unequal RC poles	. 122
		5.5.4	.4 (Optimal pole splitting of higher order PPFs	. 124
	5.	5.5	Intrins	ic PPF loss	125
		5.5.5	1 I	Loss of single-stage PPFs	. 126
		5.5.5.	.2 I	Loss of two-stage PPFs	. 126
		5.5.5	.3 I	Loss of three-stage PPFs	. 129
	5.	5.6	The eff	fect of termination impedances	130
		5.5.6	.1 (Optimum component values for a single-stage PPF	. 130
		5.5.6	.2 (Optimum component values for multiple-stage PPFs	. 132
	5.	5.7	The eff	fect of parasitic capacitance	134
		5.5.7	.1 S	Single-stage PPF	. 135
		5.5.7	2 E	Effect on optimal device values	. 136
		5.5.7	.3 N	Multi-stage PPFs	. 136
		5.5.7	.4 S	Summary of effect of parasitic capacitance	. 138
	5.	5.8	Compo	onent value deviation	138
		5.5.8	.1 0	Geometric center frequency deviation	. 139
	5.	5.9	Summa	ary and design example	140
	5.6	Com	parison ration	of frequency dividers and polyphase filters for quadrature signal	.142
6		Desi	gn exar	nples	146
	6.1	2.4-0	GHz rece	eiver for sensor applications	146
	6.	1.1	RF from	nt-end	147
		6.1.1.	1 I	NA and quadrature mixers	. 147
		6.1.1.	2 I	_O buffer	. 149
	6.	1.2	Analog	g baseband circuit	151
		6.1.2	1 0	Channel-select filter	. 151
		6.1.2	.2 N	Mixer-baseband interface	. 151
		6.1.2	.3 E	3iquad	. 151
		6.1.2	4 I	_imiter	. 152
	6.	1.3	Experi	mental results	153
	6.2	Direc	ct-conve	rsion receiver for ubiquitous communications	156

6	5.2.1	Introduction	156
6	5.2.2	RF front-end design	157
	6.2.2	.1 Low-noise amplifier	157
	6.2.2	.2 Downconversion mixer	157
	6.2.2	.3 Quadrature LO signal generation	158
ϵ	5.2.3	Baseband design	159
	6.2.3	.1 Mixer-baseband interface and channel-select filter	159
	6.2.3	.2 Limiting amplifier	159
	6.2.3	.3 Received signal strength indicator	160
6	5.2.4	Experimental results	160
6.3	Effic	ient current reuse for low-power transceivers	164
6	5.3.1	Introduction	164
6	5.3.2	Current boosting with active elements	165
6	5.3.3	Design example	166
ϵ	5.3.4	Simulation results	167
6.4	A di	rect-conversion RF front-end in a 65-nm CMOS	168
6	5.4.1	RF front-end design	168
	6.4.1	.1 Low-noise amplifier	168
	6.4.1	.2 Quadrature mixers	
	6.4.1	.3 Quadrature LO signal generation	171
	6.4.1	.4 Simulation results	173
6.5	Low	-noise amplifiers for UWB	174
6	5.5.1	Inductively degenerated CMOS low-noise amplifiers for BG1	174
6	5.5.2	Common-gate LNA for BG1 and BG3	176
ϵ	5.5.3	Conclusions	178
6.6	A du	al-band direct-conversion RF front-end for WiMedia UWB receiver	179
6	5.6.1	Introduction	179
e	5.6.2	Front-end design	179
	6.6.2	.1 First LNAs (LNA1)	
	6.6.2	.2 Buffer	181
	6.6.2	.3 Second LNA (LNA2)	
	6.6.2	.4 Downconversion mixers	
	6.6.2	.5 LO signal generation	
6	5.6.3	Experimental results	184
6.7	Sum	mary and comparison to other published designs	188

х

7	Conclusions	196
Append	lix A: General common-source LNA	198
Append	lix B: Analysis and optimization of 4-stage PPF	200

Symbols

A	amplitude
A	gain
A_{bal}	amplitude balance
A_i	insertion gain
$A_{i,max}$	maximum insertion gain
A_{v}	voltage gain
BW_{-3dB}	-3-dB bandwidth
BW _{dev}	bandwidth deviation factor
BW _{Iout}	output signal current bandwidth
BW _{rel}	relative bandwidth
BW _{rel,eff}	efficient relative bandwidth
BW _{rel,Iout}	relative output signal current bandwidth
BW _{rel,S11}	relative input matching bandwidth
BW _{S11}	input matching bandwidth
С	noise correlation coefficient
С	system capacity
C_C	coupling capacitance
C_{fb}	feedback capacitance
C_{gd}	gate-drain capacitance
C_{gs}	gate-source capacitance
C_{in}	input capacitance
CLK _{IN}	input clock signal
C _{par}	parasitic capacitance
C_{pn}	parasitic capacitance, $n = 1, 2, 3,$
C_S	shunt capacitor
C_T	total capacitance
C_V	virtual ground capacitor
$e_{n,out}$	output noise voltage
f	frequency
F	noise factor
$f_{SII=-10dB}$	frequency where S_{II} equals -10 dB
f _{S11=-10dB,HI}	upper corner frequency where S_{II} equals -10 dB
f _{S11=-10dB,LO}	lower corner frequency where S_{11} equals -10 dB

F_{DSB}	double-sideband noise factor
f _{in}	input frequency
f _{Iout,HI}	maximum output current signal frequency
f _{Iout,LO}	minimum output current signal frequency
f _{max}	maximum oscillation frequency
fout	output frequency
f _{S11,HI}	maximum input matching frequency
f _{S11,LO}	minimum input matching frequency
F _{SSB}	single-sideband noise factor
f_T	unity (current) gain frequency
F _{TOT}	total noise factor
G	power gain
G_{conv}	conversion gain
g_{d0}	output conductance at zero drain-source voltage
g_{ds}	channel conductance
g_g	induced gate noise factor
g_m	transconductance
$g_{m,eff}$	effective transconductance
$g_{m,n}$	transconductance of an NMOS transistor
$g_{m,p}$	transconductance of a PMOS transistor
g_{mb}	substrate transconductance
Н	transfer function
Ibias	bias current (source)
Iboost	boost current (source)
I_D	drain current
$\overline{i_d^2}$	drain current noise
I_E	additional current source
$\overline{i_g^2}$	gate-induced current noise
I _{LNA}	low-noise amplifier current
I_{LO}	LO buffer current
Iout	output signal current
Iout,max	maximum output signal current
I_{REF}	reference bias current
IRR _{gain}	image-reject ratio defined with magnitude balance only
$IRR_{min,n}$	minimum image-reject ratio of an <i>n</i> -stage polyphase filter

IRR _{phase}	image-reject ratio defined with phase balance only
I _{SWI}	switch transistor current
k	magnetic coupling factor
k_B	Boltzmann's constant, 1.380658 · 10 ⁻²³ J/K
k_B	capacitor ratio
k_L	load impedance coefficient
$k_{L,opt,n-stg}$	optimum load impedance coefficient of an <i>n</i> -stage polyphase filter
k_n	coefficient for NMOS transistor
k_n	relative pole frequency factor, $n = 1, 2, 3,$
k_{nC}	capacitor ratio, $n = 1, 2, 3, \ldots$
k_{nR}	resistor ratio, $n = 1, 2, 3, \ldots$
<i>k</i> _{par}	parasitic capacitance coefficient
k_S	source impedance coefficient
k_Z	ratio of polyphase filter output and input impedances
L	channel length
L	inductor
L	loss
L_{Cpar}	additional loss due to parasitic capacitance
L_d	drain inductance
L_g	gate inductance
L_{in}	input inductance
LO_{EXT}	external LO signal
LO_{VCO}	LO signal from a VCO
L_P	primary inductor
L_S	source inductance
L_S	secondary inductor
L _{tot,n-stg}	total loss of an <i>n</i> -stage polyphase filter
т	shunt-peak load design parameter
М	mutual inductance
M_{LOAD}	load transistor
M_n	MOS transistor, $n = 1, 2, 3, \ldots$
M_{Nn}	NMOS transistor, $n = 1, 2, 3, \ldots$
M_{Pn}	PMOS transistor, $n = 1, 2, 3,$
M_{SWI}	switch transistor
п	number of fingers

n	turns ratio
N_{DSB}	input-referred double-sideband noise power
NF _{DSB}	double sideband noise figure
NF_{MIN}	minimum noise figure
NF_{SSB}	single sideband noise figure
N_S	source noise power
N _{SSB}	input-referred single-sideband noise power
р	resonator design parameter
Р	power
P _{fund}	power of a fundamental signal
P _{IMDn}	power of an input-referred n^{th} -order intermodulation component
P _{in}	input power
Pout	output power
q	electron charge, 1.6021773 · 10 ⁻¹⁹ C
Q	quality factor
Q_{in}	input network quality factor
Q_L	inductor quality factor
Q_r	resonator quality factor
R	resistor
r _{ds}	drain-source resistance
R_{eq}	equivalent (input) resistance
R_{fb}	feedback resistance
RF _{in}	input RF signal
r_g	gate resistance
R _{in}	real part of an input impedance
R_L	load resistance
<i>R</i> _{Lin}	series resistance of an input inductor
R_{Ls}	inductor series resistance
R_{max}	maximum resistance value
R _{min}	minimum resistance value
r _o	output resistance
R_p	shunt resistance
R_S	source resistance
R_{sh}	silicide sheet resistance
R_{typ}	typical resistor value

S	Laplace variable
S	sensitivity level
S_{11}	input matching
$S_{11,BW}$	input matching bandwidth
SNR _{in}	signal-to-noise ratio at the input
SNR _{min}	minimum signal-to-noise ratio
SNR _{out}	signal-to-noise ratio at the output
Т	absolute temperature
Т	signal period
Т	transformer
$V_{D,sat}$	drain-source saturation voltage
V _{dc,out}	output common mode voltage
V_{DD}	supply voltage
V_{DS}	drain-source voltage
$\overline{v_g^2}$	gate voltage noise
V_{GS}	gate-source voltage
V _{in}	input voltage
V_{LO}	local oscillator amplitude
Vout	output voltage
V_{pp}	peak-to-peak voltage
V_{REF}	reference voltage
V_{RF}	RF signal amplitude
V_S	source voltage
V_T	threshold voltage
W	channel width
Ζ	impedance
Z_0	source (reference) impedance
Z_{fb}	feedback impedance
Zg	gate impedance
Z_{gd}	gate-drain impedance
Z_{gs}	gate-source impedance
Z _{in}	input impedance
$Z_{in,PPF}$	polyphase filter input impedance
$Z_{in,\omega n}$	input impedance at the frequency ω_n , $n = 1, 2, 3,$
Z_L	load impedance

$Z_{L,Cpar}$	load impedance with parasitic capacitance
Z_{LC}	LC resonator characteristic impedance
Zout	output impedance
$Z_{out, \omega n}$	output impedance at a frequency ω_n , $n = 1, 2, 3,$
$Z_{out, PPF}$	polyphase filter output impedance
Z _{rel}	relative impedance
Z_S	source impedance
α	noise parameter
α_n	n^{th} -order nonlinearity coefficient, $n = 0, 1, 2,$
β	feedback factor
ϕ	control signal
δ	gate induced noise factor
χ	ratio of a transistor substrate transconductance g_{mb} and transconductance g_m
Г	reflection coefficient
γ	drain current noise factor
ω	angular frequency
ω_0	center (angular) frequency
ω_c	corner frequency
$\omega_{c,max}$	maximum corner frequency
$\omega_{c,min}$	minimum corner frequency
$\omega_{c,n-stg}$	geometric center frequency of an <i>n</i> -stage polyphase filter
$\omega_{c,RCdev}$	center frequency, device values deviating from typical values
$\omega_{c,RCtyp}$	center frequency, typical device values
ω_{Fmin}	minimum noise factor frequency
$\omega_{Iout,max}$	maximum output signal current frequency
$\omega_{IRR,min}$	minimum image-reject ratio frequency
ω_{max}	maximum frequency
ω_{min}	minimum frequency
ω_n	pole (angular) frequency, $n = 1, 2, 3,$
ω_r	resonance (angular) frequency
$\omega_{SII=-10dB}$	angular frequency where S_{11} equals -10 dB
$\omega_{SII=-10dB,HI}$	upper angular frequency where S_{11} equals -10 dB
$\omega_{SII=-10dB,LO}$	lower angular frequency where S_{II} equals -10 dB
ω_T	unity gain (angular) frequency
$\Delta \omega_{-3dB}$	-3-dB bandwidth

$\Delta \theta$	phase deviation
ΔL	loss difference
ΔP_n	ratio between fundamental signal and the n^{th} -order intermodulation product
ΔR	resistance value deviation
ΔR_{max}	relative maximum resistor value deviation
ΔR_{min}	relative minimum resistor value deviation
ΔT	turn on time
ΔV	differential (voltage) signal

Abbreviations

ac	alternative current
ADC	analog-to-digital converter
AGC	automatic gain control
BB	baseband
BG	band group
BT	Bluetooth
BT LEE	low-end extension for Bluetooth
BW	(noise) bandwidth
BWER	bandwidth extension ratio
CD	common-drain
CDMA	code division multiple access
CG	common-gate
CLK	clock
CMOS	complementary metal oxide semiconductor
CMRR	common-mode rejection ratio
CS	common source
CSMA	carrier sense multiple access
DAC	digital-to-analog converter
dc	direct current
DCR	direct-conversion receiver
DCS	Digital Cellular System
DSB	double-sideband
ESD	electrostatic discharge
FCC	Federal Communications Commission
FDD	frequency division duplex
FDMA	frequency division multiple access
FET	field effect transistor
GFSK	gaussian frequency shift keying
g _m C	transconductance-capacitor
GPS	Global Positioning System
GSM	Global System for Mobile Telecommunications
Ι	in-phase
IC	integrated circuit

ICP	input compression point
IDCS	inductively degenerated common-source
IEEE	Institute of Electrical and Electronics Engineers, Inc
IF	intermediate frequency
IIP2	input referred second-order intercept point
IIP3	input referred third-order intercept point
IMD2	second-order intermodulation distortion
IMD3	third-order intermodulation distortion
IP2	second-order intercept point
IP3	third-order intercept point
IRR	image-reject ratio
ISM	Industrial, Scientific, and Medical
LC	inductor-capacitor
LNA	low-noise amplifier
LO	local oscillator
MBOA	Multiband OFDM Alliance
MB-OFDM	multiband OFDM
MOS(FET)	metal oxide semiconductor (field-effect transistor)
NF	noise figure
NMOS	N-channel metal oxide semiconductor transistor
OFDM	orthogonal frequency division multiplexing
P _{1dB}	input compression point
PA	power amplifier
РСВ	printed circuit board
PCS	Personal Communication Services
PDC	Personal Digital Cellular, (Pacific Digital Cellular)
PLL	phase-locked loop
PMOS	P-channel metal oxide semiconductor transistor
PPF	polyphase filter
PSD	power spectral density
PVT	process, voltage, and temperature
Q	quadrature phase
RC	resistor-capacitor
RF	radio frequency
RLC	resistor-inductor-capacitor

RSSI	received signal strength indicator
RX	receiver
SAW	surface acoustic wave
SCL	source-coupled logic
SNR	signal-to-noise ratio
SPF	single-stage polyphase filter
SSB	single-sideband
TX	transmitter
UMTS	Universal Mobile Telecommunications System
U-NII	Unlicensed National Information Infrastructure
UWB	ultrawideband
VCO	voltage controlled oscillator
VGA	variable gain amplifier
WCDMA	Wideband Code Division Multiple Access
Wi-Fi	Wireless Fidelity
WiMAX	Worldwide Interoperability for Microwave Access
WLAN	wireless local area network
WPAN	wireless personal area network

1 Introduction

During the last decade, we have seen the emergence of numerous new radio technologies and applications. The development has been rapid and the field of new innovations is wide. Lately, the development of wireless systems has diverged, with the trend now being to offer quite different data rates. High data rates and wide bandwidths are provided to enable consumers to transfer larger volumes of data in smaller amounts of time. Ultrawideband (UWB), which is capable of providing data rates up to 480 Mbps, is one example of such a system. On the other hand, there exist a class of applications that do not require such high speed or bandwidth, but would still gain benefit from a wireless connection. Wireless sensors are an example of applications where a modest, i.e. less than 1 Mbps, capacity is adequate. Sensors operate with extremely low power providing autonomous operation for several years with minimal maintenance cost and effort.

The CMOS technology evolution is mainly driven by the microprocessor industry. As line width continues to shrink, the transistors occupy less silicon area and switch faster, which is crucial for digital computing. Radio frequency (RF) designs have also gained benefit from technology scaling. The typical figures of merit, cut-off frequency f_T and maximum oscillation frequency f_{MAX} , are approximately inversely proportional to the channel length, and both have exceeded the 100 GHz limit [1]. The speed of active devices is therefore more than adequate for the common wireless systems such as GSM, WCDMA, Bluetooth, and WLAN (IEEE 802.11a/b/g), which operate at frequencies below 6 GHz. In addition, their speed is sufficient for UWB, for which maximum operational frequencies are around 10 GHz.

Clearly, the design of digital and analog circuits in modern deep-submicron process faces several new challenges. For example, gate current leakage, lowering self gain g_m/g_{ds} , device matching etc. are the primary challenge in scaled CMOS technologies [2]. Except for the switching time, the analog performance of the transistors is getting worse, along with evershrinking gate length. Thus, analog/RF ICs profit less from new technology generations than pure digital electronics. The decreasing supply voltages are making the design of analog and RF circuits more challenging, while the continuous power supply voltage reduction does not lead to lower power consumption [2]. RF circuits are usually dominated by passive components, the size of which does not scale proportionately. As a result, the RF chip area does not shrink as much as digital circuits do. Despite of all the challenges, there is a need to realize complete transceivers and baseband processors on a single CMOS chip to minimize the silicon area and costs. Lately, innovative receivers, where the digital signal processing functions are brought closer to the LNA, have been published [3], [4]. Still, most of the RF front-ends, including at least an LNA and mixer, are designed in the analog domain.

1.1 Research contribution and publications

This thesis concentrates on the design and implementation of RF circuits for low-power shortrange applications such as sensor systems and UWB. The main interest is in LNAs, downconversion mixers, and quadrature local oscillator (LO) signal generation circuits for direct-conversion receivers. The author provides solutions related to low-power and wideband applications. The thesis includes theory involved in analyzing the performance of the existing circuits and proposes new ones. It should be noted that the circuits presented here are not limited to these two main applications, but can be utilized in other kinds of wireless communication systems as well.

A large number of receivers for sensor applications and UWB have been reported in the literature. It is not the purpose of this thesis to review these comprehensively. Nor is the focus of this thesis on developing models for different IC components or on deriving requirements for different blocks based on system specifications and receiver partitioning.

This thesis is mainly based on the previously published work of the author in [P1] - [P7]. The short summary and contribution of each publication is given below. In addition, the author has authored or co-authored other publications related to the topics in [P8] - [P12], which are not included in this thesis.

Paper P1 is a journal article presenting a receiver design for a 2.4-GHz sensor system. It includes a front-end consisting of an LNA, which is merged with the downconversion mixers, LO buffers, and one baseband channel. The receiver was designed, implemented, and measured by a research team, which consisted of four members, including the author. The other members were Prof. J. Ryynänen, J. Järvinen, and D.Sc. J. Jussila. The author is responsible for the design and implementation of the RF front-end with Prof. J. Ryynänen. J. Järvinen and D.Sc. J. Jussila are responsible for designing the analog baseband parts. The idea of the presented current boosting method was originally proposed by Prof. J. Ryynänen. The other authors, D.Sc. K. Kivekäs and M. Honkanen, are mainly responsible for writing the details covering the sensor system. Prof. K. Halonen gave valuable comments on the manuscript.

Paper P2 is a conference article that deals the circuit design for the sensor system presented in [P1]. Compared to the first demonstrator [P1], the LNA and mixer are separated, the 90-degree phase shift generation circuit is included, the second baseband channel is added, and the receiver also includes improved limiters for A/D conversion, the received signal strength indicator (RSSI), and on-chip bias current generator. However, the active area from the first receiver is not increased. The contribution is same as for paper [P1].

Paper P3 is a conference paper that demonstrates the RF front-end design in a 65-nm CMOS. The front-end includes the low-noise amplifier, folded quadrature mixers, frequency divider for quadrature LO signal generation, and LO buffers. The author designed all the circuits and had the main responsibility for writing the manuscript under the supervision of Prof. J. Ryynänen.

Papers P4 and P5 are conference articles that analyze the performance of the inductively degenerated common-source and common-gate LNAs, respectively. The presented design examples are targeted for the UWB system. The theory presented in these papers was carried out by the author under the supervision of Prof. J. Ryynänen.

Paper P6 is a conference article, which covers the RF front-end design for the dual-mode UWB WiMedia receiver. The front-end consists of multi-stage LNAs, the downconversion mixers, the passive polyphase filters, and LO buffers. The author is responsible for the design, implementation, and measurements of the RF front-end, in addition to the writing of the manuscript under the supervision of Prof. J. Ryynänen. Prof. K. Halonen gave valuable comments for the manuscript.

Paper P7 is a journal article accepted for publication. It presents the analysis and design of passive polyphase filters in detail. The author is responsible for all the presented analysis and

has a major responsibility for the manuscript. K. Stadius is responsible for writing the introduction. He was also involved in the brainstorming of the needed analysis and, together with the other authors, suggested several improvements to the manuscript.

- [P1] J. A. M. Järvinen, J. Kaukovuori, J. Ryynänen, J. Jussila, K. Kivekäs, M. Honkanen, and K. Halonen, "2.4-GHz receiver for sensor applications", *IEEE Journal of Solid-State Circuits*, vol. 40, no. 7, July 2005, pp. 1426-1433.
- [P2] J. Kaukovuori, J. A. M. Järvinen, J. Ryynänen, J. Jussila, K. Kivekäs, and K. A. I. Halonen, "Direct-conversion receiver for ubiquitous communications," *Radio and Wireless Symposium (RWS'06)*, San Diego, CA, 17-19 Jan. 2006, pp. 103-106.
- [P3] J. Kaukovuori, J. Ryynänen, and K. A. I. Halonen, "A direct-conversion RF front-end in a 65-nm CMOS," *Norchip Conference*, Linköping, Sweden, 20-21 Nov. 2006, pp. 235-238.
- [P4] J. Kaukovuori, J. Ryynänen, and K. A. I. Halonen, "CMOS low-noise amplifier analysis and optimization for wideband applications," *IEEE Ph.D Research in Microelectronics and Electronics (PRIME'06)*, Otranto, Italy, 11-16 June 2006, pp. 445-448.
- [P5] J. Kaukovuori, M. Kaltiokallio, and J. Ryynänen, "Analysis and design of commongate low-noise amplifier for wideband applications," *European Conference on Circuit Theory and Design (ECCTD'07)*, Sevilla, Spain, 26-30 Aug. 2007, pp. 64-67.
- [P6] J. Kaukovuori, J. Ryynänen, and K. A. I. Halonen, "A dual-band direct-conversion RF front-end for WiMedia UWB receiver," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'07)*, Honolulu, HI, 3-5 June 2007, pp. 211-214.
- [P7] J. Kaukovuori, K. Stadius, J. Ryynänen, and K. A. I. Halonen, "Analysis and design of passive polyphase filters," *IEEE Trans. on Circuits and Systems I – Regular Papers*, accepted for publication.

Other publications related to the topic.

- [P8] K. Stadius, T. Rapinoja, J. Kaukovuori, J. Ryynänen, and K. A. I. Halonen, "Multitone fast frequency-hopping synthesizer for UWB radio," *IEEE Trans. on Microwave Theory and Techniques*, vol. 55, no. 8, Aug. 2007, pp. 1633-1641.
- [P9] M. Hotti, J. Kaukovuori, J. Ryynänen, J. Jussila, K. Kivekäs, and K. Halonen, "A dualband direct-conversion RF front-end with IIP2 calibration," *Proc. of the European Microwave Association (EuMA)*, vol. 1, no. 4, Dec. 2005, pp. 281-287.
- [P10] M. Kaltiokallio, J. Kaukovuori, and J. Ryynänen, "Analysis of feedbacks in LNA matching," *European Conference on Circuit Theory and Design (ECCTD'07)*, Sevilla, Spain, 26-30 Aug. 2007, pp. 68-71.
- [P11] M. Hotti, J. Kaukovuori, J. Ryynänen, J. Jussila, K. Kivekäs, and K. Halonen, "A direct conversion RF front-end for 2-GHz WCDMA and 5.8-GHz WLAN applications," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'03)*, Philadelphia, PA, 8-10 June 2003, pp. 45-48.
- [P12] J. Kaukovuori, M. Hotti, J. Ryynänen, J. Jussila, and K. Halonen, "A linearized 2-GHz SiGe low noise amplifier for direct conversion receiver," *IEEE International* Symposium on Circuits and Systems (ISCAS'03), Bangkok, Thailand, 25-28 May 2003, pp. II-200-203.

1.2 Organization of the Thesis

This thesis is organized into six chapters. After the introduction, Chapter 2 contains the background information related to the topic of the thesis. Receiver architectures and fundamental definitions related to the receiver design are discussed. A brief introduction to sensor networks and UWB systems is given.

Chapter 3 concentrates on low-noise amplifier design. First, narrowband LNAs based on inductively degenerated common-source and common-gate topologies are discussed. Then, several LNA topologies for wideband applications are presented and analyzed. The main emphasis is on the input matching and insertion gain bandwidth analysis and optimization. Subsequently, the issues related to the bonding pad and LNA load design are covered.

Chapter 4 is focused on the downconversion mixer design. A short overview of passive mixers is provided, but most of the chapter is devoted to active Gilbert cell mixer design. The issues related to the mixer input stage, switch quad, and load design and interfaces with other blocks are discussed.

In Chapter 5, techniques to generate a quadrature signal are discussed. The main emphasis is on the analysis and design of passive polyphase filters. The design of static and dynamic frequency dividers is also presented.

The circuit implementations and design examples are included in Chapter 6. First, directconversion receivers targeted to low-power sensor systems are presented. Then, a design example of a 2-GHz RF front-end simulated with a 65-nm CMOS is given. The wideband LNA design examples and a processed front-end IC targeted for UWB are presented. Finally, the thesis is summarized.

References

- [1] "International technology roadmap for semiconductors 2005: Radio frequency and analog/mixed-signal technologies for wireless communications," [Online]. Available: http://www.itrs.net/Links/2005ITRS/Wireless2005.pdf
- [2] A.-J. Annema, B. Nauta, R. van Langevelde, and H. Tuinhout, "Analog circuits in ultra-deep-submicron CMOS," *IEEE J. of Solid-State Circuits*, vol. 40, no. 1, Jan. 2005, pp. 132-143.
- [3] R. B. Staszewski, *et al.*, "All-digital TX synthesizer and discrete-time receiver for Bluetooth radio in 130-nm CMOS," *IEEE J. of Solid-State Circuits*, vol. 39, no. 12, Dec. 2004, pp. 2278-2291.
- [4] R. Bagheri, A. Mirzaei, S. Chehrazi, M. E. Heidari, M. Lee, M. Mikhemar, W. Tang, and A. Abidi, "An 800-MHz–6-GHz software-defined wireless receiver in 90nm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2860-2876.

2 Overview of RF IC design for wireless radio systems

The purpose of this chapter is to provide an overview to the design of wireless short-range radio systems. Fundamental issues related to receiver design are discussed at a general level to give background information for the reader and to justify the chosen solutions presented later in the thesis. The purpose is not to provide in-depth discussion on all relevant aspects, but references to the literature are comprehensively given.

First, the common receiver architectures are briefly discussed. Next, the fundamental receiver parameters needed in RF circuit design are defined. Since this thesis also includes some circuit analysis, the MOS transistor small-signal model used in calculations is described. Wireless sensor systems and UWB, which are the two main target applications for the design examples presented in Chapter 6, are then briefly discussed.

2.1 Receiver architectures

The main purpose of the RF receiver is to provide certain tasks for the received signal such as amplifying, filtering, demodulation, and analog-to-digital conversion with adequate signal-to-noise ratio (SNR) before digital signal processing. The received RF signal can be strong or extremely weak, while there can be a strong blocking signal(s) with a certain offset from the wanted frequency, which needs to be rejected. These translate into requirements in terms of dynamic range, sensitivity, blocking, and intermodulation performance [1]. The choice of the receiver architecture affects the power dissipation, requirements, and achievable performance. The integration level with the number of external components determines the cost, which is also a crucial criterion in selecting the proper receiver architecture. In addition, external filters typically require a rather low impedance level to drive them. Therefore, when the target is to design the receiver with low power consumption, the number of such filters should be minimized.

In the following, the most common radio receiver topologies, i.e. superheterodyne, directconversion, low-IF, and wideband IF, are briefly introduced. Since software defined radios [2], sub-sampling receivers [3], and super-regenerative receivers [4] – [6] differ remarkably from the typical receiver topologies regarding the implementation of RF parts, they are left out of this thesis.

2.1.1 Direct-conversion architecture

The direct-conversion topology is also called as *zero-IF* or *homodyne* architecture [7]. It was published in 1924 by F. M. Colebrook [8] and the first practical implementations were introduced in 1947 [9]. However, it was not until the beginning of 1990s that the DCR was suitable for IC integration due to difficulties of handling the problems associated with second-order nonlinearity and dc offsets. The first IC DCRs were applied on paging receivers [10].

The block diagram of a typical direct-conversion receiver is shown in Fig 2.1. Because the selectivity of the front-end usually is not adequate, the RF signal after the antenna is pre-filtered to attenuate the signals out of the reception band. The signal is amplified in the low-noise amplifier (LNA) before being downconverted to zero intermediate frequency (IF). In CDMA

and WCDMA systems, an external interstage filter after the LNA can be used to attenuate the TX signal leakage and to relax the linearity requirements of the succeeding mixer [11], [12]. For phase- and frequency-modulated signals, the downconversion must be performed in quadrature to prevent signal sidebands from aliasing on one another [13]. Because the LO is centered in the desired channel, useful signal and noise occupy both the upper and lower sidebands. The low-pass filter with a bandwidth of a half of the symbol rate removes adjacent channels at baseband. Because the filtering is performed at low frequencies, the filters can be realized on-chip without external high-Q components.



Fig 2.1. Direct-conversion receiver.

From a power consumption point of view, the direct-conversion architecture is a tempting solution. The number of internal nodes operating at RF frequency is small and there is no need for buffering the high-frequency signals off-chip. Because the RF signal is converted directly to zero intermediate frequency (IF), the image consists of the channel itself. Therefore, DCR architecture eliminates the image-reject problem existing in other radio architectures [13]. However, insufficient amplitude and phase balance between in-phase (I) and quadrature phase (Q) signals can increase the bit error rate. Compared to other architectures, the requirement for the image-rejection ratio is less stringent [11], [14].

DCR has its own drawbacks, for example, a high sensitivity to flicker noise and dc offsets [13], [15]. Wideband communications systems, e.g. WCDMA, WLAN, and UWB, reduce the influence of the flicker noise and, depending on the system specifications, the dc offset problem can be relaxed by highpass filtering without deteriorating the SNR [11]. On the other hand, in narrowband systems such as GSM, PCS, and Bluetooth, most of the signal power is located close to dc frequency and therefore highpass filtering is not possible [14], [16]. Self-mixing, LO leakage, and even-order distortion, which can deteriorate the received signal, are covered in detail in, for example, [13], [15], [17], and [18].

2.1.2 Superheterodyne receiver

The superheterodyne architecture was introduced by E. Armstrong in 1918 [19]. The block diagram of a superheterodyne receiver is shown in Fig 2.2. The pre-section filter suppresses the signals out of the reception band. The LNA output is matched to the image-reject filter, which attenuates the unwanted signals and noise at the image-frequencies. The channel-select filter after the first downconversion is usually at the fixed frequency. Therefore, the first intermediate frequency is typically set by the VCO, which frequency is adjustable to cover the whole

reception band. The first IF must be higher than half of the reception bandwidth (i.e. the bandwidth of the pre-selection filter) to keep the image signal out of the downconverted band. After the channel-select filter, the signal is amplified and divided into I and Q branches in the demodulator. The selectivity can be improved by increasing the number of IF steps, but this comes at the expense of an additional filter stage.

The main drawback of a superheterodyne is the number of external filters, which are difficult to integrate on-chip with current IC technologies. That leads to a low integration level and both the cost and power consumption are increased due to increased number of off-chip components and due to buffering high-frequency signals off-chip. However, superior sensitivity and selectivity can be achieved with superheterodyne architecture [20]. The superheterodyne architecture is the most current-consuming receiver and requires expensive external components. It is therefore not applicable to multi-mode receivers or low-power systems.



Fig 2.2. Superheterodyne receiver.

2.1.3 Low-IF architecture

Basically, the block diagram of the low-IF receiver is same as for the DCR (see Fig 2.1). The low-IF receiver downconverts the input signal directly to a low IF frequency, which is above dc, but lower than half of the reception bandwidth. Single-stage downconversion is performed in quadrature, and a low-IF receiver does not need an external intermediate filter. Thus, a high integration level is achieved. Compared to the DCR, the low-IF receiver does not suffer from the dc offset problem. Because there is not a signal at dc, the dc offsets can be filtered out without removing the signal. In addition, the flicker noise is less problematic.

The drawback of the low-IF receiver is that the mirror (image) signal, which has an offset from the LO frequency equal to the desired channel but which is on the opposite side of the LO, is different from the wanted signal. Thus, the low-IF architecture requires a good matching for image rejection [21]. The choice of the IF frequency offers design tradeoffs. A very low IF relaxes the image-rejection requirement but complicates the requirements for the frequency synthesizer [16]. In addition, if the IF frequency is too low, the signal can be corrupted by 1/*f* noise and the fractional bandwidth, where the image must be rejected, becomes larger [22]. High IF frequency, however, increases the complexity and the current consumption of the IF stages. The low-IF receiver is feasible for narrowband systems having a moderate IRR requirement, e.g. GSM [23], GPS [24], DCS-1800 [25], and Bluetooth [16]. Low-IF topology is proposed for 2.4-GHz WLAN as well [22].

2.1.4 Wideband-IF

In a wideband-IF receiver architecture, such as that shown in Fig 2.3, the signal is downconverted in two phases to zero frequency [20], [26], [27]. First, the whole reception band is downconverted with quadrature mixers such that a large bandwidth signal at IF is maintained. A simple low-pass filter is used at IF to remove any upconverted frequency components, allowing all channels to pass to the second stage of mixers [20]. At the second downconversion to zero-IF, the wanted channel is selected by adjusting the frequency of the second LO. The channel filtering is finally performed at baseband as in DCR. Although downconversion is performed in multiphase, as in the typical superheterodyne receiver, discrete filters are avoided. The image rejection is achieved with the second downmixing step, which is similar to the Weaver technique [28], [29].

Compared to DCR, wideband-IF has several advantages. First, there are no local oscillators, which operate at the same frequency as the receiver RF signal, thus minimizing problems related to time-varying dc offsets. In addition, if the channel selection is performed by tuning only the frequency of the second LO, reduction in phase noise in the first LO can be achieved [20]. In addition, the flicker noise requirement for the first mixer is relaxed. However, care has to be taken in the accuracy of the first downconversion not to deteriorate the image-reject capability and the sensitivity of the receiver [30]. In addition, the multistage realization increases the power consumption.



Fig 2.3. Wideband-IF receiver.

2.1.5 **RF front-end in different receiver topologies**

As stated earlier, the choice of the most suitable receiver topology is constrained in several ways. The most popular receiver topology candidates for CMOS integration nowadays are low-IF and direct-conversion. The demonstrator receivers presented in Chapter 6 are all targeted for direct-conversion receiver. From a RF front-end point of view, however, there is no significant difference, which the targeted receiver topology is. The LNA is required in all receiver topologies, and most of the receiver topologies include mixer(s) as well. Therefore, the LNA

and mixer design methods presented in this thesis are suitable for most of the receiver topologies. It should be noted that the mixer directly follows the LNA in the experimental RF front-ends presented in Chapter 6. Therefore, LNA output matching is not required. If the LNA load is an external filter or if there is a need to measure the performance of the LNA separately, then the LNA output needs to be matched to a certain impedance.

2.2 Definitions and general receiver design parameters

This section is devoted to giving definitions of the most typical design parameters, which are encountered in analog RF circuit analysis, simulations, and measurements and which are used in later parts of this thesis. More details related to the characteristics of wireless communication systems can be found in, for example, [1], [30], [31], and [32].

2.2.1 Input matching and stability

The input of the LNA must be matched to a specific impedance level so as not to deteriorate the performance of the preceding filter or balun. The input matching is defined with the reflection coefficient Γ , which is defined as

$$\Gamma = \frac{Z_{in} - Z_0}{Z_{in} + Z_0}$$
(2.1)

and $20 \log |\Gamma|$ is defined as the return loss. In (2.1) Z_{in} is the LNA input impedance and Z_0 is the reference (source) impedance. Usually, Z_0 is defined by the pre-select filter or balun and return loss better than -10 dB is targeted [33], [34]. The stability of the LNA is checked by ensuring that the real part of the input impedance remains positive at all frequencies [7].

For two-ports, scattering parameters (s-parameters) are typically defined and measured. For example, the s-parameter S_{II} measures the quality of the matching of port 1 when the port 2 is terminated. Commonly, the parameter S_{II} is used to measure the input matching even though the port 2 is not well-defined. As was mentioned, this is the case for example in direct-conversion receivers, where the LNA can be directly followed by a mixer. Therefore, in this thesis, S_{II} is always used to represent the quality of the input matching.

2.2.2 Voltage gain and insertion gain

For the superheterodyne receiver, where both the input and output ports are matched, the power gain is a suitable figure-of-merit for the gain. Different power gain definitions are given for example in [7] and [35]. However, for the direct-conversion and low-IF receivers, the LNA is typically the only block requiring input matching and output matching is not required. In such cases, the most common figure-of-merit for the gain is the voltage gain, which is defined as a ratio of output and input voltages:

$$A_{v} = \frac{V_{out}}{V_{in}} \,. \tag{2.2}$$

The output voltage at load impedance Z_L is $V_{out} = I_{out}Z_L$, where I_{out} is the output signal current of the input stage. Therefore, the voltage gain can be expressed as

$$A_{v} = \frac{V_{out}}{V_{in}} = \frac{I_{out}Z_{L}}{V_{in}} = g_{m,eff}Z_{L}, \qquad (2.3)$$

where $g_{m,eff}$ is the effective transconductance of the input stage, as shown in Fig 2.4. In general, at least for LNAs, $g_{m,eff}$ is different from the transconductance g_m of the input transistor due to the effect of the input matching circuit, as is discussed later. In addition, for downconversion mixers, the effective gain is different due to conversion loss. In narrowband applications, where the reception bandwidth is typically less than 100 MHz, it is sufficient to give the LNA gain at the center frequency only. Therefore, as is shown in (2.3), the LNA gain is set by appropriately designing the effective transconductance $g_{m,eff}$ and the load impedance Z_L .



Fig 2.4. General transconductance stage including source and load impedances. In the case of LNA, g_m -element includes the input matching circuit as well.

In a perfectly matched case, the input voltage V_{in} is half of the source voltage V_S . However, LNAs have a perfect input match only in a specific frequency area. Therefore, for wideband LNAs having several 100 MHz or over 1 GHz bandwidth, a more suitable merit is the insertion gain A_i^{1} defined as the ratio of the output signal voltage to half of the voltage of the source driving the LNA input [36], i.e.

$$A_i = \frac{V_{out}}{\frac{V_s}{2}} = 2\frac{V_{out}}{V_s} = 2\frac{I_{out}Z_L}{V_s}.$$
(2.4)

The insertion gain measures only the variation of the output voltage and is fundamentally limited by either the variation in Z_L or output signal current I_{out} as given in (2.4). Therefore, both the Z_L and I_{out} should have adequate bandwidths to achieve the required insertion gain over the wanted frequency area. When insertion gain is referred to later in this thesis, the definition given in (2.4) is used.

2.2.3 Linearity

Linearity is an important measure of a single block or the whole receiver. The receiver must be able to receive a weak signal in the presence of strong interfering ones. There are several measures of linearity. The most commonly used receiver linearity tests are input 1-dB compression point (ICP or P_{1dB}) and the third-order intercept point (IP3). In addition, the second-order intercept point (IP2) is a figure-of-merit for certain receiver architectures, such as

¹ In this thesis, A_i always represents the insertion gain, not the current gain.

DCRs. Linearity tests are typically performed with single-tone or two-tone sinusoidal signals. Linearity tests can be specified for modulated signals as well, but the use of such signals in simulations is tedious. Therefore, the relation between modulated test signal(s) and corresponding sinusoidal test signal(s) can be derived, but to do so is not within the scope of this thesis.

2.2.3.1 Compression and desensitization

For a memoryless nonlinear system, the input-output relationship can be approximated with a polynomial

$$y(t) = \alpha_0 + \alpha_1 x(t) + \alpha_2 x^2(t) + \alpha_3 x^3(t), \qquad (2.5)$$

where α_n are constant in a time-invariant system [7]. When $A\cos(\omega t)$ is used as an input signal for a nonlinear system depicted with (2.5), the output signal is

$$y(t) = \alpha_0 + \frac{1}{2}\alpha_2 A^2 + \left(\alpha_1 + \frac{3}{4}\alpha_3 A^3\right)\cos(\omega t) + \frac{1}{2}\alpha_2 A^2 \cos(2\omega t) + \frac{\alpha_3}{4} A^3 \cos(3\omega t). \quad (2.6)$$

The signal at the fundamental frequency decreases when α_l and α_3 have opposite signs. The input compression point measures the level of the input signal, which causes the 1-dB drop from the small-signal gain. This is graphically presented in Fig 2.5. Moreover, the largest voltage swing in the front-end is typically achieved at the mixer output node (assuming a voltage mode operation). The gain compresses due to the limited the supply voltage and the signal amplitude does not increase along with the input signal level. When this is the case, the signal clipping at the mixer output can dominate the front-end compression point.



Fig 2.5. Definition of the input 1-dB compression point.

The receiver inability to process a weak desired signal is called desensitization [37]. In practice, there are two main mechanisms causing desensitization. Blocking occurs when the gain of the wanted weak signal compresses in the presence of large interferers. Blocking is caused by third-order nonlinearity. When a two-tone input signal $A_1 \cos(\omega_1 t) + A_2 \cos(\omega_2 t)$ is fed into a nonlinear system depicted by (2.5) such that A_1 is the amplitude of a weak wanted signal and A_2 is the amplitude of a strong interferer, the gain at the wanted signal is

$$y_{\omega 1}(t) = \left(\alpha_{1} + \frac{3}{4}\alpha_{3}A_{1}^{2} + \frac{3}{2}\alpha_{3}A_{2}^{2}\right)A_{1}\cos(\omega_{1}t) \approx \left(\alpha_{1} + \frac{3}{2}\alpha_{3}A_{2}^{2}\right)A_{1}\cos(\omega_{1}t). \quad (2.7)$$
Therefore, the signal at the fundamental frequency decreases when the amplitude of the large unwanted signal increases. The second desensitization mechanism is caused by second-order nonlinearity, which causes the low-frequency noise components to be upconverted to the desired frequency [37].

2.2.3.2 Intermodulation

When two sinusoidal test signals with different frequencies are applied to a nonlinear system, the output signal consists of harmonics of the input signal and intermodulation components. When a two-tone input signal $A_1 \cos(\omega_1 t) + A_2 \cos(\omega_2 t)$ is fed into a nonlinear system of (2.5), the third-order intermodulation terms of interest are [38]

$$y_{IMD3}(t) = \frac{3}{4}\alpha_3 A_1^2 A_2 \cos(2\omega_1 t - \omega_2 t) + \frac{3}{4}\alpha_3 A_1 A_2^2 \cos(2\omega_2 t - \omega_1 t).$$
(2.8)

If the amplitudes of the test signals are equal, the amplitude of the intermodulation term depends on the third power of the input signal. In other words, the power of the third-order intermodulation component at the output increases by 3 dB when the level of the input test signals is increased by 1 dB. This is shown in Fig 2.6. The third-order intermodulation intercept point is defined as the point where the extrapolated curves of the wanted signal and third-order intermodulation distortion terms intercept in a log-log-scale.

Similarly, the second-order intermodulation terms of interest are

$$y_{IMD2}(t) = \alpha_2 A_1 A_2 \cos(\omega_1 t - \omega_2 t) + \alpha_2 A_1 A_2 \cos(\omega_2 t - \omega_1 t).$$
(2.9)

The power of the second-order intermodulation component at the output increases by 2 dB when the level of the input test signals is increased by 1 dB. As is shown in Fig 2.6, the second-order intermodulation intercept point is defined as the point where the extrapolated curves of the wanted signal and the second intermodulation distortion terms intercept in a log-log-scale.



Fig 2.6. Definition of the IIP2 and IIP3.

The IIP3 and IIP2 can be calculated as

$$IIP3 = \frac{3}{2}P_{out} - \frac{1}{2}P_{IMD3} - G = P_{in} + \frac{\Delta P_3}{2}$$
(2.10)

and

$$IIP2 = 2P_{out} - P_{IMD2} - G = P_{in} + \Delta P_2, \qquad (2.11)$$

where P_{out} is the output power, P_{in} is the input power of one tone, P_{IMD3} and P_{IMD2} are the powers of the intermodulation products at the output, G is the power gain, and ΔP_3 and ΔP_2 are the differences between the fundamental signal and the intermodulation products. The voltage gain can be used instead of the power gain in (2.10) and (2.11) [31]. For cascaded stages, the input intercept point can be calculated with an equation similar to Friis' noise formula:

$$\frac{1}{IIP3_{tot}} = \frac{1}{IIP3_1} + \frac{G_1}{IIP3_2} + \frac{G_1G_2}{IIP3_3} + \dots + \frac{\prod_{i=1}^{n-1}G_i}{IIP3_n}.$$
(2.12)

In (2.12), $IIP3_n$ and G_n are the input intercept point and power gain of the n^{th} stage. The power gain in (2.12) can be replaced with the square of the voltage gain. More general equations taking into account the selectivity of each stage is presented in [39].

In some cases, the IIP3 requirement needs to be defined by a method other than the conventional intermodulation test. For example, in the WCDMA system, the transmitter signal can leak to the LNA input as a result of finite out-of-band characteristics of the pre-filter. If there is a blocker located between the transmit and reception bands, it is possible that the interfering signal overlaps the wanted signal. Therefore, IIP3 defined for signals with different power is given as

$$IIP3 = \frac{1}{2}P_{\omega 1} + P_{\omega 2} - \frac{1}{2}P_{IMD3}, \qquad (2.13)$$

where $P_{\omega l}$ and $P_{\omega 2}$ are the input powers of the two interferers and P_{IMD3} is the power of the inputreferred third-order intermodulation component [40]. It should be noted that (2.13) is valid only when $\omega_2 > \omega_l$, i.e. when the two interfering signals are at lower frequency than the intermodulation component. Otherwise, the factors of $P_{\omega l}$ and $P_{\omega 2}$ should be interchanged.

2.2.4 Sensitivity and noise

The sensitivity determines the minimum detectable signal level. The signal quality is typically determined by bit or frame error rate. The sensitivity level is determined as

$$S = -174dBm + 10\log(BW) + SNR_{\min} + NF, \qquad (2.14)$$

where -174 dBm is the available noise power from the source at a temperature of 290 K, BW is the channel bandwidth, SNR_{min} is the minimum signal-to-noise ratio, and NF is the receiver noise figure. The SNR_{min} depends on the required bit error rate and the used modulation method. General level information related to requirements of SNR_{min} in different communication systems can be found from [31], for example.

2.2.4.1 Noise figure

The noise performance of a single block is characterized by a noise factor (F) or a noise figure ($NF = 10\log F$). The noise factor determines how much the given circuit or system degrades the signal-to-noise ratio and is defined as

$$F = \frac{SNR_{in}}{SNR_{out}} = \frac{\text{total output noise}}{\text{total output noise due to the source}},$$
 (2.15)

where SNR_{in} and SNR_{out} are the signal-to-noise ratios at the input and output, respectively. For a single block, the noise factor can be derived as

$$F = \frac{V_s^2 / (4k_B T \cdot BW \cdot R_s)}{V_{out}^2 / e_{n,out}^2} = \frac{e_{n,out}^2}{4k_B T \cdot BW \cdot R_s \left(\frac{V_{out}}{V_s}\right)^2},$$
(2.16)

where V_S is the source voltage, V_{out} and $e_{n,out}$ are the output signal and output noise voltages, k_B is Boltzmann's constant, BW is the bandwidth, and R_S is the source resistance. The Friis' equation for the noise factor of cascaded blocks is given as [41]

$$F_{tot} = F_1 + \frac{F_2 - 1}{G_1} + \frac{F_3 - 1}{G_1 G_2} + \dots + \frac{F_n - 1}{\prod_{i=1}^{n-1} G_i},$$
(2.17)

where $F_1...F_n$ are the noise factors of the successive blocks of the receiver and $G_1...G_n$ are their available power gains. Equation (2.17) is not suitable for most of the IC receivers, since power-matched interfaces are required. Therefore, the noise contribution can be rewritten in the form of

$$F_{tot} = F_1 + \frac{F_2 - 1}{A_{0,1}^2 A_{\nu_1}^2} + \frac{F_3 - 1}{A_{0,1}^2 A_{\nu_2}^2} + \dots + \frac{F_n - 1}{\prod_{i=1}^{n-1} A_{i-1,i}^2 A_{\nu_i}^2},$$
(2.18)

where $A_{\nu n}$ in the voltage gain of the n^{th} stage to high-impedance load and $A_{n-2,n-1}$ is the voltage division ratio between stages n-2 and n-1 [31]:

$$A_{n-2,n-1} = \frac{Z_{in,n-1}}{Z_{in,n-1} + Z_{out,n-2}}.$$
(2.19)

In (2.19), $Z_{in,n-1}$ and $Z_{out,n-2}$ are the input and output impedances of the cascaded stages. In practical IC realizations $Z_{in,n-1} >> Z_{out,n-2}$ and therefore (2.19) is close to unity. As a result, the total noise factor of a typical receiver can be given with voltage gains.

As is shown in (2.16), the noise figure cannot be determined without the source impedance level. Except the LNA, the inputs of the block used in DCR are not matched and therefore a better measure for noise performance is the input referred noise voltage [40]. In addition, the possible noise from the image band of the mixers should be taken into account [39]. Anyway, if Friis' formula is being used, the noise factors of blocks after the LNA can be defined as inputreferred noise voltage scaled to the noise in a hypothetical 50- Ω resistor [42]. Whichever cascaded noise factor formula is used, the overall noise contribution of the whole receiver frontend can be quickly estimated. The noise contribution of the mixers and baseband is reduced when the (voltage) gain of the LNA is increased. Therefore, the gain of the first blocks trade-off between the cascaded linearity and noise performance of the whole system.

2.2.4.2 Noise factor definitions for mixers

Because the mixer translates the signal and noise from both the main and image frequencies, two noise factors are defined for mixers, single-sideband (SSB) noise figure and double-sideband (DSB) noise figure. Which noise figure measure should be used depends mainly on the receiver architecture. The SSB noise figure is applicable to the architectures where the RF signal is downconverted to IF frequency, which is higher than half of the system (image-reject filter) bandwidth [7]. Therefore, the SSB noise figure is applicable for superheterodyne receivers, low-IF receivers, and for the first downconversion stage of the wideband-IF receiver. The DSB noise figure is applicable to the direct-conversion receiver. In [43] it is shown that mixers in the heterodyne architecture have twice as many noise contributors as that in the direct-conversion architecture. Thus, the difference between SSB and DSB noise factors can be approximated with

$$\underbrace{\left(1+\frac{N_{SSB}}{N_{S}}\right)=\left(1+\frac{2N_{DSB}}{N_{S}}\right)}_{F_{SSB}} < 2\underbrace{\left(1+\frac{N_{DSB}}{N_{S}}\right)}_{F_{DSB}} \Longrightarrow F_{SSB} < 2F_{DSB},$$
(2.20)

where N_{SSB} and N_{DSB} are the input-referred single-sideband and double-sideband noise power of the mixer, F_{SSB} and F_{DSB} are the single-sideband and double-sideband noise factors, respectively, and N_S is the noise from the source. If N_{DSB} is much larger than N_S , the SSB noise figure is approximately 3 dB higher than the DSB noise figure [43].

2.3 MOS transistor small-signal model and noise sources

Since some circuit analysis is presented later in this thesis, the MOS transistor small-signal model used in calculations is briefly described. A simple small-signal model of MOS transistor is depicted in Fig 2.7. The discussion of different MOS transistor models is not within the scope of this thesis. That issue is covered in the literature, for example in [44] and [45]. Since the circuits analyzed in this work typically include several transistors, a simple enough model is needed to achieve useful and understandable equations for RF IC design. Therefore, in this thesis, the MOS transistors are replaced with a simple unilateral small-signal model having only gate-source capacitance C_{gs} and transconductance g_m , i.e. C_{gd} capacitor, gate resistance r_g , and output (drain-source) resistance r_o are omitted for the sake of simplicity. Several equations derived in this thesis were validated with simulations with actual transistor models provided by the foundry. Despite the simplicity of the used MOS model, the discrepancy between the theory and simulations was found to be rather small when operating at the frequencies of the designed circuits presented in thesis.

The major noise sources are also included into the MOS model shown Fig 2.7. Although the in-depth noise analysis of the LNAs and mixers is mostly omitted from this thesis, an understanding of the basics of the MOS noise sources is crucial in RF IC design. They are therefore briefly described in the following.



Fig 2.7. Simple MOS transistor model including the major noise sources.

At RF frequencies, the noise is dominated by the drain and gate current sources, which are due to the thermal fluctuations of channel charge in MOSFET [46]. At low frequencies, the 1/*f* noise (flicker noise) is the dominant noise source. In downconversion mixers, the 1/*f* noise increases the low-frequency noise and in and the VCO it increases phase noise due to upconversion. In LNAs, the flicker noise can be neglected, because the operational frequencies are well above the noise corner frequency. In addition, in modern short-channel MOSFETs, the gate resistance increases the gate current noise remarkably. The spectral noise current densities of the three noise sources depicted in Fig 2.7 are

$$i_d^2 = 4k_B T g_{d0} \gamma \Delta f , \qquad (2.21)$$

$$\overline{i_g^2} = 4k_B T \delta g_g \Delta f , \qquad (2.22)$$

$$v_g^2 = 4k_B T r_g \Delta f . ag{2.23}$$

 g_{d0} is the MOSFET output conductance at zero drain-source bias and γ and δ are coefficients describing the magnitude of the noise power. The theoretical long-channel value for γ and δ in saturation are 2/3 and 4/3, respectively [47]. For short-channel transistors, γ increases partly due to thermal noise of parasitic resistances and partly due to short channel effects such as velocity saturation and channel length modulation [46]. For short channel transistors, δ enhances due to the effect of the gate resistance. The analytical derivation for γ and δ for short channel devices is complex if not impossible to obtain [48].

Occasionally, the g_{d0} is replaced by g_m in saturation. The relation between g_{d0} and g_m is given by

$$\alpha = \frac{g_m}{g_{d0}}, \qquad (2.24)$$

which is unity for long-channel devices, but can be much less than one for short-channel devices [49]. The g_g in (2.22) is given as

$$g_g = \frac{\omega^2 C_{g_s}^2}{5g_{d0}} \,. \tag{2.25}$$

Due to the same physical origin, drain and induced gate current noise sources are partially correlated. For long-channel devices, the correlation factor c is

$$c = \frac{\overline{i_g i_d^*}}{\sqrt{i_g^2 i_d^2}} \approx -j0.395 , \qquad (2.26)$$

when the noise source polarities are as indicated in Fig 2.7. For short-channel transistors some discrepancy from theoretical value has been observed [46].

The gate resistance is a significant contributor to the measured gate current noise in shortchannel devices [46], [50]. The gate resistance for a single gate finger connected at both ends is

$$r_{\rm g} = \frac{1}{12} \frac{R_{sh}}{n^2} \frac{W}{L} \,, \tag{2.27}$$

where R_{sh} is the silicide sheet resistance and *n* is the number of fingers. In addition, the resistance of a metal1-to-polysilicon via and the silicide-to-polysilicon contact gives rise to the gate resistance [46]. The gate resistance noise is heavily dependent upon the transistor layout. The noise can be decreased by optimizing the layout by using transistors with multiple fingers (known as interdigitating or folding) and by connecting the gate from both ends. If the gate was contacted only at the one end, the factor of the term in (2.27) would be $\frac{1}{3}$ instead of $\frac{1}{12}$. Too much folding, however, increases the wiring parasitic capacitances, which lowers the cut-off frequency of the transistor.

2.4 Overview of short-range radio systems

There exist several wireless personal area network (WPAN) standards, such as Bluetooth (BT) [51], ZigBee [52], and Ultrawideband (UWB) [53]. These have quite different requirements in terms of operational frequencies and bandwidths, which affect the overall system performance and design challenges. According to Shannon's theorem, the system capacity C is a logarithmic function of signal-to-noise ratio (SNR) but a linear function of bandwidth (BW), i.e.

$$C = BW \log_2(1 + SNR), \tag{2.28}$$

when a channel is perturbed by white Gaussian noise [54]. Therefore, increasing the system bandwidth has significant improvement on system capacity. UWB takes the advantage of wide operational bandwidth to achieve the high capacity. On the other hand, with modest BW, a reasonable capacity can be still achieved if the signal-to-noise ratio is relaxed. This is common for sensor networks, which can achieve several-year autonomous operation by trading the system requirements with low power consumption. A short introduction to sensor applications and UWB system, which are the targeted systems for the experimental circuits of this thesis, is provided next.

2.4.1 Wireless sensor systems

Sensors are encountered in every day life. Sensors typically monitor temperature, pressure, humidity, acceleration etc. These small devices could also benefit from wireless connection to a personal computer or a mobile terminal, for example. The envisioned use cases for wireless sensors are wide. They can provide monitoring and control capability for a number of applications such as transportation, manufacturing, biomedical, environmental monitoring, and safety and security systems [55]. For such devices, an occasional wireless interconnection with the maximum data rate of only a few tens or a few hundreds of kilobits per second is needed. In addition, the maximum connection range of a few meters only is usually required.

2.4.1.1 General requirements

The extremely low power consumption and material costs are the key requirements for sensor radio modules. To achieve over-1-year stand-alone operation, extremely small active and standby currents are required. Therefore, the sensors are at the sleep mode for most of the time, with the duty cycle in the order of 0.1 to 1 percent [56]. In addition, the wake-up time should be minimized [P1]. One-dollar bill material costs on a timescale of a few years require a small device with a minimum number of external components. A high integration level can be achieved by using, for example, the direct-conversion receiver or low-IF architecture. The demand for low cost requires the use of CMOS technology without additional process options, like high-quality resistors or capacitors. Currently, the speed of the transistors in modern deep submicron CMOS processes is not the limiting factor. However, the challenges for receiver (RX) design emerge from the poor "analog" performance of the transistor. For example, the available gain from a single transistor is only moderate. In addition, in a CMOS DCR, the flicker noise increases the NF significantly and must be considered already in the downconversion mixers. A higher NF than, for example, in cellular systems, can be tolerated because the sensors are typically short-range devices [57]. Because the bit rate is low and operating range short, the sensitivity of the receiver can be relaxed compared to typical cellular radios [57]. Due to the unique nature of sensor networks, new circuit and system solutions have been presented recently [56] - [60].

Quite a few of the current short-range radio systems fulfill requirements set by sensor networks in a satisfactory manner. Standard Bluetooth consumes too much power and suffers from relatively long device discovery and connection setup times [P1].² The IEEE 802.15.4 (ZigBee) is a standard, which overcomes some of the limitations that Bluetooth encounters. However, it does not support dynamic topologies very well, while this is an essential requirement for mobile sensor solutions and has only slight improvements in power consumption compared to standard Bluetooth [P1]. The demonstrator receivers presented in [P1] and [P2] are targeted for a system that uses Bluetooth as a basis, but has slight changes made in the BT radio parameters, as is briefly described next.

2.4.1.2 Overview of low-end extension for Bluetooth (BT LEE)

In typical radios, the current consumption increases along with the operational frequency. For that reason, operating at the sub-1-GHz frequency ISM band would be beneficial to reach the minimal power consumption [42], [56], [61]. However, the current mobile terminals usually have Bluetooth as the short-range wireless connection. Thus, to avoid the need for yet another radio, Bluetooth with slight changes in the radio parameters was utilized to overcome the shortcomings that it has from the sensor networks viewpoint [P1]. Some of the BT radio parameters, such as data rate, receiver sensitivity, and transmitted power, are relaxed to enable very low-power implementation. Table 2.1 shows the basic radio parameters for BT LEE. The details can be found in [62].

² It should be noted that an ultra low power Bluetooth specification has been recently created. [Online] Available: http://www.bluetooth.com/Bluetooth/Learn/Technology/lowpower

Physical bit rate	333 kbps		
Frequency band	2.4 GHz (ISM band)		
Modulation	2GFSK (h = 2.4, BT = 0.5)		
Duplex	TDD		
Co-existence of multiple devices			
Connection setup channel	CSMA		
Data delivery	FDMA		
Jamming avoidance	FDMA		

Table 2.1. Basic BT LEE radio parameters.

2.4.2 Ultrawideband

Recently, integrated ultrawideband RF circuits have received considerable attention [63] - [69]. According to the Federal Communications Commission (FCC), the unlicensed frequency spectrum allocated for UWB systems is 3.1 - 10.6 GHz. According to WiMedia UWB standard³, the UWB band is further divided into several narrower sub-bands each occupying a bandwidth of 528 MHz [53]. As is shown in Fig 2.8, the sub-bands are grouped such that the 12 first sub-bands form four band groups (BGs), each consisting of three sub-bands, and the last two sub-bands form a fifth BG. During the time this thesis was written, in the first ECMA-368 standard version, the bandgroup-1 (BG1) formed by the first three sub-bands was mandatory for UWB communications and the remaining BGs were optional to expand the system capability. At the moment, in the second ECMA-368 standard version, there are no mandatory BGs [53]. It should be noted that there is also a competing approach for the OFDM-based UWB communications that is based on the impulse radio technique [64]. The circuit techniques presented in this thesis, however, are towards the OFDM based solution.



Fig 2.8. WiMedia UWB band group allocation.

2.4.2.1 Target applications

The primary driving factor for UWB systems is to provide high-data-rate communication for short ranges. The target range is 10 meters for 110 Mbps. The range can be reduced for higher bit rates [63]. As envisioned in use cases, UWB is targeted to become a general solution for cable replacement between consumer electronic equipment or as a wireless link between peripheral device to laptop or mobile phone. In addition to high-throughput applications, UWB

³ Formerly known as Multiband OFDM Alliance (MBOA) standard.

is also being considered for low-data-rate sensor networks [64]. The impulse-based UWB is accepted as an alternative physical layer for IEEE 802.15.4a standard [65].

2.4.2.2 Multisystem challenges

Due to the wide operational band of UWB, interference with other existing radio systems is unavoidable. Therefore, the challenges set by other RF systems are briefly discussed. The FCC requires that the power spectral density (PSD) of UWB devices measured in 1-MHz bandwidth must not exceed the -41.25 dBm level, which is low enough not to cause interference to other wireless systems sharing the same bandwidth. This presents a serious challenge to UWB, since the other systems operating at the same band typically have much higher transmit powers. An example of a hostile environment is a mobile handset, where several other radios can be simultaneously on. The transmit powers of cellular radios can be up to +30 dBm, which is several orders of magnitude higher than UWB transmitters are permitted [63]. Fig 2.9 shows the frequency allocation for the most common cellular systems operating within 800 MHz to 8 GHz. As shown in (2.13), despite pre-filtering, non-overlapping systems like GSM900 and 2.4-GHz WLAN or Bluetooth can corrupt UWB signal reception as a result of intermodulation at BG1 frequencies. Furthermore, a WiMAX interferer (not shown in Fig 2.9 due to regulatory requirements for the RF spectrum usage in different countries) is an in-band blocker in BG1, having a larger power level than the wanted signal [69]. Therefore, UWB receivers are susceptible to strong interferers caused by other active radios.

To expand the UWB system capability, the additional BGs are considered in addition to the mandatory BG1. The frequency spectrum allocated for the whole UWB system overlaps with IEEE 802.11a bands, which are located at 5.15 - 5.875 GHz area. As a result, the strong interference caused by Wi-Fi radios in the vicinity or within the same terminal can desensitize the UWB receiver. Therefore, the coexistence of Wi-Fi radios prevents the effective use of BG2, which limits the number of sub-bands that a MB-OFDM UWB radio can cover in practice [68]. Although high operational frequency sets its own challenges, BG3 is a more viable solution for a mobile handset due to less congested frequency usage. The utilization of BG3 is not trivial, since path loss and complexity of the hardware increase along with higher operational frequencies. The circuit design and experimental results of a dual-band RF front-end designed for WiMedia UWB receiver operating in BG1 and BG3 are presented in Section 6.6.



Fig 2.9. Frequency spectrum allocation for the most common wireless systems between 800 MHz and 8 GHz.

References

- P. Orsatti, F. Piazza, and Q. Huang, "A 20-mA-receive, 55-mA-transmit, single-chip GSM transceiver in 0.25-μm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 12, Dec. 1999, pp. 1869-1880.
- [2] W. H. W. Tuttlebee, "Software-defined radio: Facets of a developing technology," *IEEE Personal Communications*, vol. 6, no. 2, April 1999, pp. 38-44.
- [3] S. Lindfors, A. Pärssinen, and K. A. I. Halonen, "A 3-V 230-MHz CMOS decimation subsampler," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 50, no. 3, March 2003, pp. 105-117.
- [4] E. Armstrong, "Some recent developments of regenerative circuits," *Proc. of the IRE*, vol. 10, no. 4, Aug. 1922, pp. 244-260.
- [5] F. X. Moncunill-Geniz, P. Palà-Schönwälder, and O. Mas-Casals, "A generic approach to the theory of superregenerative reception," *IEEE Trans. on Circuits and Systems – I: Regular Papers*, vol. 52, no. 1, Jan. 2005, pp. 54-70.
- [6] P. Favre, N. Joehl, A. Vouilloz, P. Deval, C. Dehollain, and M. J. Declercq, "A 2-V 600-μA 1-GHz BiCMOS super-regenerative receiver for ISM applications," *IEEE J. of Solid-State Circuits*, vol. 33, no. 12, Dec. 1998, pp. 2186-2196.
- [7] B. Razavi, *RF Microelectronics*, Prentice-Hall, 1998, p. 335.
- [8] F. M. Colebrook, "Homodyne," *Wireless World and Radio Review*, vol. 13, 1924, pp. 645-648.
- [9] D. Tucker, "The history of the homodyne and synchrodyne," *Journal of British Institution of Radio Engineers*, vol. 14, April 1954, pp. 143-154.
- [10] J. F. Wilson, R. Youell, T. H. Richards, G. Luff, and R. Pilaski, "A single-chip VHF and UHF receiver for radio paging," *IEEE Journal of Solid-State Circuits*, vol. 26, no. 12, Dec. 1991, pp. 1944-1950.
- [11] J. Rogin, I. Kouchev, G. Brenna, D. Tschopp, and Q. Huang, "A 1.5V 45mW directconversion WCDMA receiver IC in 0.13µm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 12, Dec. 2003, pp. 2239-2248.
- [12] H. Waite, P. Ta, J. Chen, H. Li, M. Gao, S. C. Chang, W. Redman-White, O. Charlton, Y. Fan, R. Perkins, D. Brunel, E. Soudee, N. Lecacheur, and S. Clamagirand, "A CDMA2000 zero-IF receiver with low-leakage integrated front-end," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 7, July 2004, pp. 1175-1179.
- [13] B. Razavi, "Design considerations for direct-conversion receivers," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 44, no. 6, June 1997, pp. 428-435.
- [14] P. Sivonen, J. Tervaluoto, N. Mikkola, and A. Pärssinen, "A 1.2-V RF front-end with on-chip VCO for PCS 1900 direct conversion receiver in 0.13-µm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 41, no. 2, Feb. 2006, pp. 384-394.
- [15] A. A. Abidi, "Direct-conversion radio transceivers for digital communications," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 12, Dec. 1995, pp. 1399-1410.

- [16] W. Sheng, B. Xia, A. E. Emira, C. Xin, A. Y. Valero-López, S. T. Moon, and E. Sánchez-Sinencio, "A 3-V, 0.35-µm CMOS Bluetooth receiver IC," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 1, Jan. 2003, pp. 30-42.
- [17] A. Pärssinen, J. Jussila, J. Ryynänen, L. Sumanen, and K. A. I. Halonen, "A 2 GHz wide-band direct conversion receiver for WCDMA applications," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 12, Dec. 1999, pp. 1893-1903.
- [18] K. Kivekäs, A. Pärssinen, and K. A. I. Halonen, "Characterization of IIP2 and dcoffsets in transconductance mixers," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 11, Nov. 2001, pp. 1028-1038.
- [19] E. Armstrong, "The super-heterodyne Its origin, development, and some recent improvements," *Proc. of the IRE*, vol. 12, no. 5, Oct. 1924, pp. 539-552.
- [20] J. C. Rudell, J.-J. Ou, T. B. Cho, G. Chien, F. Brianti, J. A. Weldon, and P. R. Gray, "A 1.9-GHz wide-band IF double conversion CMOS receiver for cordless telephone applications," *IEEE J. of Solid-State Circuits*, vol. 32, no. 12, Dec. 1997, pp. 2071-2088.
- [21] J. Crols and M. S. J. Steyaert, "Low-IF topologies for high-performance analog front ends of fully integrated receivers," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 45, no. 3, March 1998, pp. 269-282.
- [22] F. Behbahani, J. C. Leete, Y. Kishigami, A. Roithmeier, K. Hoshino, and A. A. Abidi, "A 2.4-GHz low-IF receiver for wideband WLAN in 0.6-µm CMOS – Architecture and front-end," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1908-1916.
- [23] S. Tadjpour, E. Cijvat, E. Hegazi, and A. A. Abidi, "A 900-MHz dual-conversion low-IF GSM receiver in 0.35-µm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 12, Dec. 2001, pp. 1992-2002.
- [24] D. K. Shaeffer, A. R. Shahani, S. S. Mohan, H. Samavati, H. R. Rategh, M. del Mar Hershenson, M. Xu, C. P. Yue, D. J. Eddleman, and T. H. Lee, "A 115-mW, 0.5-μm CMOS GPS receiver with wide dynamic-range active filters," *IEEE Journal of Solid-State Circuits*, vol. 33, no. 12, Dec. 1998, pp. 2219-2231.
- [25] M. S. J. Steyaert, J. Janssens, B. De Muer, M. Borremans, and N. Itoh, "A 2-V CMOS cellular transceiver front-end," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1895-1907.
- [26] J. Crols and M. S. J. Steyaert, "A single-chip 900 MHz CMOS receiver front-end with a high performance low-IF topology," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 12, Dec. 1995, pp. 1483-1492.
- [27] S. Mirabbasi and K. Martin, "Classical and modern receiver architectures," *IEEE Communications Magazine*, vol. 38, no. 11, Nov. 2000, pp. 132-139.
- [28] D. K. Weaver, "A third method of generation and detection of single-sideband signals," *Proc. of the IRE*, vol. 44, no. 12, Dec. 1956, pp. 1703-1705.
- [29] S. Wu and B. Razavi, "A 900-MHz/1.8-GHz CMOS receiver for dual-band applications," *IEEE J. of Solid-State Circuits*, vol. 33, no. 12, Dec. 1998, pp. 2178-2185.

- [30] A. Springer, L. Maurer, and R. Weigel, "RF system concepts for highly integrated RFICs for W-CDMA mobile radio terminals," *IEEE Trans. on Microwave Theory and Techniques*, vol. 50, no. 1, Jan. 2002, pp. 254-267.
- [31] A. Pärssinen, *Direct conversion receivers in wide-band systems*, Doctoral Thesis, Helsinki University of Technology, Espoo, 2000, p. 235.
- [32] A. Loke and F. Ali, "Direct conversion radio for digital mobile phones Design issues, status, and trends," *IEEE Trans. on Microwave Theory and Techniques*, vol. 50, no. 11, Nov. 2002, pp. 2422-2435.
- [33] P. Sivonen and A. Pärssinen, "Analysis and optimization of packaged inductively degenerated common-source low-noise amplifiers with ESD protection," *IEEE Trans. on Microwave Theory and Techniques*, vol. 53, no. 4, April 2005, pp. 1304-1313.
- [34] G. Girlando and G. Palmisano, "Noise figure and impedance matching in RF cascode amplifiers," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 46, no. 11, Nov. 1999, pp. 1388-1396.
- [35] D. Leenaerts, J. van der Tang, and C. Vaucher, *Circuit design for RF transceivers*, Springer, 2001, p. 344.
- [36] A. Rofougaran, G. Chang, J. J. Rael, J. Y.-C. Chang, M. Rofougaran, P. J. Chang, M. Djafari, J. Min, E. W. Roth, A. A. Abidi, and H. Samueli, "A single-chip 900-MHz spread-spectrum wireless transceiver in 1-µm CMOS Part II: Receiver design," *IEEE J. of Solid-State Circuits*, vol. 33, no. 4, April 1998, pp. 535-547.
- [37] R. G. Meyer and A. K. Wong, "Blocking and desensitization in RF amplifiers," *IEEE J. of Solid-State Circuits*, vol. 30, no. 8, Aug. 1995, pp. 944-946.
- [38] K. A. Simons, "The decibel relationship between amplifier distortion products," *Proc. of the IEEE*, vol. 58, no. 7, July 1970, pp. 1071-1086.
- [39] C. R. Iversen and T. E. Kolding, "Noise and intercept point calculation for modern radio receiver planning," *IEE Proc. of Communications*, vol. 148, no. 4, Aug. 2001, pp. 255-259.
- [40] J. Ryynänen, *Low-noise amplifiers for integrated multi-mode direct-conversion receivers*, Doctoral Thesis, Helsinki University of Technology, Espoo, 2004, p. 138.
- [41] H. Friis, "Noise figures of radio receivers," *Proc. of the IRE*, vol. 32, no. 7, July 1944, pp. 419-422.
- [42] S. Mahdavi and A. A. Abidi, "Fully integrated 2.2-mW CMOS front end for a 900-MHz wireless receiver," *IEEE J. of Solid-State Circuits*, vol. 37, no. 5, May 2002, pp. 662-669.
- [43] K. L. Fong and R. G. Meyer, "Monolithic RF active mixer design," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 46, no. 3, March 1999, pp. 231-239.
- [44] C. C. Enz and Y. Cheng, "MOS transistor modeling for RF IC design," *IEEE J. of Solid-State Circuits*, vol. 35, no. 2, Feb. 2000, pp. 186-201.
- [45] T. Ytterdal, Y. Cheng, and T. A. Fjeldly, *Device modeling for analog and RF CMOS circuit design*, Wiley, 2003, p. 306.

- [46] A. J. Scholten, L. F. Tiemeijer, R. van Langevelde, R. J. Havens, A. T. A. Zegers-van Duijhoven, and V. C. Venezia, "Noise modeling for RF CMOS circuit simulation," *IEEE Trans. on Electron Devices*, vol. 50, no. 3, March 2003, pp. 618-632.
- [47] A. Van der Ziel, *Noise in solid-state devices and circuits*, John Wiley & Sons, 1986, p. 306.
- [48] Y. Lin, M. Obrecht, and T. Manku, "RF noise characterization of MOS devices for LNA design using a physical-based quasi-3-D approach," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 10, Oct. 2001, pp. 972-984.
- [49] D. K. Shaeffer and T. H. Lee, "A 1.5-V, 1.5-GHz CMOS low noise amplifier," *IEEE J. of Solid-State Circuits*, vol. 32, no. 5, May 1997, pp. 745-759.
- [50] B. Razavi, R.-H. Yan, and K. F. Lee, "Impact of distributed gate resistance on the performance of MOS devices," *IEEE Trans. on Circuits and Systems I: Fundamental Theory and Applications*, vol. 41, no. 11, Nov. 1994, pp. 750-754.
- [51] Bluetooth specification version 2.1, 26 July 2007, [Online]. Available: http://www.bluetooth.com/Bluetooth/Learn/Technology/Specifications/
- [52] ZigBee specification release 13, Document 053474r13, 1. December 2006, [Online]. Available: http://www.zigbee.org/en/spec_download/download_request.asp
- [53] High rate ultra wideband PHY and MAC standard, ECMA standard 368, 2nd Ed, Dec.
 2007, [Online]. Available: http://www.ecma-international.org/publications/standards/Ecma-368.htm
- [54] C. E. Shannon, "A mathematical theory of communication," *Bell System Technical Journal*, vol. 27, July and Oct. 1948, pp. 379-423 and 623-656.
- [55] K. Bult, A. Burstein, D. Chang, M. Dong, M. Fielding, E. Kruglick, J. Ho, F. Lin, T. H. Lin, W. J. Kaiser, H. Marcy, R. Mukai, P. Nelson, F. L. Newburg, K. S. J. Pister, G. Pottie, H. Sanchez, K. Sohrabi, O. M. Stafsudd, K. B. Tan, G. Yung, S. Xue, and J. Yao, "Low power system for wireless microsensors," *Int. Symposium on Low Power Electronics and Design (ISLPED'96)*, Monterey, CA, 12-14 Aug. 1996, pp. 17-21.
- [56] C. C. Enz, A. El-Hoiydi, J.-D. Decotignie, and V. Peiris, "WiseNET: An ultralowpower wireless sensor network solution," *IEEE Computer*, vol. 37, no. 8, Aug. 2004, pp. 62-70.
- [57] G. Asada, M. Dong, T. S. Lin, F. Newberg, G. Pottie, and W. J. Kaiser, "Wireless integrated network sensors: Low power systems on a chip," *Eur. Solid-State Circuits Conference (ESSCIRC'98)*, The Hague, The Netherlands, 22-24 Sept. 1998, pp. 9-16.
- [58] J. M. Rabaey, J. Ammer, T. Karalar, L. Suetfei, B. Otis, M. Sheets, and T. Tuan, "Picoradios for wireless sensor networks: The next challenge in ultra low power design," *IEEE Int. Solid-State Circuits Conference (ISSCC'02)*, San Francisco, CA, 3-7 Feb. 2002, pp. 200-201.
- [59] B. P. Otis, Y. H. Chee, R. Lu, N. M. Pletcher, and J. M. Rabaey, "An ultra-low power MEMS-based two-channel transceiver for wireless sensor networks," *Symposium on VLSI Circuits (VLSI'04)*, Honolulu, HI, 17-19 June 2004, pp. 20-23.
- [60] T.-H. Lin, H. Sanchez, R. Rofougaran, and W. J. Kaiser, "CMOS front-end components for micropower RF wireless systems," *Int. Symposium on Low Power Electronics and Design (ISLPED'98)*, Monterey, CA, 10-12 Aug. 1998, pp. 11-15.

- [61] A.-S. Porret, T. Melly, D. Python, C. C. Enz, and E. A. Vittoz, "An ultralow-power UHF transceiver integrated in a standard digital CMOS process: Architecture and receiver," *IEEE J. of Solid-State Circuits*, vol. 36, no. 3, March 2001, pp. 452-466.
- [62] M. Honkanen, A. Lappeteläinen, and K. Kivekäs, "Low end extension for Bluetooth," in *Proc. IEEE Radio and Wireless Conf. (RAWCON'04)*, Atlanta, GA, 19-22 Sept. 2004, pp. 199-202.
- [63] A. Batra, J. Balakrishnan, G. R. Aiello, J. R. Foerster, and A. Dabak, "Design of a multiband OFDM systems for realistic UWB channel environments," *IEEE Trans. on Microwave Theory and Techniques*, vol. 52, no. 9, Sept. 2004, pp. 2123-2138.
- [64] D. Cabric, I. D. O'Donnell, M. S.-W. Chen, and R. W. Brodersen, "Spectrum sharing radios," *IEEE Circuits and Systems Magazine*, 2nd Quarter 2006, pp. 30-45.
- [65] IEEE 802.15 WPAN Low Rate Alternative PHY Task Group 4 a (TG4a), [Online]. Available: http://www.ieee802.org/15/pub/TG4a.html
- [66] B. Razavi, T. Aytur, C. Lam, F.-R. Yang, K.-Y. Li, R.-H. Yan, H.-C. Kang, C.-C. Hsu, and C.-C. Lee, "A 0.13µm CMOS UWB transceiver," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2555-2562.
- [67] C. Sandner, S. Derksen, D. Draxelmayr, S. Ek, V. Filimon, G. Leach, S. Marsili, D. Matveev, K. L. R. Mertens, F. Michl, H. Paule, M. Punzenberger, C. Reindl, R. Salerno, M. Tiebout, A. Wiesbauer, I. Winter, and Z. Zhang, "A WiMedia/MBOA-compliant CMOS RF transceiver for UWB," *IEEE Journal of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2787-2794.
- [68] A. Valdes-Garcia, C. Mishra, F. Bahmani, J. Silva-Martinez, and E. Sánchez-Sinencio, "An 11-band 3–10 GHz receiver in SiGe BiCMOS for multiband OFDM UWB communication," *IEEE J. of Solid-State Circuits*, vol. 42, no. 4, April 2007, pp. 935-948.
- [69] S. Lo, I. Sever, S.-P. Ma, P. Jang, A. Zou, C. Arnott, K. Ghatak, A. Schwartz, L. Huynh, V. T. Phan, and T. Nguyen, "A dual-antenna phased-array UWB transceiver in 0.18-μm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2776-2786.

3 Low-noise amplifier design

The subject of this chapter is the design of low-noise amplifiers (LNAs) for wireless applications. First, narrowband LNAs based on inductively degenerated common-source (IDCS) and common-gate (CG) topologies are discussed. Then, in Section 3.3, several LNA topologies for wideband applications are presented and analyzed. The main emphasis is on input matching and insertion gain bandwidth analysis and optimization. Compared with the literature, the major part of the analysis and equations presented in that section is new. The effect of the input pad is briefly discussed in Section 3.4 and a method to neutralize its effect is proposed. Finally, LNA load structures for narrowband and wideband applications are presented in Section 3.5.

A wide variety of different LNA topologies is presented in the literature [1]. The purpose is not to present and compare all of them in this thesis. The focus is on the most relevant and the most-used LNA topologies utilized in wireless applications. For example, distributed LNAs are omitted due to their high current consumption and large layout requirement.

3.1 General LNA design aspects

The LNA is usually the first amplifying stage in a receiver. According to Friis' equation, the LNA sets the minimum noise figure of the receiver. In addition, it should provide enough gain to overcome the noise of subsequent stages, which may otherwise degrade the receiver sensitivity. Simultaneously, the gain of the LNA should not overload the following stages and compromise the dynamic range. The bandwidth of the LNA should be large enough to cover the whole reception band with some design margin but narrow enough to perform some pre-filtering to suppress unwanted interferers. Typically, the linearity performance of the whole front-end or receiver is dominated by the stages that follow the LNA. However, the linearity of the LNA should be adequate to tolerate large blockers and not to produce unwanted intermodulation tones in the reception band. Sufficient input matching performance is required because the properties of the filter preceding the LNA will degrade if the input of the LNA is not properly matched to a certain impedance. As a result, to design a proper LNA, there are several figures-of-merit to be fulfilled. Moreover, some of the requirements are difficult to achieve simultaneously.

The choice between single-ended and balanced structures can already be made in the LNA input. Balanced topology gives immunity against the substrate and supply noise and interferers, but it consumes more current to achieve similar noise figures and gains compared to its single-ended counterpart. LNA typically includes one or two on-chip inductors, which dominate the overall layout area. Since differential coils are available in most of the design kits, the layout area difference between single-ended and balanced LNAs is small. Of course, keeping the number of on-chip inductors low or even zero, as is presented in, for example, [2] and [3], is essential to achieve a small layout area and to keep floor-planning as simple as possible.

Because the LNA is usually based on structures where there are only one or two stacked transistors, the minimum supply voltage of approximately 1 - 1.5 volts is easily achieved with current process technologies. Therefore, the ever-lowering supply voltage typically is not the

main problem of the LNA, but rather the problem concerns other blocks like the downconversion mixer, where more devices are stacked.

In addition to the basic design targets, the LNA can offer several other possibilities such as variable gain or the LNA can include adjustability to make it capable of operating in multisystem applications. The implemented LNAs in this work have generally been a part of a single-system RF front-end. Therefore, the issues related to the design of a multi-mode LNA are not included in this thesis. The design of variable gain methods and issues related to multi-mode LNAs have been covered in [4], for example. In addition, LNAs exploiting thermal noise cancelling have been introduced in the literature [5] - [7]. Noise cancelling LNAs, however, deserve more attention than this thesis can offer.

3.2 Narrowband LNAs

Typically, sufficient performance for narrowband wireless applications is achieved either with inductively degenerated common-source or common-gate LNA topologies, both of which are discussed in this section.

3.2.1 Inductively degenerated common-source LNA

The inductively degenerated common-source (IDCS) amplifier shown in Fig. 3.1a [8] is one of the most commonly used CMOS LNA topologies. The detailed analysis of this topology has been given in several publications [9] – [14]. It has the best noise figure and it provides both current and voltage gain, thereby reducing the noise contribution of the following stages [10]. The equivalent circuit is presented in Fig. 3.1b. The transistor M_1 is replaced with a simple small signal model having only gate-source capacitance C_{gs} and transconductance g_m . An extra C_d capacitance can be added parallel with C_{gs} of the input device, as shown in Fig. 3.1a, to optimize the noise performance [14]. However, when a simple transistor model is assumed, C_d does not affect the calculations, i.e. the C_{gs} in the following figures and formulas can be replaced by the sum of C_{gs} and C_d .



Fig. 3.1. a) Inductively degenerated common-source LNA, b) its simple small-signal equivalent circuit.

3.2.2 Input matching

For the simple input stage presented in Fig. 3.1b, the input matching can be expressed as

$$Z_{in} = s \left(L_{in} + L_s \right) + \frac{1}{sC_{gs}} + \frac{g_m L_s}{C_{gs}} \,. \tag{3.1}$$

When the Z_{in} is matched to the real-value source impedance R_s , the input matching conditions become:

$$R_{in} = \frac{g_m}{C_{es}} L_s \approx \omega_T L_s , \qquad (3.2)$$

$$\omega_0^2 = \frac{1}{\left(L_{in} + L_s\right)C_{gs}} \,. \tag{3.3}$$

In previous equations, components L_{in} , L_S , and C_{gs} can be identified from Fig. 3.1 and g_m and ω_T are the transconductance and the unity-gain frequency⁴ of input device M_I , respectively. In addition to the ideal components presented in Fig. 3.1, the LNA input is typically involved with parasitic components and the nonidealities in active and passive devices, which complicate the design and analysis. For example, the effect of the pad and ESD protection are discussed in Section 3.4. In addition, the series resistances R_{Lin} and R_{Ls} of inductors L_{in} and L_S , together with input transistors gate resistance r_g , have an effect on the input matching conditions:

$$R_{in} = \frac{g_m L_s}{C_{gs}} + r_g + R_{Lin} + R_{Ls} , \qquad (3.4)$$

$$\omega_0^2 = \frac{1 + g_m R_{Ls}}{\left(L_{in} + L_s\right) C_{gs}} \,. \tag{3.5}$$

The Q-value of an inductor having value of L and series resistance R_L is calculated as

$$Q_L = \frac{\omega L}{R_L},\tag{3.6}$$

where ω is the frequency of interest. Equation (3.6) is valid if only the series resistance R_L is taken into account in the inductor model. Clearly, models of wire-bond inductance and monolithic inductors include several lumped components, but the detailed description of those is beyond the scope of this thesis. The modeling and characterization of inductors is discussed in the literature in, for example, [15] – [20].

The advantage of inductive source degeneration is that impedance matching is achieved without the use of a physical resistor, which would increase LNA's noise. The input LC matching network can be considered lossless, which does not introduce noise of its own. However, the series resistances of inductors L_{in} and L_S have a slight effect on the NF of the LNA. Typically, L_{in} is realized as wire-bond inductance and, since L_S has low inductance value, it is implemented as an on-chip spiral inductor. Typically, the former has higher Q-values than the latter. To minimize the effect of the series resistance of the source inductor, LNAs utilizing bondwires as degeneration inductance(s) are published as well, in [21] and [22], for example.

⁴ For the sake of brevity, in this thesis the angular frequency ω is used synonymously with frequency *f*.

3.2.2.1 Gain and effective transconductance

As was discussed earlier in Section 2.2.2, the voltage gain is a suitable figure-of-merit for narrowband LNAs. The voltage gain is defined as a product of effective transconductance $g_{m,eff}$ and a load impedance Z_L . The $g_{m,eff}$ of a basic IDCS stage shown in Fig. 3.2 can be evaluated as

$$g_{m,eff} = \frac{I_{out}}{V_{in}} = \frac{g_m V_{gs}}{V_{in}} = \frac{g_m}{s C_{gs}} \frac{I_{in}}{V_{in}} = \frac{g_m}{s C_{gs}} \frac{1}{Z_{in}},$$
(3.7)

where components g_m and C_{gs} have their typical meanings and Z_{in} is calculated according to (3.1). At a resonance frequency, the $g_{m,eff}$ is

$$g_{m,eff} = \frac{1}{\omega_0 L_s} = \frac{g_m}{\omega_0 C_{gs} R_s} \approx \frac{\omega_T}{\omega_0 R_s} .$$
(3.8)

Therefore, in a matched case, the $g_{m,eff}$ is independent of the transconductance g_m of M_1 but a relation to g_m exists via the input matching criterion (3.2). Equation (3.7) can be formed to

$$g_{m,eff} = \frac{I_{out}}{V_{in}} = g_m \frac{V_{gs}}{V_{in}} = g_m Q_{in}, \qquad (3.9)$$

where the Q_{in} measures the voltage gain between the input of the matching network and the gate-source voltage of the input transistor [23]. This gain is important, because it affects both the voltage gain and the noise of the input transistor and of the overall amplifier. By increasing the Q_{in} , lower g_m is required for the input transistor. Both can be achieved by reducing the size of the input transistor. However, the input matching becomes sensitive to the component value variations if a large Q-value (>3) is utilized [23]. In addition, with small C_{gs} , the input matching circuit becomes sensitive to the parasitic capacitances at the input matching circuit. Therefore, the use of IDCS stage can be impractical with low power levels [24].



Fig. 3.2. Equivalent circuit to evaluate the effective transconductance and Q-value of an IDCS stage.

3.2.2.2 Noise optimization of IDCS LNA

The noise optimization of an IDCS LNA is widely studied in the literature, in [11] - [14] and [25] - [29], for example. The detailed noise analysis and comparison of noise formulas derived in different publications is beyond the scope of this thesis. A short overview of the topic is given in this section for the sake of clarity and completeness.

The input transistor noise performance optimization usually involves the optimization of the Q-value of the input matching circuit and aspect ratio and drain current of the input transistor [11]. A high Q-value for the input matching network is beneficial to reducing the drain current noise, but it increases the induced gate current noise. The noise factor is expressed as

$$F = 1 + \frac{R_{Lin}}{R_S} + \frac{R_g}{R_S} + \frac{\gamma}{\alpha} \frac{1}{Q_{in}} \left(1 - 2|c| \sqrt{\frac{\delta\alpha^2}{5\gamma} + \frac{\delta\alpha^2}{5\gamma}} \right) \left(\frac{\omega_0}{\omega_T} \right) + \frac{\delta\alpha}{5} Q_{in} \left(\frac{\omega_0}{\omega_T} \right), \quad (3.10)$$

where R_{Lin} , R_{g} , and R_{S} are the series resistance on input inductor, the gate resistance, and the source impedance, respectively [11] - [13]. δ , γ , and α are the typical transistor noise factors, Q_{in} is the Q-value of the input matching circuit, and ω_0 and ω_T are the operational and cut-off frequencies, respectively. In addition, c is the correlation coefficient between drain and gate noises. Equation (3.10) is valid when the polarity of the gate current noise source is as indicated in Fig 2.7. It is observed from (3.10) that there are parts that are independent of the Q_{in} or increase or decrease along with Q_{in} . Therefore, an optimum NF exists for a particular Q_{in} [11]. In addition, in general, the NF improves along with higher f_T . However, as was mentioned earlier, the noise factors γ and δ tend to increase along with shorter transistor channels. Fortunately, some parts in (3.10) are given as ratios of noise factors γ and δ , which can be considered nearly constant. This is a reasonable assumption, since both noise sources have the same physical origin [30]. Therefore, the increment of noise factors γ and δ due to short-channel effect is not as severe for the IDCS stage as it is for a common-gate stage, for example. The noise due to gate resistance R_g can be minimized through interdigitation [11], [30]. The noise of the R_{Lin} is small but non-negligible and in certain cases it can become a major noise contributor [31].

The noise figure of an IDCS stage given by (3.10) is plotted in Fig. 3.3 as a function of Q_{in} . The noise due to R_g and R_{Lin} are ignored. The noise parameters δ and γ , shown with a solid line, have classical values of 4/3 and 2/3, respectively. For a comparison, a case where both values are doubled is presented with a dashed line. In addition, the noise is plotted such that the ratio ω_0 / ω_T has the values of 0.05, 0.1, and 0.2. According to (3.10), the noise increases along with the increasing ω_0 / ω_T ratio. Doubling the noise parameters increases the minimum NF by approximately 1 dB, but the optimal Q_{in} value remains nearly unchanged.



Fig. 3.3. Noise figure of an IDCS LNA. The factor ω_0 / ω_T has the values of 0.05, 0.1, and 0.2. The NF increases along with the larger ω_0 / ω_T ratio.

A noise optimization analysis similar to [11] is given in [14], where an additional capacitor C_d is added in parallel with the intrinsic gate-source capacitance C_{gs} of the input transistor as is shown in Fig. 3.1a. As a result, there is freedom to optimize both the Q_{in} and size of the input device [14]. Thus, the latter technique allows for a better noise minimization when compared to the previously mentioned. The noise factor at a resonance is given by

$$F = 1 + \frac{\frac{\delta}{5} \left(Q_{in}^2 + \frac{1}{4} \right) P^2 + \frac{\gamma}{4}}{R_S Q_{in}^2 g_m},$$
(3.11)

where P is defined as

$$P = \frac{C_{gs}}{C_{gs} + C_d} \tag{3.12}$$

and Q_{in} is the quality factor of the input circuit. The optimum values for width of the M_1 and Q-value of the input matching circuit are also derived in [14]. In [29] it was stated that (3.12) does not take into account the overall impact of all the overlap capacitances. Therefore, P is more accurately given as

$$P \simeq \frac{C_{gs}}{C_{gs} + C_{gs,ov} \cdot \left(2 + \frac{g_{m1}}{g_{m2}}\right) + C_d},$$
(3.13)

where $C_{gs,ov}$ is the overlap capacitance between gate and source and g_{m2} is the transconductance of the cascode device [29]. Even when the value of C_d is set to zero, the intrinsic value of Pbecomes less than unity and the use of additional capacitance C_d becomes unnecessary. For example, the value of P was found as low as 0.3 in a 0.18-µm CMOS process [29].

While equations (3.10) and (3.11) provide good starting points to design the noise optimized LNA input stage, both formulas neglect the noise of the cascode stage and the noise and loss due to series resistance of the source inductor. In addition, the bonding pad and ESD protection needed at the LNA input increase the NF. Both the bonding pad and ESD protection cause a parasitic capacitance to substrate and the LNA noise increases due to two mechanisms. First, the noise and other interferers from the substrate can couple through the bondpad to the LNA input [32]. Secondly, the parasitic components at the LNA input affects the input matching circuit, causing degradation in input impedance and altering noise contribution compared to the traditional LNA noise analysis [31]. As a result, the noise minimization of the LNA is challenging. It should also be noted that the optimum NF is typically found by sweeping the width of the input device in a circuit simulator.

3.2.2.3 Linearity

In addition to the input matching, inductive degeneration induces negative feedback at input transistor's source node. In general, negative feedback reduces the distortion [33]. The IIP3 of an inductively-degenerated LNA is presented in [34], for example. Since the degeneration inductance has low impedance at low frequency, the transconductance stages using inductive degeneration are more linear than those using resistive degeneration with the same bias current [34]. In addition, the linearity of the IDCS stage is affected by the input matching network,

which amplifies the input voltage by its Q-value and decreases the linearity. The distortion analysis based on Volterra's series presented in [34] and [35], however, is tedious for hand calculations. The IIP3 of the IDCS stage can be roughly estimated with

$$IIP3_{LNA} = \frac{IIP3_{FET}}{\left(\frac{V_{gs}}{V_{in}}\right)^2} = \frac{IIP3_{FET}}{Q_{in}^2},$$
(3.14)

where $IIP3_{FET}$ is the intrinsic IIP3 of the input transistor [36]. Typically, the IIP3 improves with higher overdrive voltage. In addition, the common-source stage shows peaking, "sweet spot", when the bias point of the transistor changes from the weak inversion to the strong inversion region. Over a narrow bias range in the moderate inversion region, the drain current follows a nearly square-law behavior with minimized third-order distortion [37]. A simulation example is shown in Fig. 3.4, where a CS stage is simulated with and without inductive source degeneration. The simulation is performed with a 0.13-µm CMOS. The drain-source voltage is set to 0.4 V. A peak in IIP3 is observed with $V_{gs} \approx 0.46$ V.

Since the sweet spot typically occurs over a narrow region of bias voltage, the linearization of an LNA or any other amplifier is challenging when setting the bias point accurately to the sweet spot. Furthermore, the sweet spot is significantly affected by process variation. The location of the sweet spot changes in terms of V_{gs} voltage between process corners primarily due to variations in V_{th} , but the variation is much smaller in terms of the bias currents [37]. The sweet spot phenomenon, however, should not be completely neglected, since the peak shifts to higher drain currents as line widths scale down [38]. Therefore, the bias currents, where the sweet spot occurs, are in the region of interest in RF design [37]. As is shown in Fig. 3.4, in IDCS stages the sweet spot phenomenon is weak due to degeneration [37], [39].



Fig. 3.4. IIP3 of a CS stage. The IIP3 is shown without source inductor and with 1-nH and 2-nH source inductors.

3.2.2.4 Cascode stage

Usually, the CS amplifiers utilize a cascode stage shown in Fig. 3.5a if its usage is not restricted by low supply voltage. There are several benefits from the cascode transistor. Usually, the input impedance of the cascode transistor is smaller than the load impedance. Therefore, the Miller effect of the input stage is reduced, which improves the LNA stability. Due to the larger reverse isolation, the input stage and the load can be separately designed and optimized.

Usually, the noise contribution of a cascode stage to the overall noise of the LNA is small but non-negligible, since at high frequencies the impedance looking into the drain of M_1 is low [10]. The noise contribution of the cascode transistor is decreased by minimizing the parasitic capacitance between input and cascode transistors [40]. The substrate with high resistivity helps to keep this capacitance low [41]. In addition, the parasitic capacitance can be minimized, for example, by utilizing a dual-gate MOS device [42], [43]. This typically improves the noise performance by less than 0.3 dB. The noise contribution of the cascode device can be reduced by resonating the parasitic capacitance out with the additional inductance shown in Fig. 3.5b [44]. However, the large layout area required by on-chip inductors typically limits the use of this method. In addition, due to the finite Q-value, of the on-chip inductor adds the noise of its own. Furthermore, the resonator utilized between the input and cascode stages shown in Fig. 3.5b can be used to filter the image signal [44] or unwanted interferers [45], [P6].

The cascode stage can be used for gain control as shown in Fig. 3.5c. With additional switches, part of the current signal is steered from the LNA load. The accuracy of the gain control depends on matching of transistor sizes and is well controlled over process variations and temperature corners [46]. More details on designing variable gain in low-noise amplifiers can be found in [4], for example.



Fig. 3.5. a) IDCS LNA with a cascode stage, b) noise improvement or unwanted signal filtering with internal node resonator, c) variable gain LNA.

If the target is to design an LNA with extremely low supply voltages, the cascode amplifier can be realized with a folded structure as is shown in Fig. 3.6a and Fig. 3.6b. The former uses a PMOS current source to create a wideband high-impedance node to force the current signal to the folded cascode stage. In the latter case, a narrowband LC resonator tuned at carrier frequency f_0 is used instead of a current source. However, at resonance the impedance of the tank can be comparable to the impedance $1/g_{m2}$ seen at the source of M_2 . Due to the current division, a portion of the signal will be lost to the tank [47]. Thus, the gain of the LC-folded cascode topology is lower by a factor of

$$\varsigma \cong \frac{g_{m2}}{g_{m2} + \frac{\omega_0 C}{Q_L}}$$
(3.15)

than that of the wideband cascode topology [47]. In the previous equation, Q_L is the quality factor of the inductor. The loss is minimized having a large-value inductor with maximal Q-value and small capacitor. The supply voltage reduction is achieved at the expense of an increased number of inductors. In addition, the current consumption is increased, because the two transistors do not share the common dc current path compared to the typical cascode amplifier. The cascode transistor M_2 , however, can be biased with smaller dc current than the input device M_1 . The bias current of M_2 cannot be reduced too much, because the signal loss of the LC tank can be minimized by increasing the transconductance g_{m2} .



Fig. 3.6. Folded LNA with a) PMOS current source, b) LC resonator.

3.2.3 Common-gate topology

Common-gate (CG) LNA is another widely used topology in wireless communications [24], [48] – [56], [P1]. In this section, the basics of the CG LNA are discussed. First, the input matching and voltage gain are presented and noise figure analysis follows. The last part of this section briefly describes g_m -boosted topology, which combines the CS and CG LNA structures. The more detailed analysis of a CG stage as a function of frequency is presented in Section 3.3.7.

3.2.3.1 Input matching and voltage gain

The desired input impedance of a CG input stage is achieved by adjusting the bias current, aspect ratio, and overdrive voltage such that $1/g_m$ is close to the termination impedance Z_0 . Typically, single-ended Z_0 is 50 Ω , and therefore a g_m of approximately 20 mS is required. The simulated g_m of a single input transistor as a function of the transistor width and drain current (I_D) is shown in Section 6.1.1, Fig. 6.3. The CG stage does not suffer from the Miller effect, and thus an adequate reverse isolation can be achieved with a single transistor stage. Therefore, the input matching network and load can be designed separately. In [52] and [56], such a CG LNA

alternative, where feedback is used to adjust the input impedance and frequency transfer function simultaneously, is presented.

Large impedance towards the signal ground is needed to steer the signal into the input transistor source. This can be achieved with a current source I_{bias} shown in Fig. 3.7a. That topology is not typically utilized in the LNA since the current source I_{bias} increases the noise. A better noise performance is achieved by using a source inductor L_S as shown in Fig. 3.7b. The L_S forms a parallel LC resonator with the parasitic capacitance C_{par} associated with the source node of the M_I . When the on-wafer measurements are not applicable, the source node typically needs to be connected either to package or PCB by using a bondwire inductance L_{in} as shown in Fig. 3.7c. The L_{in} is resonated at the wanted frequency with a dc blocking capacitor C_{in} , which can be either an on-chip or an external component.



Fig. 3.7. Common-gate LNA input interfaces: a) current source, b) parallel LC resonator, and c) series and parallel LC resonators.

In Fig. 3.7b and Fig. 3.7c, the capacitor C_{par} includes the parasitic capacitances at the source node, i.e. source-body junction capacitance of M_I , substrate capacitance of L_S , and capacitance caused by the bonding pads and on-chip metal wiring. Furthermore, the value of the source inductor L_S can be decreased by adding an additional shunt capacitor C_S in parallel with L_S (the C_S is not shown in Fig. 3.7b or Fig. 3.7c). Therefore, all the capacitance at the source node can be included in a single source capacitor C_T used in the following calculations:

$$C_T = C_{gs} + C_{par} + C_S . ag{3.16}$$

The input impedance Z_{in} of a CG input stage shown in Fig. 3.7b can be calculated as

$$Z_{in} = \frac{sL_S}{1 + sL_S g_m + s^2 L_S C_T} \,. \tag{3.17}$$

The source inductance L_S resonates with the capacitance C_T at the frequency of

$$\omega_0 = \frac{1}{\sqrt{L_s C_T}} \tag{3.18}$$

and at that frequency, (3.17) simplifies to $1/g_m$. The input impedance Z_{in} of a CG input stage shown in Fig. 3.7c is

$$Z_{in} = \frac{1 + s^2 C_{in} L_{in}}{s C_{in}} + \frac{s L_S}{1 + s L_S g_m + s^2 L_S C_T} \,.$$
(3.19)

The detailed analysis of design input and source resonators is presented in Section 3.3.7.

With a perfect impedance matching $(1/g_m = R_s)$, the voltage gain of the CG amplifier becomes a division of output load versus the source impedance, i.e. Z_L/R_s . The assumption is valid if the drain-to-source resistor r_{ds} is much larger than the load resistance at the drain. Otherwise, the gain and input impedance formulas should be modified to [49]

$$A_{v} = \frac{g_{m}Z_{L}}{\left(1 + \frac{Z_{L}}{r_{ds}}\right)}$$
(3.20)

and

$$R_{in} = \frac{1}{g_m} \left(1 + \frac{Z_L}{r_{ds}} \right). \tag{3.21}$$

3.2.3.2 Noise of a CG stage

The noise factor of a CG LNA is expressed as [50]

$$F = 1 + \frac{\gamma}{\alpha} \left(\frac{1}{1+\chi}\right)^2 \frac{1}{g_m R_s},$$
(3.22)

where γ is the coefficient of channel thermal noise, g_m is the transistor transconductance, χ is the ratio of the transistor substrate transconductance g_{mb} and g_m , R_s is the source resistance and α is defined as in (2.24). According to (3.22), the noise performance of the common-gate stage is independent of the operation frequency. Because the minimum NF of an IDCS LNA increases along with the frequency, CG LNA can be a better option at very high frequencies [50]. In addition, the noise figure can be maintained with low drain currents by ensuring that proper input matching is achieved by keeping g_m at sufficient level.

When χ is neglected and perfect input matching is assumed, the minimum noise factor typically presented in the literature is achieved:

$$F = 1 + \frac{\gamma}{\alpha} = \frac{5}{3} = 2.2 \text{ dB}.$$
 (3.23)

For short-channel devices, γ can be much greater than one, and α can be much less than one. Accordingly, the minimum theoretically achievable noise figures tend to be around 3 dB or greater in practice [11]. Therefore, the NF is slightly higher compared to an inductively degenerated common-source LNA, which limits the usage of the common-gate LNA.

With imperfect input matching, the noise factor can be lowered according to [24]⁵

$$F = 1 + \gamma \frac{1 + S_{11}}{1 - S_{11}}, \qquad (3.24)$$

⁵ It should be noted that (3.24) presents the input matching dependency correctly. There is a misspelling in the original paper [24], where the '+' and '-' signs are interchanged.

where α is neglected. In mass-product applications, however, the optimum input matching performance should be targeted to meet matching specifications also in the presence of process and temperature variations [31].

In addition, (3.22) does not take into account the noise of the load. If resistive load R_L is used, taking into account its noise contribution, the noise factor becomes [7]

$$F = 1 + \frac{1}{g_m R_s} \left(\frac{\gamma}{\alpha} + \frac{(1 + g_m R_s)^2}{g_m R_L} \right) \xrightarrow{g_m = 1/R_s} 1 + \frac{\gamma}{\alpha} + \frac{4R_s}{R_L} .$$
(3.25)

Thus, the resistive load can make a significant contribution to the overall noise.

The noise equations above assume that the gate induced current noise is neglected. This is a valid assumption since, in a properly interdigitated MOS transistor, the gate resistance is small and the impedance at the gate is low. The noise of the bias current source of the CG-LNA depicted in Fig. 3.7a is thoroughly analyzed in [51]. The low noise current source is designed by minimizing its transconductance and gate (or base) resistance and by using a degeneration resistor [51].

3.2.3.3 *G_m*-boosted CG stage

The input matching requirement prevents increasing g_m of the input transistor to lower the noise factor. The link between noise factor and input matching can be separated by introducing inverting gain from the source to the gate of the input transistor [57], [58]. As a result, the effective transconductance of the boosted CG stage is

$$g_{m,eff} = (1+A)g_m,$$
 (3.26)

where A is the gain between source and gate. The input matching is achieved, when

$$g_m = \frac{1}{R_s} \frac{1}{(1+A)}$$
(3.27)

and the noise factor is reduced to [58]

$$F = 1 + \frac{\gamma}{\alpha} \frac{1}{(1+A)}.$$
(3.28)

According to (3.27) less bias current is required to achieve the required input matching and therefore less drain current noise from the input transistor is obtained.



Fig. 3.8. G_m -boosted common-gate input stage.

The inverting gain can be implemented with reactive components so as not to add additional noise. In [57], the capacitor cross-coupling method, which is suitable for differential input configurations, is presented. Due to the capacitor divider between C_{gs} and coupling capacitance C_c , the inverting gain is approximately $A = C_c / (C_c + C_{gs})$, which is always less than one [59].

To achieve inverted gain greater than one, the transformer-coupled technique shown in Fig. 3.9b is proposed in [60]. The primary and secondary inductors L_P and L_S form a transformer, which provides an anti-phase operation between gate and source terminals [60]. The method is applicable both for single-ended and fully differential structures. The input admittance is approximately

$$Y_{in} \approx \frac{1}{sL_p} + (1+nk)g_m + (1+2nk+n^2)sC_{gs}, \qquad (3.29)$$

where $k = M / \sqrt{L_P L_S}$ is the coupling coefficient and $n = \sqrt{L_S / L_P}$ is the turns ratio [60]. The inversion gain factor is $A = kn = k\sqrt{L_S / L_P}$, and therefore the transformer-coupled structure has a freedom to set the desired inversion gain. Since the gate terminal is not ac grounded, the induced gate noise must be taken into account. The noise factor of a transformed-coupled CG-LNA is [60]

$$F = 1 + \frac{\gamma}{\alpha} \frac{1}{1+nk} + \frac{\delta\alpha}{5} \left(\frac{\omega}{\omega_T}\right)^2 \frac{\left(1+2nk+n^2\right)^2}{\left(1+nk\right)^3}.$$
(3.30)

However, the nonidealities related to an on-chip transformer are not included in the noise analysis. For example, losses due to parasitic capacitances and series resistors increase the overall noise figure.

The measured noise figures of g_m -boosted CG-LNAs using capacitor cross-coupling presented in [57] and [61] are 3.0 dB and the one utilizing transformer coupling achieves an NF of 2.5 dB [60]. Therefore, the cross-coupling technique improves the NF only slightly compared to basic CG LNAs, but the transformer-coupled technique seems to benefit from the increased effective transconductance, which results in improved NF.



Fig. 3.9. a) Capacitor cross-coupled CG stage [57], b) transformer-coupled CG-stage [60].

3.3 Wideband LNA topologies

In this sub-chapter, LNA topologies for wideband applications are presented. First, the IDCS LNA, which is typically used in narrowband systems, is analyzed in detail to find its suitability and limitations for wideband applications. After that, two alternatives based on IDCS topology are described and an IDCS LNA utilizing Chebyshev input matching is briefly discussed. Then, the resistor-feedback and common-drain-feedback CS LNAs are studied. After that, the CG LNA is analyzed in detail. Finally, LNAs utilizing reactive feedback are briefly described and a summary on wideband LNA design is given.

In this section, the main emphasis is on the analysis of input matching and insertion gain bandwidths. To design a wideband LNA with flat insertion gain response, the effect of the load impedance must be taken into account as well. Different load structures are studied in Section 3.5. For the purpose of comparison, several simulation examples are shown in this section. Some of the simulations are shown with normalized frequencies, i.e. the center frequency is set to 1. When the simulation examples are presented taking into account more realistic component values, the design target is UWB BG1 and then the input matching center frequency is set to 4 GHz.

Appendix A shows a model for a general CS LNA having arbitrary gate, gate-source, gatedrain, source, and load impedances. In the following, some input impedances and insertion gains are calculated by exploiting equations presented there.

3.3.1 IDCS LNA

The performance of the IDCS stage is analyzed in detail as a function of frequency. In the following, the input matching and output signal current bandwidths are analyzed as a function of the input matching circuit Q-value at resonance frequency ω_0 . The Q_{in} is a useful parameter, since it also gives a relation to the noise performance and effective transconductance of the IDCS stage, as given in (3.9) and Fig. 3.3. The purpose of the analysis is to find the design rules and limitations, which affect the achievable bandwidth of both S_{11} and I_{out} .

3.3.1.1 Input matching bandwidth

The magnitude of the input matching S_{11} is calculated with (2.1) and (3.1) and by using the basic input matching criteria (3.2) and (3.3):

$$|S_{11}| = \frac{1}{\sqrt{1 + \left(\frac{2\omega C_{gs} Z_0}{\omega^2 C_{gs} (L_{in} + L_s) - 1}\right)^2}}.$$
(3.31)

In (3.31), Z_0 is the source impedance ($Z_0 = R_{in}$). The frequencies, where the S_{11} equals -10 dB, are:

$$\omega_{S11=-10dB} = \frac{\sqrt{Z_0^2 C_{gs}^2 + 9C_{gs} \left(L_{in} + L_s\right)} \pm Z_0 C_{gs}}{3C_{gs} \left(L_{in} + L_s\right)} \,.$$
(3.32)

Thus, the BW where S_{11} is better than -10 dB is:

$$BW_{S11} = \frac{2}{3} \frac{Z_0}{\left(L_{in} + L_S\right)}.$$
(3.33)

The result shows that the S_{II} range can be widened either by minimizing the value of input and source inductances or by increasing the input impedance Z_0 . According to (3.2), when the values of inductors are fixed, to maintain the input matching and keeping the C_{gs} unchanged (constant input device size) a larger g_m is needed. Since the aspect ratio of the input transistor cannot be altered, the larger g_m can only be achieved with increased drain current. In addition, pad and ESD protection cause parasitic capacitances at the LNA input node, which tend to lower the input impedance [31]. As a result, having LNA matched to higher Z_0 can become challenging to realize in practice.

Next, the S_{11} performance is analyzed such that Z_0 level is constant and optimum values for the input matching components, C_{gs} , L_{in} , L_s , and g_m , are calculated to achieve maximally wide matching. Perfect matching at ω_0 is assumed. For the following analysis, the Q-value of the input matching circuit at resonance frequency ω_0 is rewritten with (3.8) and (3.9) as

$$Q_{in} = \frac{g_{m,eff}}{g_m} = \frac{1}{\omega_0 L_s g_m} = \frac{1}{\omega_0 C_{gs} R_s}.$$
(3.34)

By using the input matching criteria (3.2) and (3.3), Equation (3.32) can be given as a function of Q_{in} by using (3.34):

$$\omega_{S11=-10dB} = \frac{\omega_0}{3} \left(\sqrt{9 + \frac{1}{Q_{in}^2}} \pm \frac{1}{Q_{in}} \right).$$
(3.35)

Similarly, the S_{11} bandwidth is expressed as

$$BW_{S11} = \frac{2}{3} \frac{Z_0}{L_S + L_{in}} = \frac{2}{3} \frac{\omega_0}{Q_{in}} .$$
(3.36)

Therefore, the input matching bandwidth is inversely proportional to Q_{in} . The S_{I1} of an IDCS stage is shown in Fig. 3.10. The simulation is performed such that the center frequency is tuned to 4 GHz and the source inductor is set to 1 nH. The other component values used in simulations are given in Table 3.1. The source impedance is 50 Ω . The Q_{in} has values of 1, 2, and 4. According to (3.34), when perfect matching is targeted, the C_{gs} capacitor decreases along with Q_{in} . Since L_S is kept constant, the g_m is evaluated with (3.2) to maintain the optimum input matching, while the value of the inductor L_{in} is calculated with (3.3) to maintain the center frequency. It should be noted that the S_{I1} upper and lower side curves are symmetric with respect to ω_0 in logarithmic scale. Therefore, it is beneficial to set center frequency ω_0 to the geometric average of the corner frequencies of the wanted band. For example, the frequency allocation for UWB BG1 is between 3.168 and 4.752 GHz. Hence, the geometric average frequency is 3.88 GHz and, according to (3.36), the largest Q_{in} , which covers the whole BG1 band, is 1.63.



Fig. 3.10. S_{11} of an IDCS stage versus Q_{in} .

Table 3.1. Component values for the simulation shown in Fig. 3.10.

Q_{in}	$L_{S}[nH]$	L_{in} [nH]	C_{gs} [fF]	g_{ml} [mS]
1	1.00	0.99	796	39.8
2	1.00	2.98	398	19.9
4	1.00	6.96	199	9.95

3.3.1.2 Output signal current of an IDCS stage

In Section 3.2.2.1, the effective transconductance of the IDSC stage was analyzed at the input matching frequency. In the following, the I_{out} is analyzed in details as a function of frequency. For an IDCS shown in Fig. 3.2, the I_{out} is

$$I_{out} = \frac{g_m V_s}{1 + s \left(g_m L_s + C_{gs} R_s \right) + s^2 \left(C_{gs} L_{in} + C_{gs} L_s \right)} \,. \tag{3.37}$$

By using input matching criteria (3.2) and (3.3), equation (3.37) is expressed as

$$I_{out} = \frac{g_m V_S}{1 + s \cdot 2g_m L_S + \left(\frac{s}{\omega_0}\right)^2}.$$
(3.38)

The magnitude of (3.38) achieves its maximum value at a frequency of

$$\omega_{I_{out},\max} = \omega_0 \sqrt{1 - \frac{2}{Q_{in}^2}} , \qquad (3.39)$$

where ω_{θ} is the input matching resonance frequency given by (3.3) and Q_{in} is defined in (3.34). As a result, the maximum output current frequency is always lower than the optimal input matching resonance frequency. In addition, when $Q_{in} < \sqrt{2}$, (3.39) becomes imaginary. In that case, the maximum I_{out} is achieved at the zero frequency. That case is further analyzed later. At the frequency given by (3.39) and assuming $Q_{in} > \sqrt{2}$, the maximum output current is

$$I_{out,\max} = \frac{g_{m,eff}}{2} \frac{V_S}{\sqrt{1 - \frac{1}{Q_{in}^2}}}$$
(3.40)

The frequencies, where Iout is decreased by 3 dB from its maximum value, are

$$\omega_{I_{out},\max-3dB} = \omega_0 \sqrt{1 - \frac{2}{Q_{in}^2} \pm \frac{2}{Q_{in}} \sqrt{1 - \frac{1}{Q_{in}^2}}} .$$
(3.41)

If the lower -3-dB corner frequency is evaluated, the term inside the square root of (3.41) with the '-' sign becomes negative when

$$Q_{in} = \frac{2}{\sqrt{2 - \sqrt{2}}} \approx 2.6$$
. (3.42)

If the Q_{in} is smaller than that limit, the scaled I_{out} achieves a -3-dB level already at the zero frequency, as is shown in Fig. 3.11. In that case, the -3-dB bandwidth is solely defined by the upper corner frequency, i.e. by (3.41) with a '+' sign only. The value defined by (3.42) also maximizes the value of (3.41).

When $Q_{in} < \sqrt{2}$, the maximum I_{out} is at zero frequency and then the -3-dB corner frequency is defined by another formula:

$$\omega_{I_{out},\max-3dB,Q_{in}<\sqrt{2}} = \frac{\omega_0}{\sqrt{2}}\sqrt{2 - \frac{1}{Q_{in}^2} + \sqrt{8 - \frac{4}{Q_{in}^2} + \frac{1}{Q_{in}^4}}} .$$
(3.43)

The maximum value of (3.43) is less than the value given by (3.42). Therefore, the I_{out} achieves its maximum –3-dB bandwidth when $Q_{in} \approx 2.6$. Therefore, inserting (3.42) into (3.41), the fundamental upper –3-dB limit for the I_{out} of a perfectly matched IDCS is given as

$$\omega_{I_{out},\max-3dB,\max} = \omega_0 \sqrt{\sqrt{2}} \approx 1.189 \cdot \omega_0 \,. \tag{3.44}$$

The result generally is independent of the process technology used. The scaled I_{out} is shown as a function of normalized frequency with different Q_{in} values in Fig. 3.11. The frequency, where optimum I_{out} is achieved, becomes closer to ω_0 with larger Q_{in} values, as is given by (3.39), but also the -3-dB bandwidth becomes narrower. Conclusions are drawn in Section 3.3.1.4, where a comparison to BW_{S11} is also made.



Fig. 3.11. Scaled I_{out} with several Q_{in} values.

3.3.1.3 Noise figure

The noise figure is an important design criterion as well. Since additional components are not added to the IDCS stage, the noise optimization presented in Section 3.2.2.2 still gives the minimum achievable NF. The input matching circuit Q-value must be low to achieve wide enough input matching and output signal current bandwidths. This, however, conflicts with the optimal noise figure presented in Fig. 3.3. The NF starts to increase remarkably when $Q_{in} < 2$. In addition, the minimum NF of the IDCS stage is typically achieved at a slightly lower frequency compared to the optimal input matching frequency. In narrowband applications, the noise increase from the optimal level is significant and nearly optimal NF can be achieved. However, in wideband applications, the noise performance can degrade significantly at high frequencies. This is depicted in a simulation example presented in Section 6.5.1, where an IDCS LNA for UWB BG1 is designed. The NF around 3-GHz frequency is approximately 1.7 dB and it increases to 2.5 dB at 4.7 GHz.

3.3.1.4 Conclusion and remarks

When a proper IDCS stage is designed, both the input matching and output signal current must cover the wanted frequency BW. The minimum and maximum corner frequencies of S_{11} and I_{out} for an optimally matched IDSC stage as a function of Q_{in} are shown in Fig. 3.12. In that figure, $f_{Iout,Max-3dB,HI}$ and $f_{Iout,Max-3dB,LO}$ are the upper and lower corner frequencies, where the output signal current is decreased from the maximum value by 3 dB, respectively, and $f_{S11=-10dB,HI}$ and $f_{S11=-10dB,LO}$ are the upper and lower frequencies where S_{11} equals -10-dB, as is shown in Fig. 3.10. The frequency limits at the *y*-axis are relative, i.e. corner frequencies are scaled with respect to ω_0 . As is shown in Fig. 3.12, relative bandwidths of both the S_{11} and I_{out} decrease along with higher Q_{in} . The upper corner frequencies of S_{11} and I_{out} are nearly identical when Q_{in} is around 2. Then, the lower corner frequency is limited by S_{11} and the relative BW_{S11} is approximately 33 %. With smaller Q_{in} values, the upper corner frequency of I_{out} starts to decrease rapidly thus limiting the achievable insertion gain BW. Thus, according to (3.36), the input matching can be widened unlimitedly by decreasing Q_{in} but, as shown by Fig. 3.12, the bandwidth of I_{out} cannot.



Fig. 3.12. Minimum and maximum corner frequencies of S_{11} and I_{out} as a function of Q_{in} .

Next, the IDCS LNA design possibilities for UWB BG1 are considered. The relative bandwidth BW_{rel} of BG1 is approximately 40 % and the center frequency ω_{l} is 3.96 GHz. If the IDCS stage is matched exactly at that frequency, according to (3.44), the maximum upper -3-dB corner frequency of I_{out} is 4.71 GHz, which is not adequate for UWB BG1. To achieve the insertion gain, which covers the whole BG1 with some design margin, the LNA load should be designed to compensate the droop caused by I_{out} . This is challenging to achieve in practice. Since I_{out} corner frequency linearly follows ω_0 as is given by (3.41), it would be beneficial to set the input matching frequency to a frequency higher than 3.96 GHz. That, however, conflicts with the optimum input matching frequency strategy, since, as discussed earlier, to maximize the S_{II} coverage, the ω_0 should be placed at the geometric average of the BG1 corner frequencies (3.88 GHz). To achieve -10 dB input matching covering the whole BG1, $Q_{in} < 1.63$ is needed. That leads to inadequate I_{out} performance since, according to (3.41), the -3-dB corner frequency would be approximately 4.3 GHz. Therefore, the performance of an IDCS LNA is difficult to optimize for applications having wide relative bandwidths. For example, due to higher operational frequency, the BW_{rel} of BG3 is approximately 22 %. It is possible to realize this with the IDCS stage as is presented in Section 6.6.2.1.

The example of S_{11} and insertion gain of an IDCS LNA is presented in Fig. 3.13. The input is tuned at 4-GHz center frequency and the targeted maximum insertion gain is 20 dB to 300- Ω load. The input stage is realized with Q_{in} values of 2, 2.6, and 4. To achieve the targeted insertion gain, first the effective transconductance is evaluated with (2.4) and (3.40). When $g_{m,eff}$ is known, source inductor L_S , C_{gs} capacitance and g_m of the input transistor are evaluated with (3.34). The component values are given in Table 3.2. Clearly, to achieve sufficient input matching performance and gain variation, UWB BG1 is not realizable with IDCS LNA.



Fig. 3.13. The simulated insertion gain and S_{11} of the IDCS LNA. Both the gain and S_{11} are presented with Q_{in} values of 2, 2.6, and 4. The frequency limits of UWB BG1 are also shown.

Table 3.2. Component values for the simulation shown in Fig. 3.10.

Q_{in}	L_{in} [nH]	L_{S} [nH]	C_{gs} [fF]	g_{ml} [mS]
2.0	2.60	1.38	398	14.4
2.6	3.88	1.29	306	11.8
4.0	6.73	1.23	199	8.07

As shown in this section, the S_{11} and I_{out} performances of the IDCS stage are tightly related to Q_{in} . To achieve wider relative bandwidths and to maintain sufficient overall performance, additional design parameters are needed to find other ways to separately achieve adequate Q_{in} , S_{11} , and I_{out} performances. Next, two alternatives for basic IDCS LNA are presented. The first uses RC feedback and the other uses an additional common-gate signal path.

3.3.2 IDCS LNA with RC-feedback

An IDCS LNA having RC feedback connected from the input to the output node of the cascode stage is shown in Fig. 3.14a. It has been used in several wideband LNAs, in [7] and [62], for example. In addition, the method to connect the RC feedback to the intermediate node of the shunt-peak load as is shown in Fig. 3.14b is proposed in [P4]. When the feedback is connected to the inductor instead of the output node, the parasitic capacitance at the output node is minimized. The parasitic capacitance at the node between the R_{load} and L_{load} slightly modifies the operation of the shunt-peak load, as is discussed in more detail in Section 3.5.2. According to simulations, there is no significant difference in terms of input matching or noise performance when comparing the structures shown in Fig. 3.14.



Fig. 3.14. IDCS LNA with RC-feedback topologies a) [62], b) [P4].

To simplify the analysis for LNAs shown in Fig. 3.14, the following modifications are made. According to simulations, C_{fb} has only a small effect and can be removed if it is large compared to intrinsic MOS capacitances. In addition, the cascode stage has an insignificant effect on overall LNA performance and is removed as well. Due to the feedback, the amplifier is not unilateral anymore, and therefore the load has an impact on input matching and gain performance. The simplest case is achieved when the load consists of a plain resistor only. The resulting simple model is shown in Fig. 3.15a, while its small-signal equivalent is depicted in Fig. 3.15b.



Fig. 3.15. a) Simple IDCS LNA with resistor shunt-feedback, b) its small-signal equivalent.

The input impedance and output signal current can be evaluated with (A.1) and (A.3) found in Appendix A. Despite the simple model shown in Fig. 3.15b, the component values to achieve optimal S_{II} or targeted insertion gain are difficult to present with closed-form formulas. Therefore, the effect of the feedback to the LNA performance is studied with simulations, while the component values are found by means of a manual search. The input matching is targeted at a 4 GHz frequency and the targeted maximum insertion gain is 20-dB to the load impedance of 300 Ω . Two simulation setups with the feedback resistor values of 1 k Ω and 2 k Ω are investigated. The corresponding simulated Q_{in} values at 4 GHz are 1.3 and 1.7. The other component values are given in Table 3.3. The simulation results are shown in Fig. 3.16. For the purpose of comparison, the performance of a basic IDCS stage without any feedbacks having $Q_{in} = 1.7$ is also shown with a solid line. The insertion gain and S_{11} with R_{fb} values of 1 k Ω and 2 k Ω are shown with dashed-dotted and dashed lines, respectively. When $R_{fb} = 2 k\Omega$, the S_{11} is better than -10 dB between 3.0 – 5.0 GHz. With the equal Q_{in} , the basic IDCS stage would cover S_{11} only in the frequency area of 3.3 – 4.9 GHz. When the value of R_{fb} is reduced to 1 k Ω , the BW_{S11} extends further.



Fig. 3.16. Insertion gain and S_{11} with different RC feedback resistor values.

Sim. #	Q_{in}	L_{in} [nH]	L_{S} [nH]	C_{gs} [fF]	g_m [mS]	R_{fb} [k Ω]
1)	1.7	1.91	1.48	468	15.9	-
2)	1.7	2.00	0.77	480	22.0	2.0
3)	1.3	1.40	0.52	500	31.4	1.0

Table 3.3. Component values for the simulation shown in Fig. 3.16.

The frequency where the maximum insertion gain of resistor feedback IDCS LNA is achieved is at a frequency lower than optimal input matching frequency ω_0 , as was found with the basic IDCS topology (3.39). However, the maximum gain frequency deviation from ω_0 is not as severe as it is with the basic IDCS. In addition, the gain deviation decreases along with lowering R_{fb} values, which is a consequence of negative feedback, i.e. the signal current through R_{fb} cancels part of the signal current provided by the transconductor M_1 . Flatter gain response is achieved, but the g_m needs to be increased to maintain the required insertion gain. Hence, utilizing R_{fb} offers a trade-off having wide S_{11} and insertion gain BWs, but it demands higher transconductance from the input transistor, which possibly increases the LNA current consumption. The R_{fb} value cannot be chosen arbitrarily low, since eventually it increases the NF of the whole LNA. A design example of a RC feedback IDCS LNA with actual IC models is presented in Section 6.5.1. In that design example, R_{fb} , with lower than 500 Ω value, becomes
one of the noisiest components in the LNA. In that design example, the RC feedback offers only moderate improvement to the wideband LNA, when S_{II} , insertion gain, and NF performances are considered.

3.3.3 IDCS with additional signal path

An alternative to the RC feedback is having a separate signal path as shown in Fig. 3.17a. When a simple small-signal model is used, the transistor M_2 isolates the output impedance. Therefore Z_L can be set to zero in (A.1) and the input impedance becomes

$$Z_{in} = sL_{in} + \frac{\left(1 + sg_m L_s + s^2 C_{gs} L_s\right) Z_{CG}}{1 + s\left(g_m L_s + C_{gs} Z_{CG}\right) + s^2 C_{gs} L_s}.$$
(3.45)

In (3.45), the Z_{GC} is the impedance of the additional CG path:

$$Z_{CG} = \frac{1}{g_{m2} + sC_{gs2}}.$$
(3.46)



Fig. 3.17. a) LNA with additional signal path, b) its small-signal equivalent.

The Q-value of the input matching circuit can be lowered with the CG signal path. According to simulations, this can be achieved with only a few mS transconductance g_{m2} and therefore the noise from M_2 can be kept small. The effect of the CG dual signal path is analyzed with simulations in three cases: 1) without additional feedback, 2) the g_{m2} of the CG transistor M_2 is set to 3 mS and the source inductor L_S value is set to 0.5 nH, and 3) the $g_{m,2}$ is kept at 3 mS but the L_S set to 1.0 nH. The other components are adjusted to achieve 20-dB insertion gain and optimal S_{11} and the component values are given in Table 3.4. The simulated results are shown in Fig. 3.18. The performance of the basic IDCS stage with $Q_{in} = 2$ is shown with a solid line and simulation cases 2) and 3) are shown with dashed and dashed-dotted lines, respectively. The simulated Q_{in} values at 4 GHz frequency are 2.1 and 1.5 for cases 2) and 3), respectively. According to case 2), although the Q_{in} is increased compared to case 1), the maximum insertion gain frequency is closer to optimal matching frequency and BW_{S11} is slightly wider. With the setup shown with a dashed-dotted line, the Q_{in} is further lowered to 1.5. As a result, the BW_{S11} increases but, due to larger L_S , the maximum insertion gain frequency is lower compared to case

2). Although g_{m2} is kept at 3 mS in cases 2) and 3), the S_{11} and insertion gain performances can be altered with the choice of other components. When comparing case 3) to the IDCS with resistor-feedback case 2), similar S_{11} and insertion performances are achieved and the differences between component values are small. However, compared to the IDCS LNA with the resistive feedback presented in Section 3.3.1.3, the component values can be adjusted more freely with this LNA. In addition, due to the high output impedance of the additional CG path, the reverse isolation is not degraded.



Fig. 3.18. S_{11} with different CG path simulation setups.

Table 3.4. Component values for the simulation shown in Fig. 3.18.

Sim. #	Q_{in}	L_{in} [nH]	$L_{S}[nH]$	C_{gs} [fF]	g_{ml} [mS]	g_{m2} [mS]
1)	2.0	2.60	1.38	398	14.4	-
2)	2.1	3.26	0.50	340	18.3	3.00
3)	1.5	2.00	1.00	430	21.4	3.00

3.3.3.1 Comparison of basic IDCS LNA and its alternatives

Two LNA alternatives using a basic IDCS topology as a basis were presented. RC-feedback or an additional CG signal path provides an additional degree of freedom to optimize the S_{11} , gain, and noise performances. A design example, which compares the basic IDCS LNA topology to these two alternatives with actual IC component models, is presented in Section 6.5.1. The input stage Q-value can be decreased by lowering the feedback resistor value or by increasing the g_m of the additional CG path. In that design example, wider S_{11} is achieved at a cost of decreased gain and increased noise. In addition, only one UWB band group can be covered with a single input and the two alternatives can offer only a moderate improvement to the basic topology. If there is a need to cover wider bandwidth with a single LNA, IDCS topology or its alternatives do not offer sufficient performance and other options should be found.

3.3.4 LC ladder matching network

Fig. 3.19 shows one of the first CMOS LNAs designed for the UWB [63]. The core of that LNA is also based on IDCS topology. To have a wide operation BW, a Chebyshev input matching technique and a shunt-peaked load are utilized. The drawback of using wideband input matching is the number of on-chip inductors. Wideband input matching circuit in Fig. 3.19 becomes more troublesome when designing a balanced or differential LNA. Then, all inductors at the input matching cannot be differential, which leads to an even larger number of inductors. To minimize the noise from the input matching circuit and to prevent the current signal leakage into the resonator L_2C_2 , inductors having high Q-value are required. As a result, the area of the input matching circuit becomes large. The input impedance, gain, and noise performance of the LNA with Chebyshev input matching are studied in, for example, [63] and [64]. Despite its shortcoming, the Chebyshev input matching technique is an option to be considered, when wide relative gain and S_{II} bandwidths are needed.



Fig. 3.19. Wideband IDCS-LNA with Chebyshev input matching network [63].

3.3.5 CS with shunt-resistor feedback

In Section 3.3.1 it was found that IDCS LNA cannot provide adequate S_{II} and gain performance for applications having large relative BWs. In addition, neither of its variants presented in Sections 3.3.2 and 3.3.3 could improve the performance significantly, since the source inductor limits the achievable bandwidths and causes the maximum output signal current to be at lower frequency compared to optimal S_{II} frequency. When the source inductor is removed, other input matching techniques are needed. In noise sensitive applications, the resistive input termination is not an option. Therefore, the LNA input matching realization by using feedback is studied. The simplest feedback topology is a resistor (shunt) feedback connected from the gate to the drain of the input transistor, as is shown in Fig. 3.20. For simplicity, the overlap capacitor C_{gd} of M_I and the parasitic capacitance at the output node are neglected. Although the LNA shown in Fig. 3.20 seems to be a special case of the one shown in Fig. 3.15 having $L_S = 0$, the removal of the source inductor greatly alters the performance of the whole LNA and its input matching circuit. Therefore, it is justified to analyze the LNA of Fig. 3.20 separately.



Fig. 3.20. a) Shunt-resistor feedback LNA with resistive load, b) its simple small-signal equivalent circuit.

Typically, the reactive components shown in Fig. 3.20a are neglected when the gain and input matching equations of the shunt-feedback amplifier are presented in the literature. However, the reactive components should be taken into account, since they play a critical role what comes to the performance of the LNA. The capacitor C_{gs} includes the intrinsic capacitance of the input transistor M_I , the parasitic capacitances caused by the pad (not shown in Fig. 3.20) and metal wiring etc. The input inductor L_{in} can be realized with a bondwire, for example. The input impedance of the LNA presented in Fig. 3.20 can be calculated by using a general input impedance equation (A.1):

$$Z_{in} = sL_{in} + \frac{R_{fb} + R_L}{1 + g_m R_L + sC_{gs} \left(R_{fb} + R_L\right)}.$$
(3.47)

The rightmost part of (3.47) has an imaginary part, which can be resonated out with L_{in} . As a result, the input matching can have real value without the source inductance. The input matching is met with the following criteria:

$$R_{in} = \frac{L_{in}}{C_{gs}} \left(\frac{1 + g_m R_L}{R_{fb} + R_L} \right),$$
(3.48)

$$\omega_0^2 = \frac{1}{L_{in}C_{gs}} \left[1 - \frac{L_{in}}{C_{gs}} \left(\frac{1 + g_m R_L}{R_{fb} + R_L} \right)^2 \right] = \frac{1}{L_{in}C_{gs}} \left(1 - \frac{C_{gs}}{L_{in}} R_{in}^2 \right).$$
(3.49)

The maximum output current signal is achieved at the frequency of (3.49) and is

$$I_{out} = \frac{C_{gs}}{L_{in}} \left(\frac{g_m R_{fb} - 1}{g_m R_L + 1} \right).$$
(3.50)

According to the equations above, R_{in} , ω_b and I_{out} depend on the reactive and resistive component values and on the g_m of the input transistor. Compared to typical IDCS LNAs, there are more components, which affect the relevant design parameters. Since C_{gs} , L_{in} , resistors R_{fb} and R_L , and g_m do not track each other, resistor-feedback LNA is more sensitive to component value deviation from nominal values than basic IDCS topology.

Next, the S_{II} and output signal current bandwidths of the shunt-feedback CS are studied. The corner frequencies, where the output signal current is decreased by 3 dB from the maximum value and S_{II} achieves the –10-dB limit, are

$$\omega_{I_{out},\max-3dB} = \sqrt{\omega_0^2 \pm 2R_{in} \frac{1}{L_{in}\sqrt{L_{in}C_{gs}}}} = \omega_0 \sqrt{1 \pm \frac{2}{Q_{in} - \frac{1}{Q_{in}}}},$$
(3.51)

$$\omega_{S_{11}=-10dB} = \sqrt{\omega_0^2 \pm \frac{2}{3} R_{in} \frac{1}{L_{in} \sqrt{L_{in} C_{gs}}}} = \omega_0 \sqrt{1 \pm \frac{2}{3} \cdot \frac{1}{Q_{in} - \frac{1}{Q_{in}}}},$$
(3.52)

respectively. In previous equations, Q_{in} is the Q-value of the input network at ω_0

$$Q_{in} = \left| \frac{V_{gs}}{V_{in}} \right|_{\omega = \omega_0} = \sqrt{\frac{C_{gs}}{L_{in}}} \left(\frac{R_{fb} + R_L}{1 + g_m R_L} \right) = \sqrt{\frac{L_{in}}{C_{gs}}} \frac{1}{R_{in}}.$$
(3.53)

Fig. 3.21 presents a simulation example of the LNA shown in Fig. 3.20. The input inductor L_{in} has the values of 1 nH, 2 nH, and 4 nH and the load resistor is set to 300 Ω . The center frequency is tuned to 4 GHz and the maximum insertion gain level is set to 20 dB. The component values are given in Table 3.5. The maximum insertion gain is achieved at the same frequency with optimum S_{II} frequency. Both the insertion gain and S_{II} have wide bandwidths and the -3-dB insertion gain corner frequency always exceeds the -10-dB corner frequencies of the S_{II} . According to (3.51) and (3.52), both the insertion gain and S_{II} bandwidth increase with smaller L_{in} values. This is also predicted by the Q-value of the input network (3.53). When the value of L_{in} is small enough, the term inside the square root in (3.51) and (3.52) becomes negative when the '-' sign is applied. In that case, the I_{out} and S_{II} corner frequencies are already achieved at the zero frequency, as shown in Fig. 3.21. Then, the I_{out} and S_{II} corner frequencies given by (3.51) and (3.52) are limited by the upper corner frequency, i.e. the '+' sign is only valid.

To maximize both S_{II} and insertion gain bandwidths, the value of L_{in} should be as small as possible. However, when L_{in} is decreased, the value of the feedback resistor should be lowered to maintain the optimum matching. Due to the negative feedback, the resulting overall output signal current decreases along with lower R_{fb} values. Thus, the transconductance of M_I should be increased to maintain the targeted insertion gain. The noise of a CS LNA with resistor feedback can be approximated with [7]

$$F = 1 + \frac{R_{fb}}{R_s} \left(\frac{1 + g_{m1}R_s}{1 - g_{m1}R_{fb}}\right)^2 + \frac{1}{R_s R_{load}} \left(\frac{R_{fb} + R_s}{1 - g_{m1}R_{fb}}\right)^2 + \frac{\gamma g_{m1}}{\alpha R_s} \left(\frac{R_{fb} + R_s}{1 - g_{m1}R_{fb}}\right)^2.$$
 (3.54)

According to (3.54), the input referred noise from both the load and feedback loop increase along with lowering R_{fb} when other component values are kept constant. Equation (3.54) is suggestive only, since it does not take into account the effect of Q_{in} nor the induced gate current noise.



Fig. 3.21. The simulated insertion gain and S_{II} of the resistor feedback CS LNA. The insertion gain is shown with a solid line and S_{II} with a dashed line. Center frequency is tuned to 4 GHz and insertion gain level is set to 20 dB. Both the gain and S_{II} are presented with L_{in} values of 1 nH, 2 nH, and 4 nH.

Table 3.5. Q_{in} and component values for the simulation shown in Fig. 3.21.

Q_{in}	L_{in} [nH]	C_{in} [fF]	$R_L[\Omega]$	$R_{fb} [\Omega]$	g_m [mS]
1.1	1	319	300	622	45.7
1.4	2	398	300	810	33.5
2.2	4	317	300	1375	18.8

It can be concluded that wider S_{II} and insertion gain bandwidths can be obtained with resistor feedback CS LNA compared to basic IDCS. Since a source inductor is absent, it can also be realized with smaller die area. A disadvantage of this is that it results in high current consumption and probably degraded NF performance. An example of wideband LNA utilizing resistive feedback can be found in, for example, [65] and [66].

3.3.6 CS with common-drain feedback

Another active feedback loop is achieved by connecting a source follower (common-drain stage) output node to the input. In a general case shown in Fig. 3.22a, impedances Z_g , Z_L , and Z_{fb} represent the gate, load, and feedback impedances, respectively. The input impedance can be evaluated as

$$Z_{in} = Z_{g} + \frac{1 + g_{m2}Z_{fb} + sC_{gs2} \left(Z_{L} + Z_{fb}\right)}{g_{m2} \left(1 + g_{m1}Z_{L}\right) + sC_{gs1} \left(1 + g_{m2}Z_{fb}\right) + sC_{gs2} \left[\left(1 + g_{m1}Z_{L}\right) + sC_{gs1} \left(Z_{L} + Z_{fb}\right)\right]}.$$
 (3.55)



Fig. 3.22. CS amplifier utilizing common-drain feedback, a) general topology, b) Z_G is replaced with bondwire inductance and Z_{GD} and Z_L are replaced with resistors, c) small-signal representation of b).

In a simple case shown in Fig. 3.22b, Z_L and Z_{fb} are realized with resistors R_L and R_{fb} , respectively. The right-hand side without term Z_g has both real and imaginary parts. The real value input impedance can be achieved by resonating the imaginary part with input inductance L_{in} . If C_{gs2} is neglected for the simplicity, the input matching is met with the following criteria:

$$R_{in} = g_{m2} \frac{L_{in}}{C_{gs1}} \left(\frac{1 + g_{m1} R_L}{1 + g_{m2} R_{fb}} \right),$$
(3.56)

$$\omega_0^2 = \frac{1}{C_{gs1}L_{in}} \left[1 - g_{m2}^2 \frac{L_{in}}{C_{gs1}} \left(\frac{1 + g_{m1}R_L}{1 + g_{m2}R_{fb}} \right)^2 \right] = \frac{1}{C_{gs1}L_{in}} \left(1 - \frac{C_{gs1}}{L_{in}} R_{in}^2 \right).$$
(3.57)

The maximum output signal current is achieved at a frequency given by (3.57). In a perfectly matched case this is

$$I_{out} = \frac{g_{m1}}{g_{m2}} \frac{1 + g_{m2} R_{fb}}{1 + g_{m1} R_L} \sqrt{\frac{C_{gs1}}{L_{in}}} .$$
(3.58)

The maximum insertion gain $A_{i,max}$ can be re-expressed with the load resistor R_L and by using (3.56) and (3.57):

$$A_{i,\max} = R_L I_{out} = R_L \frac{g_{m1}}{R_{in}} \sqrt{\frac{L_{in}}{C_{gs1}}} = g_{m1} R_L \sqrt{1 + \left(\frac{L_{in}\omega_0}{R_{in}}\right)^2} .$$
(3.59)

Now, there are six parameters to set three equations. The input matching frequency is tuned at the wanted frequency with L_{in} and C_{in} by (3.57). Then, the maximum gain is set to the required level with g_{m1} and R_L (3.59). Finally, the input matching is set to the desired input impedance with g_{m2} and R_{fb} according to (3.56).

In Fig. 3.23, a simulation example of the LNA shown in Fig. 3.22 is shown. The input matching is tuned at 4 GHz frequency and the input bondwire has the values of 1 nH, 2 nH, and 4 nH, respectively. As in the previous LNA examples, the load resistor value is set to 300 Ω and the other component values are calculated such that the maximum insertion gain of 20 dB is achieved. The result is shown in Fig. 3.23. As was the case with resistor feedback CS LNA, the

maximum bandwidths are obtained by choosing the inductor L_{in} value as small as possible. The choice of g_{m2} or R_{fb} does not affect the gain or input matching bandwidths. Of course, the noise contributions of M_2 and R_{fb} should be taken into account. The noise formulas for a feedback LNA are given in the literature, in [2], [67], and [68], for example. However, the presented noise analyses give suggestive results only, since all the noise sources are not included. Therefore, to achieve a sufficient noise figure, optimum sizing for M_2 and R_{fb} is found with a circuit simulator. Typically, the transistor M_1 makes the major noise contribution, while the effect of M_2 or R_{fb} is small but non-negligible.



Fig. 3.23. The simulated insertion gain and S_{II} of the common-drain feedback CS LNA. The insertion gain is shown with a solid line and S_{II} with a dashed line. Center frequency is tuned to 4 GHz and insertion gain level is set to 20 dB. Both the gain and S_{II} are presented with L_{in} values of 1 nH, 2 nH, and 4 nH.

L_{in} [nH]	C_{in} [fF]	$R_L[\Omega]$	$R_{fb}[\Omega]$	g_{ml} [mS]	g_{m2} [mS]
1	319	300	300	29.8	3.10
2	398	300	300	23.5	1.96
4	317	300	300	14.8	0.93

Table 3.6. Component values for the simulation shown in Fig. 3.23.

When Fig. 3.21 and Fig. 3.23 are compared, the gain and S_{II} curves are found to be identical. Therefore, in a fully matched case, the bandwidths of resistor feedback and common-drain feedback CS LNAs are set by the reactive input matching components L_{in} and C_{in} , i.e. as is given by (3.51), (3.52), and (3.53). The common-drain (CD) feedback LNA requires smaller transconductance than the resistor feedback LNA to achieve similar gain level. In addition, there is more freedom to choose the appropriate component values for a CD feedback LNA compared to resistor feedback CS LNA. Therefore, CD feedback LNA is better suited for wideband applications. It is utilized, for example, in [2], [67], [68], and [69].

3.3.7 Wideband common-gate LNA

As was shown in Section 3.3.1, the drawback of a typical IDCS LNA is its restricted input matching capabilities, which limit its usage in wideband solutions. To mitigate that problem, the Chebyshev input matching network can be used, but the use of several on-chip inductors degrades the noise performance. The CG input stage, however, offers rather simple input matching realization. The input resistance at the MOSFET source is inversely proportional to the transconductance g_m and the resulting impedance match is wideband [70]. Due to the simple input matching circuit, the NF difference between wideband CG and IDCS LNAs becomes smaller than in narrowband systems. Therefore, CG LNA is a viable option for applications requiring wide operational bands, for example UWB [52], [55].

In Section 3.2.3, the performance of the CG LNA was analyzed at the center frequency only and the operation as a function of frequency was neglected. In this section, the wideband CG LNA design is analyzed in detail taking into account the effect of the input matching network. The CG-LNA is analyzed with the circuit shown in Fig. 3.24. The transistor M_I is replaced with a simple small-signal model consisting of gate-source capacitor C_{gs} and transconductance g_m . The g_m includes also the substrate transconductance g_{mb} , i.e. the effective transconductance $g_{m.eff}$ of a CG stage is $g_{m.eff} = g_m + g_{mb}$. For simplicity, the $g_{m.eff}$ is marked with a g_m only in all figures and equations. The capacitor C_{par} includes the parasitic capacitances at the source node and all the capacitances at the source node are included in a single capacitance C_T , as is given in (3.16).



Fig. 3.24. CG stage small-signal model used in analysis.

Next, design parameters related to the source and series input resonators are defined. The Q-value of an ideal lossless parallel resonator formed by L_S and C_T is infinite, but resonators characteristic impedance is expressed as

$$Z_{LC} = \sqrt{\frac{L_s}{C_T}} . \tag{3.60}$$

By combining (3.18) and (3.60), the Z_{LC} can be given as

$$Z_{LC} = \omega_0 L_S = \frac{1}{\omega_0 C_T} \,. \tag{3.61}$$

Furthermore, when the relative characteristic impedance of the LC source resonator is scaled with respect to the source impedance Z_0 , an additional parameter Z_{rel} is introduced as

$$Z_{rel} = \frac{Z_{LC}}{Z_0}.$$
 (3.62)

In a matched case, i.e. $g_m = 1/Z_0$, when (3.34) and (3.62) are compared, there seems to be a relation $Z_{rel} = 1/Q_{in}$ between Z_{rel} and the Q-value of an IDCS stage. However, this is not a case. Z_{rel} depends on the values of source components L_S and C_T and on input matching frequency ω_0 and typically it has values between 0.5 and 2. However, as defined earlier, the Q_{in} is a ratio between gate-source and input voltages. In Section 3.3.7.1, Q_{in} of a CG stage is shown to be 1 at optimum input matching frequency. In that sense, Q_{in} is not a suitable parameter to analyze CG stage. Thus, in the following analysis, Z_{rel} is used instead.

Next, the input resonator components L_{in} and C_{in} are related to source resonator components L_S and C_T . The input inductance value L_{in} is scaled with respect to L_S with a design parameter p, such that $L_{in} = L_S / p$. To maximize the S_{11} and I_{out} bandwidths, the series resonator should be tuned at the same frequency with the source parallel resonator. Therefore $C_{in} = pC_T$ is needed. As a result, the input impedance of the circuit shown in Fig. 3.24 given in (3.19) is expressed as

$$Z_{in} = \frac{1 + s^2 L_S C_T}{spC_T} + \frac{sL_s}{1 + sL_s g_m + s^2 L_S C_T}.$$
(3.63)

Next, the Q-value of the input matching network, the input matching and output signal current BWs, and noise performance of the CG stage are analyzed with design parameters Z_{rel} and p.

3.3.7.1 Input network Q-value

As was discussed earlier, the input matching and output signal current bandwidths are typically related to the Q-value of the input matching network at center input matching frequency ω_0 . For an IDCS LNA, for example, the input matching network is a series resonator and Q_{in} is typically larger than two. The fundamental difference of the CG-LNA is that the matching network is a parallel resonator, and therefore its Q-value is lower than that of an IDCS stage [58]. For the CG stage shown in Fig. 3.7b, the gate-source voltage of M_1 is equal to input voltage signal, i.e. Q_{in} is simply one. However, the input series resonator shown in Fig. 3.7c has an effect on the overall Q-value of the CG input. The Q-value of the CG input shown in Fig. 3.24 can be solved as

$$Q_{in,CG} = \frac{V_S}{V_{in}} = p \left(\frac{\omega}{\omega_0}\right)^2 \frac{1}{\sqrt{\left(1 - \left(2 + p\right)\left(\frac{\omega}{\omega_0}\right)^2 + \left(\frac{\omega}{\omega_0}\right)^4\right)^2 + Z_{rel}\frac{\omega}{\omega_0}\left(1 - \left(\frac{\omega}{\omega_0}\right)^2\right)^2\right)^2}}.$$
 (3.64)

The $Q_{in,CG}$ always equals one at the center frequency ω_0 regardless of p or Z_{rel} values. This is clear, since at ω_0 , input and resonator resonators are short-circuit and open circuit, respectively. Although the $Q_{in,CG}$ is one at ω_0 , the performance of the CG stage can be affected by the design of input and source resonator, i.e. with p or Z_{rel} variables. By calculating the local optimums of (3.64), it can be shown that, when $Z_{rel} < 2\sqrt{p}$, the input matching network offers voltage gain. The maximum Q-value is obtained at

$$\omega_{\mathcal{Q}_{m,CG},\max} = \frac{\omega_0}{2} \sqrt{\left(4 + 2p - Z_{rel}^2\right) \pm \sqrt{\left(2p - Z_{rel}^2\right) \left(8 + 2p - Z_{rel}^2\right)}}$$
(3.65)

and at those frequencies the maximum Q-value is

$$Q_{CG_{in},\max} = \frac{2p}{Z_{rel}\sqrt{4p - Z_{rel}^2}}.$$
(3.66)

The Q-values of a CG input is plotted in Fig. 3.25 as a function of frequency with several p values such that $Z_{rel} = 1$. The center frequency is scaled to 1. The peaking causes linearity degradation according to (3.14). This is depicted in the design example shown in Section 6.5.2. The effect of $Q_{in,CG}$ on the noise performance is discussed later in Section 3.3.7.4.



Fig. 3.25. Q-value of the input matching network. The input matching network is simulated with the following design values: $Z_{rel} = 1$ and p has values of $\frac{1}{2}$, 1, and 2.

3.3.7.2 Input matching bandwidth

First, the effect of the series LC resonator formed by C_{in} and L_{in} is neglected, and only the effect of the source resonator in considered. The input impedance is given by (3.17). Assuming a perfect input matching ($g_m=1/Z_0$) at ω_0 , the corner frequencies where the typical -10 dB S_{11} limit is achieved can be calculated as

$$\omega_{S11=-10dB} = \frac{\omega_0}{3} \sqrt{9 + 2Z_{rel}^2 \left(1 \pm \sqrt{1 + \frac{9}{Z_{rel}^2}}\right)}.$$
(3.67)

In (3.67), Z_{rel} is given by (3.62). The bandwidth, where S_{II} is better than -10 dB, is achieved by subtracting the two frequencies given by (3.67). The result is scaled with respect to center frequency ω_0 giving the relative input matching bandwidth $BW_{rel,SII}$:

$$BW_{rel,S11} = \frac{BW_{S11}}{\omega_0} = \frac{2}{3}Z_{rel} .$$
(3.68)

Because the center frequency ω_0 and source impedance Z_0 are fixed, the relative bandwidth can be widened by increasing the Z_{LC} value. According to (3.61), the maximum value of Z_{LC} is achieved when the value of C_T is minimized, i.e. the additional shunt capacitor C_S is omitted and the value of L_S is increased such that the resonator is tuned at the wanted frequency. In that case, the center frequency is susceptible to additional parasitic capacitance caused by layout, for example. In addition, it should be noted that the series resistance of the inductor L_S is not taken into account in the simple model shown in Fig. 3.24. Since the impedance level at the source node is low due to $1/g_m$ of the input transistor, the finite Q-value of the L_S has only a slight effect on the input matching bandwidth, as is shown in Section 6.5.2. According to simulations, $Q_{LS} > 5$ is sufficient not to cause significant error in the BW result achieved by the theory. Such an inductor Q-value is easily achieved with current technology. As is discussed later in Section 3.3.7.4, the choice of the inductor L_S value plays a significant role regarding CG stage noise performance.

An example of BW_{SII} is shown in Fig. 3.26. The S_{II} performance is shown with a solid line for a case where the source parallel resonator is taken into account and $Z_{rel} = 1$.



Fig. 3.26. S_{II} bandwidth as a function of normalized frequency. The solid line shows the performance when the source parallel resonator is taken into account ($Z_{rel} = 1$) and dashed and dashed-dotted lines show the performance parameter *p* values of 1 and 2, respectively.

Next, the effect of input parallel resonator formed by L_{in} and C_{in} is taken into account. The input impedance is given by (3.63). The frequencies where the $-10 \text{ dB } S_{II}$ limit is achieved are

$$\omega_{S11=-10dB} = \frac{\omega_0}{2\sqrt{3}} \sqrt{F_1 + F_2 \pm \sqrt{2}\sqrt{(F_1 + F_2)F_2 + F_3}} , \qquad (3.69)$$

where

$$F_1 = -3F_4 + 12 , (3.70)$$

$$F_2 = \sqrt{9F_4^2 + 16p^2} , \qquad (3.71)$$

$$F_3 = -4(9F_4 + 2p^2), \qquad (3.72)$$

and

$$F_4 = \left(\frac{p - Z_{rel}^2}{Z_{rel}}\right)^2.$$
 (3.73)

The relative input matching bandwidth is shown in Fig. 3.27 as a function of p with several Z_{rel} values.



Fig. 3.27. Relative S_{11} bandwidth as a function of resonator design parameter p with several source resonator relative impedance Z_{rel} values.

The maximum $BW_{rel,S11}$ is achieved, when

$$p = 2Z_{rel}^2$$
, (3.74)

and then

$$BW_{rel,S11,\max} = \sqrt{\frac{1}{6} \left(\sqrt{73} - 3\right)} Z_{rel} \approx 0.96 \cdot Z_{rel} .$$
(3.75)

Fig. 3.26 shows the S_{II} , when $Z_{rel} = 1$. A case, where both the input and source resonators have equal component values (p = 1) is shown with a dashed line. According to (3.74), when $Z_{rel} = 1$, the $BW_{rel,SII}$ achieves its maximum with p = 2. This is depicted with a dashed-dotted line in Fig. 3.26.

Fig. 3.28 presents the $BW_{rel,SII}$ as a function of Z_{rel} . The solid line shows the effect of the source parallel resonator only as is given by (3.68). The performance of the whole matching circuit with several series resonator design parameter p values is shown with a dashed line according to (3.69). With certain p and Z_{rel} values, it is possible to achieve wider $BW_{rel,SII}$ with both resonators than with the source resonator only. To extend the $BW_{rel,SII}$, the Z_{rel} of the source resonator should be increased. In addition, the design parameter p should be chosen appropriately not to limit the overall $BW_{rel,SII}$. The Z_{rel} increases proportionally to the source inductor value. Because the optimal value for p depends quadratically on Z_{rel} , the value of input inductor L_{in} is inversely proportional to the value of L_S . The choice of optimal Z_{rel} and p values is not unambiguous when $BW_{rel,SII}$ is known. For example, the minimum and maximum inductor values are typically limited by the technology used.



Fig. 3.28. Relative input matching bandwidth as a function of source resonator relative impedance Z_{rel} with several resonator design parameter p values. The solid line shows the effect of the source parallel resonator only.

3.3.7.3 Output signal current of CG stage

The CG stage transforms the voltage signal V_s sensed at the source node to the drain current. At a resonance frequency ω_0 , the output current $I_{out} = g_m V_s = g_m V_{in}$, where V_{in} is the voltage signal fed to the LNA input. Both the input and source resonators, however, affect the I_{out} as a function of frequency. The output signal current of the CG stage achieves its maximum value at ω_0 . Then such corner frequencies are calculated where the output signal current is decreased by 3 dB compared to the signal current at the resonance frequency. When the effect of the source resonator is considered only, the -3-dB corner frequencies are

$$\omega_{lout,\max-3dB} = \omega_0 \left(\sqrt{1 + Z_{rel}^2} \pm Z_{rel} \right). \tag{3.76}$$

When the -3-dB corner frequencies are scaled with respect to ω_0 and the BW is calculated by subtracting the two frequencies given by (3.76), the relative -3-dB output signal current BW is:

$$BW_{rel,lout} = \frac{BW_{lout}}{\omega_0} = 2Z_{rel}.$$
(3.77)

According to (3.77), the output signal current BW improves linearly with source resonator impedance level. An example of a BW_{lout} of a CG stage having $Z_{rel} = 1$ is shown in Fig. 3.29 with a solid line.



Fig. 3.29. Bandwidth of I_{out} as a function of normalized frequency. The solid line shows the performance when the source parallel resonator is taken into account $(Z_{rel} = 1)$ and dashed and dashed-dotted lines show the performance with parameter *p* having values of 1 and 6, respectively.

When the effect of the series resonator is taken into account, the relative -3-dB output current BW is

$$BW_{rel,lout} = \frac{1}{2}\sqrt{F_5 - F_6 \pm \sqrt{2}}\sqrt{\left(F_6^2 + 4F_7 - 4\right) - \frac{F_6}{F_5}\left(F_6^2 + 8F_7 + 8\right)} , \qquad (3.78)$$

where

$$F_5 = \sqrt{16p^2 + F_4^2} , \qquad (3.79)$$

$$F_6 = F_4 - 4 , (3.80)$$

$$F_7 = F_4 + 2p^2 - 3, (3.81)$$

and F_4 is given by (3.73). The $BW_{rel,lout}$ is plotted as a function of p with several Z_{rel} values in Fig. 3.30. The maximum $BW_{rel,lout}$ is achieved, when

$$p = 6Z_{rel}^2, \qquad (3.82)$$

which results in

$$BW_{rel,lout,\max} = \sqrt{\frac{1}{2} \left(\sqrt{1201} - 25\right)} Z_{rel} \approx 2.2 \cdot Z_{rel} .$$
(3.83)

Compared to the $BW_{rel,SII}$, the $BW_{rel,Iout}$ is wider with equal Z_{rel} and p values. Therefore, it is more challenging to achieve S_{II} better than -10 dB than to gain a flatness of 3 dB. However, when the effect of the load impedance is taken into account, the gain can be limited at high frequencies due to the parasitic capacitance at the output node, as is shown in the design example presented in Section 6.5.2.



Fig. 3.30. Relative -3-dB output signal current bandwidth as a function of resonator design parameter p with several Z_{rel} values.

The $BW_{rel,lout}$ is shown as a function of Z_{rel} in Fig. 3.31. The solid line shows the effect of the source parallel resonator only (3.77) and the performance of the whole matching circuit with several series resonator design parameter p values is shown with a dashed line according to (3.78). For example, if both the input and source resonators have equal component values (p = 1), the $BW_{rel,lout}$ is approximately 1.4 when Z_{rel} is 1. As is presented in Fig. 3.29 and Fig. 3.31, in that case the $BW_{rel,lout}$ is narrower than with the source resonator case only. According to (3.82), the $BW_{rel,lout}$ is at its maximum when p = 6 and Z_{rel} is 1. This is depicted with a dashed-dotted line in Fig. 3.29. As can be seen in Fig. 3.29 and Fig. 3.31, when the effect of both resonators is taken into account, BW_{lout} is only slightly wider compared to the case where only the effect of the source resonator in considered.



Fig. 3.31. Relative bandwidth of I_{out} as a function of source resonator relative impedance Z_{rel} with several resonator design parameter p values. The solid line shows the effect of the source parallel resonator only.

3.3.7.4 Noise figure

In Section 3.2.3.2, the noise of a CG stage was considered at a center frequency only. In this sub-section, the effect of source and input resonators to CG stage noise behavior in the frequency domain are analyzed. First, the noise caused by input transistor is considered only as shown in Fig. 3.32a. Then, the effect of the finite source inductor series resistance is taken into account as depicted in Fig. 3.32b. Finally, the noise of the CG stage is analyzed in a case when the input series resonator presented in Fig. 3.32c is included but the inductors' series resistances are ignored. The gate induced noise, the substrate transconductance, and the noise due to load are ignored in all cases for the sake of simplicity.



Fig. 3.32. a) Equivalent circuit to analyze the noise of CG transistor only, b) the effect of finite source inductor series resistance is taken into account, c) the effect of the input series resonator is taken into account.

When the effect of source resonator is taken into account (Fig. 3.32a), assuming $\alpha = 1$, the noise factor of a CG LNA is

$$F = 1 + \frac{\gamma}{g_m R_s} \left[1 + \left(\frac{R_s}{L_s \omega}\right)^2 \left(1 - \left(\frac{\omega}{\omega_0}\right)^2 \right)^2 \right].$$
(3.84)

The noise figure of a CG stage is presented in Fig. 3.33 such that $\gamma = 2/3$, $R_S = 50 \Omega$, and $g_m = 1/R_S$. L_S had values of 1 nH, 2 nH, and 4 nH. The capacitor values of the source resonator are calculated such that the resonance is achieved at $f_0 = 4$ GHz. The minimum noise figure is independent of the source resonator characteristic impedance at the resonance frequency. At other frequencies, however, the source resonator plays a significant role. The frequency bandwidth, where NF is kept at a particular limit, can be increased by maximizing the L_S value, i.e. by having as large a source resonator characteristic impedance as possible.



Fig. 3.33. Noise factor as a function of frequency. $\gamma = 2/3$, $f_0 = 4$ GHz, and $L_s = 1$ nH, 2 nH, and 4 nH.

The NF equations above assume that no signal loss is caused at the transistor source node. In practice, when the source is equipped with an LC resonator, the finite series resistance of L_S causes both signal leakage to the resonator and increases the noise of its own (Fig. 3.32b). When the effect of the finite series resistance R_{Ls} of the source inductor is taken into account, assuming $\alpha = 1$, the noise factor of a CG stage can be expressed as

$$F = 1 + \frac{\gamma}{g_m R_s} \left[1 + \frac{R_s^2}{R_{Ls}^2 + L_s^2 \omega^2} \left(1 - \left(\frac{\omega}{\omega_0}\right)^2 \right)^2 + \frac{R_{Ls} R_s}{R_{Ls}^2 + L_s^2 \omega^2} \left(2 + R_{Ls} g_m C_T^2 \omega^2 \right) \right] + \frac{R_{Ls} R_s}{R_{Ls}^2 + L_s^2 \omega^2} .$$
(3.85)

The last term in (3.85) is the noise due to the series resistance R_{Ls} of the source inductor itself. In addition, several terms inside the brackets are also affected by R_{Ls} . Therefore, the R_{Ls} affects the noise performance of a CG stage due to modified noise transfer functions. For example, the finite R_{Ls} affects the frequency where the minimum noise figure is eventually achieved. If $g_m R_{Ls} / \gamma + 2C_T R_{Ls}^2 / L_S + 2R_{Ls} / R_S <<1$, the minimum noise factor frequency is approximately

$$\omega_{F_{\min}} \approx \frac{1}{\sqrt{L_S C_T}} \sqrt{1 + R_{LS} \left(\frac{g_m}{2\gamma} + \frac{1}{R_S}\right)}$$
(3.86)

and at that frequency the minimum noise factor is approximately

$$F_{\min} \approx 1 + \frac{\gamma}{g_m R_s} + \frac{C_T R_{Ls}}{g_m L_s} \left(2\gamma + \gamma \frac{C_T R_{Ls} R_s}{L_s} + g_m R_s \right).$$
(3.87)

In a matched case $(1/g_m = R_S)$, (3.87) can be re-expressed as

$$F_{\min} = 1 + \gamma + \frac{1}{Q_L Z_{rel}} \left(1 + 2\gamma + \frac{\gamma}{Q_L Z_{rel}} \right), \tag{3.88}$$

where Q_L is the Q-value of the source inductor defined in (3.6) and Z_{rel} is the characteristic impedance of the source resonator as given by (3.62). The minimum noise figure of a CG LNA

with γ values 2/3, 1, and 3/2 is shown in Fig. 3.34 as a function of $Q_L Z_{rel}$. It is crucial to have both a high inductor Q-value and large source impedance value to minimize the NF.



Fig. 3.34. Noise figure of a CG LNA described by source resonator performance. The noise parameter γ has values of 2/3, 1, and 3/2.

The noise factor of a CG stage is plotted in Fig. 3.35 with the same component values as in Fig. 3.33, except the source inductor Q-value is fixed at the moderately low value of 5 at 4 GHz frequency. According to Fig. 3.34, the minimum noise factor improves along with larger source inductor values. Due to finite R_{Ls} , the minimum noise figure frequency also shifts to higher frequency according to (3.86).



Fig. 3.35. Noise factor as a function of frequency. $\gamma = 2/3$, $L_s = 1$ nH, 2 nH, and 4, nH. The capacitor value is calculated such that $f_0 = 4$ GHz. The Q-value of the inductor is set to 5 at $f_0 = 4$ GHz.

The series resonator (Fig. 3.7c) affects the noise figure of the CG-state as well. In a matched case ($g_m = 1/R_S$) and assuming $\alpha = 1$, considering the noise of the input transistor only (the series resistances of inductors are neglected for simplicity), the noise factor for the circuit shown in Fig. 3.24 is expressed as

$$F = 1 + \gamma \left[1 + \frac{\left(1 - \left(\frac{\omega}{\omega_0}\right)^2\right)^2 \left(1 + \left(\frac{p^2}{Z_{rel}^2} - 2 - 2p\right) \left(\frac{\omega}{\omega_0}\right)^2 + \left(\frac{\omega}{\omega_0}\right)^4\right)}{\left(p^2 \left(\frac{\omega}{\omega_0}\right)^4\right)} \right].$$
 (3.89)

At $\omega = \omega_0$ (3.89) simplifies to 1+ γ . The noise figure is plotted as a function of frequency such that $Z_{rel} = 1$ and p has values of 1, 2, and 4. With certain p and Z_{rel} values, the input matching circuit offers voltage gain and the noise factor of a CG stage is less than the classical limit of 1+ γ . There are two frequencies, where the minimum noise factors are achieved:

$$\omega_{F_{\rm min}} = \frac{\omega_0}{2Z_{rel}} \sqrt{\left(4Z_{rel}^2 + 2pZ_{rel}^2 - p^2\right) \pm \sqrt{\left(2pZ_{rel}^2 - p^2\right)\left(8Z_{rel}^2 + 2pZ_{rel}^2 - p^2\right)}} \ . \tag{3.90}$$

At the frequencies of (3.90), the minimum noise factor is

$$F_{\min} = 1 + \gamma \frac{p}{Z_{rel}^2} \left(1 - \frac{p}{4Z_{rel}^2} \right).$$
(3.91)

Equations (3.90) and (3.91) are valid if $p < 2Z_{rel}^2$. Otherwise, the noise factor minimum of $1+\gamma$ is achieved at ω_0 . When (3.65) and (3.90) are compared, the noise factor minimum frequencies differ from the input Q-value maximum frequencies. The noise transfer functions for the noise of the source resistance and the drain current noise of the input device are different, which causes the difference in the maximum Q-value and noise factor minimum frequencies.



Fig. 3.36. NF of a wideband CG LNA ($\gamma = 2/3$). The CG-LNA is simulated with following input matching network design values: $Z_{rel} = 1$ and p has values of 1, 2, and 4.

The series resistances of the both inductors were omitted from the analysis. If the L_{in} is realized with a wirebond inductance and L_S is an on-chip inductor, the noise contribution of the former

is much smaller compared to the latter. The finite series resistance of the source inductor causes additional noise and loss, as is given by (3.85). In addition, the noise factor optimums are higher than predicted by (3.91). The effect of an input matching circuit on CG LNA noise can be observed in the design example presented in Section 6.5.2.

3.3.7.5 Conclusion and remarks

In this section, the CG LNA was analyzed in the frequency domain. The previous LNA analyses were mainly based on the Q-value of the input matching circuit at the center input matching frequency ω_0 . Since Q_{in} of the CG stage is always one at ω_0 , two design parameters, Z_{rel} and p, were introduced as suitable design/optimization variables for a CG stage. Relevant merits, such as the input matching network Q-value, the input matching and output signal current BWs, and noise performance of the CG stage, were analyzed in detail using parameters Z_{rel} and p. The analysis showed that it is more challenging to meet the -10-dB S_{11} performance than the 3-dB variation in output signal current. In addition, with certain Z_{rel} and p values the input matching network can have voltage gain, which affects both the linearity (IIP3) and noise figure performance. A design example, which shows the wideband CG LNA design procedure utilizing the presented analysis, is shown in Section 6.5.2. A wideband LNA can be realized with the CG stage, but the disadvantage is the high noise figure. If the high impedance at the source node is realized using an LC resonator, it is crucial to utilize a large-value inductor having a high Q-value to minimize the NF.

3.3.8 Reactive feedback

Recently, LNAs having multiple-GHz bandwidths have been realized with reactive feedback, e.g. [71] - [74]. The use of reactive feedback circumvents the problem associated with the large layout area of a CS LNA having the LC ladder network or noise performance of a CG LNA.

The simplified schematic of the first stage of the LNA presented in [71] is shown in Fig. 3.37a. The gate and drain inductors L_G and L_D having mutual inductance M resonate the output capacitor C_L and capacitances associated transistors M_1 and M_2 . The input matching design details can be found in [71]. Although the first stage has wideband S_{11} performance, it has gain peaks at resonance frequencies. Therefore, an additional amplifier stage is required to boost the gain at midband frequencies. This comes with a penalty of reduced linearity.

The simplified schematic of the first stage of the LNA presented in [72] is shown in Fig. 3.37b. The input impedance is approximately [72]

$$Z_{in} \approx \frac{1}{\left(1+\beta\right)\beta g_m},\tag{3.92}$$

where g_m is the transistor M_1 transconductance and β is the feedback factor, which is the inverse of the transformer's effective turns ratio (k/n, where k is the magnetic coupling factor and n is the physical turns ratio) [72]. In [72], the input transformer T_1 has approximately n of 4.5:1 and k of 0.85. The second stage is also utilized in [72] to widen the overall amplifier passband and to mitigate the Miller effect for the input stage. Current-reuse is used to minimize the current consumption. In Fig. 3.37c, the idea of replacing source and load inductors with a single feedback transformer is presented [73]. As an advantage, smaller inductor values are required and the feedback transformer boosts the gain at the source by a factor of $1 + k\sqrt{L_D/L_S}$ and an almost flat frequency response is achieved [74].



Fig. 3.37. Simplified schematics of LNAs utilizing reactive feedback, a) [71], b) [72], c) [73].

3.3.9 Comparison of wideband LNA topologies

In this sub-section, several different wideband LNA topologies are presented. In single-system narrowband applications, sufficient input matching BW is typically achieved with IDCS LNA. However, when the target is a system having large relative bandwidth, the use of an IDCS input stage can be limited. A slight improvement can be achieved with RC-feedback or additional CG signal path. The input matching BW improves along with lower feedback resistor value or increased g_m of the additional CG path. As a disadvantage, the overall gain and noise performance degrades, and therefore the two alternatives can offer only a moderate improvement to the basic topology. A design example, which compares the basic IDCS LNA topology to these two alternatives with actual IC component models, is presented in Section 6.5.1. According to that design example, the maximum BW achieved with an IDCS LNA is less than 2 GHz and thus only one UWB band group can be covered, for example.

A basic CS-LNA with shunt-resistor or common-drain feedback can offer wideband input matching and output signal current performance. Compared to the IDCS stage, the maximum output signal current is achieved at the input matching frequency. In addition, because sufficient input matching is achieved without source inductance, the LNA occupies a small layout area. However, the use of a resistor-feedback CS stage is challenging, since both the gain and S_{11} depend on the actual values of all components and thus that topology is sensitive to component-value deviation. Similar performance to the resistor feedback CS-LNA can be achieved with a common-drain feedback. It is less sensitive to the component values, and similar gain performance compared to a resistor feedback CS LNA can be achieved with a lower current consumption.

A wide operational band can be achieved with a CG stage. The NF, however, is significantly higher compared to the IDCS LNA and therefore the usage of a CG stage is limited to applications that can tolerate high NF. A design example of a wideband CG LNA is presented in Section 6.5.2.

If multi-GHz BWs are required with a single LNA, an IDCS stage with Chebyshev input matching network is one possibility. Due to several on-chip inductors it consumes significant silicon area and suffers from higher NF than a basic IDCS LNA. This can be circumvented by using reactive feedback, for example. This can offer low NF simultaneously with a small die area and low power consumption. However, it is another matter how an appropriate and well-modeled transformer is realized in practice, since transformers typically are not available in foundry-provided design kits.

As a conclusion, none of the presented LNA structures cannot be declared as the winner, but all of them have the advantages and disadvantages of their own. Therefore, the choice of a proper LNA topology depends on the requirements and it can be limited by the technology used. For example, as presented in Section 6.6, the input stage for the LNA operating in UWB BG1 was realized with the IDCS stage having an additional CG path.

3.4 Effect of bonding pad

The integrated circuits are almost always mounted either in a package or connected on a PCB. An input pad is needed to provide the interface between the chip and the outside world. Typically, in direct-conversion and low-IF receivers, the LNA input is the first and only RF port having an off-chip interface. The typical pad structure consists of a metal plate and ESD protection, as is shown in Fig. 3.38a. The equivalent circuit shown in Fig. 3.38b is achieved by replacing the transistor M_1 with a simple small signal model consisting of gate-source capacitance C_{GS} and transconductance g_m . Bondwire L_{in} is needed to connect the pad to the package lead frame or to the PCB microstrip line.



Fig. 3.38. a) Typical pad structure, b) simple equivalent circuit of a typical pad, c) modified pad structure, and d) simple equivalent circuit of a modified pad.

The capacitor C_P shown in Fig. 3.38b includes the parasitic capacitances at the gate node, i.e. capacitance caused by the bonding pads, ESD diodes, bondwire, and on-chip metal wiring. In a CG LNA, the parasitic capacitance can be included in a source resonator and therefore it can be resonated out with the source inductor. In a CS amplifier, the input parasitic capacitance has a more significant effect. The C_p transforms the real part of the LNA input impedance given by (3.2) downwards. Then, the real part of the impedance, shown in Fig. 3.38b, is approximately [31]

$$R_{in} \approx \left(\frac{C_{gs}}{C_{gs} + C_P}\right)^2 \frac{g_m L_s}{C_{gs}}, \qquad (3.93)$$

where C_{gs} , C_p , g_m , and L_s can be identified from Fig. 3.38b and R_{eq} is the equivalent input resistance. Therefore, the R_{eq} should be designed to a higher level to maintain the overall optimal input impedance level. When the input reactive components values will not be altered, the higher R_{eq} is achieved by increasing the transconductance g_m . In addition, if the overdrive voltage is kept constant to maintain the linearity of the input stage, the increasing of g_m is mainly achieved with a larger drain current. The C_p should be minimized so as not to increase the current consumption. The on-chip metal wiring can be affected with a proper layout. The ESD diodes are typically provided by the foundry, and minimizing of C_p by modifying those is difficult.

A UWB front-end design example is presented in Section 6.6. When the LNA for BG3 was designed, it was found that the effect of parasitic capacitance can be minimized by connecting the lower pad metal layer (shield) to the source of the input transistor M_I , as is shown in Fig. 3.38c. The corresponding small-signal equivalent is shown in Fig. 3.38d. The parasitic capacitance C_{PI} is in parallel with the intrinsic C_{GS} capacitance and C_{PI} can be absorbed into the input matching network. When additional capacitances C_{PI} and C_{P2} are taken into account, the input matching criteria are modified to

$$R_{in} \approx \frac{g_m L_s \left[\left(L_{in} + L_s \right) \left(C_{gs} + C_{p1} \right) + C_{p2} L_s \right]}{\left(L_{in} + L_s \right) \left(C_{gs} + C_{p1} \right)^2} \approx \frac{g_m L_s}{C_{gs} + C_{p1}},$$
(3.94)

$$\omega_0^2 \approx \frac{1}{\left(L_{in} + L_S\right)\left(C_{gs} + C_{p1}\right) + C_{p2}L_S} \approx \frac{1}{\left(L_{in} + L_S\right)\left(C_{gs} + C_{p1}\right)},$$
(3.95)

when $(L_{in}+L_s)^2 >> 4C_{p2}L_{in}L_s$. When (3.94) is compared to (3.93) the real part value degradation is smaller. As a disadvantage, the capacitors C_{p1} and C_{p2} are not modeled in design kit and therefore capacitance extraction from the layout is required. Typically, the shield is realized with the lowermost metal layer and therefore the distance between pad contact (uppermost metal layer) and shield is much larger than the distance between the shield and substrate. Therefore, $C_{p2} > C_{p1}$. That is a desired result, since it is beneficial to have C_{p1} as small as possible to neutralize the effect of the pad.

It should be noted that the simple model shown in Fig. 3.38d ignores the influence of the ESD diode. Therefore, some parasitic capacitance to ground is still remaining, but its effect is significantly reduced. In addition, the simple model shown in Fig. 3.38d results also in another resonance at the frequency of

$$\omega_0^2 \approx \frac{\left(L_{in} + L_S\right)}{C_{p2}L_{in}L_S} \tag{3.96}$$

and at that frequency the input resistance is approximately

$$R_{in} \approx -\frac{g_m L_s}{C_{gs} + C_{p1} + C_{p2}}.$$
(3.97)

Since the input impedance is negative, the network input is unstable. In practice, however, other parasitic capacitors and resistors not included to the simple model have an affect on overall input impedance and attenuate the effect on negative resistance. In addition, the frequency of (3.96) is much higher than (3.3) and therefore its effect can be neglected.

The modified pad structure was used in the BG3 input of the UWB front-end design example presented in Section 6.6.

3.5 Load design

In this sub-chapter, different load structures are studied. For narrowband applications, sufficient performance is typically achieved either with a resistive or with a damped RLC resonator. Because the bandwidths of those loads are limited, different wideband structures are briefly discussed. The main interest is on the shunt-peak load and its alternatives. The wideband load design for two-stage amplifiers is also considered. The bandwidth shrinkage in such two-stage amplifiers, where both stages utilize separate shunt-peak loads, is discussed. Finally, a modified shunt-peak load using only single on-chip coil is proposed for two-stage amplifiers.

The different load alternatives described in this sub-chapter are not limited use with LNAs only. Obviously, same load structures can be utilized with local oscillator (LO) buffers or separate mixer transconductance stages as well. In a well-designed load, the effect of the following stage is also taken into account. For example, when a proper LNA or LO buffer load is designed, the loading effect of the mixer input stage and switch transistors should be considered, respectively.

3.5.1 Narrowband loads

The simplest load alternative is an ideal resistance, which has no parasitic capacitances thus being wideband by nature. In practice, however, all monolithic devices have parasitic capacitances and therefore the resistive amplifier output load is actually an RC-load. The capacitive part is the parasitic capacitance formed by the load resistance and the output (cascode) transistor and of the input parasitic capacitance of the following stage. At the RC-pole frequency, the impedance of the load is degraded by 3 dB from the low frequency value. Therefore, the resistive loads are typically restricted to applications, where the highest operational frequency is less than 2 GHz. Another disadvantage of the resistive load is the voltage drop, which can be problematic with low supply voltages.

The parasitic capacitance can be resonated out with a parallel inductor. Typically, the load inductor is an on-chip component, which requires a significant layout area compared to the resistor load only. The simplified schematic of the amplifier having a resonator load is shown in Fig. 3.39, where the input stage is replaced with a simple transconductance element. The output

resistance of the transconductance stage is assumed much larger than the load impedance. Taking into account the series resistance R_L of the load inductor L, the resonance frequency is

$$\omega_r = \frac{1}{\sqrt{LC}} \sqrt{1 - \frac{R_L^2 C}{L}} = \frac{1}{\sqrt{LC}} \frac{Q_L}{\sqrt{1 + Q_L^2}},$$
(3.98)

where Q_L is the Q-value of the load inductor defined in (3.6) and all the parasitic capacitances are included into C. Due to finite Q_L , the resonance frequency is lower than with an ideal inductor but the resonance frequency is close to $\omega = 1/\sqrt{LC}$ when Q_L is large. At the resonant frequency (3.98), the impedance is

$$Z_{L,\omega=\omega_r} = \frac{1}{\frac{1}{R} + \frac{CR_L}{L}} = R \left\| (\omega_r Q_L L) \right\|.$$
(3.99)

One measure of the resonator tank is its Q-value, which is usually determined with -3-dB bandwidth, i.e.

$$Q_r = \frac{\omega_r}{\Delta \omega_{-3dB}} \,. \tag{3.100}$$

For the resonator shown in Fig. 3.39, the Q-value is approximately [75]

$$Q_r \approx \frac{\omega_r RL}{\omega_r^2 L^2 + R_L R} = \frac{R}{\omega_r L + \frac{R}{O_r}}.$$
(3.101)

The load impedance at resonant frequency can be expressed with (3.101) as

$$Z_{L,\omega=\omega_r} = \left(1 - \frac{Q_r}{Q_L}\right) R = Q_r \omega_r L .$$
(3.102)

Therefore, when the desired bandwidth and impedance level of the resonator are known, the required component values can be calculated with (3.102). According to (3.99) and (3.101) the load resistor R and inductor L affect differently the impedance level and bandwidth of the resonator. The impedance level decreases when either the value of R or L is decreased. However, the Q-value of the resonator is lowered when the value of the load resistor R is decreased or the load inductor value is increased. In the latter case, the shunt-capacitance value needs to be decreased to keep the resonant frequency unchanged. The maximum bandwidth is obtained, when the load capacitor C consists only of the parasitic capacitance associated with the output node. In that case, the resonance frequency is susceptible to additional parasities due to layout and the parasitics caused by metal wiring etc. should be extracted carefully.



Fig. 3.39. Amplifier with a resonator load.

An example of RLC resonator impedance as a function of frequency is shown in Fig. 3.40. For the purpose of comparison, an actual and an ideal inductor were used. The Q-values are shown in Fig. 3.40. The response when an actual 5-nH on-chip inductor having Q_L of approximately 11 is used is shown with a solid line. The Q_L of practical on-chip inductors has a maximum value at self-resonant frequency. The response with an ideal 5-nH inductor, which has Q-value of 12 at 4 GHz, is shown with a dashed line. The center frequency is set with an ideal capacitor to 4 GHz and the impedance is damped with an ideal 300- Ω parallel resistor. When the impedances of loads using accurate and simple inductor models are compared, the difference is insignificant. Therefore, the resonator impedance and Q-value can be accurately predicted by knowing the Q_L at the resonance frequency. According to (3.99), the inductor impedance is approximately 1.5 k Ω , and the impedance of the whole resonator is 250 Ω at the resonance frequency. The -3-dB bandwidth is approximately 2 GHz ranging from 3.1 to 5.1 GHz. Thus, the resonator Q-value is about 2 and the value predicted by (3.101) is 1.99.



Fig. 3.40. Resonator load impedance and Q-value of the load inductor as a function of frequency. The values with an actual 5-nH on-chip inductor are shown with solid lines, while dashed lines show the responses with an ideal 5-nH inductor having Q-value of 12 at 4 GHz.

3.5.2 Wideband load structures

To design a wideband amplifier with a flat insertion gain response, both the output signal current of the input stage and the load should have wide enough bandwidths to achieve required gain deviation. In a narrowband amplifier, for example in an IDCS LNA, a resonator load is typically used. Although the RLC load of Fig. 3.40 has nearly 2-GHz BW, using it as a load for a maximally wide IDCS stage such as that presented in Section 3.3.1.2 would lead to an LNA that has approximately 6-dB gain variation over UWB BG1 band, for example. Generally, that is an undesirably large value. For low-frequency applications, which have a wide operation band, the resistive load could be a sufficient alternative if the voltage drop is tolerated. When the requirement of a wide relative wideband is combined with high center frequency, both the RLC and RC loads typically do not offer adequate performance.

For wideband and high-frequency applications, the shunt-peak load shown in Fig. 3.41 is a commonly used load structure. It consists of a series connection of inductance and L and resistor R with a shunt capacitor C. The frequency response is of the shunt-peak load is analyzed with the factor m, which is the ratio of L/R and RC time constants, i.e. $m = L/R^2C$ [76]. Then, the inductance value is expressed as

$$L = mR^2C. (3.103)$$

The load impedance Z_L of a shunt-peak load is

$$Z_{L}(s) = \frac{R+sL}{1+sCR+s^{2}CL} = R\frac{1+smCR}{1+sCR+s^{2}mC^{2}R^{2}} = R\frac{1+m\frac{s}{\omega_{0}}}{1+\frac{s}{\omega_{0}}+m\frac{s^{2}}{\omega_{0}^{2}}},$$
(3.104)

where $\omega_0 = 1/RC$ is the RC pole frequency.



Fig. 3.41. Amplifier with a shunt-peak load.

The magnitude of the load impedance scaled to 0 dB is illustrated in Fig. 3.42 with several *m* values. When m = 0, there is no shunt peaking, and the response has a -3-dB corner frequency at a scaled frequency of 1. With $m = \sqrt{2} - 1 \approx 0.414$, the maximally flat response is achieved, and then the normalized -3-dB corner frequency is approximately 1.72. In the following, the term bandwidth extension ratio (BWER) is used. This is the ratio of the -3-dB corner frequency to the unity RC pole frequency [59]. The maximum BWER is achieved with $m = 1/\sqrt{2} \approx 0.707$, and then the normalized -3-dB corner frequency is approximately 1.85. In that case, approximately 1.5 dB peaking occurs at 0.88 normalized frequency.



Fig. 3.42. Scaled magnitude of a shunt-peak load.

The performance of a shunt-peak load can be modified by adding an additional capacitor C_B in parallel with *L*, as is shown in Fig. 3.43. As a result, a bridged-shunt peaking load is formed [59].



Fig. 3.43. Bridged-shunt peaking load.

The load impedance of a bridged shunt-peak load is

$$Z_{L}(s) = R \frac{1 + m\frac{s}{\omega_{0}} + k_{B}m\frac{s^{2}}{\omega_{0}^{2}}}{1 + \frac{s}{\omega_{0}} + (k_{B} + 1)m\frac{s^{2}}{\omega_{0}^{2}} + k_{B}m\frac{s^{3}}{\omega_{0}^{3}}},$$
(3.105)

where $k_B = C_B/C$, $\omega_0 = 1/RC$, and *m* is evaluated with (3.103). The shunt capacitor C_B adds another pole and zero to Z_L . The -3-dB corner frequency and gain peaking are depicted in Fig. 3.44 as a function of *m* and with different k_B values. The RC-pole frequency is normalized to 1. Compared to the typical shunt-peak load ($k_B = 0$), the maximum -3-dB cut-off frequency is not increased (BWER ≈ 1.85) but is achieved with lower peaking. However, the maximum BWER is achieved with a smaller value of *m*, thus leading to a smaller inductance value. As a result, inductor *L* occupies a smaller die area and it has higher self-resonant frequency [59]. When high ω_0 frequency is targeted, the load capacitance *C* should be small. The minimum value of *C* is achieved when it is formed by the capacitive load of the next stage and parasitic capacitances of monolithic components connected to the output node. In that case, the design of a small-value capacitor C_B can be challenging. In practice, the load resistor and inductor and a possible feedback path (for example, see Fig. 3.14b and Fig. 3.17a) add parasitic capacitance at node A and a small C_B is present even without a physical capacitor. The value of capacitor C_B (or the parasitic capacitance at node A) should be kept less than the value of load capacitance *C*, since the operation of the simple bridged shunt-peak load shown in Fig. 3.43 becomes sensitive to component values and the BWER starts to decrease if $k_B > 1$.



Fig. 3.44. -3-dB corner frequency and gain peaking of a shunt-peak load ($k_B = 0$) and a bridged-shunt load with $k_B = 1/8$, $k_B = 1/4$, and $k_B = 1/2$.

When an inductor is inserted such that the load capacitances are separated, the series-peaked load shown in Fig. 3.45a is formed [59]. The maximum –3-dB BW without peaking is approximately 2.52. In addition, the method is utilized, for example between the output of the CS input stage and a cascode stage as is shown in Fig. 3.45b, to minimize the effect of parasitic capacitances [77]. As a disadvantage, the transimpedance of the load shown in Fig. 3.45a is sensitive to the capacitor ratios, which define the BW and the ripple at the pass band.



Fig. 3.45. a) Series peak load, b) a design example of series-peaking [77].

Other wideband loads can be designed by combining the presented structures, as is presented in [59] and [78], for example. Although BWERs higher than 4 can be reached with some of the techniques, there is also ripple at in-band frequencies. Clearly, complicated load impedance is challenging to realize in practice. A slight deviation from the nominal device values and the presence of parasitic components can significantly degrade the passband ripple and limit the -3-dB corner frequency. In addition, one disadvantage is the number of on-chip inductors, which is especially troublesome with balanced amplifiers. For example, balanced series-peaking loads cannot be realized with differential inductors. Thus, among various types of wideband load structures, shunt-peak loads are preferred in single-stage balanced amplifiers, since the number of on-chip inductors is small and its impedance is least sensitive to device values or ratios.

3.5.3 Load design for two-stage wideband amplifiers

With modern deep-submicron CMOS processes the intrinsic gain from a single transistor is only moderate. As a result, the gain achieved from a single-stage amplifier may not be sufficient and multi-stage amplifiers are needed to achieve adequate gain to suppress the noise of the following stages. The design of a two- or multi-stage amplifier is not so straightforward either. Although gain can be increased with cascaded amplifier stages, the overall bandwidth is limited by the stage that has the narrowest bandwidth. If the bandwidths of each stage are nearly equal, the overall bandwidth shrinks due to the cumulative roll-off of cascaded stages. Clearly, the number of on-chip inductors should be low to keep the layout area as small as possible. Next, the operation of a two-stage amplifier, where both stages utilize separate shunt-peak loads, is discussed. Then, a modified shunt-peak load using only single on-chip coil is proposed for two-stage amplifiers.

3.5.3.1 Cascaded shunt-peak loads

When two amplifiers are cascaded as shown in Fig. 3.46, the bandwidth typically shrinks compared to a single-stage amplifier. The bandwidth shrinkage of multiple-stage amplifiers is discussed in, for example, [78] and [79]. For example, when the amplifier is considered as having a single dominant pole, the normalized -3-dB frequency of the two-stage amplifier decreases to 0.643. Although the single dominant pole model is simple, the BW of the cascaded amplifier can be roughly estimated.



Fig. 3.46. Two-stage amplifier with shunt-peak loads.

In the following, the effective BW of a two-stage amplifier utilizing shunt-peak loads is analyzed. For simplicity, only the effect of the cascaded loads is considered and the bandwidth of the amplifier (i.e. g_m -stage in Fig. 3.46) is assumed infinite. Despite of the simplified model, since the impedance of the shunt-peak load consists of a zero and two poles, the analysis of cascaded shunt-peak loads is not so straightforward as it is for single-pole model. For example, dissimilar loads could be designed such that the other load peaks at a certain frequency, which compensates the droop caused by the other load. Thus, higher BWER could be obtained than with a single-stage shunt-peak load. The analysis of two-stage amplifiers having dissimilar shunt-peak loads with closed-form equations is tedious. Therefore, the -3-dB corner frequency is simulated and presented in Fig. 3.47. In that figure, m_1 in the x-axis is the design parameter of the first shunt-peak load and the values from 0 to 1 with incremental steps of 0.2 are values of the design parameter m_2 of the second stage. The order of shunt-peak loads can be interchanged without altering the -3-dB corner frequency. The maximum corner frequency of approximately 1.61 is achieved, when m_1 and m_2 values are around 0.8 ... 0.9. If $m_1 = m_2 = m$, the accurate values for m and BWER are $1/\sqrt{\sqrt{2}} \approx 0.84$ and $\sqrt{2^{1/4} + \sqrt{2}} \approx 1.61$, respectively. Therefore, the *m*-value, which yields the maximum BWER, is slightly different from the single-stage shuntpeak load. Eventually, the maximum corner frequency starts to decrease, when m_1 and m_2 are larger than 1. Therefore, the BWER of the two-stage amplifier utilizing shunt-peak loads is limited to 1.61.6 Compared to the single-stage case, the BWER is decreased by approximately a factor of 1.15. Clearly, taking into account the effect of the finite BW of the input stage, the bandwidth shrinkage can be significantly worsened.



Fig. 3.47. -3-dB corner frequency of a two-stage amplifier having dissimilar shunt-peak loads. The shunt-peak load design parameter m_1 is shown in *x*-axis, and values from 0 to 1 with steps of 0.2 are values of the design parameter m_2 of the second stage.

 $^{^{6}}$ In this case, it would be possible to define the BWER of a two-stage amplifier as a ratio of -3-dB corner frequency to the cascaded RC-pole frequency (0.643). However, in this thesis, the BWER is always rationed to the unity normalized frequency for the sake of clarity.

The use of large *m* values significantly increases the gain peaking as is presented in Fig. 3.48. For example, with maximum BWER ($m_1 = m_2 = 0.84$), the gain peaking is 4.7 dB, while with $m_1 = m_2 = 1/\sqrt{2}$, the gain peak is only 3 dB, and the BWER of a two-stage amplifier would be 1.60. Hence, the difference to maximal –3-dB corner frequency is insignificant, but is achieved with smaller peaking.

The maximally flat response can be achieved with several m_1 and m_2 combinations. However, according to simulations, the maximally wide response without peaking is achieved when *m*-values of both loads are identical. The maximally flat response is achieved with $m = \sqrt{2} - 1 \approx 0.414$, which is the same result as with a single-stage shunt-peak load. Then, the resulting normalized -3-dB corner frequency is approximately 1.33. Compared to the single-stage case, the BWER is decreased by approximately a factor of 1.3.



Fig. 3.48. -3-dB corner frequency of a two-stage amplifier having dissimilar shunt-peak loads. The shunt-peak load design parameter m_1 is shown in x-axis, and values from 0 to 1 with steps of 0.2 are values of the design parameter m_2 of the second stage.

3.5.3.2 Cross-connected shunt-peak load

As presented earlier, the maximum –3-dB normalized corner frequency of a two-stage amplifier utilizing shunt-peak loads is limited to 1.6. The BWER could be enhanced with the more complex load presented in Section 3.5.2. However, the number of differential inductors should be as small as possible for a compact layout. Therefore, a structure where the load inductor is re-used in cascaded stages is shown in Fig. 3.49. The resulting transimpedance of the modified shunt-peak load is

$$Z_{L}(s) = \frac{g_{m2}(R_{1}R_{2} + sR_{1}L + sR_{2}L) - sL}{1 + s[g_{m2}L + C_{1}R_{1} + C_{2}R_{2}] + s^{2}[L(C_{1} + C_{2}) + C_{1}C_{2}(R_{1}R_{2} + sR_{1}L + sR_{2}L)]}.$$
 (3.106)



Fig. 3.49. Two-stage amplifier with a cross-connected shunt-peak load.

For simplicity, resistors and capacitors are assumed identical, i.e. $R_1 = R_2 = R$ and $C_1 = C_2 = C$, and *L* is replaced with (3.103). In addition, the g_{m_2} is assumed to be constant as a function of frequency, i.e. it has infinite BW. Then, (3.106) can be modified to the form

$$Z_{L}(s) = R \cdot \frac{G_{2} + (2mG_{2} - m)\frac{s}{\omega_{0}}}{1 + (2 + mG_{2})\frac{s}{\omega_{0}} + (1 + 2m)\frac{s^{2}}{\omega_{0}^{2}} + 2m\frac{s^{3}}{\omega_{0}^{3}}},$$
(3.107)

where $\omega_0 = 1/RC$ is the *RC* pole frequency and $G_2 = g_{m2}R$ is the dc gain of the second stage. Compared to a typical shunt-peak load, there are now three complex poles and a zero. An example of characteristic impedance (3.107) is shown in Fig. 3.50, where $G_2 = 5$ and with *m* values of 0.5, 1, and 2. The impedance is scaled to 0 dB at unity frequency. By choosing component values appropriately, the two local points (i.e. the points where the derivative of the magnitude curve equals zero) can be equalized and thus flat band gain at the wanted frequency area can be achieved.



Fig. 3.50. Scaled magnitude of a modified shunt-peak load with $G_2 = 5$ and with several *m* values.

The analysis of the operation of a modified shunt-peak load with closed-form equations is complicated. The target is to equalize the gains at two local points to have flat gain over a wide operation frequency. Therefore, a design value for m is sought as a function of G_2 . The first-order formula could not sufficiently model the behavior of the modified shunt-peak load. Therefore, an approximate second-order formula is given to achieve a simple design criterion for m:

$$m = \frac{1}{0.2 + 0.3G_2 - 0.005G_2^2} \,. \tag{3.108}$$

The operation of the modified shunt-peak load is shown in Fig. 3.51 with different G_2 values and (3.108) is used to calculate *m*. At low frequencies, the frequency response shows the roll-off due to the first pole. The transimpedance is scaled to 0 dB in the flat band area. With larger G_2 values, the bandwidth of the flat gain area increases and the band of interest shifts to a higher frequency.



Fig. 3.51. Magnitude of the modified shunt-peak load with different G_2 values.

The performance of the modified shunt-peak load with $G_2 = 5$ is compared to a two-stage amplifier utilizing identical shunt-peak loads in Fig. 3.52. The modified shunt-peak load is shown with a solid line, while the traditional shunt-peak load with maximally flat and maximally wide bandwidth are presented with dashed and dashed-dotted lines, respectively. Due to quite different frequency behavior, comparison of all three cases is not trivial. The magnitudes of the traditional shunt-peak load are scaled to 0 dB at zero frequency and the magnitude of the modified shunt-peak load is scaled to 0 dB in the flat band area. The gain flatness of the modified shunt-peak load is better in the normalized frequency area of 1–1.5. The corner frequency, where the gain is decreased by 3 dB from the band of interest, is approximately 2.23. As a result, the modified shunt-peak load offers a trade-off between the gain level and having flat frequency response over a wide bandwidth with a single coil. The design example and usage of such circuit is shown later in Section 6.6.2.5.



Fig. 3.52. Frequency response of two-stage amplifier. A solid line shows the response of a modified shunt-peak load ($G_2 = 5$). Dashed and dashed-dotted lines present the responses of a typical shunt-peak load with maximally flat and maximally wide bandwidths, respectively.

References

- E. A. M. Klumperink, F. Bruccoleri, and B. Nauta, "Finding all elementary circuits exploiting transconductance," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 11, Nov. 2001, pp. 1039-1053.
- [2] J.-H. C. Zhan and S. S. Taylor, "A 5GHz resistive-feedback CMOS LNA for low-cost multi-standard applications," *IEEE Int. Solid-State Circuits Conference (ISSCC'06)*, San Francisco, CA, 6-9 Feb. 2006, pp. 721-722.
- [3] Adiseno, H. Magnusson, and H. Olsson, "A 1.8-V wide-band CMOS LNA for multiband multistandard front-end receiver," *European Solid-State Circuits Conf.* (ESSCIRC'03), Lisbon, Portugal, 16-18 Sept. 2003, p. 141-144.
- [4] J. Ryynänen, Low-noise amplifiers for integrated multi-mode direct-conversion receivers, Doctoral Thesis, Helsinki University of Technology, Espoo, Finland, 2004, p. 138.
- [5] F. Bruccoleri, E. A. M. Klumperink, and B. Nauta, *Wideband low noise amplifiers exploiting thermal noise cancellation*, Springer, 2005, p. 182.
- [6] R. Bagheri, A. Mirzaei, S. Chehrazi, M. E. Heidari, M. Lee, M. Mikhemar, W. Tang, and A. A. Abidi, "An 800-MHz–6-GHz software-defined wireless receiver in 90nm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2860-2876.
- [7] C.-F. Liao and S.-I. Liu, "A broadband noise-canceling CMOS LNA for 3.1–10.6-GHz UWB receivers," *IEEE J. of Solid-State Circuits*, vol. 42, no. 2, Feb. 2007, pp. 329-339.
- [8] R. E. Lehmann and D. D. Heston, "X-band monolithic series feedback LNA," *IEEE Trans. on Microwave Theory and Techniques*, vol. MTT-33, no. 12, Dec. 1985, pp. 1560-1566.
- [9] H. Hashemi and A. Hajimiri, "Concurrent multiband low-noise amplifiers Theory, design, and applications," *IEEE Trans. on Microwave Theory and Techniques*, vol. 50, no. 1, Jan. 2002, pp. 288-301.
- [10] J. C. Rudell, J.-J. Ou, T. B. Cho, G. Chien, F. Brianti, J. A. Weldon, and P. R. Gray, "A 1.9-GHz wide-band IF double conversion CMOS receiver for cordless telephone applications," *IEEE J. of Solid-State Circuits*, vol. 32, no. 12, Dec. 1997, pp. 2071-2088.
- [11] D. K. Shaeffer and T. H. Lee, "A 1.5-V, 1.5-GHz CMOS low noise amplifier," *IEEE J. of Solid-State Circuits*, vol. 32, no. 5, May 1997, pp. 745-759.
- [12] —, "Corrections to "A 1.5-V, 1.5-GHz CMOS low noise amplifier"," IEEE J. of Solid-State Circuits, vol. 40, no. 6, June 2005, pp. 1397-1398.
- [13] —, "Corrections to "A 1.5-V, 1.5-GHz CMOS low noise amplifier"," *IEEE J. of Solid-State Circuits*, vol. 41, no. 10, Oct. 2006, p. 2359.
- [14] P. Andreani and H. Sjöland, "Noise optimization of an inductively degenerated CMOS low noise amplifier," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 9, Sept. 2001, pp. 835-841.
- [15] S. S. Mohan, M. del Mar Hershenson, S. P. Boyd, and T. H. Lee, "Simple accurate expressions for planar spiral inductances," *IEEE J. of Solid-State Circuits*, vol. 34, no. 10, Oct. 1999, pp. 1419-1424.
- [16] J. Gil and H. Shin, "A simple wide-band on-chip inductor model for silicon-based RF ICs," *IEEE Trans. on Microwave Theory and Techniques*, vol. 51, no. 9, Sept. 2003, pp. 2023-2028.
- [17] O. H. Murphy, K. G. McCarthy, C. J. P. Delabie, A. C. Murphy, and P. J. Murphy, "Design of multiple-metal stacked inductors incorporating an extended physical model," *IEEE Trans. on Microwave Theory and Techniques*, vol. 53, no. 6, June 2005, pp. 2063-2072.
- [18] I. C. H. Lai and M. Fujishima, "A new on-chip substrate-coupled inductor model implemented with scalable expressions," *IEEE J. of Solid-State Circuits*, vol. 41, no. 11, Nov. 2006, pp. 2491-2499.
- [19] H.-Y. Lee, "Wideband characterization of a typical bonding wire for microwave and millimetre-wave integrated circuits," *IEEE Trans. on Microwave Theory and Techniques*, vol. 43, no. 1, Jan. 1995, pp. 63-68.
- [20] H. Patterson, "Analysis of ground bond wire arrays for RFICs," *IEEE MTT-S International Microwave Symposium (IMS'97)*, Denver, CO, 8-13 June 1997, pp. 765-768.
- [21] A. Pärssinen, J. Jussila, J. Ryynänen, L. Sumanen, and K. A. I. Halonen, "A 2-GHz wide-band direct conversion receiver for WCDMA applications," *IEEE J. of Solid-State Circuits*, vol. 34, no. 12, Dec. 1999, pp. 1893-1903.
- [22] F. S. Lee and A. P. Chandrakasan, "A BiCMOS ultra-wideband 3.1–10.6-GHz frontend," *IEEE J. of Solid-State Circuits*, vol. 41, no. 8. Aug. 2006, pp. 1784-1791.

- [23] Q. Huang, F. Piazza, P. Orsatti, and T. Ohguro, "The impact of scaling down to deep submicron on CMOS RF circuits," *IEEE J. of Solid-State Circuits*, vol. 33, no. 7, July 1998, pp. 1023-1036.
- [24] H. Darabi and A. A. Abidi, "A 4.5-mW 900-MHz CMOS receiver for wireless paging," *IEEE J. of Solid-State Circuits*, vol. 35, no. 8, Aug. 2000, pp. 1085-1096.
- [25] J.-S. Goo, H. T. Ahn, D. J. Ladwig, Z. Yu, T. H. Lee, and R. W. Dutton, "A noise optimization technique for integrated low-noise amplifiers," *IEEE J. of Solid-State Circuits*, vol. 37, no. 8, Aug. 2002, pp. 994-1002.
- [26] J. Janssens and M. Steyaert, "MOS noise performance under impedance matching constraints," *Electronics Letters*, vol. 35, no. 15, 22nd July 1999, pp. 1278-1280.
- [27] P. Leroux, J. Janssens, and M. Steyaert, "A 0.8-dB NF ESD-protected 9-mW CMOS LNA operating at 1.23 GHz," *IEEE J. of Solid-State Circuits*, vol. 37, no. 6, June 2002, pp. 760-765.
- [28] K.-J. Sun, Z.-M. Tsai, K.-Y. Lin, and H. Wang, "A noise optimization formulation for CMOS low-noise amplifiers with on-chip low-Q inductors," *IEEE Trans. on Microwave Theory and Techniques*, vol. 54, no. 6, June 2006, pp. 1554-1560.
- [29] P. Rossi, F. Svelto, A. Mazzanti, and P. Andreani, "Reduced impact of induced gate noise on inductively degenerated LNAs in deep submicron CMOS technologies," *Analog Integrated Circuits and Signal Processing*, vol. 42, no. 1, Jan. 2005, pp. 31-36.
- [30] A. J. Scholten, L. F. Tiemeijer, R. van Langevelde, R. J. Havens, A. T. A. Zegers-van Duijhoven, and V. C. Venezia, "Noise modeling for RF CMOS circuit simulation," *IEEE Trans. on Electron Devices*, vol. 50, no. 3, March 2003, pp. 618-632.
- [31] P. Sivonen and A. Pärssinen, "Analysis and optimization of packaged inductively degenerated common-source low-noise amplifiers with ESD protection," *IEEE Trans. on Microwave Theory and Techniques*, vol. 53, no. 4, April 2005, pp. 1304-1313.
- [32] J. T. Colvin, S. S. Bhatia, and K. K. O, "Effects of substrate resistances on LNA performance and a bondpad structure for reducing the effects in a silicon bipolar technology," *IEEE J. of Solid-State Circuits*, vol. 34, no. 9, Sept. 1999, p. 1339-1344.
- [33] W. Sansen, "Distortion in elementary transistor circuits," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 46, no. 3, March 1999, pp. 315-325.
- [34] K. L. Fong and R. G. Meyer, "High-frequency nonlinearity analysis of commonemitter and differential-pair transconductance stages," *IEEE J. of Solid-State Circuits*, vol. 33, no. 4, April 1998, pp. 548-555.
- [35] R. A. Baki, T. K. K. Tsang, and M. N. El-Gamal, "Distortion in RF CMOS shortchannel low-noise amplifiers," *IEEE Trans. on Microwave Theory and Techniques*, vol. 54, no. 1, Jan. 2006, pp. 46-56.
- [36] A. A. Abidi, G. J. Pottie, and W. J. Kaiser, "Power-conscious design of wireless circuits and systems," *Proc. of IEEE*, vol. 88, no. 10, Oct. 2000, pp. 1528-1545.
- [37] B. Toole, C. Plett, and M. Cloutier, "RF circuit implications of moderate inversion enhancement linear region in MOSFETs," *IEEE Trans. on Circuits and Systems I: Regular Papers*, vol. 51, no. 2, Feb. 2004, 319-328.

- [38] P. H. Woerlee, M. J. Knitel, R. van Langevelde, D. B. M. Klaassen, L. F. Tiemeijer, A. J. Scholten, and A. T. A. Zegers-van Duijinhoven, "RF-CMOS performance trends," *IEEE Trans. on Electron Devices*, vol. 48, no. 8, Aug. 2001, pp. 1776-1782.
- [39] T. Tikka, J. Ryynänen, M. Hotti, and K. Halonen, "Design of a high linearity mixer for direct-conversion base-station receiver," *IEEE International Symposium on Circuits* and Systems (ISCAS'06), Kos, Greece, 21-24 May 2006, pp. 2321-2324.
- [40] W. Guo and D. Huang, "Noise and linearity analysis for a 1.9 GHz CMOS LNA," *IEEE Asia–Pasific Conference on Circuits and Systems (APCCAS'02)*, Bali, Indonesia, 28-31 Oct. 2002, vol. 2, pp. 409-414.
- [41] C. Tinella and J. M. Fournier, "Design of a SOI fully integrated 1V, 2.5 GHz front-end receiver," *IEEE International SOI Conference (SOI'01)*, Durango, CO, 1-4 Oct. 2001, pp. 139-140.
- [42] F. Behbahani, J. C. Leete, Y. Kishigami, A. Roithmeier, K. Hoshino, and A. A. Abidi, "A 2.4-GHz low-IF receiver for wideband WLAN in 0.6-µm CMOS – Architecture and front-end," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1908-1916.
- [43] R. Fujimoto, K. Kojima, and S. Otaka, "A 7-GHz 1.8-dB NF CMOS low-noise amplifier," *IEEE Journal of Solid-State Circuits*, vol. 37, no. 7, July 2002, pp. 852-856.
- [44] T. H. Lee, H. Samavati, and H. R. Rategh, "5-GHz CMOS wireless LANs," *IEEE Trans. on Microwave Theory and Techniques*, vol. 50, no. 1, Jan. 2002, pp. 268-280.
- [45] A. Bevilacqua, A. Vallese, C. Sandner, M. Tiebout, A. Geroa, and A. Neviani, "A 0.13μm CMOS LNA with integrated balun and notch filter for 3-to-5-GHz UWB receivers," *IEEE Int. Solid-State Circuits Conference (ISSCC'07)*, San Francisco, CA, 11-15 Feb. 2007, pp. 420-421.
- [46] M. Zargari, M. Terrovitis, S. H.-M. Jen, B. J. Kaczynski, M. Lee, M. P. Mack, S. S. Mehta, S. Mendis, K. Onodera, H. Samavati, W. W. Si, K. Singh, A. Tabatabaei, D. Weber, D. K. Su, and B. A. Wooley, "A single-chip dual-band tri-mode CMOS transceiver for IEEE 802.11a/b/g wireless LAN," *IEEE J. of Solid-State Circuits*, vol. 39, no. 12, Dec. 2004, pp. 2239-2249.
- [47] E. Abou-Allam, J. J. Nisbet, and M. C. Maliepaard, "Low-voltage 1.9-GHz front-end receiver in 0.5-μm CMOS technology," *IEEE J. of Solid-State Circuits*, vol. 36, no. 10, Oct. 2001, pp. 1434-1443.
- [48] A. Rofougaran, G. Chang, J. J. Rael, J. Y.-C. Chang, M. Rofougaran, P. J. Chang, M. Djafari, J. Min, E. W. Roth, A. A. Abidi, and H. Samueli, "A single-chip 900-MHz spread-spectrum wireless transceiver in 1-μm CMOS Part II: Receiver design," *IEEE J. of Solid-State Circuits*, vol. 33, no. 4, April 1998, pp. 535-547.
- [49] A. Rofougaran, J. Y.-C. Chang, M. Rofougaran, and A. A. Abidi, "A 1 GHz CMOS RF front-end IC for a direct-conversion wireless receiver," *IEEE Journal of Solid-State Circuits*, vol. 31, no. 7, July 1996, pp. 880-889.
- [50] X. Guan and A. Hajimiri, "A 24-GHz CMOS front-end," IEEE J. of Solid-State Circuits, vol. 39, no. 2, Feb. 2004, pp. 368-373.

- [51] K. Bhatia, S. Hyvönen, and E. Rosenbaum, "A compact, ESD-protected, SiGe BiCMOS LNA for ultra-wideband applications," *IEEE J. of Solid-State Circuits*, vol. 42, no. 5, May 2007, pp. 1121-1130.
- [52] G. Cusmai, M. Brandolini, P. Rossi, and F. Svelto, "A 0.18-µm CMOS selective receiver front-end for UWB applications," *IEEE J. of Solid-State Circuits*, vol. 41, no. 8, Aug. 2006, pp. 1764-1771.
- [53] S. Mahdavi and A. A. Abidi, "Fully integrated 2.2-mW CMOS front end for a 900-MHz wireless receiver," *IEEE J. of Solid-State Circuits*, vol. 37, no. 5, May 2002, pp. 662-669.
- [54] A.-S. Porret, T. Melly, D. Python, C. C. Enz, and E. A. Vittoz, "An ultra-low power UHF transceiver integrated in a standard digital CMOS process: Architecture and receiver," *IEEE J. of Solid-State Circuits*, vol. 36, no. 3, March 2001, pp. 452-466.
- [55] B. Razavi, T. Aytur, C. Lam, F.-R. Yang, K.-Y. Li, R.-H. Yan, H.-C. Kang, C.-C. Hsu, and C.-C. Lee, "A UWB CMOS transceiver," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2555-2562.
- [56] A. Liscidini, M. Brandolini, D. Sanzogni, and R. Castello, "A 0.13 μm CMOS frontend for DCS1800/UMTS/802.11b-g with multiband positive feedback low-noise amplifier," *IEEE J. of Solid-State Circuits*, vol. 41, no. 4, April 2006, pp. 981-989.
- [57] W. Zhuo, S. Embabi, J. Pineda de Gyvez, and E. Sánchez-Sinencio, "Using capacitive cross-coupling technique in RF low noise amplifiers and downconversion mixer design," *Eur. Solid-State Circuits Conf. (ESSCIRC'00)*, Stockholm, Sweden, 19-21 Sept. 2000, pp. 116-119.
- [58] D. J. Allstot, X. Li, and S. Shekhar, "Design considerations for CMOS low-noise amplifiers," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'04)*, Fort Worth, TX, 6-8 June 2004, pp. 97-100.
- [59] S. Shekhar, J. S. Walling, and D. J. Allstot, "Bandwidth extension techniques for CMOS amplifiers," *IEEE J. of Solid-State Circuits*, vol. 41, no. 11, Nov. 2006, pp. 2424-2439.
- [60] X. Li, S. Shekhar, and D. J. Allstot, "G_m-boosted common-gate LNA and differential Colpitts VCO/QVCO in 0.18-μm CMOS," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2609-2619.
- [61] W. Zhuo, X. Li, S. Shekhar, S. H. K. Embabi, J. Pineda de Gyvez, D. J. Allstot, and E. Sánchez-Sinencio, "A capacitor cross-coupled common-gate low-noise amplifier," *IEEE Trans. on Circuits and Systems II: Express Briefs*, vol. 52, no. 12, Dec. 2005, pp. 875-879.
- [62] C.-W. Kim, M.-S. Kang, P. T. Anh, H.-T. Kim, and S.-G. Lee, "An ultra-wideband CMOS low noise amplifier for 3–5-GHz UWB system," *IEEE J. of Solid-State Circuits*, vol. 40, no. 2, Feb. 2005, pp. 544-547.
- [63] A. Bevilacqua and A. M. Niknejad, "An ultra-wideband CMOS LNA for 3.1 to 10.6 GHz wireless receivers," *IEEE J. of Solid-State Circuits*, vol. 39, no. 12, Dec. 2004, pp. 2259-2268.
- [64] A. Ismail and A. A. Abidi, "A 3–10.6-GHz low-noise amplifier with wideband LCladder matching network," *IEEE J. of Solid-State Circuits*, vol. 39, no. 12, Dec. 2004, pp. 2269-2277.

- [65] Y. Park, C.-H. Lee, J. D. Cressler, and J. Laskar, "The analysis of UWB SiGe HBT LNA for its noise, linearity, and minimum group delay variation," *IEEE Trans. on Microwave Theory and Techniques*, vol. 54, no. 4, April 2006, pp. 1687-1697.
- [66] M. Ranjan and L. E. Larson, "A low-cost and low-power CMOS receiver front-end for MB-OFDM ultra-wideband systems," *IEEE J. of Solid-State Circuits*, vol. 42, no. 3, March 2007, pp. 592-601.
- [67] J. Borremans, P. Wambacq, G. Van den Plas, Y. Rolain, and M. Kuijk, "A bondpadsize narrowband LNA for digital CMOS," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'07)*, Honolulu, HI, 3-5 June 2007, pp. 677-680.
- [68] S. Andersson, C. Svensson, and O. Drugge, "Wideband LNA for a multistandard wireless receiver in 0.18 µm CMOS," *Eur. Solid-State Circuits Conf. (ESSCIRC'03)*, Lisbon, Portugal, 16-18 Sept. 2003, p. 655-658.
- [69] H. Knapp, D. Zöschg, T. Meister, K. Aufinger, S. Buguth, and L. Treitinger, "15 GHz wideband amplifier with 2.8 dB noise figure in SiGe bipolar technology," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'01)*, Phoenix, AZ, 20-22 May 2001, pp. 287-290.
- [70] D. B. Estreich, "A monolithic wide-band GaAs IC amplifier," *IEEE J. of Solid-State Circuits*, vol. SC-17, no. 6, Dec. 1982, pp. 1166-1173.
- [71] C.-T. Fu and C.-N. Kuo, "3–11-GHz CMOS UWB LNA using dual feedback for broadband matching," *IEEE Radio Frequency Integrated Circuits Symposium* (*RFIC'06*), San Francisco, CA, 11-13 June 2006, pp. 67-70.
- [72] M. T. Reiha and J. R. Long, "A 1.2 V reactive-feedback 3.1–10.6 GHz low-noise amplifier in 0.13-μm CMOS," *IEEE J. of Solid-State Circuits*, vol. 42, no. 5, May 2007, pp. 1023-1033.
- [73] S. Lee, J. Bergervoet, K. S. Harish, D. Leenaerts, R. Roovers, R. van de Beek, and G. van der Weide, "A broadband receiver chain in 65nm CMOS," *IEEE Int. Solid-State Circuits Conference (ISSCC'07)*, San Francisco, CA, 11-15 Feb. 2007, pp. 418-419.
- [74] R. Roovers, D. M. W. Leenaerts, J. Bergervoet, K. S. Harish, R. C. H. van de Beek, G. van der Weide, H. Waite, Y. Zhang, S. Aggarwal, and C. Razzell, "An interference-robust receiver for ultra-wideband radio in SiGe BiCMOS technology," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2563-2572.
- [75] R. Kaunisto, *Monolithic active resonator filters for high frequencies*, Doctoral Thesis, Helsinki University of Technology, Espoo, Finland, 2000, p. 125.
- [76] S. S. Mohan, M. del Mar Hershenson, and T. H. Lee, "Bandwidth extension in CMOS with optimized on-chip inductors," *IEEE J. of Solid-State Circuits*, vol. 35, no. 3, March 2000, pp. 346-355.
- [77] B. Analui and A. Hajimiri, "Bandwidth enhancement for transimpedance amplifiers," *IEEE J. of Solid-State Circuits*, vol. 39, no. 8, Aug. 2004, pp. 1263-1270.
- [78] T. H. Lee, *The design of CMOS radio-frequency integrated circuits*, Cambridge University Press, 1998, p. 598.
- [79] R. P. Jindal, "Gigahertz-band high-gain low-noise AGC amplifiers in fine-line NMOS," *IEEE J. of Solid-State Circuits*, vol. SC-22, no. 4, Aug. 1987, pp. 512-521.

4 Downconversion mixer design

The mixer is required in almost all receiver and transmitter topologies. It is a nonlinear circuit, the primary target of which is to perform frequency translation. In principle, that can be accomplished with either nonlinear devices (diodes, transistors) or time-varying elements (switches). Mixers can in general be categorized as passive or active mixers. Typically, active mixers can provide conversion gain, while the passive mixers are lossy but more linear. In addition, mixers can be divided into double-balanced, single-balanced, and unbalanced designs [1]. The kind of mixer required depends on the related application and system requirements.

This chapter deals with the design of the downconversion mixers for DCRs. First, general mixer design guidelines are discussed. In this thesis, only active mixers were implemented. Therefore, the detailed analysis of passive mixers is omitted from this thesis and only a short overview is given in Section 4.2. After that, the most common active mixer topology, the Gilbert cell mixer, is described and the issues related to the mixer input stage, switch quad, load design, and interfaces with other blocks are discussed. Finally, different mixer design aspects, such as folded mixers, mixers utilizing current boosting, and quadrature mixer design alternatives are considered. In addition to the mixer structures presented in this chapter, there are different kinds of mixers, which are not covered in this thesis. For example, switched transconductance mixers [2] and bulk driven mixers [3] can be found in the literature.

4.1 General mixer design aspects

To design a proper downconversion mixer for a receiver, the most important specifications are gain, noise, and linearity. In DCRs, the mixer interfaces are usually not matched. Therefore, the conversion gain (or loss) is measured with voltage gain. Typically, a wireless front-end should provide approximately 25-35 dB voltage gain to suppress the noise of the following stages. Because the practical voltage gain achieved from the LNA(s) is normally 15-25 dB, the mixer should provide the rest of the needed gain.

The design of a proper mixer is challenging due to linearity and noise requirements. In a DCR, dc offsets and 1/f noise of the mixer can severely degrade the sensitivity of narrow band systems, where most of the useful signal is located close to dc. The wide bandwidth of the communication system, e.g. WCDMA or UWB, reduces the influence of the flicker noise. Because the mixer handles the output signal of the LNA, its linearity (both ICP and IIP3) should be higher than the LNAs linearity by the gain of the LNA in order not to become a bottleneck in the front-end. In practice, high linearity is challenging to achieve without consuming a substantial amount of current. Thus, the mixer typically dominates the whole front-end linearity. In addition to the ICP and IIP3 requirements, the mixer is usually a dominant source for the second-order nonlinearity [4]. The IIP2 is especially problematic if a direct-conversion or low-IF architecture is used. The analysis and calibration of the second-order intermodulation is comprehensively studied in the literature, for example, in [5] – [8].

4.2 Passive mixers

A typical passive commutating mixer utilizing four NMOS transistors as analog switches is shown in Fig. 4.1. It does not need bias current and therefore it would be an ideal low-power circuit. Since there is no dc current flowing thorough the mixers, passive mixers are often claimed to be free of 1/*f* noise. This is not completely true, since flicker noise can be observed with passive mixers also [9], [10]. Its magnitude is proportional to the input signal amplitude and inversely proportional to the slope of the gate voltage waveforms at LO transitions [10]. Nevertheless, the level of flicker noise of the passive mixers is substantially lower than that of the active ones.



Fig. 4.1. Passive mixer.

The dc bias level of the switch gate plays a significant role in mixer performance. Typically the gate voltage is set close to the threshold of conduction to achieve the lowest on-resistance. As a result, noise from the switch quad is minimized and good linearity (IIP3) is achieved [11]. In addition, overlapping on-periods are prevented [12]. The overlapping on-periods result in decreased conversion gain and increased noise from the LO port [12]. The conversion gain can be increased with overlapping off-periods but, in that case, the linearity suffers [13]. Furthermore, linearity performance (both IIP2 and IIP3) is maximized, when the voltage swing across the switch transistors is kept as small as possible. Such passive mixer topology is discussed, for example, in [14].

Due to their flicker noise and linearity properties, passive mixers are potential alternatives. The disadvantage of the mixer shown in Fig. 4.1 is its non-unilateral nature, i.e. the signal flows bidirectionally from input to output, thus causing several problems. First, when a passive mixer is connected to the output of the LNA, the circuits after the mixer can load the LNA and degrade its gain and selectivity [15]. In addition, two mixers downconverting a common input by quadrature LO signal phases would tend to load each other's outputs during the overlap period when switches are on [16]. As a result, the output impedance of the mixers is lowered, which increases the noise contribution of the first baseband stage. The overlap time can be minimized by increasing the LO amplitude, as discussed later in Section 4.3.3.1. In addition, there is a second mechanism, which lowers the mixer's output impedance [9]. With larger switches, there will be larger source/drain-junction capacitances, which results in larger amounts of stored noise. Due to the switched capacitor effect, these capacitances cause equivalent resistor, the value of which is inversely proportional to the LO frequency and the value of parasitic capacitance [9], [11]. Therefore, from the noise point of view, there is a trade-off between having a switch transistor that is large enough for low on-resistance but small enough to achieve good noise performance.

An additional example of a passive mixer is presented in Fig. 4.2 [17]. Compared to the passive mixer shown in Fig. 4.1, the RF signal is fed to the switch transistors' gates and V_{DS} varies with the LO signal. The transistors are in the triode region and therefore resistance is inversely proportional to the RF signal. The capacitors C_V are needed to filter out the high-frequency currents injected into the virtual ground nodes. Thus, the op-amp needs only to have enough bandwidth to handle the low-frequency components [17]. The IF voltage signal is available at the output of the feedback resistors. The mixer shown in Fig. 4.2 has excellent linearity, but also a disadvantage in the form of large required LO swing and the rather poor noise performance.



Fig. 4.2. Highly linear passive mixer [17].

4.3 Active downconversion mixers

The most common mixer for IC designers is probably the Gilbert-cell mixer and its variants. It was originally proposed by Barrie Gilbert as a four-quadrant multiplier in 1968 [18]. The schematic of a typical CMOS Gilbert mixer is shown in Fig. 4.3a. It consists of a balanced grounded transconductance stage, which converts the input voltage signal to current-mode. A differential pair could be used as a transconductance stage as well. After the input transconductor, the signal is fed to the switching quad, which is driven by a large (i.e. several 100 mV_{pp}) LO signal. The output current of the mixer is typically driven to either resistive, reactive, or active load.

Usually the Gilbert-cell mixer is driven with balanced RF and LO signals, i.e. it is a doublebalanced structure, as is shown in Fig. 4.3a. If either of the RF or LO signals is supplied alone, there is no output signal. However, either of the input signals can be inputted single-endedly. Then, the other side of the input or switch quad is ac-grounded. For example, the output signal of the single-ended LNA can be brought to the mixer in that way [19]. Then, the mixer conversion gain is decreased, because only half of the mixer input stage transconductance is utilized. Double-balanced mixers are usually preferred in DCRs because they generate less even-order distortion, provide high port-to-port isolation, and usually have better noise performance than their single-balanced counterparts. A single-balanced mixer is shown in Fig. 4.3b. It does not require a differential RF input signal and requires only half of the current of a balanced topology. The main problem in using single-balanced mixers is the large LO signal feedthrough at its output that may saturate the following stage [16]. Single-balanced mixers are suited for the first downconversion in receivers utilizing multi-step downconversion, see, for example, [15]. Next, different parts of the downconversion mixer and interfaces to other blocks are discussed.



Fig. 4.3. a) Double-balanced mixer, b) single-balanced mixer.

4.3.1 Interface to LNA

The LNA output is usually ac coupled to the mixer input. Typically, the load of the LNA is a resonator and therefore the dc level of the LNA output is close to the positive supply voltage (when NMOS transistors are used as LNA input devices and resonator is connected towards positive supply). Thus, the LNA output dc level is not sufficient for the mixer input stage and with ac-coupling the mixer can be biased independently. In addition, due to ac-coupling, the second-order distortion generated by the LNA is filtered out, and the main IIP2 source of the front-end is the mixer itself.

4.3.2 Mixer input stage

Since the noise performance of mixers is generally rather poor, the transconductance is needed for signal amplification. In addition, a separate input stage improves the reverse isolation and separates the switch transistors from the LNA output. In the following, different possibilities to implement the mixer input stage are discussed. It should be noted that downconversion mixers without a separate transconductor stage are published as well, for example, in [P1], [P2], [20], and [21]. Such structures, where the LNA is merged with the downconversion mixer, are briefly introduced in Section 4.4.4.

In this thesis, stand-alone mixers were not implemented but the mixers were implemented together the LNA. Therefore, input matching techniques for mixers are not discussed. Nevertheless, there may be a need to match the mixer input to a certain impedance, for example due to the interstage SAW filter, for example. In that case, input matching methods similar to those presented for the LNA, would, in general, be applied to mixers as well.

Usually, the mixer input stage is composed of a balanced grounded CS amplifier or a differential pair with a tail current source, as shown in Fig. 4.4a and Fig. 4.4b, respectively. Both cases have their benefits and drawbacks. With the grounded CS stage, the available voltage headroom is relaxed by V_{DS} . This is crucial, especially in low-voltage designs. The third-order nonlinearity of both common-source and differential pair transconductors generally improve with larger overdrive voltages. In addition, when biased at the same current and device dimensions, the grounded CS stage has better third-order nonlinearity performance than the differential pair with a tail current source [22], [23]. However, a tail current source offers a common-mode rejection ratio, and the trade-off of the grounded CS stage is greater sensitivity to supply noise [22]. The tail current source can be replaced with an LC resonator tuned at the fundamental frequency to achieve high impedance to ground [24]. The LC resonator can also be tuned at the second harmonic to suppress common mode and second harmonic signals [25]. The inductor provides a dc bias path and thus dc voltage drop is avoided. As a drawback, an on-chip inductor consumes significant silicon area.

The linearity of the input stage can be improved by source degeneration. Degeneration can be implemented with resistive or reactive components. Using resistors leads to small layout area, but resistors are noisy and cause a voltage drop. If inductors are utilized instead, better noise performance is achieved, but at the cost of increased layout area. In addition, the transconductance stage with inductive degeneration has better IIP3 than those with resistive degeneration [26]. In a case of quadrature mixers, the source inductor can be shared between I- and Q-input stages to save the valuable silicon area [27].



Fig. 4.4. Mixer transconductance stage utilizing a) grounded CS stage, b) differential pair with a tail current source, or c) source-coupled pair with a narrow-band LC resonator.

The noise due to the mixer transconductor stage is briefly discussed at a general level. The thermal noise due to the transconductance stage is at the same frequency as the input RF signal and, in DCRs, it will be downconverted to dc frequency, just like the useful signal [28]. It should be noted that any periodic LO waveform downconverts the noise from the odd-harmonics of the LO signal also. The noise at each frequency is uncorrelated and the noise contributions are added as the mean square [28]. However, the noise at the fundamental frequency accounts for a major part of the noise, even if the noise from the input stage is wideband [29]. If inductive source degeneration or input matching is used, the frequency response of the transconductance stage becomes frequency selective. In that case, noise from fewer harmonics should be taken into account [29]. The flicker noise of the transconductance transistors is up-converted to LO frequency and its harmonics. Typically, the flicker noise corner frequency of the transconductance MOSFETs is much lower than the LO frequency. Thus, they contribute only thermal noise at the mixer output [28].

When second-order intermodulation is taken into account, the fully differential transconductor (Fig. 4.4b) has higher second-order linearity than the grounded CS transconductor [5]. The second-order nonlinearity is mainly determined by threshold voltage mismatch. It improves with increasing biasing currents and decreasing device widths, i.e. for larger overdrive voltages [5]. Alternative input stage possibilities minimizing the second-order intermodulation are shown in Fig. 4.5. Figure 4.5a presents a mixer where the transconductance stage is ac-coupled to the switch quad to filter out the IM2 distortion generated by the input stage. The achieved IIP2 improvement is only moderate [30]. Moreover, the transconductance stage and switch quad require different quiescent current levels for optimal performance. With separate dc paths, the performance of both stages can be optimized separately, but more current is consumed compared to the conventional structure [31]. Figure 4.5b presents a transconductor that simultaneously obtains large degeneration at low frequencies and low degeneration at RF frequencies [32]. As a result, high IIP2 and IIP3 are achieved. Figure 4.5c shows a simple biasing technique that cancels the IM2 distortion in the CS transconductor [33]. Ideally, if the transconductor is excited differentially and all the devices are matched, no IM2 distortion is generated. Based on Monte Carlo mismatch simulations, a mixer with an IIP2 over +70 dBm was achieved in [33] with actual components. In addition, the transconductor of Fig. 4.5c is suitable for low supply voltages. However, it has slightly lower IIP3 compared to the conventional CS transconductor, but the difference decreases with large overdrive voltages [33].



Fig. 4.5. Mixer inputs to minimize the second-order intermodulation: a) ac-coupled input stage, b) degeneration with triode-region FET [32], and c) biasing technique for IM2 cancellation [33].

4.3.3 Switch quad

After the input transconductance stage, the RF signal is fed to a switching quad, which performs the frequency translation. The performance of the switch quad is crucial with regard to the gain and noise of mixers. A switch quad typically causes additional loss due to frequency conversion. In addition to loss, the switch transistors generate thermal noise in a way similar to other active components [29]. In DCRs, where the signal is directly transferred to the intermediate frequency, the challenge of CMOS switching transistors is the high intrinsic flicker noise [23], [29]. Furthermore, although the linearity of the input transconductor is typically the dominant nonlinearity source of the mixer, also the switch quad nonlinearity should be taken into account [34].

4.3.3.1 Conversion gain

An example of a simple mixer LO pair and transistor M_1 and M_2 currents versus LO voltage are shown in Fig. 4.6. When there is no LO signal, i.e. $V_{LO} = 0$, the bias current is evenly shared between M_1 and M_2 . When the switch pair is excited by an LO voltage V_{LO} , the tail current is steered to either of the branches. For the following simulation example, the transistors M_1 and M_2 are replaced with voltage-controlled current sources obeying classical long-channel transistor equations, i.e.

$$I_{D} = \begin{cases} k_{n} \left(V_{GS} - V_{T} \right)^{2}, V_{GS} > V_{T}, V_{DS} \ge V_{GS} - V_{T} \\ 2k_{n} \left[\left(V_{GS} - V_{T} \right) V_{DS} - \frac{1}{2} V_{DS}^{2} \right], V_{GS} > V_{T}, V_{DS} \le V_{GS} - V_{T} \\ 0, V_{GS} < V_{T} \end{cases}$$
(4.1)

The design values of $k_n = 30 \text{ mA/V}^2$ and $V_T = 0.4 \text{ V}$ were chosen⁷. The drain current of LO pair transistor M_I is shown in Fig. 4.6 relative to I_{bias} for cases, where the overdrive $(V_{gs}-V_T)$ voltages are set to 50 mV, 100 mV, and 200 mV and the LO signal amplitude is swept. The drain-source voltage is approximately 600 mV. As can be seen, the required differential LO amplitude should be at least $\sqrt{2}$ times the overdrive voltage to completely drive the bias current to either of the branches. In that case, the other transistor of the LO pair acts as a cascode device and the other transistor is switched off.



Fig. 4.6. Mixer LO pair and transistor M_1 and M_2 currents relative to I_{bias} versus LO amplitude.

The quiescent current of switch transistor M_1 is shown in Fig. 4.7 in cases where the LO pair is driven with a sinusoidal LO signal having different amplitudes. The overdrive voltage is set to 100 mV. A solid line shows switching, which is performed with LO signal amplitude that is too low. The transistor is not off at any moment and incomplete switching occurs. With dashed and dashed-dotted lines are shown switching with larger LO voltages. The transition from on to off state is sharper with larger LO amplitude. Fast switching is required to keep the time period, when both transistors are on, as short as possible.

⁷ These values do not accurately model any particular transistor in deep submicron CMOS processes. The only purpose of using them is to achieve a simple and useful transistor model for simulation purposes.



Fig. 4.7. Quiescent current of LO pair transistor M_1 with different LO signal amplitudes.

The conversion gain of the switching quad is calculated by averaging the gain over one period of the LO signal. This is analyzed in literature, for example, in [35] - [38]. For a fundamental RF frequency the conversion gain of a single-balance and double-balanced mixers are given by

$$G_{conv} = \frac{1}{\pi} \sin\left(\frac{\pi\Delta T}{T}\right) \frac{T}{\Delta T},$$
(4.2)

where T is the LO signal period and ΔT is the turn on time. When the LO signal has a typical 50-% duty cycle, i.e. $\Delta T = 0.5T$, the voltage gain of a mixer is given by

$$A_V = \frac{2}{\pi} g_m R_L, \qquad (4.3)$$

where g_m is the transconductance of the single input transistor and R_L is the load resistance [23]. As a result, the switching causes conversion loss of 3.9 dB. Equation (4.3) assumes that the LO signal amplitude is large enough for complete switching. A too-low LO signal causes incomplete switching, which increases the conversion loss and noise. Next, the conversion loss of an ideal double-balanced mixer is simulated as a function of LO signal level. The input transistors of a mixer shown in Fig. 4.3 are replaced with ideal voltage-controlled current sources having a constant transconductance and the switch transistors are replaced with voltage-controlled current sources obeying (4.1). The switch quad conversion gain is shown as a function of LO signal amplitude for different overdrive voltages. With increasing overdrive voltages, larger LO amplitude is required to approach the theoretical limit of 3.9 dB loss. When the single-ended LO signal amplitude is equal to overdrive voltage, the conversion loss is approximately 5 dB in this example. Furthermore, as is observed from Fig. 4.8, when the differential LO signal amplitude is $\sqrt{2}$ times the overdrive voltage, the achieved conversion loss is approximately 0.5 dB from the theoretical limit.



Fig. 4.8. Conversion gain vs. LO signal amplitude with several switch transistor overdrive voltages.

4.3.3.2 Noise and linearity

The noise of commutating mixers is widely presented in the literature, for example, in [28], [29], and [37] - [39]. In the following, the common methods of minimizing the noise of the mixer apply to the LO signal amplitude and shape, device sizing and type, and quiescent current. Clearly, trade-offs between mixer noise and linearity and overall front-end power consumption are encountered.

LO signal amplitude: The noise of the switch transistor can be decreased by raising the amplitude of sinewave LO, which sharpens the transition and minimizes the time when both switches are on [23]. In addition, transition time can be shortened replacing the sine-waveform by a more square-like one [37]. However, if the LO voltage drives the switch transistor deep into triode region, mixer nonlinearity worsens because of the nonlinear output resistance of the transconductance FETs [40]. The rail-to-rail LO signal with short rise and fall times can be achieved by driving the sinusoidal signal through an inverter chain. This, however, can consume a substantial amount of current in the LO signal chain.

Switch sizing: The choice of the size of the switch transistors is a trade-off between the parasitic capacitance for the LO buffers and switching time. For a given LO swing, switch transistors are simultaneously on for a shorter period of time as their width increases [22]. In addition, the spectral density of input-referred noise goes down linearly with MOS transistor gate area [28]. However, increasing the switch transistor gate area is impractical due to the higher parasitic capacitances, which increase the current consumption of the LO buffer. In addition, the capacitance at the source of the switch transistor slows down the switching speed and degrades the SNR [37], [28]. The tail capacitance can be tuned out with the inductor between the transconductor stage and switch quad. As a result, the resonator lowers the effect of the mixer noise [20].

Quiescent current: To minimize the 1/f noise, the drain current of the switch transistor should be decreased [28]. However, when the switch transistor drain current is reduced, the impedance seen at the switch transistor sources increases. As a result, more RF current will be shunted by

the parasitic capacitance at the switch transistors' source node, which degrades the signal-tonoise ratio [28]. Therefore, there exists a certain drain current level for 1/f and thermal noise optimum, which depends on the used technology and baseband bandwidth. Finally, although the switch quad usually has a minor effect on overall mixer linearity, the linearity of the switch transistor typically worsens along with small quiescent currents [34].

Device type: The 1/f noise can also be mitigated by using PMOS switches instead of NMOS ones, as is achieved, for example, in [41] – [44]. The PMOS switch quad is usually used in folded mixers, where the transconductor stage uses NMOS devices. The folded mixers are briefly discussed in Section 4.4.2.

4.3.4 Load design and interface to the baseband

The design of mixer load is important, since in DCRs the mixer load forms the interface to the baseband. The typical voltage-mode load for the mixer is realized with an RC-impedance, as is shown in Fig. 4.9a. The load also acts as a first filtering stage for baseband. The pole improves the mixer out-of-band blocking performance and reduces the out-of-band linearity requirements of the following baseband channel. The mixer and the first stage of the baseband circuitry are usually dc coupled. The ac-coupling between the mixer and baseband is possible in those communications systems where the modulation method allows high-pass filtering. When the dc coupling is utilized, the dc level of the mixer output must be adequate for the following baseband stage. In the case of a double-balanced mixer, if resistive load is used, the switch transistor's dc current I_{swi} and the value of the load resistor R_L are related, i.e.

$$V_{dc,out} = V_{DD} - 2I_{swi}R_L \tag{4.4}$$

To improve the voltage gain of the mixer, for example to suppress the noise contribution of the following blocks, the load resistor value can be increased by adding parallel current sources with the resistors loads as presented in Fig. 4.9b. It should be noted that, in practice, the linearity of the analog baseband circuit limits the value of R_L [45]. In addition, the largest voltage swing in the front-end is typically achieved at the mixer output node (assuming a voltage mode operation). Thus, the maximum achievable voltage swing can be limited by the available voltage headroom and the signal clipping at the mixer output can dominate the front-end compression point. The mixer load can be also realized with floating resistors, as is shown in Fig. 4.9c [36]. The PMOS current sources provide the dc current to the mixer. The current sources, however, add new noise sources to the mixer output. The optimum resistor value and mixer load design is analyzed in, for example, [45].



Fig. 4.9. Different voltage-mode mixer load structures, a) RC-impedance, b) parallel resistor and current source, and c) floating resistors with PMOS current sources.

The mixer can be designed in current mode, i.e. to drive low-impedance node, as is shown in Fig. 4.10. This can be achieved by connecting the mixer outputs to the virtual grounds of the operational amplifier (op-amp), as is shown in Fig. 4.10a [44], [47], [48]. Then, the IF output operates in current-mode instead of voltage mode and the voltage swing at the output is very small. Since the output voltage signal modulating the switch transistor is significantly decreased, the nonlinearity of the switches is minimized and the blocking performance of the mixer is improved. The current-mode interfaces are suitable for low-voltage applications, since there is no voltage signal at the mixer output [45].

The cascode or folded cascode transistors can be used to implement a current-mode interface [45]. An example of such an interface is shown in Fig. 4.10b [P1]. Compared to the topology shown in Fig. 4.10a, the folded cascode topology requires larger current through the current source I_B , thus increasing noise. However, the structure shown in Fig. 4.10b is implemented without op-amp and low power consumption can be achieved in mixer-baseband interface.



Fig. 4.10. Current-mode interfaces using a) an op-amp [47] and b) folded cascode transistors [P1].

4.4 Active mixer design techniques and alternatives

In the following, several mixer design options are presented. First, two alternatives to realize the quadrature mixer are presented and compared. Then, folded mixers and a mixer utilizing current boosting are discussed. Finally, mixers merged with the LNA are briefly described.

4.4.1 Quadrature mixers

In typical receivers, the downconversion to zero or IF frequency is usually performed with quadrature (I/Q) mixers. For I/Q-mixers, there are different possibilities to implement the interface between input transconductor and switch quad. The first topology is shown in Fig. 4.11. The RF signal is inputted to two separate input stages, which drive the switch quads of their own. Another possibility is to utilize a single input stage, which drives both switch quads, as is shown in Fig. 4.12.



Fig. 4.11. Two separate Gilbert mixers driven by quadrature LO signals.



Fig. 4.12. Quadrature mixer with single input stage.

The conversion loss of a typical mixer and the one driving quadrature switch quads is shown in Fig. 4.13a. For that simulation, ideal mixers having an ideal input transconductor and switch transistor obeying (4.1) were used. The quiescent current of the switch transistor is 250 μ A. The conversion loss is plotted as a function of the LO amplitude. The value of the input transconductance is equal for both mixers. Since the transconductor of the mixer shown in Fig. 4.12 drives both switch quads, the conversion gain is by $\sqrt{2}$, i.e. 3 dB lower than the that of a basic Gilbert cell mixer of Fig. 4.11. In addition, complete switching requires larger LO amplitude when quadrature switch quads are driven from a single transconductor.

However, the comparison shown in Fig. 4.13a is not fair, since the drain current of the input stage of I/Q mixer of Fig. 4.12 is double compared to the ones of Fig. 4.11. Therefore, with reasonable accuracy, the size of the input transistor can be doubled to achieve equal capacitive loading for the preceding stage and overdrive voltage equal to the ones in Fig. 4.11. As a result, the transconductance of input devices in Fig. 4.12 can be doubled. The corresponding comparison is shown in Fig. 4.13b. The conversion gains are normalized such that the conversion gain of a typical switch quad is 0 dB. As a result, with equal current consumption and capacitive loading for the preceding stage, the better conversion gain can be achieved with the quadrature mixer.

When the noise of the transconductors is considered, the quadrature mixer has a 3-dB advantage over a pair of typical mixers [49]. However, the quadrature mixer probably has a noise disadvantage due to the noise of the switch quads, i.e. noise from Q-branch can be observed at the output of I-branch, and vice versa. In addition, parasitic capacitance at the node between the transconductor and switch quad is expected to be larger in Fig. 4.12 than in Fig. 4.11 resulting in worsened SNR. The fair comparison of noise and linearity performance of mixers shown in Fig. 4.11 and Fig. 4.12 is nontrivial, since the mixing performance depends on the dimensioning and biasing of switch transistors, used LO signal level, and device matching. Which mixer structure alternative should be chosen depends heavily on the used circuit configuration and requirements.



Fig. 4.13. a) Conversion gain of a typical Gilbert cell mixer (solid line) and the mixer driving quadrature switch quads (dashed line). b) Conversion gain of a typical mixer is normalized to 0 dB and the transconductance of I/Q mixer is doubled compared to a).

For extremely low-power applications, for example [21] and [P1], the quadrature mixer shown in Fig. 4.12 is a preferred option, since the desired conversion gain is achieved with the smaller current consumption. In addition, on-chip inductors can be used as a part of downconversion mixers, for example to optimize the noise of the mixer [20] or to realize the high-impedance LC resonator load for the folded mixer (see, for example, Fig. 4.14b). Then, separate transconductor stages would require on-chip inductors of their own, which would result in increased layout area and complicated floor planning.

4.4.2 Folded mixers

In a typical Gilbert cell mixer, input stage, switching stage, and load are stacked. Due to shrinking supply voltages, the implementation of such a mixer is challenging, because all stages require sufficient voltage headroom to achieve adequate performance. This can be mitigated by using folded mixer structures, as shown in Fig. 4.14. In addition, since the dc current of both the transconductor and switch stages can be adjusted separately, the gain, linearity, and noise performance of both stages can be optimized separately.

High impedance needed for folding is obtained either with current sources, as in [50] and [51] (Fig. 4.14a), or by LC resonator tuned at the center frequency [24], [44], [52] (Fig. 4.14b). The current sources require voltage headroom for a proper operation and they add noise, but the LC resonator occupies layout area. Since the resonator has high impedance at the wanted frequency and low impedance at other frequencies, the frequency selective performance of the resonator can be used to improve the linearity of the mixer. For example, the standing wave at double LO frequency produced by the mixing pair is attenuated [24]. Thus, the voltage swing across the drain-source terminals of both mixer input stage and switch transistors is lowered, which results in improved IIP3 [44]. Since the resonator has low impedance at the frequency of $|f_{RFI} - f_{RF2}|$, the second-order intermodulation generated by the input stage of the mixer is attenuated. As a result, the IIP2 of the mixer is also improved [52]



Fig. 4.14. Folded mixer realized with a) current sources, b) LC resonator.

4.4.3 Mixers with current boosting

A current boosting method is utilized in a mixer presented in Fig. 4.15a [53]. The optimum bias for the input stage and switching stage can be optimized separately with current boosting. Due to proper gain and linearity performance, the input transconductance stage should be biased with rather higher current. However, the performance of the switch quad may require quite low current level for optimum operation. With smaller quiescent current, the flicker noise of the switch transistors is lowered, which is important in DCRs. To reduce the LO overdrive required for complete switching, the drain current of switch transistors is reduced. The conversion gain is increased for two principal reasons: the mixer requires a lower LO swing to switch completely and a larger load resistor value can be used to increase the voltage gain. Alternatively, if the load resistor value is kept unchanged, mixer design for lower supply voltages is alleviated with current boosting, since the voltage drop at the resistive load is reduced.

The disadvantage of the switch transistor current reduction is that the impedance seen at the switch transistor sources is raised, which degrades the signal-to-noise ratio, as has been discussed earlier in Section 4.3.3.2. In addition, the boost current source adds noise of its own and increases the mixer white noise figure. However, the overall mixer performance usually benefits from properly designed current boosting.

The current boosting technique can be modified in such a way that the PMOS boost source is used as an additional transconductor, as is shown in Fig. 4.15b [54]. Then, the effective trans-

conductance of the mixer input stage is the combination of transconductances, $g_{m,n1} + g_{m,p2}$. As a result, slightly improved gain and noise properties can be achieved. However, the gain and noise performance improvement depends heavily on the used circuit configuration.



Fig. 4.15. a) Current boosting with constant current source, b) current boosting comprising additional transconductance.

In [55], a mixer with dynamic current injection was presented. Instead of feeding a fixed dc current, the dynamic PMOS current sources inject current only at the switching events. As a result, the flicker noise is reduced. That technique, however, requires a large LO signal swing and good matching between the NMOS dc current sources and PMOS dynamic current sources.

The additional capacitance due to current boosting can be avoided by feeding the boost current through the differential inductor, as shown in Fig. 4.16. In addition, the noise contribution of the boost current source is minimized, since its noise is common mode at the mixer output [P1]. The dc current source I_{boost} can be replaced with active blocks utilized in the receiver, as is discussed in Section 6.3. As a result, the overall current consumption can be reduced efficiently.



Fig. 4.16. Current boosting to inductor center tap [P1].

4.4.4 Merged LNA and mixer

Fig. 4.17 shows a structure where the LNA and mixer are merged. Compared to separate LNA and mixers, the number of nodes operating at RF frequency can be reduced, which minimizes the current consumption [20]. However, due to the Miller effect, the reverse isolation of this topology is poor and the drain node of the LNA is susceptible to the double-frequency LO beat typically generated by a quadrature switch quad [20]. The reverse isolation can be improved by adding a cascode stage or by using a CG input stage instead [21], [P1]. The problem of the CG input is the high noise figure, since switch transistors act as a low impedance load for the input transistors, and therefore there is only small voltage gain provided before the switch quad. However, the merged CG-LNA and mixer topology is suitable for applications requiring extremely low power consumption.



Fig. 4.17. LNA merged with a mixer.

References

- K. L. Fong and R. G. Meyer, "Monolithic RF active mixer design," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 46, no. 3, March 1999, pp. 231-239.
- [2] E. A. M. Klumperink, S. M. Louwsma, G. J. M. Wienk, and B. Nauta, "A CMOS switched transconductor mixer," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 8, Aug. 2004, pp. 1231-1240.
- [3] G. Kathiresan and C. Toumazou, "A low voltage bulk driven downconversion mixer core," *IEEE Int. Symposium on Circuits and Systems (ISCAS'99)*, Orlando, FL, 30 May – 2 June 1999, pp. II-598-601.
- [4] K. Kivekäs, A. Pärssinen, and K. A. I. Halonen, "Characterization of IIP2 and dcoffsets in transconductance mixers," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 11, Nov. 2001, pp. 1461-1473.

- [5] D. Manstretta, M. Brandolini, and F. Svelto, "Second-order intermodulation mechanisms in CMOS downconverters," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 4, March 2003, pp. 394-406.
- [6] E. E. Bautista, B. Bastani, and J. Heck, "A high IIP2 downconversion mixer using dynamic matching," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1934-1941.
- [7] I. Elahi, K. Muhammad, and P. T. Balsara, "IIP2 and DC offsets in the presence of leakage at LO frequency," *IEEE Trans. on Circuits and Systems II: Express Briefs*, vol. 53, no. 8, Aug. 2006, 647-651.
- [8] M. Brandolini, M. Sosio, and F. Svelto, "A 750-mV 15kHz 1/f noise corner 51dBm IIP2 direct-conversion front-end for GSM in 90nm CMOS," *IEEE Int. Solid-State Circuits Conference (ISSCC'06)*, San Francisco, CA, 5-9 Feb. 2006, pp. 470-471.
- [9] W. Redman-White and D. M. W. Leenaerts, "1/f-noise in passive CMOS mixers for low and zero IF integrated receivers," *European Solid-State circuits Conference (ESSCIRC'01)*, Villach, Austria, 18-20 Sept. 2001, pp. 68-71.
- [10] S. Chehrazi, R. Bagheri, and A. A. Abidi, "Noise in passive FET mixers: A simple physical model," *IEEE Custom Integrated Circuits Conference (CICC'04)*, Orlando, FL, 3-6 Oct. 2004, pp. 375-378.
- [11] M. Valla, G. Montagna, R. Castello, R. Tonietto, and I. Bietti, "A 72-mW CMOS 802.11 direct conversion front-end with 3.5-dB NF and 200-kHz *1/f* noise corner," *IEEE Journal of Solid-State Circuits*, vol. 40, no. 4, April 2005, pp. 970-977.
- [12] N. Poobuapheun, W.-H. Chen, Z. Boos, and A. M. Niknejad, "A 1.5-V 0.7–2.5-GHz CMOS quadrature demodulator for multiband direct-conversion receivers," *IEEE Journal of Solid-State Circuits*, vol. 42, no. 8, Aug. 2007, pp. 1669-1677.
- [13] A. R. Shanani, D. K. Shaeffer, and T. H. Lee, "A 12-mW wide dynamic range CMOS front-end for a portable GPS receiver," *IEEE Journal of Solid-State Circuits*, vol. 32, no. 12, Dec. 1997, pp. 2061-2070.
- [14] R. Bagheri, A. Mirzaei, S. Chehrazi, M. E. Heidari, M. Lee, M. Mikhemar, W. Tang, and A. A. Abidi, "An 800-MHz–6-GHz software-defined wireless receiver in 90nm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2860-2876.
- [15] S. Mahdavi and A. A. Abidi, "Fully integrated 2.2-mW CMOS front end for a 900-MHz wireless receiver," *IEEE Journal of Solid-State Circuits*, vol. 37, no. 5, May 2002, pp. 662-669.
- [16] A. A. Abidi, G. J. Pottie, and W. J. Kaiser, "Power-conscious design of wireless circuits and systems," *Proceedings of the IEEE*, vol. 8, no. 10, Oct. 2000, pp. 1528-1545.
- [17] J. Crols and M. S. J. Steyaert, "A 1.5 GHz highly linear CMOS downconversion mixer," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 7, July 1995, pp. 736-742.
- [18] B. Gilbert, "A precise four-quadrant multiplier with subnanosecond response," *IEEE Journal of Solid-State Circuits*, vol. SC-3, no. 4, Dec. 1968, pp. 365-373.
- [19] J. Ryynänen, K. Kivekäs, J. Jussila, L. Sumanen, A. Pärssinen, and K. Halonen, "Single-chip multi-mode receiver for GSM900, DCS1800, PCS1900, and WCDMA," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 4, April 2003, pp. 594-602.

- [20] H. Sjöland, A. Karimi-Sanjaani, and A. A. Abidi, "A merged CMOS LNA and mixer for a WCDMA receiver," *IEEE J. of Solid-State Circuits*, vol. 38, no. 6, June 2003, pp. 1045-1050.
- [21] A.-S. Porret, T. Melly, D. Python, C. C. Enz, and E. A. Vittoz, "An ultralow-power UHF transceiver integrated in a standard digital CMOS process: Architecture and receiver," *IEEE J. of Solid-State Circuits*, vol. 36, no. 3, March 2001, pp. 452-466.
- [22] S. Wu and B. Razavi, "A 900-MHz/1.8-GHz CMOS receiver for dual-band applications," *IEEE Journal of Solid-State Circuits*, vol. 33, no. 12, Dec. 1998, pp. 2178-2185.
- [23] A. Rofougaran, J. Y.-C. Chang, M. Rofougaran, and A. A. Abidi, "A 1 GHz CMOS RF front-end IC for a direct-conversion wireless receiver," *IEEE Journal of Solid-State Circuits*, vol. 31, no. 7, July 1996, pp. 880-889.
- [24] E. Abou-Allam, J. J. Nisbet, and M. C. Maliepaard, "Low-voltage 1.9-GHz front-end receiver in 0.5-μm CMOS technology," *IEEE J. of Solid-State Circuits*, vol. 36, no. 10, Oct. 2001, pp. 1434-1443.
- [25] M. A. Copeland, S. P. Voinigescu, D. Marchesan, P. Popescu, and M. C. Maliepaard, "5-GHz SiGe monolithic radio transceiver with tunable filtering," *IEEE Trans. on Microwave Theory and Techniques*, vol. 48, no. 2, Feb. 2000, pp. 170-181.
- [26] K. L. Fong and R. G. Meyer, "High-frequency nonlinearity analysis of commonemitter and differential-pair transconductance stages," *IEEE Journal of Solid-State Circuits*, vol. 33, no. 4, April 1998, pp. 548-555.
- [27] T. Tikka, J. Mustola, V. Saari, J. Ryynänen, M. Hotti, J. Jussila, and K. Halonen, "A 1-to-4 channel receiver for WCDMA base-station applications," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'06)*, San Francisco, CA, 11-13 June 2006, pp. 23-26.
- [28] H. Darabi and A. A. Abidi, "Noise in RF-CMOS mixers: A simple physical model," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 1, Jan. 2000, pp. 15-25.
- [29] M. T. Terrovitis and R. G. Meyer, "Noise in current-commutating CMOS mixers," *IEEE Journal of Solid-State Circuits*, vol. 34, no. 6, June 1999, pp. 772-783.
- [30] K. Kivekäs, *Design and characterization of downconversion mixers and the on-chip calibration techniques for monolithic direct conversion radio receivers*, Doctoral Thesis, Helsinki University of Technology, Espoo, Finland, 2002, p. 67.
- [31] P. Zhang, T. Nguyen, C. Lam, D. Gambetta, T. Soorapanth, B. Cheng, S. Hart, I. Sever, T. Bourdi, A. Tham, and B. Razavi, "A 5-GHz direct-conversion CMOS transceiver," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 12, Dec. 2003, pp. 2232-2238.
- [32] M. Brandolini, P. Rossi, D. Sanzogni, and F. Svelto, "A +78 dBm IIP2 CMOS direct downconversion mixer for fully integrated UMTS receivers," *IEEE Journal of Solid-State Circuits*, vol. 41, no. 3, March 2006, pp. 552-559.
- [33] P. Sivonen, A. Vilander, and A. Pärssinen, "Cancellation of second-order intermodulation distortion and enhancement of IIP2 in common-source and common-emitter transconductors," *IEEE Trans. on Circuits and Systems I: Regular Papers*, vol. 52, no. 2, Feb. 2005, pp. 305-312.

- [34] M. T. Terrovitis and R. G. Meyer, "Intermodulation distortion in current-commutating CMOS mixers," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 10, Oct. 2000, pp. 1461-1473.
- [35] J. C. Rudell, J.-J. Ou, T. B. Cho, G. Chien, F. Brianti, J. A. Weldon, and P. R. Gray, "A 1.9-GHz wide-band IF double conversion CMOS receiver for cordless telephone applications," *IEEE J. of Solid-State Circuits*, vol. 32, no. 12, Dec. 1997, pp. 2071-2088.
- [36] S. Zhou and M. C. F. Chang, "A CMOS passive mixer with low flicker noise for lowpower direct-conversion receiver," *IEEE Journal of Solid-State Circuits*, vol. 40, no. 5, May 2005, pp. 1084-1093.
- [37] T. Melly, A.-S. Porret, C. C. Enz, and E. A Vittoz, "An analysis of flicker noise rejection in low-power and low-voltage CMOS mixers," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 1, Jan. 2001, pp. 102-109.
- [38] —, "Addition to "An analysis of flicker noise rejection in low-power and low-voltage CMOS mixers"," *IEEE Journal of Solid-State Circuits*, vol. 37, no. 8, Aug. 2002, p. 1090.
- [39] C. D. Hull and R. G. Meyer, "A systematic approach to the analysis of noise in mixers," *IEEE Trans. on Circuits and Systems I: Fundamental Theory and Applications*, vol. 40, no. 12, Dec. 1993, pp. 909-919.
- [40] F. Behbahani, J. C. Leete, Y. Kishigami, A. Roithmeier, K. Hoshino, and A. A. Abidi, "A 2.4-GHz low-IF receiver for wideband WLAN in 0.6-µm CMOS – Architecture and front-end," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1908-1916.
- [41] D. Manstretta, R. Castello, and F. Svelto, "Low 1/f noise CMOS active mixers for direct conversion," *IEEE Trans. on Circuits and Systems – II: Analog and Digital Signal Processing*, vol. 48, no. 9, Sept. 2001, pp. 846-850.
- [42] P. Choi, H. C. Park, S. Kim, S. Park, I. Nam, T. W. Kim, S. Park, S. Shin, M. S. Kim, K. Kang, Y. Ku, H. Choi, S. M. Park, and K. Lee, "An experimental coin-sized radio for extremely low-power WPAN (IEEE 802.15.4) application at 2.4 GHz," *IEEE J. of Solid-State Circuits*, vol. 38, no. 12, Dec. 2003, pp. 2258-2268.
- [43] Z. Xu, S. Jiang, Y. Wu, H.-y. Jian, G. Chu, K. Ku, P. Wang, N. Tran, Q. Gu, M.-z. Lai, C. Chien, M. F. Chang, and P. S. Chow, "A compact dual-band direct-conversion CMOS transceiver for 802.11a/b/g WLAN," *IEEE Int. Solid-State Circuits Conference* (ISSCC'05), San Francisco, CA, 6-10 Feb. 2005, pp. 98-99.
- [44] P. Sivonen, J. Tervaluoto, N. Mikkola, and A. Pärssinen, "A 1.2-V RF front-end with on-chip VCO for PCS 1900 direct conversion receiver in 0.13-µm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 2, Feb. 2006, pp. 384-394.
- [45] J. Jussila, *Analog baseband circuits for WCDMA direct-conversion receivers*, Doctoral Thesis, Helsinki University of Technology, Espoo, Finland, 2003, p. 247.
- [46] A. Rofougaran, G. Chang, J. J. Rael, J. Y.-C. Chang, M. Rofougaran, P. J. Chang, M. Djafari, J. Min, E. W. Roth, A. A. Abidi, and H. Samueli, "A single-chip 900-MHz spread-spectrum wireless transceiver in 1-µm CMOS Part II: Receiver design," *IEEE J. of Solid-State Circuits*, vol. 33, no. 4, April 1998, pp. 535-547.

- [47] P. M. Stroet, R. Mohindra, S. Hahn, A. Schuur, and E. Riou, "A zero-IF single-chip transceiver for up to 22Mb/s QPSK 802.11b wireless LAN," *IEEE Int. Solid-State Circuits Conference (ISSCC'01)*, San Francisco, CA, 4-8 Feb. 2001, pp. 204-205.
- [48] J. Ryynänen, M. Hotti, V. Saari, J. Jussila, A. Malinen, L. Sumanen, T. Tikka, and K. A. I. Halonen, "CDMA multicarrier receiver for base-station applications," *IEEE Journal of Solid-State Circuits*, vol. 41, no. 7, July 2006, pp. 1542-1550.
- [49] J. Harvey and R. Harjani, "Analysis and design of an integrated quadrature mixer with improved noise, gain, and image rejection," *IEEE Int. Symposium on Circuits and Systems (ISCAS'01)*, Sydney, Australia, 6-9 May 2001, pp. IV-786-789.
- [50] I. Vassilou, K. Vavelidis, T. Georgantas, S. Plevridis, N. Haralabidis, G. Kamoulakos, C. Kapnistis, S. Kavadias, Y. Kokolakis, P. Merakos, J. C. Rudell, A. Yamanaka, S. Bouras, and I. Bouras, "A single-chip digitally calibrated 5.15–5.825-GHz 0.18-μm CMOS transceiver for 802.11a wireless LAN," *IEEE J. of Solid-State Circuits*, vol. 38, no. 12, Dec. 2003, pp. 2221-2231.
- [51] M. S. J. Steyaert, J. Janssens, B. De Muer, M. Borremans, and N. Itoh, "A 2-V CMOS cellular transceiver front-end," *IEEE J. of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1895-1907.
- [52] T. W. Kim, B. Kim, and K. Lee, "Highly linear receiver front-end adopting MOSFET transconductance linearization by multiple gated transistors," *IEEE J. of Solid-State Circuits*, vol. 39, no. 1, Jan. 2004, pp. 223-229.
- [53] W. Sansen and R. Meyer, "An integrated wide-band variable-gain amplifier with maximum dynamic range," *IEEE J. of Solid-State Circuits*, vol. SC-9, no. 8, Aug. 1974, pp. 159-166.
- [54] S.-G. Lee and J.-K. Choi, "Current-reuse bleeding mixer," *Electronics Letters*, vol. 36, no. 8, 13th April 2000, pp. 696-697.
- [55] H. Darabi and J. Chiu, "A noise cancellation technique in active RF-CMOS mixers," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2628-2632.

5 Quadrature signal generation

This chapter concentrates on circuit techniques for quadrature (I/Q) local oscillator (LO) signal generation. The chapter is organized as follows. First, general quadrature signal generation aspects are discussed. Then, the design of frequency dividers is briefly presented. Next, the passive quadrature signal generation structures are described. The main emphasis of this chapter is on the analysis and design of polyphase filters. Since the input LO signal in the experimental circuits presented in Chapter 6 was external, the design aspects of the VCO and PLL are beyond the scope of this thesis. The quadrature LO signal generation by cross-connected VCOs or ring oscillators are directly related to VCO design, and therefore those I/Q signal generation methods are left out of this thesis as well.

5.1 General quadrature signal generation aspects

Quadrature local oscillator signals are required in direct-conversion (and low-IF) receivers to drive downconversion mixers. Most of the modern communication standards utilize phase or frequency modulated signals, which require mixing with in-phase and quadrature signals to prevent information overlapping [1]. Therefore, the generation of quadrature signals is an essential part of the front-end design. In typical receivers, the quadrature phasing is performed on the LO signal path. Quadrature phasing can be implemented on an RF signal path as well [2] - [4]. That method, however, inevitably increases the noise figure of the receive path due to loss caused by the passive network.

There are several techniques to generate the 90° phase shift [5]. Commonly, the in-phase (I) and quadrature-phase (Q) signals are generated with passive phase shifter [6], divide-by-two circuit [7], or cross-coupled VCOs [8]. The selection of the applied quadrature signal generation method depends on the targeted system, selected radio architecture, and applied IC process. The quadrature signals always experience some imbalance due to mismatches among the active and passive components and due to the layout of the circuit as well. Amplitude and phase balance of the generated I and Q signals affect the image rejection of the receiver and thus the quality of reception.

Despite the applied quadrature signal generation method, the LO buffers are typically needed to drive the mixers. LO buffers are used to minimize the amplitude imbalance between I and Q LO signal branches and to compensate for loss caused by passive phase shifters, for example. In addition, the frequency dividers usually are sensitive to capacitive loading and cannot directly drive the gate capacitance of the mixer switch transistor. There are several ways to implement LO buffers. For example, an LO signal having a large swing can be achieved by using simple CMOS inverters. In general, the typical amplifier structures presented in LNA and mixer chapters apply as LO buffers, as well. The design of LO buffers is not covered in detail in this thesis. Nevertheless, a couple of implemented LO buffers are presented in Chapter 6. In a well-designed LO buffer special attention is paid to guaranteeing operation with large-amplitude signals. When the signal at the LO buffer input is increased, the LO swing will be eventually saturated to a certain level either due to compression of the transconductance stage or signal clipping at the buffer output. The achieved LO buffer output swing should be large enough for

proper mixer switching performance. This can be challenging to achieve with low current levels and thus the LO buffers can carry a significant part of the whole receiver current consumption.

5.2 Static frequency dividers

The frequency dividers (divide-by-2) are widely used to generate quadrature signals. The static divide-by-2 circuits are based on the cross-connected master-slave D-latch configuration presented in Fig. 5.1, see, for example, [7], [9] – [20]. When the differential input signal has a 50-% duty cycle, the output signals have 90° phase difference.

The advantage of the frequency dividers as quadrature generation circuits is that they provide a large output amplitude swing. In addition, static frequency dividers can operate from dc to the maximum toggle frequency. The maximum operation frequencies of different frequency dividers are studied in, for example, [10]. By using a frequency divider the LO signal is not at the same frequency as the RF signal, which reduces the LO leakage and LO pulling in DCRs [7]. As a drawback, the current consumption of the divider becomes large if high operational frequency is needed. Due to device mismatches the output swings of the I and Q branches may have amplitude and phase imbalance.



Fig. 5.1. Block diagram of static divide-by-two circuit.

5.2.1 D-latch structures

A typical D-latch circuit diagram is presented in Fig. 5.2a. A source-coupled logic (SCL) latch consists of an input tracking stage, where M_{NI} and M_{N2} are utilized to sense and track the data variation, while a cross-coupled regenerative pair formed by M_{N3} and M_{N4} is used to store the data [11]. The operation mode of the latch is determined by the input signal (*CLK*_{IN}) driving the input differential pair M_{N5} and M_{N6} . With a high clock period, the tail current entirely flows to the tracking circuit and vice versa. Typically, NMOS devices are utilized because of their higher speed compared to PMOS transistors [12].

Nowadays, the realization of the latch presented in Fig. 5.2a can be challenging with the low supply voltages, since three transistors and a resistive load are stacked. The supply voltage limitation can be mitigated by removing the tail current source M_{N7} as shown in Fig. 5.2b [13].

In that case, however, the drain currents of the clocked transistors are defined by the clock signal. Without a constant current source, the amplitude of the clock signal decides the currents in the latch circuit, which also affect the output voltage swing and power consumption [13]. Since the static dc point is not well defined, the total current of the divider fluctuates in time and unpredictable output swings can result [7], [14]. In addition, one possibility to design a low-voltage latch is to utilize inductive loading instead of resistive load [15].



Fig. 5.2. Schematic of a CMOS latch a) with tail current source, b) without tail current source.

A low-voltage D-latch can also be implemented by shifting the clocked transistors as shown in Fig. 5.3a [16]. Due to tail current sources, the operation of the clocked transistors M_{N5} and M_{N6} and tracking (M_{N1}, M_{N2}) and holding (M_{N3}, M_{N4}) transistors is better controlled than the latch of Fig. 5.2b. When CLK is high, M_{N5} steers all the tail current from M_{N7} . Then, M_{N1} and M_{N2} are off and the latch is in the hold mode. When CLK signal is low, the reverse occurs. However, when CLK is low, M_{N5} is not completely off, and part of the tail current of M_{N7} flows through M_{N5} and loading of the output RC node through M_{N1} or M_{N2} is insufficient. To improve the operation in tracking mode, resistor R_{L2} shown in Fig. 5.3b can be added to steer a larger part of the tail current to either M_{N1} or M_{N2} .



Fig. 5.3. a) Low-voltage latch, b) low-voltage latch with improved tracking mode operation.

The maximum operational speed of a latch shown in Fig. 5.2a is limited by tracking mode, and therefore the tail current must be sufficiently high [11]. The performance of the latch can be improved, for example, by using separate tail currents for tracking and latching circuits [11]. In addition, the divider speed can be improved with the dynamic loading shown in Fig. 5.4 [17], [18]. The PMOS transistors M_{Pl} and M_{P2} are used as dynamic loads by varying their resistance with clock signals [18]. When the CLK signal is high, M_{N5} is turned on and PMOS load devices are operated in the linear region and have small impedance. As a result, the output RC time constant is small and the tracking speed is maximized. When CLK is low, PMOS devices have a small gate-to-source voltage (cut-off region) thus having larger impedance. Then, the RC time constant is large and maximum gain is achieved for latching mode.



Fig. 5.4. Latch utilizing dynamic load [18].

Another resistorless frequency divider utilizing only two stacked transistors is presented in Fig. 5.5a [19]. When CLK is high, M_{N5} and M_{N6} are off and the master latch is in the sense mode and the slave is in the store mode. Having only two stacked transistors, this topology can be useful in designs with low supply voltage. Since all devices are connected to the output node, the maximum operational frequency is limited. To improve the toggle speed, the store transistors M_{N3} and M_{N4} can be stacked above data transistors M_{N1} and M_{N2} as is shown in Fig. 5.5b [20].



Fig. 5.5. a) Schematic of a low supply voltage D-latch [19], b) D-latch with smaller output capacitance [20].

5.3 Dynamic frequency dividers

In addition, a class of dynamic or regenerative dividers is presented in the literature, for example in [21] – [26]. The principle of regenerative frequency division is shown in Fig. 5.6 [21]. The double-frequency input signal is fed to the mixer, which is followed by a low-pass filter and an amplifier. The output signal of the mixer contains the frequency f_{LO} and its harmonics $3f_{LO}$, $5f_{LO}$, ... [10]. When the input frequency f_{in} is high enough, the lowpass filter removes the high-order components and the output signal f_{LO} is fed back to the second mixer input. Without low-pass filtering the divider could lock to an undesired frequency. However, since the frequency response of the active mixer conversion gain shows low-pass behavior, an additional filter is not required [22]. The maximum toggle frequency of a dynamic divider is higher than that of a static frequency divider [10]. The disadvantage of regenerative frequency dividers is that they have a minimum operation frequency. This is determined by the input frequency at which the sum frequency at the mixer output, $f_{in} + f_{out}$, passes the low-pass filter [23].



Fig. 5.6. Principle of regenerative frequency divider.

The basic regenerative divider shown in Fig. 5.6 does not provide quadrature outputs. To achieve divide-by-2 with quadrature outputs, the regenerative divider requires two mixers connected as shown in Fig. 5.7a. A positive feedback loop is created such that each multiplier stage contributes 90 degrees of phase shift to the divided output signal [24]. Although higher toggle frequency can be achieved with dynamic dividers with lower power consumption compared to static dividers [10], they are quite seldom used as quadrature signal generators. A few design examples can be found, for example in [25] and [26]. In both designs, bias tuning is needed to control the phase difference between I- and Q-outputs. The schematic of a CMOS dynamic divider similar to the one designed with bipolar devices in [25] is shown in Fig. 5.7b.



Fig. 5.7. a) Dynamic divider with quadrature outputs [24], b) its CMOS implementation example.

5.4 First-order RC-CR networks

The first-order RC-CR network capable of providing 90° phase shift is presented in Fig. 5.8a. The ratio of the I and Q output signals is

$$\frac{\Delta V_{lout}\left(s\right)}{\Delta V_{Qout}\left(s\right)} = \frac{1}{sRC} \,. \tag{5.1}$$

Therefore, the phase difference between I and Q outputs is exactly 90° with all frequencies, but the outputs are balanced only at a single frequency of $\omega = 1/RC$. Thus, the equal amplitude frequency shifts if the absolute value of RC varies due to process and temperature.

Another RC-CR network is shown in Fig. 5.8b [30], [31]. The ratio of I and Q outputs is

$$\frac{\Delta V_{lout}(s)}{\Delta V_{Qout}(s)} = \frac{1 - sRC}{1 + sRC}.$$
(5.2)

Thus, the amplitude balance is unity at all frequencies, but the phase has exactly 90° difference only at $\omega = 1/RC$. In practical RF IC realizations, both networks are quite useless due to the limited amplitude or phase accuracy. The amplitude balance can be improved by, for example, using a limiting amplifier after the RC-CR networks. A better amplitude or phase balance can be achieved by cascading polyphase filters, as is discussed in the next section.



Fig. 5.8. a) Constant phase RC-CR network, b) constant amplitude RC-CR network.

5.5 Passive polyphase filters

Passive polyphase RC filters for quadrature generation was originally proposed by Gingell in the 1970s [32]. Passive polyphase filter (PPFs) IC realizations have been published since 1995 [6]. However, their detailed analysis in the literature is still limited [32] - [43] and explicit formulas given do not cover all relevant issues. In this section, the characteristics of an *n*-stage polyphase RC network are thoroughly discussed. For example, the transfer functions, frequency responses, image rejection, and loss are analyzed in detail. Based on this analysis design guidelines for optimum configuration and dimensioning will be presented. In addition, the impact of device tolerances and the effect of parasitic capacitance will be considered. The presented analysis is mainly focused on PPFs with a maximum of three stages, because the PPFs implemented on IC seldom have more stages. The equations can be derived for higher order PPFs too with the methods presented in this section.

Based on the input signal feeding technique two variants of PPFs can be separated. These are depicted in Fig. 5.9 and Fig. 5.10. In this thesis, these variants are called Type I and Type II PPF, respectively⁸. Since this slight change on input feeding has significant impact on the PPF performance, the analysis of both variants is carried out throughout the text. The material presented here is based on publication [P7].



Fig. 5.9. Type I polyphase filter.



Fig. 5.10. Type II polyphase filter.

5.5.1 Introduction to the analysis of polyphase filters

The analysis of polyphase filters starts by calculating the output signals and the frequency responses of a single PPF stage. An n^{th} stage of a polyphase filter is shown in Fig. 5.11a. The output is loaded with impedance Z_{Ln} representing the input impedance of the following stage. This can be a next PPF stage, a mixer switch, or an amplifier input.

⁸ As far as the author is aware, there are no standard names for the PPF structures. The purpose is not to give names for the two PPF topologies, but a short legend is needed to be able to separate the two structures in some way.



Fig. 5.11. a) An n^{th} PPF stage with output load impedance Z_{Ln} , b) load impedance of n^{th} PPF stage, when the load is a next PPF stage. Z_{Ln+1} represents the loading of the $n+1^{\text{th}}$ stage.

When a balanced input signal is applied to I input, the Q input nodes are virtual grounds whatever the source impedance is, and vice versa. When the load is the next PPF stage, the load impedance Z_{Ln} is presented with an equivalent circuit shown in Fig. 5.11b. The output voltages of n^{th} PPF stage shown in Fig. 5.11a can be calculated by voltage division and superposition rules. For example, the output node voltage $I_{out+,n}$ is calculated as follows

$$V_{Iout+,n} = V_{Iin+,n} \frac{(sC_n)^{-1} || Z_{Ln}}{R_n + (sC_n)^{-1} || Z_{Ln}} + V_{Qin-,n} \frac{R_n || Z_{Ln}}{(sC_n)^{-1} + R_n || Z_{Ln}}.$$
(5.3)

The first right-hand side term is achieved by applying the voltage signal to the input node $I_{in+,n}$ and grounding all the other inputs. Thus, the capacitor C_n connected to the input node $Q_{in-,n}$ is in parallel with the load impedance Z_{Ln} of the output node $I_{out+,n}$. The resulting output signal is calculated with voltage division. The other output voltages can be calculated similarly. In the general case, when each output node is loaded with impedance Z_{Ln} , the differential I- and Q-output signals are

$$\Delta V_{lout,n} = \frac{Z_{Ln}}{R_n + Z_{Ln} + sR_nC_nZ_{Ln}} \left(\Delta V_{lin,n} - sC_nR_n\Delta V_{Qin,n} \right),$$
(5.4)

$$\Delta V_{Qout,n} = \frac{Z_{Ln}}{R_n + Z_{Ln} + sR_nC_nZ_{Ln}} \left(sC_nR_n\Delta V_{lin,n} + \Delta V_{Qin,n} \right), \tag{5.5}$$

respectively. Equations (5.4) and (5.5) are then represented in an useful matrix notation.

$$\begin{bmatrix} \Delta V_{lout,n} \\ \Delta V_{Qout,n} \end{bmatrix} = \frac{Z_{Ln}}{R_n + Z_{Ln} + sC_nR_nZ_{Ln}} \begin{bmatrix} 1 & -sC_nR_n \\ sC_nR_n & 1 \end{bmatrix} \begin{bmatrix} \Delta V_{lin,n} \\ \Delta V_{Qin,n} \end{bmatrix}$$
(5.6)

When the n^{th} PPF stage is followed by a subsequent PPF stage, the input impedance of stage n+1 acts as the load Z_{Ln} . From Fig. 5.11b, the equivalent load impedance Z_{Ln} for stage n can be calculated as

$$Z_{Ln} = \frac{R_{n+1} + Z_{Ln+1} + sC_{n+1}R_{n+1}Z_{Ln+1}}{1 + sC_{n+1}(R_{n+1} + 2Z_{Ln+1})}.$$
(5.7)

 R_{n+1} , C_{n+1} , and $Z_{Ln}+1$ are the resistor values, capacitor values, and the load impedance of the n+1th stage, respectively.

5.5.2 About PPF topologies, terminology, and notation

The manner of feeding the input signal to the first stage affects the operation of the PPF. In the case shown in Fig. 5.9, the Q-inputs of the first stage are signal grounds, i.e. $\Delta V_{Qin,I} = 0$. In Fig. 5.10 the input signal is injected in a dual-feed manner, i.e. $\Delta V_{Qin,I} = \Delta V_{lin,I}$. For the rest of this paper, the former topology is referred to as Type I PPF and the latter topology as Type II PPF. In the following equations, the upper index ()^{*I*} or ()^{*II*} is used to indicate the type of the PPF.

5.5.3 PPF frequency responses and balances

The output voltages of an n-stage PPF can be calculated by multiplying the matrix (5.6) of each PPF stage, and by using (5.7) when taking into account the effect of the load of the next stage. At first, differential output voltages of a single-stage PPF Type I are calculated as

$$\Delta V_{lout}^{I}(s) = \frac{Z_{L1}}{R_{1} + Z_{L1} + sR_{1}C_{1}Z_{L1}} \Delta V_{lin}^{I}(s), \qquad (5.8)$$

$$\Delta V_{Qout}^{I}(s) = \frac{sR_{1}C_{1}Z_{L1}}{R_{1} + Z_{L1} + sR_{1}C_{1}Z_{L1}} \Delta V_{lin}^{I}(s).$$
(5.9)

The ratio between I and Q output signals is

$$\frac{\Delta V_{lout}^{I}(s)}{\Delta V_{Qout}^{I}(s)} = \frac{1}{sR_{1}C_{1}}.$$
(5.10)

According to (5.10), the output signals have exactly the same magnitude at the single angular frequency of $\omega_l = 1/R_lC_l$. Furthermore, (5.10) is purely imaginary ($s = i\omega$). Thus, the phase difference between I and Q outputs is exactly 90° with all frequencies and with all R_l and C_l values. In addition, the result is independent of the load impedance if identical load impedance at every output node is assumed.

A wider amplitude balance with Type I PFF is achieved by cascading several stages. For example, the transfer functions of I and Q channels of a two-stage Type I PPF are calculated with (5.6):

$$H_{2-stg,I}^{I}(s) = \frac{\Delta V_{lout}^{I}}{\Delta V_{lin}^{I}} = \frac{Z_{L1}Z_{L2}(1-s^{2}C_{1}R_{1}C_{2}R_{2})}{(R_{1}+Z_{L1}+sC_{1}R_{1}Z_{L1})(R_{2}+Z_{L2}+sC_{2}R_{2}Z_{L2})},$$
(5.11)

$$H_{2-stg,Q}^{I}\left(s\right) = \frac{\Delta V_{Qout}^{I}}{\Delta V_{lin}^{I}} = \frac{Z_{L1}Z_{L2}s\left(C_{1}R_{1} + C_{2}R_{2}\right)}{\left(R_{1} + Z_{L1} + sC_{1}R_{1}Z_{L1}\right)\left(R_{2} + Z_{L2} + sC_{2}R_{2}Z_{L2}\right)}.$$
(5.12)

The amplitude balance then becomes

$$A_{bal,2-stg}^{I} = \frac{\left|H_{2-stg,I}^{I}\left(s\right)\right|}{\left|H_{2-stg,Q}^{I}\left(s\right)\right|} = \frac{1+\omega^{2}R_{1}C_{1}R_{2}C_{2}}{\omega\left(R_{1}C_{1}+R_{2}C_{2}\right)}.$$
(5.13)

Thus, unity gain balance is achieved at $\omega_1 = 1/R_1C_1$ and $\omega_2 = 1/R_2C_2$. In addition, the phase balance is always 90° at all frequencies, R and C values, and load impedances. For higher order PPFs the transfer functions can be calculated similarly.

Next, we will repeat the previous analysis for Type II PPF. The output voltages can be calculated in a manner similar to that for Type I PPF but, in this case, the Q input signals are

 $V_{Qin+} = V_{Iin+}$ and $V_{Qin-} = V_{Iin-}$ for the first stage. The transfer functions of the single-stage Type II PPF are

$$H_{1-sig,I}^{II}(s) = \frac{Z_{L1}}{R_1 + Z_{L1} + sR_1C_1Z_{L1}} (1 - sC_1R_1), \qquad (5.14)$$

$$H_{1-sig,\mathcal{Q}}^{II}\left(s\right) = \frac{Z_{L1}}{R_1 + Z_{L1} + sR_1C_1Z_{L1}} \left(1 + sC_1R_1\right).$$
(5.15)

The ratio of I and Q outputs is

$$\frac{H_{1-stg,I}^{''}(s)}{H_{1-stg,Q}^{''}(s)} = \frac{1-sR_1C_1}{1+sR_1C_1}.$$
(5.16)

Thus, the amplitude balance is unity with all component R_I and C_I values, all frequencies, and load impedances, but the phase is exactly 90° only at $\omega_I = 1/R_IC_I$. For a two-stage Type II PPF, the transfer functions are

$$H_{2-sig,I}^{II}(s) = \frac{Z_{L1}Z_{L2}\left(1 - s\left(C_{1}R_{1} + C_{2}R_{2}\right) - s^{2}C_{1}R_{1}C_{2}R_{2}\right)}{\left(R_{1} + Z_{L1} + sC_{1}R_{1}Z_{L1}\right)\left(R_{2} + Z_{L2} + sC_{2}R_{2}Z_{L2}\right)},$$
(5.17)

$$H_{2-stg,Q}^{II}(s) = \frac{Z_{L1}Z_{L2}\left(1 + s\left(C_{1}R_{1} + C_{2}R_{2}\right) - s^{2}C_{1}R_{1}C_{2}R_{2}\right)}{\left(R_{1} + Z_{L1} + sC_{1}R_{1}Z_{L1}\right)\left(R_{2} + Z_{L2} + sC_{2}R_{2}Z_{L2}\right)}.$$
(5.18)

The ratio between I and Q outputs is

$$\frac{H_{2-sig,l}^{II}(s)}{H_{2-sig,Q}^{II}(s)} = \frac{1 - s(R_1C_1 + R_2C_2) - s^2R_1C_1R_2C_2}{1 + s(R_1C_1 + R_2C_2) - s^2R_1C_1R_2C_2}.$$
(5.19)

The previous analyzes can be extended for higher order PPFs as well. It turns out that Type II PPF has always unity amplitude balance and the phase is 90° only at each RC pole. Respectively, Type I PPF has ideal phase balance and amplitude balance is unity at each RC pole frequency.

5.5.4 Image-reject ratio

In this section, image-rejection ratios (IRR) of both PPF topologies are studied. In a receiver context, the IRR is defined as the relation of the desired sideband to the suppression of the image sideband. The IRR was first calculated by Norgaard [43] and it is expressed as

$$IRR = \frac{1 + 2A_{bal}\cos(\Delta\theta) + A_{bal}^{2}}{1 - 2A_{bal}\cos(\Delta\theta) + A_{bal}^{2}}.$$
(5.20)

In (5.20), A_{bal} and $\Delta\theta$ define the amplitude ratio of I and Q outputs and phase deviation from an ideal 90° between I and Q branches, respectively. The IRR contours (in decibels) are plotted as a function of A_{bal} and $\Delta\theta$ in Fig. 5.12. For example, to achieve an IRR better than 30 dB, A_{bal} and $\Delta\theta$ should be better than 0.55 dB and 3.6°, respectively.



Fig. 5.12. IRR contours (in decibels) as a function of A_{bal} and $\Delta \theta$.

If the IRR is separately defined with magnitude balance (IRR_{gain}) and phase deviation (IRR_{phase}) factors, (5.20) simplifies to

$$IRR_{gain} = IRR_{\Delta\theta=0} = \left(\frac{1+A_{bal}}{1-A_{bal}}\right)^2,$$
(5.21)

$$IRR_{phase} = IRR_{A_{bal}=1} = \cot^2\left(\frac{\Delta\theta}{2}\right).$$
(5.22)

Thus, the amount of phase and amplitude imbalance provided by a PPF can be converted into equal IRR. Such an IRR is a figure-of-merit for a PPF used for I/Q generation.

This section is divided into two parts. The IRR performances of both PPF topologies with equal and unequal RC pole frequencies are studied separately. In the former case, equal component values are utilized in all filter stages, while, in the latter, the resistor and capacitor values of different stages are adjusted separately.

5.5.4.1 Equal RC poles

Type I PPF has ideal phase response resulting in IRR = IRR_{gain} . Based on (5.10), the amplitude balance is simply $A_{bal} = \omega_l / \omega$, where $\omega_l = 1/R_l C_l$. According to (5.21), the IRR_{gain} becomes

$$IRR_{gain,1-stg}^{I} = \left(\frac{\omega + \omega_{1}}{\omega - \omega_{1}}\right)^{2}.$$
(5.23)

Respectively, Type II PPF has ideal gain balance. Therefore, IRR = IRR_{phase} , and solving the phase of (5.16) and using the definition of (5.22), the IRR_{phase} becomes

$$IRR_{phase,1-stg}^{II} = \cot^{2}\left[\frac{1}{2}\arctan\left(\frac{(\omega-\omega_{1})(\omega+\omega_{1})}{2\omega\omega_{1}}\right)\right].$$
(5.24)

It can be proven with trigonometric functions that (5.24) equals (5.23), i.e. the IRR performance is equal for both PPF types. That holds for multi-stage PPFs, too. Although both PPF topologies have equal IRR performance, the choice of PPF type may be eventually constrained by amplitude and phase imbalance specifications required by the system.
According to (5.23) and (5.24), the IRR depends only on pole frequency. Therefore, the IRR of a passive polyphase network cannot be improved by the choice of topology. For the following IRR analysis, (5.23) is used.

PPF stages with equivalent R_1 and C_1 values can be cascaded. The IRR of an *n*-stage PPF with equal poles is

$$IRR_{gain,n-stg} = \left(\frac{\omega + \omega_{\rm l}}{\omega - \omega_{\rm l}}\right)^{2n}.$$
(5.25)

The IRRs of PFFs with 1 to 4 stages are depicted in Fig. 5.13, where the frequency axis is scaled with respect to ω_l . The relative bandwidth BW_{rel} , which defines the ratio of maximum and minimum frequencies where a specific IRR is achieved, is shown in Fig. 5.13 and is defined as

$$BW_{rel} = \frac{\omega_{\max}}{\omega_{\min}} \,. \tag{5.26}$$

The BW_{rel} for an *n*-stage PPF is calculated as

$$BW_{rel,n-stg} = \left(\frac{IRR^{1/2n} + 1}{IRR^{1/2n} - 1}\right)^2.$$
(5.27)

As a result, the IRR as a function of BW_{rel} becomes

$$IRR = \left(\frac{\sqrt{BW_{rel}} + 1}{\sqrt{BW_{rel}} - 1}\right)^{2n}.$$
(5.28)

Equation (5.28) directly relates the targeted IRR, BW_{rel} , and the number of PPF stages together. The IRRs of PPFs with 1 to 4 stages are shown in Fig. 5.14 as a function of relative bandwidth. For example, if the designed PPF should have a BW_{rel} of 2, the minimum IRR increases approximately 15 dB per PPF stage.



Fig. 5.13. IRR of equal RC pole PPFs with 1 to 4 stages as a function of frequency. The frequency axis is scaled with respect to $\omega_l = 1/R_lC_l$.



Fig. 5.14. IRR vs. relative bandwidth for 1- to 4-stage PPFs with equal RC pole frequencies.

5.5.4.2 Two-stage PPF with unequal RC poles

The IRR for a two-stage PPF calculated with (5.13) and (5.21) is given by

$$IRR_{2-stg} = \left(\frac{\omega + \omega_1}{\omega - \omega_1}\right)^2 \left(\frac{\omega + \omega_2}{\omega - \omega_2}\right)^2,$$
(5.29)

where $\omega_l = 1/R_lC_l$ and $\omega_2 = 1/R_2C_2$. The corresponding curve is plotted in Fig. 5.15.



Fig. 5.15. IRR of a 2-stage PPF. According to (5.31), an IRR of 40 dB is achieved with pole-splitting factor $k_2 = 121/81$.

For the following analysis, without losing any generality, the pole frequency ω_l is assumed to be higher than ω_2 . Figure 5.15 depicts that with unequal RC poles there is a minimum IRR (*IRR_{min}*) locating between ω_l and ω_2 . The minimum IRR frequency is $\omega_{lRR,min} = 1/\sqrt{R_1C_1R_2C_2}$.

To simplify the following calculations, the ratio of RC poles ω_1 and ω_2 is defined with a pole-splitting factor k_2 as follows:

$$k_2 = \frac{\omega_1}{\omega_2} = \frac{R_2 C_2}{R_1 C_1} > 1.$$
(5.30)

Then, the minimum IRR of a two-stage PPF between ω_1 and ω_2 is expressed as

$$IRR_{\min,2-stg} = \left(\frac{\sqrt{k_2} + 1}{\sqrt{k_2} - 1}\right)^2.$$
 (5.31)

According to (5.29), the IRR depends only on the RC pole frequencies and, according to (5.31), the minimum IRR is defined by the ratio of RC pole frequencies. In the case of equal RC poles ($k_2 = 1$), (5.31) goes to infinity, since there is no inter-pole minimum. The corner frequencies $\omega_{c,min}$ and $\omega_{c,max}$ define points, where an IRR equal to *IRR*_{min} is achieved (see Fig. 5.15).

$$\omega_{c,\min\&\max} = \frac{\omega_1}{4} \frac{(k_2 + 1)^2 \pm (k_2 - 1)\sqrt{k_2^2 + 6k_2 + 1}}{k_2\sqrt{k_2}}$$
(5.32)

As a result, the relative bandwidth BW_{rel} , where the minimum IRR is reached, is given by

$$BW_{rel,2-stg} = \frac{\omega_{c,\max}}{\omega_{c,\min}} = \frac{(k_2+1)^2 + (k_2-1)\sqrt{k_2^2 + 6k_2 + 1}}{(k_2+1)^2 - (k_2-1)\sqrt{k_2^2 + 6k_2 + 1}}.$$
(5.33)

Therefore, if BW_{rel} is known, k_2 can be expressed as

$$k_{2} = \frac{1 + BW_{rel} + \left(\sqrt{BW_{rel}} - 1\right)\left(\sqrt{BW_{rel}} + 1\right)}{\sqrt{BW_{rel}}} - 1.$$
(5.34)

The relation between BW_{rel} and minimum IRR according to (5.31) for a two-stage PPF is

$$BW_{rel,2-stg} = \frac{\left(1 + \sqrt{IRR}\right)^2 + 2\sqrt{2}IRR^{1/4}\sqrt{IRR+1}}{\left(1 + \sqrt{IRR}\right)^2 - 2\sqrt{2}IRR^{1/4}\sqrt{IRR+1}}.$$
(5.35)

When comparing (5.35) to (5.27), it can be noted that the BW_{rel} of a 2-stage PPF with unequal poles is always wider compared to a PPF with equal poles. However, the difference decreases if a high IRR is required, because then small pole-splitting factor k_2 is required. For example, with an IRR target of 30 dB and 40 dB, the BW_{rel} of unequal poles is approximately 36 % and 18 % larger than with equal poles, respectively.

5.5.4.3 Three-stage PPF with unequal RC poles

For a three-stage PPF, the IRR as a function of frequency is

$$IRR_{3-stg} = \left(\frac{\omega + \omega_1}{\omega - \omega_1}\right)^2 \left(\frac{\omega + \omega_2}{\omega - \omega_2}\right)^2 \left(\frac{\omega + \omega_3}{\omega - \omega_3}\right)^2.$$
 (5.36)

As in the two-stage case, the following analysis assumes $\omega_l > \omega_2 > \omega_3$. The RC poles ω_2 and ω_3 are related to ω_l with pole-splitting factors k_2 and k_3 . k_2 is defined as in (5.30) and k_3 is

$$k_3 = \frac{\omega_1}{\omega_3} = \frac{R_3 C_3}{R_1 C_1} > 1.$$
(5.37)

When (5.36) is plotted, two IRR minimum notches between ω_3 and ω_1 are obtained, as is shown in Fig. 5.16. First, such a relation between k_2 and k_3 is calculated such that the minimum IRR of the two notches are equivalent. The *IRR_{min}s* are not located at geometric average frequencies of $\sqrt{\omega_1\omega_2}$ and $\sqrt{\omega_2\omega_3}$, but at

$$\omega_{IRR\min} = \frac{\omega_1}{\sqrt{2}} \sqrt{\frac{F_1(k_2, k_3) \pm \sqrt{(k_2 + 1)(k_2 + k_3)(k_3 + 1)(F_1(k_2, k_3) - 6k_2k_3)}}{k_2k_3(k_2 + k_2k_3 + k_3)}},$$
(5.38)

where

$$F_1(k_2,k_3) = k_2(k_2k_3 + k_2 + 1) + k_3(k_2k_3 + k_3 + 1).$$
(5.39)

To achieve equivalent IRR_{min} s at those frequencies, there are actually three optimum relations between k_2 and k_3 :

$$k_3 = \frac{1}{k_2}$$
, $k_3 = \sqrt{k_2}$, and $k_3 = k_2^2$.

Based on definitions of pole locations, $k_3 > k_2$ is required and only the last result always obeys this. From here on, the relation according to (5.40) is used in all calculations.

$$k_3 = k_2^2 \tag{5.40}$$

The exact IRR_{min} formula at frequencies according to (5.38) becomes far too complicated to present here, but it can be calculated and included into a design formula set. To simplify the calculations, the IRR minimum presented here is calculated at geometric average of RC poles to achieve a simpler equation. As a result, an IRR minimum of a three-stage PPF is

$$IRR_{\min,3-stg} = \left(\frac{\sqrt{k_2}+1}{\sqrt{k_2}-1}\right)^3 \left(\frac{k_2-\sqrt{k_2}+1}{k_2+\sqrt{k_2}+1}\right).$$
(5.41)



Fig. 5.16. IRR of a 3-stage PPF. An IRR of 40 dB is achieved with pole-splitting factor $k_2 \approx 1.84$ and $k_3 = k_2^2 \approx 3.39$.

Equation (5.41) slightly overestimates the IRR compared to the exact calculated IRR minimum. The maximum error is approximately 0.23 dB when k_2 is close to unity and decreases with a larger k_2 . Thus, the error is insignificant and (5.41) can be used to calculate k_2 if the IRR target is known. Due to the lack of a closed-form solution of the quintic formula, k_2 cannot be analytically solved from (5.41). Therefore, a rough value for k_2 can be sketched from Fig. 5.17 or k_2 can be calculated numerically.

Next, the relative bandwidth of a 3-stage PFF is calculated. The frequencies $\omega_{c,min}$ and $\omega_{c,max}$ shown in Fig. 5.16 present corners frequencies, where the *IRR_{min}* is achieved and are given as

$$\omega_{c,\min} = \frac{\omega_3}{2\sqrt{k_2}} \frac{\left(1 + k_2\right)^2 \left(1 + k_2^2\right) - \sqrt{F_2(k_2)}}{\left(2 + k_2 + k_2^2\right)},$$
(5.42)

$$\omega_{c,\max} = \frac{\omega_1}{2k_2\sqrt{k_2}} \frac{\left(1+k_2\right)^2 \left(1+k_2^2\right) + \sqrt{F_2(k_2)}}{\left(1+k_2+2k_2^2\right)},$$
(5.43)

where $\omega_l = 1/R_1C_1$, $\omega_3 = \omega_l/k_2$, and

 $F_2(k_2) = 1 + 4k_2 - 10k_2^4 + 4k_2^7 + k_2^8.$ (5.44)

The relative bandwidth calculated with (5.42) and (5.43) is approximated with the following formula, so that k_2 can be easily solved if BW_{rel} is known.

$$BW_{rel,3-stg} \approx 2k_2^2 - 1.9k_2 + 0.9 \tag{5.45}$$

Equation (5.45) predicts the exact relative bandwidth with an accuracy better than 1.5 % when $k_2 < 2.8$. The *IRR_{min}* and the relative bandwidth, where *IRR_{min}* is covered, are shown as a function of k_2 in Fig. 5.17. For example, to achieve better than 40-dB IRR, k_2 should not exceed 1.84. Then, the *BW_{rel}* becomes approximately 4.2, which is a significant improvement compared to the *BW_{rel}* of a two-stage PFF (1.78) with the same IRR.

5.5.4.4 Optimal pole splitting of higher order PPFs

An n-stage PPF with unequal RC poles has n-1 IRR minimum notches. When equivalent $IRR_{min}s$ are targeted, the optimum RC pole frequencies for PFFs having more than three stages are not achieved with a generic formula of $k_n/k_{n-1} = k_2$ (n>3). For example, the calculation of an optimal four-stage PPF is presented in Appendix B.

The IRR and relative bandwidth as a function of k_2 for optimal 2-, 3-, and 4-stage PPFs are presented in Fig. 5.17. Furthermore, the IRR as a function of relative bandwidth for 2-, 3-, and 4-stage PPFs is shown in Fig. 5.18.



Fig. 5.17. IRR and relative bandwidth BW_{rel} as a function of pole-splitting factor k_2 . The optimum pole splitting is calculated according to (5.40) and (B.5) for 3- and 4-stage PPFs, respectively.



Fig. 5.18. IRR as a function of relative bandwidth BW_{rel} for 2-, 3-, and 4-stage PPFs with unequal RC pole frequencies.

5.5.5 Intrinsic PPF loss

When PPFs are used for quadrature signal generation in the LO signal path, the loss due to PPF is usually compensated with on-chip buffers. Often, the current consumption of the LO buffers becomes a remarkable part of RF front-end power budget. Therefore, it is crucial to be aware of the methods to minimize the PPF loss. As a rule of thumb the loss due to PPF is usually estimated to be 3 dB/stage. That holds when all PPF stages have equal RC poles, the loss is calculated at $\omega = 1/RC$, and the PPF is terminated with infinite load impedance.

The previous section described the optimization of IRR bandwidth. The analysis showed that there was no difference in IRR performance between Type I and II PPFs. In this section, the loss of both PPF types is analyzed as a function of the number of stages and pole-splitting factor. In addition, in the case of a multi-stage PPF, the optimal device scaling is discussed. In the following analysis, the source impedance is assumed zero and the termination impedance of the last PPF stage is infinite. This section discusses the loss caused by the PPF only, i.e. intrinsic loss. The effect of the finite input and output impedances are considered in Section 5.5.6.

5.5.5.1 Loss of single-stage PPFs

The differential output signals of a single-stage PPF is calculated from (5.6) by setting $Z_{LI} = \infty$. In the case of Type I PPF, I and Q output voltages have low-pass and high-pass frequency characteristics, respectively. The outputs have an equal magnitude at $\omega_l = 1/R_l C_l$ calculated as

$$H_{1-sig,\omega_{1}}^{I} = \left| \frac{1}{1+sC_{1}R_{1}} \right|_{s=i/R_{1}C_{1}} = \frac{1}{2} \left| 1-i \right| = \frac{1}{\sqrt{2}},$$
(5.46)

which corresponds to 3-dB loss. For the Type II PPF the loss at ω_l is given by

$$H_{1-sig,\omega_{1}}^{II} = \left| \frac{1 - sC_{1}R_{1}}{1 + sC_{1}R_{1}} \right|_{s=i/R_{1}C_{1}} = \left| -i \right| = 1,$$
(5.47)

which is 3 dB less than the loss of Type I PPF.

5.5.5.2 Loss of two-stage PPFs

The loss for two-stage Type I PFF can be calculated by cascading (5.6) for n = 1 and n = 2 and by applying

$$Z_{L1} = \frac{R_2 + Z_{L2} + sC_2R_2Z_{L2}}{1 + sC_2(R_2 + 2Z_{L2})} \xrightarrow{Z_{L2} \to \infty} \frac{1}{2} \left(R_2 + \frac{1}{sC_2} \right)$$
(5.48)

for the load impedance Z_{L1} of the first PPF stage. The last form of (5.48) is used in the following analysis because Z_{L2} is infinite. In the case of Type I PFF, the I-output has a loss maximum and the Q-output has a loss minimum at the geometric average of the RC poles, i.e. $\omega = 1/\sqrt{R_1C_1R_2C_2}$, as is shown in Fig. 5.19.



Fig. 5.19. Loss of two-stage PPFs as a function of frequency. The loss is shown for both PPF types. The pole-splitting factor $k_2 = 3$ ($k_{2R} = 3$ and $k_{2C} = 1$).

At that frequency, the I and Q output losses become

$$L_{2-stg,lout,\max}^{I} = \frac{\left(C_{1}R_{1} + 2C_{2}R_{1} + R_{2}C_{2}\right)^{2}}{4C_{1}C_{2}R_{1}R_{2}},$$
(5.49)

$$L_{2-stg,Qout,\min}^{I} = \left(\frac{C_{1}R_{1} + 2C_{2}R_{1} + R_{2}C_{2}}{C_{1}R_{1} + C_{2}R_{2}}\right)^{2},$$
(5.50)

respectively. In the case of equal RC poles, (5.49) and (5.50) will lead to 6-dB loss (i.e. 3 dB/stage) at both outputs. However, in the general case, RC poles are not equal. It is possible to design the pole ω_2 to the wanted frequency with the pole-splitting parameter k_2 by scaling either the value of R_2 or C_2 or both. Since the minimum (or maximum) loss is a function of R_1 , C_1 , R_2 , and C_2 as is shown in (5.49) and (5.50), there are optimal component values, which minimize the PPF loss. To calculate the optimum device scaling, the parameter k_2 is further divided into two parts

$$k_2 = k_{2R} k_{2C} \,, \tag{5.51}$$

where k_{2R} and k_{2C} denote the ratio of the first- and second-stage resistors and capacitors, i.e. $k_{2R} = R_2/R_1$ and $k_{2C} = C_2/C_1$. Equation (5.49) can be modified into

$$L_{2-stg,lout,\max}^{I} = \frac{\left(1 + 2k_{2C} + k_{2C}k_{2R}\right)^{2}}{4k_{2C}k_{2R}} = \frac{\left(1 + 2k_{2C} + k_{2}\right)^{2}}{4k_{2}}.$$
(5.52)

According to the last form, if k_2 is fixed, for example due to the IRR requirement, the loss is minimized when k_{2C} is as small as possible. In other words, when the component values of the lower pole ω_2 are chosen, the resistor value should be larger and capacitor value smaller compared to the component values of the higher pole ω_1 (i.e. $R_2 > R_1$ and $C_2 < C_1$). This holds also for higher order PPFs. In practice, when a PPF with unequal RC poles is designed, it is beneficial to dimension all the capacitors with a fixed value. Then, $k_{2C} = 1$ and the pole-splitting factor k_2 is defined by the resistor ratio k_{2R} only. The following analysis is performed for cases where equal capacitor values are used in all stages, i.e. $k_{2C} = 1$ if not otherwise mentioned. Then, the losses for Type I PPF I and Q outputs as a function of k_2 are achieved by

$$L_{2-stg,\max,I}^{I} = \frac{\left(k_{2}+3\right)^{2}}{4k_{2}},$$
(5.53)

$$L_{2-stg,\min,Q}^{I} = \left(\frac{k_{2}+3}{k_{2}+1}\right)^{2},$$
(5.54)

respectively. The losses of both outputs are plotted in Fig. 5.20 as a function of k_2 . With equal RC poles ($k_2 = 1$), both (5.53) and (5.54) lead to a loss of 6 dB. In addition, it is observed that $k_2 \ge 1$ is required to minimize the losses. Therefore, the RC poles should be placed such that the highest RC pole is the first stage in the signal path, i.e. the impedance level increases along with signal path. Then, the average loss is less than the 3 dB/stage.

Next, a similar analysis is carried out for Type II PPF. According to (5.19), I and Q outputs of a two-stage Type II PFF always have balanced amplitude. As is shown in Fig. 5.19, the maximum loss is achieved at $\omega = 1/\sqrt{R_1C_1R_2C_2}$ and is

$$L_{2-stg,\max}^{II} = \frac{\left(C_{1}R_{1} + 2C_{2}R_{1} + R_{2}C_{2}\right)^{2}}{C_{1}^{2}R_{1}^{2} + 6C_{1}R_{1}C_{2}R_{2} + C_{2}^{2}R_{2}^{2}} = \frac{\left(1 + 2k_{2C} + k_{2}\right)^{2}}{1 + 6k_{2} + k_{2}^{2}}.$$
(5.55)

The loss is minimized by choosing the capacitor scaling factor k_{2C} as small as possible. The loss of Type II PPF is plotted in Fig. 5.20 as a function of k_2 , and $k_{2C} = 1$. When $k_2 = 1$, the loss becomes 3 dB. The loss does not have an optimum point as a function of k_2 and it approaches 0 dB when k_2 increases.



Fig. 5.20. Maximum loss of both PPF types (two-stage) as a function of pole-splitting factor k_2 . For Type I PPF, both the I and Q outputs are shown separately.

5.5.5.3 Loss of three-stage PPFs

The losses of three-stage PPFs as a function of frequency are shown in Fig. 5.21, where $k_2 = k_{2R} = 2$, and $k_{3C} = k_{2C} = 1$. Both the I and Q outputs of Type I PPF have loss maximums between ω_1 and ω_3 . The frequencies of loss maximums are too lengthy to present here as closed-form formulas. Therefore, the loss of a 3-stage PPF is calculated at $\omega_2 = 1/R_2C_2$, which is the geometric average of ω_1 and ω_3 . At that frequency the loss becomes

$$L_{3-stg,\omega_{2}}^{I} = 2\left(1 + \frac{4}{\left(1 + k_{2}\right)^{2}}\right)^{2}.$$
(5.56)

The calculated and simulated loss maximums are plotted in Fig. 5.22 as a function of k_2 . Loss is reduced with increased pole splitting. When $k_2 < 2$, (5.56) predicts the loss of Type I with an accuracy better than 0.1 dB.

PPF Type II has a loss maximum exactly at ω_2 and it is expressed as

$$L_{3-stg,\omega_{2}}^{II} = \left(1 + \frac{4}{\left(1 + k_{2}\right)^{2}}\right)^{2}.$$
(5.57)

Comparison of (5.56) and (5.57) reveals that the intrinsic loss of Type II PPF is 3 dB smaller than that of Type I at ω_2 . The 3-dB difference in loss between PPF types is achieved at every pole frequency independent of the number of stages.



Fig. 5.21. Loss of three-stage PPFs as a function of frequency. Loss is shown for both PPF types. The pole-splitting factor $k_2 = 2$.



Fig. 5.22. Loss of three-stage PPF at ω_2 as a function of pole-splitting factor k_2 . Loss of PPF types I and II are calculated according to (5.56) and (5.57), respectively. The simulated maximum loss of PPF Type I is also shown.

5.5.6 The effect of termination impedances

In this section the effects of input and output impedances on the PPF loss are analyzed. Finite termination impedances cause voltage division both at the input and output of the PPF. In previous sections, the optimal pole frequencies and device scaling were analyzed. The analysis so far, however, does not define the resistor and capacitor values that should be chosen for the PPF. This chapter shows how the impedance level of the PPF should be chosen to minimize the overall loss if finite source (Z_s) and load impedances (Z_L) are known a priori. In addition, the optimum impedance level depends on the PPF topology and on the number of stages.

5.5.6.1 Optimum component values for a single-stage PPF

The differential input impedance of a Type I PPF is

$$Z_{in,1-stg}^{I} = 2\frac{R_1 + Z_{L1} + sC_1R_1Z_{L1}}{1 + sC_1(R_1 + 2Z_{L1})},$$
(5.58)

where Z_{LI} is the load impedance of a single output node (see Fig. 5.11a). The output impedance can be calculated with (5.58) by replacing Z_{LI} with Z_S . In the case of PPF Type I, the differential input and output impedance simplifies to R_I +1/s C_I at the pole frequency of $\omega_I = 1/R_IC_I$. The result is independent of Z_{LI} or Z_S .

In the general case, the signal loss due to voltage division at the input and output are calculated as

$$L_{input} = \frac{V_{in,PPF}}{V_{source}} = \left| \frac{Z_{in,PPF}}{Z_{in,PPF} + Z_S} \right|,$$
(5.59)

$$L_{output} = \frac{V_{load}}{V_{out,PPF}} = \left| \frac{Z_L}{Z_{out,PPF} + Z_L} \right|,$$
(5.60)

where $Z_{in,PPF}$ and $Z_{out,PPF}$ are differential input and output impedances of the PPF and Z_S and Z_L are differential source and load impedances, respectively. When Z_S is finite, the voltage division at the input is minimized by maximizing the value of R_1 . However, for the minimal voltage division at the output, the value of R_1 should be minimized when Z_L is finite. Thus, there is an optimum PPF impedance level, which minimizes the overall loss. The absolute values of Z_S and Z_L are related to the resistor value R_1 of the first PPF stage with parameters k_S and k_L :

$$\left|Z_{S}\right| = \frac{R_{\rm l}}{k_{\rm S}},\tag{5.61}$$

$$\left|Z_{L}\right| = R_{1}k_{L} \,. \tag{5.62}$$

A parameter k_Z is used to relate the output and input impedances.

$$\frac{|Z_L|}{|Z_S|} = k_S k_L = k_Z \tag{5.63}$$

As was shown earlier, the intrinsic PPF loss does not depend on the component values. Therefore, for a single-stage PPF, the optimum impedance level can be calculated by setting voltage divisions L_{input} and L_{output} equal. For Type I PPF the optimum output impedance relation k_L becomes

$$k_{L,opt,1-stg}^{I} = \sqrt{2 \left| \frac{Z_{L}}{Z_{S}} \right|} = \sqrt{2k_{Z}}$$
 (5.64)

The optimum R_I can then be calculated using (5.62). It should be noticed that $Z_{in,PPF}$, $Z_{out,PPF}$, Z_S , and Z_L are differential, when using equations above.

Due to the dual feed structure, the input impedance of the PPF Type II is half of input impedance of (5.58). When the analysis is repeated, the optimum k_L becomes

$$k_{L,opt,1-stg}^{II} = \sqrt{k_Z} . (5.65)$$

The optimum impedance level of Type I PPF is $\sqrt{2}$ times larger than of Type II PPF. Therefore, if k_Z is known, the resistor value R_I of Type II PPF should be $\sqrt{2}$ times larger than in PPF Type I. Accordingly, to maintain pole frequency ω_I , the capacitor value C_I of PPF Type II is $\sqrt{2}$ smaller than in PPF Type I.

The total losses of single-stage Type I and Type II PPFs taking into account both the intrinsic PPF loss and termination losses are expressed as a function of k_z as

$$L_{tot,1-stg}^{I} = \left(\frac{\sqrt{2} + 2\sqrt{k_{Z}} + \sqrt{2}k_{Z}}{k_{Z}}\right)^{2},$$
(5.66)

$$L_{tot,1-stg}^{II} = \left(\frac{2 + 2\sqrt{k_Z} + k_Z}{k_Z}\right)^2,$$
(5.67)

respectively. The total losses and the difference of the losses are plotted as a function of k_Z in Fig. 5.23. Due to finite input and output impedances, the difference between loss performance of PPF types is less than 3 dB.



Fig. 5.23. Overall loss of a single-stage PPF. Loss is shown for both PPF types as a function of k_Z (ratio of output and input impedances). In right *y*-axis, the difference of the losses is shown.

5.5.6.2 Optimum component values for multiple-stage PPFs

The optimization of a two-stage Type I PPF is first considered. When the input and output impedances are calculated, it turns out that the differential input impedance of PPF Type I at ω_l is R_l+1/sC_l regardless of the number of PPF stages. With multi-stage PFF, the input impedance at ω_2 is

$$Z_{in,\omega_2} = \frac{R_2 \left[\left(C_1 R_1 + C_2 R_2 \right)^2 + 2R_1 R_2 \left(C_1^2 + C_2^2 \right) \right]}{C_1^2 \left(R_1 + R_2 \right)^2 + R_2^2 \left(C_1 + C_2 \right)^2} - i \frac{R_2 \left[\left(C_1 R_1 + C_2 R_2 \right)^2 + 2C_1 C_2 \left(R_1^2 + R_2^2 \right) \right]}{C_1^2 \left(R_1 + R_2 \right)^2 + R_2^2 \left(C_1 + C_2 \right)^2}$$
(5.68)

At ω_2 , the output impedance of PPF Type I (Z_{out,ω_2}) is R_2+1/sC_2 and, at ω_1 , the output impedance Z_{out,ω_1} can be achieved with (5.68) by switching $R_1 \leftrightarrow R_2$ and $C_1 \leftrightarrow C_2$.

In the analysis of intrinsic PPF loss, we learnt that it is favorable to perform the pole splitting by scaling the resistor values to minimize the loss. Therefore, k_2 is again divided into two parts $(k_2 = k_{2R}k_{2C})$ to find both optimal component values and scaling ratio when finite termination impedances are taken into account. The optimum loss is found by making overall input and output voltage divisions at ω_1 and ω_2 equal. This results in the following equation:

$$\frac{Z_{in,\omega_1}}{Z_{in,\omega_1} + Z_S} \left| \frac{Z_L}{Z_L + Z_{out,\omega_1}} \right| = \left| \frac{Z_{in,\omega_2}}{Z_{in,\omega_2} + Z_S} \right| \left| \frac{Z_L}{Z_L + Z_{out,\omega_2}} \right|.$$
(5.69)

For simplicity, termination impedances Z_S and Z_L are assumed frequency independent. Equation (5.69) results an optimum k_L as

$$k_{L,opt,2-stg}^{I} = \sqrt{2k_{2R}k_{Z}} . (5.70)$$

A similar analysis can be derived for Type II PFF too, and then

$$k_{L,opt,2-stg}^{II} = \sqrt{k_{2R}k_Z} \ . \tag{5.71}$$

When the overall PPF loss is calculated with the result given by (5.70) or (5.71), the analysis again shows that pole splitting is beneficial to perform by increasing the resistor values. Therefore, the loss of the whole two-stage PPF is shown in Fig. 5.24 in a case where $k_{2C} = 1$ and resistor values (k_{2R}) are increased from 1 to 3. The Type II PPF has typically 1-2 dB lower loss than Type I PPF and the difference between PPF losses decreases with small k_Z values.



Fig. 5.24. Overall loss of a two-stage PPF. Loss is shown for both PPF types as a function of termination impedance ratio k_z . The pole-splitting factor k_2 has values of 1, 2, and 3. The arrow shows the direction of increasing k_2 .

Finally, the loss of a three-stage PPF is considered. For a Type I PPF, the I and Q outputs signals are balanced at ω_1 , ω_2 , and ω_3 . The optimum component values are achieved when voltage divisions due to termination impedances are equal at ω_1 and ω_3 , as they are in (5.69). Then, the optimum k_L values for Type I and II PPFs are calculated with

$$k_{L,opt,3-stg}^{I} = k_{2R}\sqrt{2k_{Z}} , \qquad (5.72)$$

$$k_{L,opt,3-stg}^{II} = k_{2R} \sqrt{k_Z} , \qquad (5.73)$$

respectively. Therefore, the optimum input impedance of PPF Type II is $\sqrt{2}$ smaller than that of Type I PPF for 1-, 2-, and 3-stage PPFs. The optimal total loss of a 3-stage PPF as a function of k_Z is shown in Fig. 5.25 for both filter types. The loss is shown in a case where $k_{2C} = 1$ and resistor values ($k_2 = k_{2R}$) are 1, 2, and 3.



Fig. 5.25. Overall loss of a three-stage PPF. Loss is shown for both PPF types as a function of k_z . The pole-splitting factor k_2 has values 1, 2, and 3. The result $k_2 = 1$ for Type II overlaps with $k_2 = 2$ for Type I. The arrow shows the direction of increasing k_2 .

5.5.7 The effect of parasitic capacitance

The realistic monolithic components always have parasitic capacitance to the substrate. For simplicity, in the following analysis, the parasitic capacitances of the PPF components are combined into a single parasitic capacitor C_{par} . In addition, it is assumed that the value of C_{par} is equal for each PPF node. Precisely speaking, this is not exactly true since larger resistors of small-valued devices implemented using several parallel resistors introduce more parasitic capacitance. However, the main contributors of parasitic capacitances are the capacitors attached to corresponding node, not the resistors. In a well-designed PPF, capacitor values are equal in all PPF stages. Thus, the error introduced by setting C_{par} fixed is small.

The output signals of n^{th} PPF stage are now expressed as

$$\begin{bmatrix} \Delta V_{lout,n} \\ \Delta V_{Qout,n} \end{bmatrix} = \frac{Z_{Ln}}{R_n + Z_{Ln} + sR_n Z_{Ln} \left(C_n + C_{par}\right)} \begin{bmatrix} 1 & -sC_n R_n \\ sC_n R_n & 1 \end{bmatrix} \begin{bmatrix} \Delta V_{lin,n} \\ \Delta V_{Qin,n} \end{bmatrix}.$$
 (5.74)

The load impedance Z_{Ln} is calculated from a network depicted in Fig. 5.26 and the corresponding formula is given in (5.75).



Fig. 5.26. The equivalent load network for calculation of PPF with parasitic capacitance.

$$Z_{Ln,Cpar} = \frac{R_{n+1} + Z_{Ln+1} + sC_{n+1}R_{n+1}Z_{Ln+1} + sC_{par}R_{n+1}Z_{Ln+1}}{1 + sC_{n+1}(R_{n+1} + 2Z_{Ln+1}) + sC_{par}(R_{n+1} + 2Z_{Ln+1} + s2C_{n+1}R_{n+1}Z_{Ln+1} + sC_{par}R_{n+1}Z_{Ln+1})}$$
(5.75)

The C_{par} is in parallel with the load impedance Z_L . As was calculated in (5.10) and (5.16), the load impedance does not have an effect on IRR. If the value of C_{par} is equal at each PPF node, it does not affect the IRR either. Therefore, only the loss caused by the parasitic capacitance C_{par} needs to be studied in this section. The amount of extra loss due to C_{par} is similar for both I and Q outputs of Type I PPF. It can be noted that C_{par} causes the same amount of loss for both PPF types, too. The effects of C_{par} on intrinsic loss and voltage division due to finite termination impedances are considered separately. Furthermore, how the C_{par} affects optimal device values is briefly discussed.

5.5.7.1 Single-stage PPF

For a single-stage PPF the additional intrinsic loss at $\omega_l = 1/R_I C_I$ is

$$L_{Cpar,1-stg} = \frac{1}{2} \left(\frac{C_1^2 + (C_1 + C_{par})^2}{C_1^2} \right) = \frac{1}{2} (1 + k_{par}^2),$$
(5.76)

where

$$k_{par} = \frac{C_1 + C_{par}}{C_1} = 1 + \frac{C_{par}}{C_1}.$$
(5.77)

For example, a 10-% relative parasitic capacitance causes approximately 0.45-dB additional loss. In addition, C_{par} decreases the input impedance of PPF at ω_l . Due to lower PPF input impedance, the voltage division at the input increases. Additional loss caused by voltage division at the input is

$$L_{Cpar,in} = 1 + \frac{C_{par}}{2C_1^2} \cdot \left(\frac{2C_1 + C_{par}}{1 + 2k_2 + 2k_s^2}\right) = 1 + \frac{k_{par}^2 - 1}{2\left(1 + 2k_2 + 2k_s^2\right)},$$
(5.78)

where k_s denotes the relation between R_l and differential source impedance according to (5.61). For example, with $k_s=1$, a 10-% relative parasitic capacitance causes less than 0.1 dB additional loss at input. In practice, $k_s > 1$, and therefore $L_{Cpar,in}$ has an insignificant effect compared to (5.76).

The parasitic capacitance decreases also the output impedance, which improves the voltage division at the output. The additional output voltage division due to C_{par} is less than 1 according to

$$L_{Cpar,out} = 1 - 2\frac{k_{par}^2 - 1}{k_{par}^2 + 1} \cdot \frac{1 + k_L}{2 + 2k_L + k_L^2}.$$
(5.79)

In (5.79), k_L is defined according to (5.62). Also (5.79) is quite insignificant compared to (5.76) if $k_L > 10$. Actually, by using optimal k_L values given by (5.64) and (5.65), it can be shown that, when the effect of losses $L_{Cpar,out}$ and $L_{Cpar,in}$ are combined, the result is always less than unity, i.e. parasitic capacitance decreases the termination losses. Therefore, as a worst case, the loss due to C_{par} can be predicted by (5.76).

5.5.7.2 Effect on optimal device values

In addition to the increased loss, the C_{par} modifies the optimal component values calculated with (5.64) and (5.65). The analysis of the optimal component values due to C_{par} using closed-form expressions is complicated. Therefore, the effect of C_{par} is presented by simulation results. First, the C_{par} is divided into dependent and independent parts of C_I as

$$C_{par} = \theta C_1 + C_{fixed} . ag{5.80}$$

 θ denotes the relative parasitic capacitance of C_I (per plate) and C_{fixed} presents the fixed parasitic capacitance of resistors and metal wiring of the layout etc. The effect of C_{par} was studied in two extreme cases: C_{par} is dependent only on C_I or C_{par} is independent of C_I . In the former case, decreasing the value of C_I and increasing the value of R_I minimizes the overall PPF loss. This is quite understandable because then C_{par} minimizes along with C_I . In the latter case, it is optimal to increase the value of C_I and decrease the value of R_I . This results from (5.76), which suggests that the value of C_I should be increased compared to C_{par} to minimize the k_{par} and additional intrinsic loss. In a realistic case, C_{par} has both independent and dependent parts of C_I and thus a starting point for finding the optimal component values is given by (5.64) and (5.65). This holds for the higher order PPFs, too.

5.5.7.3 Multi-stage PPFs

The loss caused by parasitic capacitance to a two-stage PPF is considered next. Only the intrinsic PPF loss is analyzed. As was shown for the single-stage PPF, it gives the worst case assumption. The additional loss as a function of frequency can be calculated with

$$L_{Cpar,2-stg}(\omega) = 1 + \frac{\left(k_{par}^{2} - 1\right)\left(\frac{\omega}{\omega_{l}}\right)^{2}\left(9 + 2k_{2} + k_{2}^{2} + 4k_{2}^{2}k_{par}^{2}\left(\frac{\omega}{\omega_{l}}\right)^{2}\right)}{1 + \left(9 + 4k_{2} + k_{2}^{2}\right)\left(\frac{\omega}{\omega_{l}}\right)^{2} + k_{2}^{2}\left(\frac{\omega}{\omega_{l}}\right)^{4}},$$
 (5.81)

where $\omega_l = 1/R_lC_l$ and k_{par} is calculated according to (5.77). In Fig. 5.27, (5.81) is plotted in two cases as a function of frequency. In the first case, no pole splitting is utilized ($k_2 = 1$, double RC pole at ω_l) and, in the second case, pole splitting with $k_2 = 3$ was utilized. Figure 5.27 was plotted with C_{par} values being 5 %, 10 %, 15 %, and 20 % of C_l . The figure depicts that the additional intrinsic loss of the PPF increases along with the frequency. In addition, the pole splitting enhances the additional loss.



Fig. 5.27. Additional loss due to C_{par} of a two-stage PPF as a function of frequency. In the first case, no pole splitting is utilized (double RC pole at ω_l) and, in the second case, PPF was designed with pole splitting of $k_2 = 3$. The C_{par} values used in plotting were 5 %, 10 %, 15 %, and 20 % of C_l .

The following analysis concentrates on calculating the loss at the highest pole ω_l , where the worst-case results are achieved. At ω_l and without pole splitting ($k_2 = 1$) the additional loss due to C_{par} is

$$L_{Cpar,2-stg,\omega=\omega_1,k_2=1} = \frac{1}{4} \left(1 + k_{par}^2 \right)^2,$$
(5.82)

which is familiar from (5.76). Actually, it can be calculated that the loss of an *n*-stage PPF at ω_l and with equal RC poles is

$$L_{Cpar,n-stg,\omega=\omega_1,k_2=1} = \frac{1}{2^n} \left(1 + k_{par}^2 \right)^n .$$
 (5.83)

Taking into account $k_2 > 1$, the additional loss for a two-stage PPF at ω_1 can be calculated as

$$L_{Cpar,2-stg,\omega=\omega_1,k_2>1} = 1 + \frac{1}{2} \left(k_{par}^2 - 1 \right) \left(1 + \frac{4 + 4k_{par}^2 k_2^2}{5 + 2k_2 + k_2^2} \right).$$
(5.84)

For the higher order PPFs, similar equations become quite lengthy and only a simulated result is presented. The additional loss due to C_{par} at ω_l is shown in Fig. 5.28 as a function of k_2 for 2- and 3-stage PPFs. The parasitic capacitance has values of 5 %, 10 %, 15 %, and 20 % of C_l . The loss of 3-stage PPF worsens quite dramatically along with pole splitting.



Fig. 5.28. Additional loss due to C_{par} of a 2- and 3-stage PPFs at ω_l as a function of pole-splitting factor k_2 . The parasitic capacitances used in figure were 5 %, 10 %, 15 %, and 20 % of C_l .

5.5.7.4 Summary of effect of parasitic capacitance

In this section, the effect of parasitic capacitance is analyzed with some simplifying assumptions. It is found that, if there is equal parasitic capacitance at each PPF stage output node, it does not affect the IRR performance. Because C_{par} modifies the input and output impedances of PPF, the optimal component values also change in a minor way. In addition, C_{par} increases the intrinsic loss, which worsens at higher frequencies and with pole splitting.

5.5.8 Component value deviation

This section analyzes the effect of the resistor and capacitor value deviation. In a typical PPF, the resistors and capacitors are in close proximity and the mismatch among resistor and capacitor values is usually well controlled in most IC processes. Therefore, the component mismatch is neglected in this analysis and only the minimum and maximum component deviations are considered. The effect of component mismatch is analyzed in [34] and [35], for example.

The relative maximum (ΔR_{max}) and minimum (ΔR_{min}) resistor value deviation from the typical value (R_{typ}) are defined in the following way:

$$\Delta R_{\max} = \frac{R_{\max} - R_{typ}}{R_{typ}} \rightarrow R_{\max} = R_{typ} \left(1 + \Delta R_{\max} \right), \qquad (5.85)$$

$$\Delta R_{\min} = \frac{R_{typ} - R_{\min}}{R_{typ}} \rightarrow R_{\min} = R_{typ} \left(1 - \Delta R_{\min} \right).$$
(5.86)

The capacitor deviation is defined in a similar manner. Typically, the component deviation is symmetrical, i.e.

$$\Delta R_{\max} \approx \Delta R_{\min} = \Delta R . \tag{5.87}$$

Therefore, the resistor R_n of the n^{th} PPF stage could be replaced with $R_n \to R_n(1\pm\Delta R_n)$ in all presented equations. The same can be done for the capacitors, too, $C_n \to C_n(1\pm\Delta C_n)$. Clearly, the pole frequencies shift due to the device-value variation.

$$\frac{1}{R_n C_n} \to \frac{1}{R_n \left(1 \pm \Delta R_n\right)} \frac{1}{C_n \left(1 \pm \Delta C_n\right)}$$
(5.88)

It is possible that the pole frequency remains unchanged because resistor and capacitor variations do not track each other. In the worst case, both the capacitor and resistor values vary in the same direction, i.e. only '+' or '-' signs apply. However, the pole-splitting factor remains nearly unchanged if the relative deviation is independent of the original component value. That holds when all the devices are made of the same material and have similar geometry. For example, in the case of a two stage-PPF, if $\Delta R_1 \cong \Delta R_2$ and $\Delta C_1 \cong \Delta C_2$, the pole-splitting factor k_2 is

$$k_{2} = \frac{R_{2}C_{2}}{R_{1}C_{1}} \to \frac{R_{2}(1 \pm \Delta R_{2})}{R_{1}(1 \pm \Delta R_{1})} \frac{C_{2}(1 \pm \Delta C_{2})}{C_{1}(1 \pm \Delta C_{1})} \approx k_{2}.$$
(5.89)

The unchanged pole-splitting factor holds for the higher order PPFs, too. As a result, the overall IRR remains unchanged but shifts in the frequency domain. To achieve the required IRR over the whole wanted bandwidth, and taking into account the device variation, the relative frequency BW_{rel} , defined as $\omega_{max}/\omega_{min}$, as in (5.27), (5.35), and (5.45), should be multiplied by bandwidth deviation factor (BW_{dev}) defined as:

$$BW_{dev} = \frac{R_{\max}C_{\max}}{R_{\min}C_{\min}} = \frac{\left(1 + \Delta R_{\max}\right)\left(1 + \Delta C_{\max}\right)}{\left(1 - \Delta R_{\min}\right)\left(1 - \Delta C_{\min}\right)}.$$
(5.90)

Therefore, the efficient relative factor $BW_{rel,eff}$ is defined:

$$BW_{rel,eff} = BW_{rel}BW_{dev}.$$
(5.91)

For example, a typical 25 % ΔR and ΔC variation leads to $BW_{dev} = 2.78$. Therefore, the required relative bandwidth BW_{rel} nearly triples.

5.5.8.1 Geometric center frequency deviation

When the PPF minimum (ω_{min}) and maximum (ω_{max}) operational frequencies are known, the geometric center frequency ω_c can be calculated as

$$\omega_c = \sqrt{\omega_{\min}\omega_{\max}} = \sqrt{BW_{rel}}\omega_{\min} = \frac{\omega_{\max}}{\sqrt{BW_{rel}}} .$$
(5.92)

The center frequencies of 1-, 2-, and 3-stage PPFs are $\omega_{c,l-stg} = \omega_l$, $\omega_{c,2-stg} = \omega_l/\sqrt{k_2}$, and $\omega_{c,3-stg} = \omega_l/k_2 = \omega_2$, respectively.

IRR performance of a 2-stage PPF ($k_2=2$) with minimum (RC_{min}), typical (RC_{typ}), and maximum (RC_{max}) component values ($\Delta R = \Delta C = 0.25$) are plotted in Fig. 5.29. The frequency where IRRs with minimum and maximum deviations overlap slightly differs from the geometric center frequency of a PPF with typical values. In the general case, the relative center frequency deviation can be calculated as

$$\Delta \omega_{RCdev} = \frac{\omega_{c,RClyp}}{\omega_{c,RCdev}} \,. \tag{5.93}$$

In (5.93), $\omega_{c,RCtyp}$ is the geometric center frequency of the PPF with typical component values, $\omega_{c,RCdev}$ is the geometric center frequency due to device variations, and thus $\Delta \omega_{RCdev}$ becomes



Fig. 5.29. IRR of a two-stage PPF with 25 % component variation.

The last form of (5.94) holds when device-value deviation is symmetrical according to (5.87). For example, with 25 % device variation, $\Delta \omega_{RCdev}$ is approximately 0.94. All the original pole frequencies ω_n of a PPF with nominal device values should be multiplied by a factor $\Delta \omega_{RCdev}$ to achieve the required IRR over the whole wanted bandwidth and taking into account the device variation.

5.5.9 Summary and design example

In this section the results are summarized and a design example is provided to clarify the PPF design procedure. Below is a list of major design principles of an optimum PPF. Reasoning for each item is given in the analysis, but for sake of brevity is not repeated here. A well-designed multi-stage PPF obeys the following rules:

- RC poles are split.
- Optimal pole splitting is calculated according to BW_{rel,eff}.
- The capacitors are of equal value in each stage.
- The impedance level (i.e. resistor values) increase along the signal chain.
- Component value deviation is taken into account when the effective relative bandwidth $BW_{rel.eff}$ is calculated.
- The resistor values are calculated taking into account the termination impedances.
- Parasitic capacitance is kept equal in each node of a PPF stage.

For the design example given next, an IRR of 40-dB is required, the wanted relative bandwidth BW_{rel} is 1.5, and component deviation $\Delta R = \Delta C = 0.25$. The source impedance (Z_S) is 100 Ω differentially and the load (Z_L) is a buffer with differential input impedance of 2 k Ω . The maximum corner frequency $\omega_{c,max}$ is scaled to 1. Then, the minimum corner frequency $\omega_{c,max} = 0.666$.

The design of a PPF starts by checking the needed number of stages. That depends on the IRR requirement and device-value variation.

- 1. *Effective relative bandwidth*: According to (5.90), the bandwidth deviation BW_{dev} is 2.78. Therefore, the effective required bandwidth (5.91) becomes 4.17.
- 2. Equal RC poles: According to (5.28), when $BW_{rel,eff} = 4.17$, the *n*-stage PPF with equal RC poles offers approximately *n*·9.31-dB IRR performance. Therefore, a 5-stage PPF would be needed.
- 3. Unequal RC poles: According to (5.34), in a two-stage PPF $k_2 = 2.69$ is needed to fulfill $BW_{rel,eff}$ requirement leading to less than 25-dB IRR (5.31). When a 3-stage PPF is utilized instead, with (5.45) k_2 can be solved to be 1.838. Equation (5.41) gives an IRR of 40.2 dB, which is adequate. The requirement for the number of stages could be quickly checked from Fig. 5.17.

When the number of PPF stages is known, the pole frequencies are calculated. The effect of component deviation is also taken into account.

- 4. Pole frequencies: The geometric center frequency ω_2 can be calculated with (5.92), $\omega_2 \approx 0.816$. The other poles, ω_1 and ω_3 , are achieved with (5.30), (5.37), and (5.40): $\omega_1 \approx 1.501$, $\omega_3 \approx 0.444$.
- 5. *Pole shifting*: The device-value deviation causes the geometric center frequency deviation. According to (5.93), the relative center frequency shift factor $\Delta \omega_{RCdev}=0.9375$. Therefore, all the pole frequencies are shifted to lower frequencies by that factor. The final pole frequencies are therefore $\omega_l \approx 1.407$, $\omega_2 \approx 0.765$, and $\omega_3 \approx 0.416$.

When termination impedances are known, the optimal device value can be calculated. In addition, the choice of PPF type is made.

6. Optimal device values: Based on termination impedance relation (5.63) $k_Z = 20$. The capacitor values are assumed to have only one value through the whole PPF, and therefore pole splitting is performed by scaling the resistor values only. The optimum k_L -values are given by (5.72) and (5.73), and they are $k_{L,opt}^I = 9.714$ and $k_{L,opt}^{II} = 6.869$ for Type I and Type II PPFs, respectively. Therefore, the R_I values calculated with (5.61) and (5.62) are 233 Ω and 330 Ω for Type I and Type II PPFs, respectively. The other resistor values are calculated with (5.30) and (5.37), i.e. R_2 and R_3 are 429 Ω and 788 Ω for Type I PPF, and 606 Ω and 1115 Ω for Type II PPF, respectively. The capacitor values are finally dimensioned with the pole frequencies

defined in Step 5. $C_1 = C_2 = C_3 = 1/\omega_l R_l$ i.e. 3.05 mF and 2.15 mF for Type I and Type II, respectively.

- 7. Loss: The loss without the effect of parasitics can be checked from Fig. 5.25. When $k_z = 20$ and $k_2 \approx 1.838$, the overall loss of Type I and Type II PPFs are approximately 10 dB and 8.4 dB, respectively. Therefore, it is optimal to choose Type II PPF to minimize the LO signal loss if the system allows frequency deviation of 1.15°.
- 8. The effect of parasitics: With modern IC processes, the bottom plate capacitance of capacitors is rather small, less than 10 %. According to Fig. 5.28, the additional loss at ω_l is approximately 2 dB with a three-stage PPF with k_2 of 1.838 regardless of PPF type.

5.6 Comparison of frequency dividers and polyphase filters for quadrature signal generation

In this chapter, the presented quadrature signal generation methods were mainly concentrated on static frequency dividers and passive polyphase filters. The advantages and disadvantages of both methods are briefly summarized and compared.

The main advantage of the passive polyphase filter is its simplicity. The bandwidth, where required amplitude and phase balance are achieved, can be increased by cascading several RC-CR stages. That, however, comes at the cost of increased LO signal loss, which needs to be compensated with amplifiers before the downconversion mixers. Therefore, to keep the signal loss at reasonable limits, in practical realizations, the number of PPF stages is typically two or three. To keep the required BW as narrow as possible, the passive components having as small deviation as possible are wanted. Clearly, the component deviation is set by the used technology and typically the resistors have stronger dependence on process parameters than the capacitors. The resistors should have well-controlled sheet resistance, which can lead to the need for additional process masks, and that is not always a desired option.

Passive phase shifters operate at the VCO frequency. Therefore, in DCRs the PA can disturb the VCO running at the same frequency, thus resulting in an effect known as LO pulling [7]. That can be avoided by using frequency dividers in quadrature signal generation, since VCO operates at double frequency. Divide-by-two circuits typically have a wide operational band and can provide a large output swing. However, the current consumption of the divider can become large if operation at high frequency is targeted. Although amplifiers are not needed to compensate the LO signal loss, the frequency dividers may require buffering before mixers as well. When the overall power consumption is considered, there is no general rule by which a quadrature signal generation method leads to an optimal result. From the layout size point of view, passive components such as resistors and capacitors usually consume more die area compared to MOSFETs. Thus, the area consumed by the PPFs typically is larger than that of frequency dividers. However, when the die area of the power supply stabilization capacitors especially required by divide-by-two circuits is taken into account, the overall layout areas of these two alternatives are comparable.

References

- [1] A. A. Abidi, "Direct-conversion radio receivers for digital communications," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 12, Dec. 1995, pp. 1399-1410.
- [2] C.-Y. Wu and C.-Y. Chou, "A 5-GHz CMOS double-quadrature receiver front-end with single-stage quadrature generator," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 3, March 2004, pp. 519-521.
- [3] M. Wiklund, S. Nilsson, C. Björk, and S. Mattison, "A 2GHz image-reject receiver in a low IF architecture fabricated in a 0.1µm CMOS technology," *IEEE International Symposium on Circuits and Systems (ISCAS'03)*, Bangkok, Thailand, 25-28 May 2003, vol. 2, pp. II-161-164.
- [4] C.-W. Kim and S.-G. Lee, "A 5.25-GHz image rejection RF front-end receiver with polyphase filters," *IEEE Microwave and Wireless Components Letters*, vol. 16, no. 5, May 2006, pp. 302-304.
- [5] A. Y. Valero-Lopez, S. T. Moon, and E. Sánchez-Sinencio, "Self-calibrated quadrature generator for WLAN multistandard frequency synthesizer," *IEEE Journal of Solid-State Circuits*, vol. 41, no. 5, May 2006, pp. 1031-1041.
- [6] J. Crols and M. S. J. Steyaert, "A single-chip 900 MHz CMOS receiver front-end with a high performance low-IF topology," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 12, Dec. 1995, pp. 1483-1492.
- [7] B. Razavi, *RF Microelectronics*, Prentice-Hall, 1998, p. 335.
- [8] A. Rofougaran, J. Rael, J. M. Rofougaran, and A. Abidi, "A 900MHz CMOS LCoscillator with quadrature outputs," *IEEE Int. Solid-State Circuits Conference* (ISSCC'96), San Francisco, CA, 8-10 Feb. 1996, pp. 392-393.
- [9] M. Suzuki, K. Hagimoto, H. Ichino, and S. Konaka, "A 9-GHz frequency divider using Si bipolar super self-aligned process technology," *IEEE Electron Device Letters*, vol. EDL-6, no. 4, April 1985, pp. 181-183.
- [10] W. Fang, A. Brunnschweiler, and P. Ashburn, "An analytical maximum toggle frequency expression and its application to optimizing high-speed ECL frequency dividers," *IEEE Journal of Solid-State Circuits*, vol. 24, no. 4, Aug. 1990, pp. 920-931.
- [11] P. Heydari and R. Mohanavelu, "Design of ultrahigh-speed low-voltage CMOS CML buffers and latches," *IEEE Trans. on Very Large Scale Integration (VLSI) Systems*, vol. 12, no. 10, Oct. 2004, pp. 1081-1093.
- [12] H.-D. Wohlmuth, D. Kehrer, and W. Simbürger, "A high sensitivity static 2:1 frequency divider up to 19 GHz in 120 nm CMOS," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'02)*, Seattle, WA, 2-4 June 2002, pp. 231-234.
- [13] Z. Gu and A. Thiede, "18 GHz low-power CMOS static frequency divider," *Electronics Letters*, vol. 29, no. 20, 2nd Oct. 2003, pp. 1433-1434.
- [14] C. Cao, Y. Ding, and K. K. O, "A 50-GHz phase-locked loop in 0.13-μm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 42, no. 8, Aug. 2007, pp. 1649-1656.
- [15] M. Terrovitis, M. Mack, K. Singh, and M. Zargari, "A 3.2 to 4GHz, 0.25µm CMOS frequency synthesizer for IEEE 802.11a/b/g WLAN," *IEEE Int. Solid-State Circuits Conference (ISSCC'04)*, San Francisco, CA, 15-19 Feb. 2004, pp. 98-99.

- [16] G. Schuppener, C. Pala, and M. Mokhtari, "Investigation on low-voltage low-power silicon bipolar design topology for high-speed digital circuits," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 7, July 2000, pp. 1051-1054.
- [17] H. M. Wang, "A 1.8V 3mW 16.8GHz frequency divider in 0.25µm CMOS," *IEEE Int. Solid-State Circuits Conference (ISSCC'00)*, San Francisco, CA, 7-9 Feb. 2000, pp. 196-197.
- [18] J. M. C. Wong, V. S. L. Cheung, and H. C. Luong, "A 1-V 2.5-mW 5.2-GHz frequency divider in a 0.35-µm CMOS process," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 10, Oct. 2003, pp. 1643-1648.
- [19] B. Razavi, K. F. Lee, and R. H. Yan, "Design of high-speed, low-power frequency dividers and phase-locked loops in deep submicron CMOS," *IEEE Journal of Solid-State Circuits*, vol. 30, no. 2, Feb. 1995, pp. 101-109.
- [20] R. Y. Chen, "High-speed CMOS frequency divider," *Electronics Letters*, vol. 33, no. 22, 23th Oct. 1997, pp. 1864-1865.
- [21] R. D. Miller, "Fractional-frequency generators utilizing regenerative modulation," *Proceedings of the IRE*, vol. 27, no. 7, July 1939, pp. 446-457.
- [22] H. Knapp, M. Wurzer, T. F. Meister, K. Aufinger, J. Böck, S. Boguth, and H. Schäfer, "86 GHz static and 110 GHz dynamic frequency dividers in SiGe bipolar technology," *IEEE MTT-S International Microwave Symposium (IMS'03)*, Philadelphia, PA, 8-13 June 2003, pp. 1067-1070.
- [23] M. Kurisu, G. Uemura, M. Ohuchi, C. Ogawa, H. Takemura, T. Morikawa, and T. Tashiro, "A Si bipolar 28-GHz dynamic frequency divider," *IEEE Journal of Solid-State Circuits*, vol. 27, no. 12, Dec. 1992, pp. 1799-1804.
- [24] J. R. Long, M. A. Copeland, S. J. Kovacic, D. S. Malhi, and D. L. Harame, "RF analog and digital circuits in SiGe technology," *IEEE Int. Solid-State Circuits Conference* (ISSCC'96), San Francisco, CA, 8-10 Feb. 1996, pp. 82-83.
- [25] J. P. Maligeorgos and J. R. Long, "A low-voltage 5.1–5.8-GHz image-reject receiver with wide dynamic range," *IEEE Journal of Solid-State Circuits*, vol. 35, no. 12, Dec. 2000, pp. 1917-1926.
- [26] A. Chung and J. R. Long, "A 5–6 GHz bipolar quadrature-phase generator," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 10, Oct. 2004, pp. 1737-1745.
- [27] J. Lee and B. Razavi, "A 40-GHz frequency divider in 0.18-µm CMOS technology," *IEEE J. of Solid-State Circuits*, vol. 39, no. 4, April 2004, pp. 594-601.
- [28] M. Fujishima, K. Asada, Y. Omura, and K. Izumi, "Low-power ½ frequency dividers using 0.1-µm CMOS circuits built with ultrathin SIMOX substrates," *IEEE Journal of Solid-State Circuits*, vol. 28, no. 4, April, 1993, pp. 510-512.
- [29] A. Engelstein, J.-M. Fournier, V. Knopik, C. Raynaud, "A comparison of 10 GHz frequency dividers in bulk and SOI 0.13 µm CMOS technologies," *IEEE International* SOI Conference (SOI'05), Honolulu, HI, 3-6 Oct. 2005, pp. 47-49.
- [30] M. D. Donald, "A 2.5GHz BiCMOS image-reject front-end," *IEEE Int. Solid-State Circuits Conference (ISSCC'93)*, San Francisco, CA, 24-26 Feb. 1993, pp. 143-144.

- [31] S. Otaka, M. Ashida, M. Ishii, and T. Itakura, "A +10-dBm IIP3 SiGe mixer with IIP3 cancellation technique," *IEEE Journal of Solid-State Circuits*, vol. 39, no. 12, Dec. 2004, pp. 2333-2341.
- [32] M. J. Gingell, "Single sideband modulation using sequence asymmetric polyphase networks," *Electrical Communication*, vol. 48, no. 1 and 2, 1973, pp. 21-25.
- [33] R. C. V. Macario and I. D. Mejallie, "The phasing method for sideband selection in broadcast receivers," *EBU Review (Technical Part)*, no. 181, 1980, pp. 119-125.
- [34] S. H. Galal, H. F. Ragaie, and M. S. Tawfik, "RC sequence asymmetric polyphase networks for RF integrated transceivers," *IEEE Trans. on Circuits and Systems II: Analog and Digital Signal Processing*, vol. 47, no. 1, Jan. 2000, pp. 18-27.
- [35] F. Behbahani, Y. Kishigami, J. Leete, and A. A. Abidi, "CMOS mixers and polyphase filters for large image rejection," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 6, June 2001, pp. 873-887.
- [36] K. Wada and Y. Tadokoro, "RC polyphase filter with flat gain characteristic," *IEEE Int. Symp. on Circuits and Systems (ISCAS'03)*, Bangkok, Thailand, 25-28 May 2003, pp. I-537-540.
- [37] H. Kobayashi, T. Kitahara, S. Takigami, and H. Sadamura, "Explicit transfer function of RC polyphase filter for wireless transceiver analog front-end," *IEEE Asia-Pacific Conference on ASIC (AP-ASIC'02)*, Taipei, Taiwan, 6-8 Aug. 2002, pp. 137-140.
- [38] N. Yamaguchi, H. Kobayashi, J. Kang, Y. Niki, and T. Kitahara, "Analysis of RC polyphase filters High-order filter transfer functions, Nyquist charts, and parasitic capacitance effects," *IEICE Technical Meeting of Circuits and Systems*, Wakayama, Japan, Jan. 2003, p. 6.
- [39] Y. Niki, J. Kang, H. Kobayashi, N. Yamaguchi, and T. Kitahara, "Analysis of RC polyphase filters Input impedance, output termination, component mismatch effects, flat-passband filter design," *IEICE Tech. Meeting of Circuits and Systems*, Wakayama, Japan, Jan. 2003, p. 6.
- [40] J. W. Archer, J. Granlund, and R. E. Mauzy, "A broad-band UHF mixer exhibiting high image rejection over a multidecade baseband frequency range," *IEEE Journal of Solid-State Circuits*, vol. SC-16, no. 4, Aug. 1981, pp. 385-392.
- [41] E. Tiiliharju and K. Halonen, "A broadband quadrature-modulator in standard 0.13 μm CMOS," *Analog Integrated Circuits and Signal Processing*, vol. 46, no. 1, Jan. 2006, pp. 37-46.
- [42] W. B. Mikhael, "Sequence discriminators and their use in frequency division multiplex-communication systems," *IEEE Trans. on Circuits and Systems*, vol. CAS-26, no. 2, Feb. 1979, pp. 117-129.
- [43] D. E. Norgaard, "The phase-shift method of single-sideband signal reception," *Proceedings of the IRE*, vol. 44, no. 12, Dec. 1956, pp. 1735-1743.

6 Design examples

In this chapter, design examples concerning the design of the LNA, downconversion mixers, LO buffers, and 90-degree phase shift circuits are presented. The main interest is the circuit design for sensor systems and the ultrawideband system (UWB). The design examples are given mainly as they were published in the original manuscripts listed in Chapter 1. The first design examples are involved in the design of a sensor radio IC. The two experimental receivers were presented in [P1] and [P2]. The baseband circuit design details are partially omitted from those two papers, since the author's contribution is not significant in that research area. Some material related to baseband design is nevertheless included here for the sake of the clarity and to gain a better understanding of the whole receiver operation. In Section 6.3, a current reuse method for transceivers is presented. In addition, a design example, where the proposed technique is used, is shown. Then, a design example of a 2.4-GHz RF front-end simulated with a 65-nm CMOS process is given [P3]. The rest of the chapter is devoted to the design of RF circuits for WiMedia UWB receivers. The wideband LNA design examples given in publications [P4] and [P5] are included in Section 6.5. Next, an experimental front-end IC designed and published in [P6] is presented. Finally, the summary of the design examples and comparison to the other published designs is given at the end of this chapter.

6.1 2.4-GHz receiver for sensor applications

This section describes a receiver designed to meet the stringent power consumption requirements for sensor radio, which operates in the 2.4-GHz ISM band with Bluetooth. To enable the reusability of the Bluetooth system, slight changes are made in the radio parameters. The details of the radio system concept can be found in [1].

The block diagram of the fabricated direct-conversion receiver (DCR) is presented in Fig. 6.1. The direct-conversion demonstrator receiver includes a low-noise amplifier, which is merged with quadrature downconversion mixers, local oscillator buffers, and one analog baseband (BB) channel with 1-bit limiter for analog-to-digital conversion. The measured active current consumption of the whole RX with a single baseband channel is 2.75 mA from a 1.2-V supply. With two BB channels, the current consumption would be 2.83 mA.



Fig. 6.1. Block diagram of the receiver.

6.1.1 RF front-end

6.1.1.1 LNA and quadrature mixers

The LNA, which is merged with I/Q-mixers, is shown in Fig. 6.2. The I-mixer is followed by a baseband channel, while the Q-channel mixer is loaded with resistors, which enables the separate characterization of the RF front-end in a dc operation point equal to the I-channel mixer. Compared to the solution where separate LNA and mixers are used, the current consumption of a merged structure with the same performance can be decreased by the amount of the dc current, which flows through the mixer switch transistors. In this receiver, the simulated current consumption of the combined LNA and mixer including the biases is 1.4 mA. If separate LNA and mixers were used, the current consumption would be at least 30 % higher.



Fig. 6.2. LNA and I/Q-mixers.

The LNA is a balanced common-gate (CG) amplifier with a moderate-Q inductor load (L_1). The CG configuration is used instead of the common-source topology because of the improved reverse isolation. In addition, in the CG configuration the Miller effect is avoided. Therefore, no cascode stage is needed, which is crucial in low-voltage applications. The differential input matching to 100 Ω is implemented with two LC resonators. To save the silicon area, the value of the source inductor L_S is decreased with shunt capacitance C_S . Furthermore, a small area inductor structure was chosen for L_S . When the area of L_S is decreased, the series resistance R_{L_S} increases, resulting in the lower quality value Q_{L_S} . As a consequence, part of the input signal leaks into the source resonator, thus worsening the signal-to-noise ratio (SNR). The R_{L_S} also contributes a noise of its own, as is predicted by (3.85). In addition, the value of L_S defines the source impedance level Z_{LC} and Z_{rel} according to (3.61) and (3.62), respectively. According to (3.88), Z_{rel} together with Q_{L_S} define the minimum achievable noise figure of a CG stage. Therefore, both parameters should be decreased carefully so as not to degrade the overall noise performance, but still some trade-off between sufficient noise performance and small die area can be made.

In addition, the finite Q-value of the inductor L_S affects the input matching. Because the drainsource resistor r_{ds} of the input transistor is much larger than the impedance of the load of the LNA, the load has an insignificant effect on the input matching. Thus, the input matching is

$$Z_{IN} = \left(sL_{IN} + \frac{1}{sC_{IN}} \right) + \left(\frac{sL_s + R_{Ls}}{sL_s \left(sC_T + g_m \right) + R_{Ls} \left(sC_T + g_m \right) + 1} \right), \tag{6.1}$$

where

$$C_T = 2C_S + C_{gs} + C_{par} \,. \tag{6.2}$$

 C_{gs} is the gate-source capacitance of the input transistor, g_m is the effective transconductance of the input transistor, and C_{par} is the capacitance at the source of the input transistor including all parasitic capacitances. At the resonance frequency, the input resistance R_{in} becomes

$$R_{in} = \frac{L_S}{L_S g_m + C_{tot} R_{LS}} = \frac{1}{g_m + \frac{1}{Z_{LC} Q_{LS}}}.$$
(6.3)

According to (6.3), the optimum S_{II} is achieved with a g_m having a value of less than $1/R_{in}$ due to its term, which is defined by the finite Q-value of the source inductor Q_{Ls} and the source resonator characteristic impedance level Z_{LC} . The width of the LNA transistor is chosen to optimize the g_m by using as small a dc current as possible. The simulated effective transconductance of a single input transistor as a function of the transistor width and drain current (I_D) is shown in Fig. 6.3. The transistor length is selected as the minimum needed to maximize the transistor f_T and g_m . The drain-source voltage (V_{DS}) of the input transistor was set to 300 mV. As can be seen in Fig. 6.3, to reach a g_m of 20 mS, which corresponds to 50 Ω , a current of at least 0.75 mA is needed, even with a wide input device. Because of a stringent current budget and a non-zero resistance R_{Ls} , the g_m of less than 20 mS was chosen, which is shown with a dot in Fig. 6.3. When optimizing for a small current consumption, the subthreshold region in Fig. 6.3, where the g_m/I_D ratio is at the maximum, can be exploited. The input referred third-order intercept point (IIP3) of the input transistor improves with a higher overdrive voltage. However, in this design, the input transistor linearity is not critical because the out-of-channel IIP3 is limited by the baseband.



Fig. 6.3. The effective transconductance (15 mS, 20 mS, and 25 mS) of the input transistor as a function of transistor width and drain current. The selected width and I_{ds} of the input transistor are shown with a dot.

The disadvantage of CMOS switching transistors is the high intrinsic flicker noise, which is the largest noise contributor in this design. According to simulations, the flicker noise of the switching transistors increases the NF of the whole receiver by 6 dB. To minimize the 1/f noise, the drain current of the switch transistor should be minimized, the active gate area of the mixer switch should be maximized, and the LO amplitude should be maximized [2], [3]. However, not all of these methods are suitable for a low-power low-voltage realization. For example, achieving a large LO swing with a sharp slope is current consuming in a DCR due to the high frequency of the LO signal. In addition, increasing the switch transistor gate area is impractical due to the higher parasitic capacitances, which increases the current consumption of the LO buffer [4]. Thus, the minimum gate length was chosen to keep the mixer gate capacitance small. In addition, the capacitance at the source of the switch transistor degrades the SNR [3]. The tail capacitance can be tuned out with the moderate-Q inductor L_1 between the LNA and the mixer. As a result, the resonator lowers the effect of the mixer flicker noise [5]. According to simulations, the inductor improves the NF and the voltage gain of the receiver by approximately 7-8 dB.

The mixer noise performance can be additionally improved by minimizing the drain current of the switch transistors, because the mixer output noise is directly related to the amount of the dc current of the switches [3]. Typically, the voltage gain and linearity improve with a larger switch current but, in this design, the noise of the switches is the most critical parameter because of the low gain of the LNA. As a result, very different currents are required in the switch transistors and in the LNA transistors to optimize the performance of the front-end and the whole receiver. In this design, the dc current that flows through the switch transistors is reduced with an additional current source I_{Boost} , which is approximately 70 % of the total drain current of the LNA input transistors. Compared to existing solutions [6] and [7], where separate current boosts are used, this design has a single boost current at the center tap of the load inductor L_l , which is a virtual ground. As a result, the thermal noise generated by the boost current source I_{Boost} will cause only common-mode noise at the output of the mixer. If separate boost current sources were used at the output nodes of the LNA, the parasitic capacitance would increase in that node. As a result, the Q-value of the LC resonator would be lower. It follows that the voltage gain would degrade and the following stages would have more effect on the output noise. In addition, separate current sources would cause an output noise voltage, which is comparable to the noise generated by the input transistors. Because the NF in this design is dominated by the flicker noise of the switches, the current boost to the center tap of L_1 has a relatively small effect on the total performance. However, the significance of the presented current boost method on the total output noise depends heavily on the selected process technology and the topology of the receiver.

6.1.1.2 LO buffer

The LO signals are amplified and buffered using balanced LO buffers of the kind shown in Fig. 6.4. The input stage of the LO buffer is a combination of NMOS and PMOS transistors. With typical LO input signal levels (i.e.> -10 dBm), the buffer operates in the large signal region. When the large LO signal has the maximum or minimum amplitude, it latches the input transistors such that the cascoded input transistors are not on simultaneously. The output signal is a sum of V_{DS} voltages of the complementary transistors (M_{NI} and M_{P2} , or M_{N2} and M_{PI}). When

compared to a conventional NMOS differential pair, the presented structure offers an approximately 4-5 dB larger voltage signal to the mixer gates with the same dc current. The NMOS-PMOS transistor pairs between the input and output are used for biasing.



Fig. 6.4. LO buffer.

The load of the LO buffer, which consists of an inductor L, a capacitor C, and capacitors $C_I - C_3$, is shown in Fig. 6.5. The resonator having a high Q-value was designed to achieve a sufficient LO signal swing with a small current consumption. As a result, the bandwidth of the LO buffer is narrow and the center frequency is sensitive to process variations. With a tunable capacitor load, the LO signal can be adjusted. Three-bit tuning with a range of ± 13 % covers all process corners. The tuning was realized by adding capacitors $C_I - C_3$ to the LC resonator. When the maximum capacitance setup is used, the Q-value of the resonator is lowered, which degrades the available signal swing. Thus, an extra current source I_E was added to increase the output signal swing of the LO buffer when the maximum capacitance setup is used. As a result, the LO signal amplitude variation can be kept within three decibels. The simulated total current consumption of one LO buffer including biases is 0.4 mA. With the additional current source I_E the current consumption increases to 0.6 mA.



Fig. 6.5. Load of the LO buffer.

6.1.2 Analog baseband circuit

6.1.2.1 Channel-select filter

The realized channel-select filter is a third-order all-CMOS filter. Figure 6.6a shows one half of the filter. The filter is realized with two single-ended circuits, which is also called a pseudo-differential structure. This structure does not need a common-mode feedback circuit, which helps the design when using low supply voltages [8]. When low power consumption and low power supply is targeted, a simple realization of the filter is preferred [9]. The third-order filter can be separated into a real and imaginary part [10]. First, there is a mixer-baseband interface with a real pole and servo loop. The real pole is followed by a biquad stage. The filter was designed for a -3-dB frequency of 550 kHz. The total current consumption of the filter including biases is 65 μ A.

6.1.2.2 Mixer-baseband interface

The mixer-baseband interface shown in Fig. 6.6a is the most critical part at baseband since it limits the dynamic range and dominates power dissipation of the baseband. The effective transconductance of the RF front-end is low. Thus, the noise contributed by the interface becomes significant. In addition, the output impedance of the mixer is small whereas a higher impedance level is preferred in the filter to minimize power dissipation. Because of the low transconductance of the RF front-end, small load impedance at the mixer output would require a high supply current in the following baseband stage to keep the input-referred noise of that stage sufficiently small. Another possibility is to increase the mixer output impedance by using cascode devices. Hence, compared to the commonly used resistive mixer load, an interface based on cascode transistors was used. An NMOS cascode device M_1 stacked with the mixer increases the mixer output impedance without additional current. Another cascode stage implemented as a folded PMOS transistor M_4 is needed to achieve sufficiently high output impedance to the NMOS load M_6 . The bias current of the PMOS cascode must be larger than the peak signal current caused by any out-of-channel interfering signal to avoid clipping. Steering some of the bias current through the NMOS current source M_5 increases the gain. The load transistor M_6 and the NMOS-capacitor M_7 form the real pole of the filter.

6.1.2.3 Biquad

The biquad stage of the channel-select filter shown in Fig. 6.6a is derived from a g_mC biquad prototype of the kind shown in Fig. 6.6b. To enable the use of low supply voltages, the transconductors are replaced with common-source NMOS transistors instead of using, for example, differential pairs. This filter structure is suitable for applications requiring moderate linearity. MOS capacitors are used because they provide a high capacitance per unit area and sufficient linearity [11].



Fig. 6.6. a) 3^{rd} -order all-CMOS channel-select filter and b) a prototype g_mC biquad.

6.1.2.4 Limiter

The 1-bit analog-to-digital conversion is realized with a limiter shown in Fig. 6.7. It consists of five stages, which give a total voltage gain of 47.5 dB. The differential signal at the channel-select filter output is transformed to a single-ended signal with a differential-to-single-ended converter. The single-ended limiter follows the differential-to-single-ended converter. The limiter consists of similar cascaded amplifier stages. By selecting a proper number of amplifiers, the power dissipation of the limiter can be minimized [12].



Fig. 6.7. Limiter.

6.1.3 Experimental results

The chip was fabricated with a 0.13-µm CMOS technology. The active chip area is 1.0 mm^2 . In addition to the actual receiver, the chip includes test structures and additional test pads that increase the total chip area to 3.3 mm^2 . The chips were bonded directly onto a printed circuit board (PCB). The chip micrograph is shown in Fig. 6.8.

The measured active current consumption of the whole receiver with one baseband channel is 2.75 mA from a 1.2-V supply. The measured and the simulated input matchings are shown in Fig. 6.9. The measured S_{II} is better than -10 dB within the wanted frequency band. The simulated S_{II} agrees well with the measured one. The voltage gain, NF, linearity (IIP3, IIP2), and receiver start-up time were measured from the analog test output of the baseband channel. During the measurements, it was found that the LO tuning was sensitive to noise and glitches generated on the chip and PCB. Thus, to confirm the validity of the results, the receiver measurements were primarily performed with the setup where the LO resonator is at the lowest frequency of operation (LO buffer in reset mode). As a result, the maximum performance could not be obtained. The measured RF response is shown in Fig. 6.10. The maximum voltage gain of 47 dB is achieved at 2.0 GHz. The corresponding frequency response of the channel-select filter is shown in Fig. 6.11 where the measured and simulated maximum voltage gains are scaled to 0 dB for the purpose of comparison. The NF measured from the analog test output is 28 dB. In addition, the receiver gain and NF were measured at 2.0 GHz as a function of supply voltage and the result is shown in Fig. 6.12. The measurement shows that the receiver operates down to 1.0-V supply. The minimum supply voltage is limited by the mixer-baseband interface, where four transistors are stacked.



Fig. 6.8. Chip micrograph.



Fig. 6.9. Measured and simulated input matching.



Fig. 6.10. Measured and simulated RF responses.



Fig. 6.11. Measured and simulated frequency responses of the channel-select filter.

The receiver IIP3 was measured by using test signals at 3.0-MHz and 5.8-MHz offsets from a 2-GHz LO. The measured IIP3 is -21 dBm at the 100- Ω input impedance. The IIP2 was measured with test tones having 2.8-MHz and 3.0-MHz frequency offsets from the LO, and the results varied between +18 and +26 dBm. For the purpose of comparison, the measured voltage gain with the optimum LO buffer setup and the simulated result are also shown in Fig. 6.10. With the optimum LO buffer setup, the maximum voltage gain is achieved at 2.35 GHz. The measured voltage gain is 52 dB and the NF is 24.5 dB. The start-up time plays an important role in sensor systems when the target is to minimize power consumption. The channel select filter output has settled after 50 µs. The start-up time is limited by the charging of the large capacitance at servo loop of the mixer-baseband interface. The measured duty cycle is given in Table 6.1. The measured group delay with a 200-kHz input signal is 105 ns.



Fig. 6.12. Measured receiver voltage gain and NF vs. supply voltage.
The merged LNA and mixer with the LO buffers consume approximately 94 % of the receiver supply current. In stand-by mode, the measured current consumption is 18 μ A. According to simulations, this is mainly due to the leakage current of the bonding pad ring. Table 6.1 summarizes the measured performance of the demonstrator. The measured performance with the optimized LO signal is given in parenthesis.

Parameter	Unit	Result
Supply voltage	V	1.2
Supply current (active)	mA	2.75
Supply current (stand-by)	μΑ	18
Voltage gain, RF + BB	dB	47 (52)
Voltage gain, LNA + Mixer	dB	12.5 (14.5)
Maximum gain frequency	GHz	2.0 (2.35)
NF	dB	28 (24.5)
Out-of-channel IIP3	dBm	-21
Out-of-channel IIP2	dBm	+18
S ₁₁	dB	-10
Filter –3-dB corner frequencies	kHz	86, 493
Start-up time	μs	~50
Limiter group delay @ 200 kHz	ns	105
Duty cycle (Up/Period) @ 100 kHz		49.0
300 kHz	%	47.5
500 kHz		46.5

Table 6.1. Summarized performance of the receiver.

6.2 Direct-conversion receiver for ubiquitous communications

6.2.1 Introduction

A direct-conversion receiver (RX) for a 2.4-GHz sensor network is described in this section. The receiver includes an LNA, downconversion mixers, a 90-degree phase shift circuit, analog filters, a 1-bit analog-to-digital converter, and a received signal strength indicator (RSSI). Compared to the previous prototype [P1], the separate LNA and mixers are used and a 90-degree phase shift circuit is added. In addition, the RX includes two baseband channels with improved selectivity. The wake-up logic and RSSI are added and limiters are fully differential. Biasing is implemented by using an on-chip current reference. The receiver consumes 4.1 mA from a 1.2-V power supply and it achieves 43-dB voltage gain, 25-dB noise figure, -22-dBm IIP3, and +11-dBm IIP2.

The RX is part of a single-chip transceiver circuit, which was fabricated with a 0.13-µm CMOS process. The RX block diagram is presented in Fig. 6.13. The RX has a shared RF input/output port with the transmitter (TX). In addition, the RX and TX use the same VCO circuit for the local oscillator (LO) signal generation.



Fig. 6.13. Block diagram of the fabricated receiver (RX). Part of the transmitter (TX) and the RX interface with the TX are shown with a dashed line.

6.2.2 RF front-end design

6.2.2.1 Low-noise amplifier

The low-noise amplifier shown in Fig. 6.14a uses a common-gate (CG) topology with a cascode stage. The input transistors are biased into the subthreshold region to maximize the transconductance versus drain current ratio. The LNA uses an LC resonator tuned at 2.45-GHz center frequency as a load. The RX shares the same time-divided RF input-output port with the TX. The CG input allows a high impedance path towards RX in the transmit mode. Thus, a proper matching performance is achieved without using the switches in the signal path.

In the previously published prototype receiver, the front-end and the baseband interface consisted of four stacked transistors [P1]. As a result, the drain-to-source voltage V_{DS} of the LNA was susceptible to the supply voltage V_{DD} . The deviation from the nominal V_{DD} changed the LNA drain current, which altered the bias condition of the mixers and the baseband interface. With a separate LNA and mixer, the performance dependence of the supply voltage is reduced. In addition, the input transistors of a separate LNA have a higher V_{DS} , which improves the gain and noise performance. However, with a separate LNA and mixer, the whole receiver consumes more current compared to [P1]. In this design, the LNA consumes approximately 2.3 mA including biasing, which is half of the total receiver current.

6.2.2.2 Downconversion mixer

The mixer schematic is shown in Fig. 6.14b. The mixer does not have a typical common-source (CS) input stage due to the strict current budget. The simulated current consumption of the whole mixer is 410 μ A, which is 10 % of the receiver total current. With such a small current a

CS amplifier realized with deep submicron CMOS transistors would not improve the gain or noise performance. Hence, the LNA output signal is fed directly on the sources of the switch transistors to maximize the linearity. The switches are designed with minimum transistor length to minimize the load of the LO buffers [4].



Fig. 6.14. a) LNA, b) mixer.

6.2.2.3 Quadrature LO signal generation

The quadrature LO signal generation circuit is shown in Fig. 6.15. When the transceiver operates normally, the LO signal is generated on-chip (LO_{VCO}) and is fed into the first buffer. However, in the receiver measurements the VCO is shut down to be able to measure the receiver performance only. Thus, an external LO signal (LO_{EXT}) is required. It is fed to the output of the first LO buffer. The first buffer uses an LC resonator as a load for better LO swing, but the buffers driving the mixers do not include inductors to save the silicon area.

The 90-degree phase shifting is realized with a single-stage polyphase filter (SPF). While the amplitude balance of the SPF is poor and susceptible to process variations, it was chosen to minimize the LO signal loss and hence the power consumption. According to the simulations, the loss due to the SPF was 5 dB in this design. The target current consumption for the LO buffers was approximately 1 mA only. Thus, the compensation of the LO signal loss caused by two or more filter stages would increase the current of LO buffers to an unacceptable level. The first buffer consumes approximately 0.46 mA and one second-stage buffer 0.33 mA supply current.



Fig. 6.15. Schematic of the LO buffers and 90-degree phase shifter.

6.2.3 Baseband design

6.2.3.1 Mixer-baseband interface and channel-select filter

The realized channel-select filter is a fourth-order all-CMOS filter with a -3-dB corner frequency of 600 kHz. Figure 6.16 shows one half of the filter. The selectivity of the filter is improved from [P1] by adding a 2nd real pole. Furthermore, the value of the servo capacitor M_{FB} was halved to decrease the silicon area. Since the dc gain of the servo loop was also halved with a constant current source M_{I} , the -3-dB frequency of the resulting highpass filter is unchanged.



Fig. 6.16. Mixer-baseband interface and a channel-select filter (one half).

6.2.3.2 Limiting amplifier

The block diagram of the limiter is shown in Fig. 6.17. The limiter consists of a unity gain input stage, four cascaded amplifiers with a total small signal gain of 47 dB, a balun, and a flip-flop that samples the output signal. The limiter amplifier is a differential pair with PMOS input devices and cross-coupled NMOS devices in parallel with diode-connected NMOS transistors to increase the load impedance and thus the gain of the amplifier.



Fig. 6.17. Limiter architecture.

To reduce the limiter input-referred dc-offset, a dc feedback loop is required. To achieve a large time constant without large resistors and capacitors, the dc feedback loop is implemented by using digital circuitry. The dc feedback loop consists of a digital 9-bit up/down counter and a 9-bit current-steering DAC to filter out the dc-offsets [13].

6.2.3.3 Received signal strength indicator

The designed received signal strength indicator (RSSI) is shown in Fig. 6.18. The RSSI has separate full-wave rectifiers for I- and Q-branches, but the following two-stage filter and the latched comparator are common. The full-wave rectifier is designed to eliminate common-mode components at the input of the RSSI. For this purpose, a common-mode feedforward technique is utilized [14]. The ripple of the rectified signal is attenuated by using a two-stage low-pass filter formed by transistors $M_I - M_4$. The targeted input signal level is -60-dBm.



Fig. 6.18. Received signal strength indicator (RSSI).

6.2.4 Experimental results

The chip was fabricated with a 0.13-µm CMOS process. The active area of the RX is approximately 1.0 mm². The chips were directly bonded onto a printed circuit board (PCB) made of FR-4 substrate. Measurements were performed both from the analog test output of the baseband circuit and from the limiter output. The performance was characterized by using the

on-chip current reference. According to the measurements from several samples, the noise figure (NF) in the Q-channel was approximately 3 dB higher than in the I-channel. In addition, the IIP2 result of the Q-channel was degraded compared to the I-channel. The imbalance and the noise leakage are probably due to a systematic error in the layout. Thus, the NF and IIP2 performance of the receiver are determined according to the I-channel measurement results only. For the other measured parameters such significant performance difference between the I- and Q-branches was not observed.

The measured S_{II} is shown in Fig. 6.19. Since the circuit model from the latest TX output was not available, the resulting input matching of the RX was inadequate, as is indicated with the dashed line in Fig. 6.19. Thus, the S_{II} of the RX was improved by adding 2.2-nH series inductances on PCB. As a result, S_{II} better than -10 dB is achieved between 2.05–2.55 GHz. However, the input matching circuit and RX/TX interface suffers from the parasitic capacitances, which cause signal loss and most of the gain degradation. The parasitic capacitance also increases the NF, because the gain is lacking before the mixers, which have the largest effect on the total output noise.



Fig. 6.19. Measured S_{II} . The dashed line shows the original S_{II} and the solid line presents the S_{II} after adding the 2.2-nH series inductors on PCB.

The measured original RF response and the one with the corrected input matching are depicted in Fig. 6.20. The input matching correction improves the gain by 1–2 dB in the wanted frequency area. The maximum voltage gain of 43 dB is achieved at 2.45 GHz. The NF measured from the analog test output is 25 dB. The measured voltage gain and NF versus the supply voltage are presented in Fig. 6.21, which shows that the receiver operates down to onevolt supply voltage. The mixer and the interface with the baseband set the limit because of four stacked transistors.



Fig. 6.20. Measured RF responses before (dashed line) and after (solid line) the correction of the RF input.



Fig. 6.21. Measured voltage gain and NF versus supply voltage.

The measured IIP3 of the receiver is -22 dBm with the test tones at 3-MHz and 5.8-MHz offsets from the LO. Because the linearity is dominated by the mixer-baseband interface, the measured IIP3 is 3 dB higher than simulated due to the lower gain at RF. The IIP2 was measured with 5.8-MHz and 6-MHz test signals. The IIP2 result varied between +11...+21 dBm in the I-channel. The Q-channel IIP2 was approximately 0 dBm.

The simulated and measured frequency responses (three samples, six channels) of the channelselect filter are shown in Fig. 6.22, where the measured and simulated maximum voltage gains are scaled to 0 dB for the purpose of comparison. The measured -3dB-corner frequency variations are within the specified ± 10 %.



Fig. 6.22. Measured and simulated channel-select filter responses.

The measured wake-up times of the whole receiver with and without the wake-up logic were 15 μ s and 80 μ s, respectively. The measured RSSI threshold varied between -59...-55 dBm. The threshold is larger than the designed value due to the lower voltage gain of the front-end. The limiter was characterized by measuring the duty cycle with input signal frequencies from 100 kHz to 600 kHz, and input signal levels from -77 dBm to -27 dBm. In all cases, the duty-cycle variation was between ± 1 % from the nominal value of 50 %. The measured limiter group delay at 200 kHz input signal frequency was 82 ns.

The measured current consumption of the whole receiver was 4.1 mA. Although the RX was measured alone, the whole transceiver could not be completely shut down due to the TX on the same chip. Thus, the measured stand-by current consumption of 440 μ A is arbitrarily high compared to the simulated value of 20 μ A, which includes the measured current consumption of the on-chip bias generator (6.2 μ A). The measured receiver performance is summarized in Table 6.2. The chip micrograph is shown in Fig. 6.23.



Fig. 6.23. Chip micrograph.

Parameter	Unit	Result
Supply voltage	V	1.2
Supply current (active)	mA	4.1
Supply current (stand-by)	μΑ	440
Voltage gain	dB	43
NF	dB	25
Out-of-channel IIP3	dBm	-22
Out-of-channel IIP2	dBm	+11
ICP	dBm	-35
S11	dB	<-10
Filter high-pass –3-dB frequency	kHz	66
Filter low-pass –3-dB frequency	kHz	546
Wake-up time	μs	15
Duty cycle	%	49
Limiter group delay @ 200kHz	ns	82
RSSI threshold	dBm	-55

Table 6.2. The measured receiver performance.

6.3 Efficient current reuse for low-power transceivers

Current boosting is a method where the performance of an active circuit block is optimized by placing a constant current source in parallel with the active signal path to provide optimal biasing for different components. In this sub-section, a technique to replace the constant dc current source with active building blocks typically required in transceivers is proposed. By using this method the total current consumption of the transceiver can be efficiently reduced without modifying its performance. A design example where the proposed technique reduces the receiver current consumption by 45 % is given.

6.3.1 Introduction

In a sensor radio, the RF circuits are usually in the dominant role when it comes to the current consumption. For example, in the design examples presented in Sections 6.1 and 6.2, the current consumption of the RF parts is approximately 90 % of the total receiver current consumption. For that reason, power optimization should be concentrated on the blocks and elements operating at the RF. In [5] and [15], the downconversion mixer is stacked on top of the LNA such that the same dc current is re-used in both blocks, thus reducing the total power consumption. However, the stacking of individual blocks is difficult in modern deep submicron CMOS processes due to decreased supply voltages.

The current in the different circuit blocks or stages can be separately adjusted with current boosting, which enables the optimization of the overall noise, linearity, and gain performance [6]. For example, the optimum performance of Gilbert cell downconversion mixers is typically achieved by biasing the switch quad with smaller quiescent current than the transconductance

input stage. In a merged LNA and mixer topology shown in Fig. 6.2, the constant boost current I_{boost} is approximately 70 % of the dc current of the input stage [P1]. A possible way to exploit the PMOS boost source is to use it as an additional transconductance stage [16]. If that method was utilized for the design shown in Fig. 6.2, the boost current source would have been split into two separate devices. As a result, the capacitive loading at the source of the switch transistor increases, which degrades the signal-to-noise ratio [3]. In addition, the gain and noise performance improvement depends heavily on the used circuit configuration. To use the current source more efficiently, a technique where the constant current source is replaced with active blocks used elsewhere in the receiver is described. Compared to the other published current boost solutions, the active blocks used as boost current source are not related to the block having the current boosting. With the proposed solution, the current consumption is efficiently reduced while still maintaining the original performance.

6.3.2 Current boosting with active elements

The LNA merged with a downconversion mixer is shown in Fig. 6.24. Compared to the one presented in Fig. 6.2, the dc current boosting source is replaced by active circuits, which can be any other blocks typically used in a receiver, i.e. LO buffers or baseband circuitry. Because the dc current fed by the active circuits flows to the center tap of the inductor L_I , the possible interference or noise coming from the active boosting will be common-mode at the output nodes of the inductor and thus will not deteriorate the performance of the LNA or mixer. A stabilization capacitor to ground can be included in the center tap of the L_I to further minimize the interference. In addition, the RF signal at the output of the LNA will be zero at the L_I center tap, because it is a virtual ground. Therefore, the disturbance between active elements operating in both sides of L_I center tap is insignificant.



Fig. 6.24. Current boosting with active elements.

The drawback of the solution is that the effective supply voltage of the blocks used in active boosting is reduced by the drain-source voltage (V_{DS}) of input transistor M_I . In addition, the finite series resistances of the inductors L_I and L_s cause a slight voltage drop. Thus, only those blocks of which the performance does not significantly degrade due to a decreased supply voltage should be placed within the boost block. Furthermore, the total dc current consumption

of the boosting blocks should not exceed the value of the corresponding dc current source I_{boost} . An additional current source I_E shown in Fig. 6.24 can be added to maintain the original biasing of the LNA and switch transistors. Taking into account the process, temperature, and supply voltage (PTV) variations, the dc operation point of different blocks can be adjusted with I_E . Furthermore, it is possible to trade the dc current between the boosting blocks and additional current source I_E . The proposed method is not limited to be used in mixers and receivers only. It can be used in all circuits that use current boosting, while the boost current is injected into the node that is a signal ground.

6.3.3 Design example

The RF front-end presented in Section 6.1 was the basis for this design example. The simulated current consumptions of the merged LNA and mixer and LO buffers were approximately 1.4 mA and 1.1 mA, respectively. The quadrature downconversion mixers are resistively loaded in this design example. The resistor values were chosen such that the dc level at the output node is compatible with the mixer/baseband interface [P1]. Two LO buffers driving quadrature mixers are used as active boosting elements, as is shown in Fig. 6.25.

The design of the LNA is not changed due to the new boosting principle. However, the dc operation point deviation from the nominal state due to process variations requires more attention. If the V_{DS} of the input transistors deviates significantly from the nominal state due to process variations, both the biasing of the input transistors and the operation of the switching stage and boosting blocks may be altered. The quiescent current of the mixer should be accurately controlled, since it also affects the properties of the baseband interface. Because the center tap of the inductor L_{I} is a signal ground, a switch transistor can be biased with a current mirror using the center tap of L_1 as a reference node as is shown in Fig. 6.25. As a result, the gate-source voltages of the diode-connected current mirror $(M_{bias,SWI})$ and switch transistor (M_{SWI}) are equal (except the minor voltage drop due to finite series resistance on L_1). The transistors $M_{bias,SWI}$ and M_{SWI} have equal aspect ratios. The slight deviation from 1-to-1 current mirroring is due to non-equal drain voltages. According to the simulations, the drain current variation of a switch transistor in different process corners is less than 7 % from the nominal value and thus the mixer performance is not significantly altered. Thus, the biasing method alleviates the problem related to adequate biasing of the merged LNA and mixer structure described in Section 6.2.2.1.

The LO buffers driving the quadrature mixers are similar to the ones used in [P1] except the tail current sources are omitted. Because both the LNA and mixer switch transistors are biased by current mirrors, the dc current through one LO buffer (I_{LO}) is achieved by subtraction:

$$I_{LO} \approx \frac{1}{2} \left(2I_{LNA} - 9I_{SWI} \right).$$
 (6.4)

 I_{LNA} and I_{SWI} are the dc currents of a single LNA and a switch transistor, respectively, as shown in Fig. 6.25. Because the LO buffers are identical, the dc current is evenly divided between them. If an additional block was placed in parallel with the LO buffers, a serial dc current source would be required to control its current independently. The LO buffer utilized in this design is a self-biasing structure and therefore a separate bias circuit is not needed. In addition, with the proposed biasing method the operational points of LO buffer devices are nearly unchanged and the performance is not degraded compared to that of the original design.



Fig. 6.25. Simplified biasing scheme for the LNA, the switch quad, and the LO buffers.

6.3.4 Simulation results

The total dc current of the LNA was kept at 1.4 mA from a 1.2-V supply. Two LO buffers consume approximately 1.1 mA, which corresponds to nearly 75 % of LNA current. Taking into account the biasing currents, the total current consumption of the design with the active current boosting method is 1.5 mA. If "traditional" dc current boosting was used having LO buffers at a separate dc current path, the total current consumption would be 2.7 mA. Therefore, 45 % current saving is achieved in this design example. The biasing of all blocks and component dimensions remained nearly unchanged compared to the original design [P1]. As a result, according to simulations, a significant difference in conversion gain, input matching, or noise performance is not observed. In addition, no remarkable difference was observed in linearity (IIP3, IIP2) performance, either. While the measurement results are not provided, the simulation results of this and [P1] are comparable, since device sizing, biasing, and modeling are nearly unchanged.

Two LO buffers are operating in a 90-degree phase shift, and a double LO frequency signal is observed at the center tap of L_1 when all the components in LNA, LO buffers, and mixers are fully matched. The signal is at such a high frequency and has such a small amplitude (<1mV_p) that its effect on the circuit operation is insignificant. However, in the case of finite device matching in switch quad or LO buffers, a small current signal leaks to the center tap of the inductor at the fundamental LO frequency and causes a several mV signal at the center tap of L_1 . That can be attenuated to a negligible level by stabilizing the center tap with a capacitor to ground, as mentioned earlier. The common-mode rejection ratio (CMRR) of the LO signal path is checked because the tail current sources of LO buffers are now omitted. The simulated CMRR in the LO signal path is reduced by only 3 dB, this being negligible degradation compared to the original design.

6.4 A direct-conversion RF front-end in a 65-nm CMOS

This sub-section demonstrates 65-nm CMOS RF circuit design for wireless applications operating in the 2-GHz frequency area. It should be noted that the presented design example is not directly related to the subject of the other two main applications (short-range sensors or UWB). However, it contains slightly different, but nevertheless relevant, circuit design details compared to the other design examples. It therefore completes the material presented in this chapter.

The simplified block diagram of the direct-conversion RF front-end is shown in Fig. 6.26. The RF front-end consists of a low-noise amplifier (LNA), folded quadrature mixers, a local oscillator (LO) divider, and LO buffers. The front-end consumes 29.3 mA from a 1.2-V power supply and, according to simulations, it achieves 39-dB voltage gain, 1.5-dB minimum spot noise figure, and -17-dBm IIP3. Because of the low-resistivity CMOS substrate, all the signal paths are balanced to achieve immunity to the substrate noise and interferers.



Fig. 6.26. Block diagram of the RF front-end.

6.4.1 RF front-end design

6.4.1.1 Low-noise amplifier

The LNA shown in Fig. 6.27 is a balanced inductively degenerated common-source amplifier with a cascode stage. The LNA consumes 5.4 mA without biasing. The minimum channel length was chosen for the LNA input transistors to optimize the f_T of the transistor and to test the process gain and noise performance. The input matching of the LNA is realized with an onchip source inductor L_S and shunt capacitor C_S between the gate and source of the input transistor, and an off-chip capacitor C_{in} and inductor L_{in} . The input matching better than -10 dB is achieved in the band of 2.1 GHz – 2.6 GHz. The load of the LNA is a resonator consisting of a 1.9-nH differential inductor and 2-pF capacitance. Two separate 1-pF capacitors units were used in parallel to achieve symmetrical parasitics at output nodes. In addition, the resonator Q-value is decreased with a shunt resistor to widen the bandwidth of the resonator.



Fig. 6.27. Simplified schematic of the LNA.

The voltage gain and noise figure (NF) of the LNA are shown in Fig. 6.28. According to simulations, a 1.2-dB noise figure is achieved with 22-dB voltage gain at 2.45-GHz frequency. The simulated IIP3 and ICP of the LNA are approximately 0 dBm and -5 dBm, respectively. According to simulations, no severe noise, gain, or linearity penalty was incurred due to minimum channel length devices when using this topology.



Fig. 6.28. Simulated LNA voltage gain and noise figure.

6.4.1.2 Quadrature mixers

The folded structure shown in Fig. 6.29 was chosen to avoid the stacking of different mixer blocks. As a result, sufficient voltage headroom was achieved for input stage, switch stage, and load. In addition, to optimize the gain, linearity, and noise performance, the dc current of both the mixer input and switch stages can be adjusted separately.

The input stage of the mixer consists of a common-source amplifier. Both the amplifier and cascode stages have 140-nm channel length to achieve better linearity performance compared to minimum length devices. The input stage consumes 13 mA without biasing. A pretty large dc current is needed to achieve a sufficient linearity and voltage gain before the switching stage.

The high impedance needed for folding could be obtained with current sources, but due to limited supply voltage device stacking should be avoided, thus leading to the use of an LC

resonator. The resonator consists of an on-chip 1.9-nH inductor, 1.2-pF shunt capacitance between the outputs (two 600-fF capacitor units in parallel for symmetrical parasitics), and two 1.2-pF capacitors from the output to the V_{DD} . With the latter capacitors, the effect of the parasitic capacitance of the following switching stage is not dominating. The inductor tunes the resonator to the wanted frequency and it also minimizes the noise contributed at the source node of the switch transistors [3].



Fig. 6.29. Simplified schematic of the quadrature mixer.

The scaled RF response at the mixer resonator output is shown in Fig. 6.30. The resonator has high impedance at the wanted frequency, but the impedance is lower at other frequencies. As a result, the standing wave at double LO frequency produced by the mixing pair is attenuated. Thus, the voltage swing across the drain-source terminals of both mixer input stage and switch transistors is lowered which results in improved IIP3 [17]. In addition, also IIP2 of the mixer is improved, because the resonator has low impedance at the frequency of $|f_{RF1} - f_{RF2}|$, thus attenuating the second-order intermodulation generated by the input stage of the mixer [18].

The PMOS transistors are used as switches to minimize the flicker noise. The 1/f noise is further optimized by using as small a quiescent current as possible. However, too small a dc current can degrade the linearity performance. Eventually, for a single switch transistor, the nominal dc current of approximately 320 μ A was chosen. In addition, the channel length of the switch transistors was optimized as 220-nm to improve the 1/f noise performance but to not deteriorate the operation of the LO buffers. Furthermore, device matching is better with non-minimum channel length transistors.



Fig. 6.30. Scaled voltage gain at the mixer resonator output.

This mixer was designed to drive voltage mode output. Therefore, RC-impedances are used as a load to improve the mixer out-of-band blocking characteristics and to relax the linearity requirements for the following blocks. The floating capacitor is used to minimize the value and the area of the load capacitor. The RC pole frequency is approximately 10 MHz. It was chosen pretty arbitrarily, because this prototype chip is not intended for any particular application. For better isolation and larger voltage gain, the RC-loads were separated from the switch transistors with PMOS cascode transistors M_C . The cascode devices have negligible effect on overall noise performance. In transient simulations, it was found that the double-frequency LO signal at the drain node of the switch transistors modulates the V_{DS} voltage of the cascode transistor. For that reason, an additional capacitor was added before the cascode stage. As a result, the double-frequency LO signal over the drain-source terminals of cascode stage is attenuated by 10 dB.

Finally, part of the quiescent current of the switch transistors is removed before the cascode stage with NMOS current sources. As a result, the load resistor value can be chosen more freely to have a sufficiently large voltage gain but not to cause voltage drop. In addition, according to simulations, also the IIP3 and NF results are improved. The length of the current source transistors was set to 4 μ m so that the effect of the 1/*f* noise is insignificant. In this design, a 1.2-k Ω load resistor value was chosen to achieve adequate gain, noise, and linearity.

6.4.1.3 Quadrature LO signal generation

The quadrature LO signal was generated from an external LO source by using the frequency divider presented in Fig. 6.31 [19]. With a double-frequency LO signal, several problems related to direct-conversion, such as LO leakage to the RF input, can be minimized. The divider topology was chosen in this design because only two transistors are stacked and a large 1.8-V_{pp} differential LO-signal swing is achieved. In addition, it does not require resistors, thus being more tolerant to process variations. The minimum length was chosen to maximize the speed of transistors. The nominal current consumption of the divider is 3.2 mA. In typical use (f_{in} <5GHz), the divider requires a rather small (<50mV_{pp}) input signal for injection locking. By

increasing the current consumption by 65 % and LO input signal amplitude to 170 mV_{pp} , the divider can operate with input frequencies as high as 9 GHz.



Fig. 6.31. Schematic of the LO divider.

The divider is sensitive to capacitive load. Usually, it cannot directly drive the gate capacitance of the switch transistor and therefore LO signals need to be buffered before mixers. In this design, the balanced LO buffers shown in Fig. 6.32 were used [P1]. The transistors M_n and M_p have minimum channel length to minimize the parasitics. Due to relatively high V_T , careful optimization is required not to force the current source M_5 into linear region. The capacitor C_2 is required to decrease the effect of the noise of the LO buffer. Otherwise, the buffer upconverts the low-frequency noise coming from the bias circuit to the LO frequency. The current consumption of one buffer is nominally 0.5 mA without biasing.



Fig. 6.32. Schematic of the LO buffer.

6.4.1.4 Simulation results

The voltage gain, IIP3, and ICP of the whole front-end simulated from the mixer output are approximately 39 dB, -17 dBm, and -29 dBm, respectively. The spot noise figure as a function of intermediate frequency (IF) is shown in Fig. 6.33 with a solid line. The minimum noise figure is 1.5 dB. The 1/f noise corner frequency is approximately 400 kHz, and thus it can increase the total NF in narrowband applications. For a comparison, an integrated NF starting from 100 Hz is shown with a dashed line in Fig. 6.33. As can be seen, a sub-3dB total NF, for example, is achieved with applications having an IF bandwidth of approximately 4.7 MHz. The summary of the front-end performance is given in Table 6.3.



Fig. 6.33. NF as a function of IF frequency. The spot NF is presented with a solid line and the integrated NF starting from 100 Hz is shown with a dashed line.

Parameter	Unit	Result
Supply voltage	V	1.2
Supply current	mA	29.3
Voltage gain	dB	39
NF (minimum)	dB	1.5
NF (100 Hz 2.5 MHz)	dB	3.9
NF (100 Hz 4.7 MHz)	dB	3.0
1/f noise corner frequency	kHz	400
Out-of-channel IIP3	dBm	-17
ICP	dBm	-29
S ₁₁	dB	<-10

Table 6.3. The simulated front-end performance.

6.5 Low-noise amplifiers for UWB

In this sub-section design examples of low-noise amplifiers for UWB are given. The first design example is based on inductively degenerated common-source (IDCS) LNA. The LNA operates in the 3.1 - 4.7 GHz band. The purpose of this design example is to compare the basic IDCS topology to the two alternatives presented in Section 3.3.2 and 3.3.3. The second design example covers the design of a common-gate LNA. The center frequency is set to 5 GHz and the target is to cover the frequency band of 3 - 8 GHz. The CG LNA theory presented in Section 3.3.7 was utilized to design the input matching resonators.

6.5.1 Inductively degenerated CMOS low-noise amplifiers for BG1

Fig. 6.34 shows three balanced IDCS LNAs designed for BG1: i) without any feedback, ii) with RC feedback, and iii) with an additional CG signal path. The balanced structure was chosen to achieve immunity to the substrate noise and interferers. The simulations were performed with a 0.13-µm CMOS process. The input and cascode transistor size, biasing, and the value of inductors were kept unchanged to achieve comparable results. The LNA core consumes approximately 7 mA from a 1.2-V supply. In case iii) the CG transistor has approximately 3-mS g_m . The feedback resistor in case ii) was chosen as 700 Ω to obtain a comparable NF with case iii). The bonding pads were also included in the simulations to achieve realistic parasitic capacitance at the input node.



Fig. 6.34. IDCS LNA i) without any feedbacks, ii) with RC-feedback, iii) with additional CG signal path.

The S_{II} , insertion gain, and NF were simulated and the results are shown in Fig. 6.35 and Fig. 6.36. It can be seen from the results that the basic IDCS LNA without any feedback has the narrowest BW_{SII} . In addition, it is the most sensitive to the parasitic capacitance of the input pad. The LNA with additional CG path or RC feedback have approximately 30 % wider BW_{SII} compared to the basic IDCS LNA. In addition, both topologies cover the whole BG1. However, the RC feedback reduces the reverse isolation by 5-6 dB compared to the basic IDCS topology with these component values. The LNA with CG path does not suffer from reverse isolation degradation.



Fig. 6.35. S_{II} with different input matching setups.

The effect of the input matching circuit to the gain BW can be observed from Fig. 6.36. The S_{II} mismatch for the basic topology causes a slight gain peaking compared to other two topologies. That also affects the NF, but the NF difference of the basic IDCS LNA compared to two variants is small. The LNA with RC-feedback has the narrowest BW. To achieve wider operation, the feedback resistor should be larger, but that would cause narrower BW_{SII} . The LNA with an additional CG path has the best gain flatness over BG1 frequencies. In addition, its maximum insertion gain frequency is closer to the input matching frequency when compared to the other two LNAs. That was also detected in Section 3.3.3, where IDCS LNA variants were simulated with ideal components.

In this design example, the LNA with dual CG path gives the most promising performance, when S_{II} , gain, and NF should cover the BW between 3.1 - 4.7 GHz. Finally, with component values used in this example all cases had similar IIP3 performance.



Fig. 6.36. Insertion gain and NF with different input matching setups.

6.5.2 Common-gate LNA for BG1 and BG3

The design of a CG LNA, which covers the UWB band groups #1 and #3, i.e. 3 - 8 GHz, is presented. For simplicity, a single-ended structure was chosen, but the analysis presented in Section 3.3.7 can be utilized for a balanced CG LNA, too. The schematic of the LNA is shown in Fig. 6.37. The bonding pad was included in the simulations to achieve realistic parasitic capacitance at the input node. The transistor M_1 was biased with a 1.2-mA current from a 1.2-V supply voltage. The source impedance Z_0 was 50 Ω . The simulations were performed with a 0.13-µm CMOS process.



Fig. 6.37. Simplified schematic of CG LNA.

The desired band is 3-8 GHz and the center frequency is approximately 5 GHz. Therefore, the relative bandwidth is 1. First, the design criteria for source and input resonators are estimated from Fig. 3.27 and Fig. 3.28. To achieve better than $-10 \text{ dB } S_{II}$, which covers the whole band with some design margin, the required minimum relative impedance Z_{rel} for the source resonator is 1.5. The input series resonator should be dimensioned with a *p* value higher than 2.

According to (3.61), the source resonator tuned at 5 GHz frequency should have at least a 2.4-nH inductor for the desired Z_{rel} . The source resonator was realized without additional shunt capacitance and the resonance at 5 GHz frequency was achieved with a 3.0-nH inductor. The input matching without the series input resonator is shown in Fig. 6.38 with a solid line. The source resonator gives S_{11} better than -10 dB from 2.7 GHz to 8.8 GHz. The relative bandwidth is 1.22, which is close to the theoretical value 1.26 given by (3.61). The Z_{rel} of the source resonator is evaluated with (3.68): $Z_{rel} = 1.83$.

According to (3.74), the optimal value for p would be 6.7 to maximize the overall $BW_{rel,SII}$. Because the Z_{rel} was slightly overdesigned, a smaller p value can be chosen and still achieve the required S_{II} . The S_{II} of the whole LNA was simulated with the p values of 2, 3, and 4. The results are shown in Fig. 6.38 with dashed lines. The BW_{rel} increases along with p according to the theory. The p value between 2.5 to 3 gives the required S_{II} bandwidth. With larger p values, the BW_{SII} is improved, but the value of the input inductor L_{in} gets smaller. Therefore, it is possible that optimal L_{in} is not reached due to limitations set by the bond wire inductor to package or PCB.



Fig. 6.38. Simulated S_{11} . The effect of the source resonator is shown with a solid line. The dashed line shows the whole S_{11} with several p values.

The insertion gain and NF are shown in Fig. 6.39 with p values of 2, 3, and 4. The gain and NF are approximately 12 dB and 4 dB over the whole wanted band. Both the gain and NF bandwidths increase along with p. However, the highest operational frequency is limited by the shunt-peak load and the parasitic capacitance at the output node. Therefore, scaling the p does not improve the BW at high frequencies. The gain is decreased at 8 GHz frequency by approximately 1 dB from the maximum value. To increase the operational BW at high frequencies, the value of the load resistor should be lower. As a result, the gain would degrade as well.



Fig. 6.39. The solid line presents the simulated insertion gain and the dashed line shows the NF with p values of 2, 3, and 4. Both the gain and NF BW increase along with p.

The Q-value of the input circuit and the IIP3 of the LNA is presented in Fig. 6.40. Both parameters are given with p values of 2, 3, and 4, and the results are shown with solid, dashed, and dashed-dotted lines, respectively. As can be seen, with p value of 2, the additional gain due to input matching resonators is low, which is predicted by (3.66). The Q_{in} increases along with larger p values. This also decreases the IIP3 by about the same amount.



Fig. 6.40. Simulated Q-value of the input resonators and IIP3 with *p* values of 2 (solid line), 3 (dashed line), and 4 (dashed-dotted line).

In this design example, the implementation of a sufficient input matching network was limited by the minimum length of the input bond wire. The upper limit for the gain was set by the parasitic capacitance of the output node. According to the design example, it is possible to achieve a unity relative BW at 5 GHz center frequency with a simple CG LNA. The simulation results are in good agreement with the theory.

6.5.3 Conclusions

According to the two presented LNA design examples, several points can be noted. The design of a wide input matching circuit is easier with a CG topology, because wider bandwidth can be achieved and the parasitic capacitance associated with the input node can be embedded in the input matching network. The current consumption of a single branch of the balanced IDCS LNA was roughly triple compared to the single-ended CG LNA. The major disadvantage of the CG LNA is the higher noise figure compared to IDCS LNA. In addition, the gain of the CG LNA is several dBs lower, which can increase the noise contribution of the following stages. Therefore, if the target is to cover just a single BG, the IDCS LNA usually offers better NF and gain performance.

6.6 A dual-band direct-conversion RF front-end for WiMedia UWB receiver

6.6.1 Introduction

This section describes a direct-conversion RF front-end designed for dual-band WiMedia UWB receiver. The front-end operates in band groups BG1 and BG3, i.e. 3.1 - 4.8 GHz and 6.3 - 7.9 GHz, respectively. The UWB receiver is targeted for a mobile handset, where several other radios can be simultaneously on. Therefore, special attention was paid to minimizing the interference from different wireless systems. The front-end achieves approximately 26-dB gain and 4.9 - 5.6-dB noise figure across three sub-bands of BG1. In BG3 mode, it obtains 23 - 26-dB gain and 6.9 to 7.7-dB NF. The front-end consumes 48.1 mA and 42.7 mA from a 1.2-V supply voltage in BG1 and BG3 operation modes, respectively.

6.6.2 Front-end design

The block diagram of the direct-conversion front-end is presented in Fig. 6.41. It includes lownoise amplifiers (LNA), quadrature downconversion mixers, a passive polyphase filter for quadrature local oscillator (LO) signal generation, and LO buffers. All the signal paths are balanced to achieve immunity to substrate noise and interferers. The front-end is targeted to cooperate with the analog baseband filter similar to the one presented in [20].

The use of a wideband LNA covering the whole UWB band is questionable in a multi-system environment, since a strong unwanted signal at any frequency in the reception band can desensitize the whole front-end. In this design, the first LNA stages are separate for BG1 and BG3 operation modes. The significant benefit achieved at the cost of individual pre-selection filters utilized for both inputs is the improved suppression of strong interferers. The other radios operating in the 5-GHz frequency band and at lower frequencies (common RF systems around and below 2 GHz) are more efficiently attenuated than with a single wideband pre-filter covering the whole UWB reception band. In addition, since the systems operating around 2.4 GHz are relatively close to the lowest frequency of BG1, the LNA1 includes a high-Q 2.4-GHz notch filter to further improve the attenuation.

With modern deep-submicron CMOS processes it is challenging to design wideband amplifiers with several GHz bands. The intrinsic gain from a single transistor is moderate and typical resonator loads used in narrowband applications must be replaced with wideband structures. As a result, the gain achieved from a single-stage amplifier may not be sufficient. Therefore, multi-stage LNAs were implemented to have adequate bandwidth and gain before the mixers.



Fig. 6.41. Block diagram of the UWB front-end.

6.6.2.1 First LNAs (LNA1)

The schematic of the BG1-LNA1 is shown in Fig. 6.42. The LNA core is based on a typical inductively degenerated common-source (IDCS) structure. It was chosen because it offers satisfactory noise, linearity, and gain performance. However, it is challenging to achieve proper input matching for a UWB system with this topology, even though the matching would be only needed for one BG at time, as in this design. As was shown in (3.33), the achievable input matching bandwidth (BW_{S11}) is proportional to the source impedance Z_0 and inversely proportional to the sum of input (L_{in}) and source inductor (L_S) values. The input is matched to 100 Ω differentially, and therefore Z_0 cannot be altered. When the effect of process deviation and practical component values are taken into account, an S_{11} sufficiently wide for the BG1-LNA with IDCS topology could not be obtained. Therefore, an additional common-gate (CG) transistors (M_{CG} s) connected to the LNA input were added. As was shown in Section 3.3.3 and in the design example presented in Section 6.5.1, the transconductance of M_{CG} s improve BW_{S11} , but the cost is slightly increased noise. In addition, due to the low-impedance signal path caused by CG transistors, the resulting input matching circuit is less sensitive to the parasitic capacitance caused by the pad with ESD protection diodes than a basic IDCS input stage.



Fig. 6.42. Schematic of the BG1-LNA1.

The BG1-LNA1 includes an additional LC notch filter tuned at 2.45 GHz to further suppress interferers at this frequency. For example, due to the 3rd-order nonlinearity, the interferer at 2.4 GHz together with 900-MHz GSM can corrupt UWB signal reception. The notch includes a cross-connected NMOS pair to improve the Q-factor of the resonator [21]. The notch is connected at the IDCS stage, which is the main source of the nonlinearity. Thus, although the CG stage bypasses the notch filter, the interferers leaking through this path do not degrade the performance.

The BG3-LNA1 is also based on the basic IDCS topology. The additional CG signal path is not utilized in BG3 because the parasitic capacitance at the input would become too large, thus deteriorating the input matching. The sum input bond wire and source inductor values are less than 3 nH per input. As a result, according to (3.33), the input matching bandwidth is approximately 2 GHz. Thus, BG3 input can be covered without any feedback or additional signal paths. However, the BG3 input is rather sensitive to the parasitic capacitance at the input node. To minimize the capacitance caused by the bonding pad, the modified pad structure presented in Section 3.4 was utilized.

Both BG1-LNA1 and BG3-LNA1 utilize a shunt-peak load. Different load structures were presented in Section 3.5.2. While various wideband load structures exist, a shunt-peak load is typically used in single-stage balanced amplifier since the number of on-chip inductors should be kept as small as possible.

6.6.2.2 Buffer

The LNA1 cannot provide enough gain before the mixers, and therefore a second amplifier (LNA2) is needed. The design of the LNA2 is a trade-off between having large enough input devices to achieve the required gain but not to cause capacitive loading for the LNA1. To maintain the freedom to optimize the performance of both LNA stages, a buffer shown in Fig. 6.44a is used between them.

The buffer has approximately unity voltage gain with a wide frequency band without inductors. The output impedance of the buffer is low, and therefore it can sustain large output capacitance. The voltage gain of LNA1 is moderate, and thus the linearity of the buffer is not the limiting factor on the whole signal chain. In addition, its contribution to the total front-end noise is less than 0.3 dB. The simulated buffer response is shown in Fig. 6.43. The response of the buffer is shown, with and without feedback capacitor C_{fb} , with solid and dashed lines, respectively. The gain is scaled to 0 dB at zero frequency for the purpose of comparison. Without the C_{fb} , the buffer is a traditional source follower and M_2 acts as a current source only. Then, the –3-dB band hardly exceeds 2.5 GHz frequency. The operational bandwidth is extended by several GHz by adding a feedback loop from the drain of M_1 to the gate of M_2 .



Fig. 6.43. Simulated frequency response of the buffer with and without capacitor C_{fb} .

6.6.2.3 Second LNA (LNA2)

The purpose of the LNA2 is to combine the signal paths and to provide the remaining gain before the mixers. The simplified schematic of a resistor-feedback LNA used as LNA2 is shown in Fig. 6.44b. The analysis of the basic operation principle of this type of amplifier is presented in, for example, [22] and [23]. This amplifier topology was chosen because it achieves wide operational BW and parasitic capacitance at the node *LNA2_{out}* causes only slight gain degradation. Therefore, additional buffers before mixers are not needed.



Fig. 6.44. a) Schematic of the buffer between LNA1 and LNA2 (single-side view only), b) schematic of the second LNA (single-side view only).

Compared to those presented in the literature, this design includes a shunt-peak load to widen the operational bandwidth of basic resistor-feedback LNA. In addition, the capacitor C_C was added to prevent the dc current flowing from input nodes to the feedback loop. Otherwise, the bias current of the input transistor would be sensitive to the process variations. The signal path is combined at the drains of the input devices and the feedback path from gate of the input device to the source of the output device has an additional switch to isolate the non-operational input from the operational one.

6.6.2.4 Downconversion mixers

The schematic of the downconversion mixers is shown in Fig. 6.45. The quadrature mixer is based on a basic Gilbert cell, where the output signal of the LNA2 is directly fed to the switching stage. Therefore, the quadrature mixer is a low-impedance load for the LNA2. With this kind of LNA/mixer interface, a wider operational BW is achieved compared to a mixer with a typical CS input stage.

The quiescent current of the switch transistors is set by current sources. The dc levels of the IF output nodes are set to 0.7 V such that the compatibility with targeted baseband filter [20] is achieved.



Fig. 6.45. Schematic of downconversion mixers.

6.6.2.5 LO signal generation

The local oscillator (LO) generation chain is designed such that the operational band uniformly covers the BG1 and BG3 frequencies, i.e. 3 - 8 GHz. The quadrature signal was generated from an external LO signal source with a three-stage passive polyphase filter (PPF). According to the simulations, the PPF causes approximately 8-dB loss for the LO signal. The losses were compensated with the LO buffer shown in Fig. 6.46. It consists of a two-stage amplifier, where the first stage is a CS amplifier with a cascode stage. Current source I_B was utilized to boost part of the cascode current. As a result, the dc drop at the output of the first stage is decreased and sufficient performance is maintained in process corners. If the switch transistors of the mixer were directly connected to the output of the first LO buffer stage, the capacitive loading would deteriorate the BW. Therefore, to be able to drive the mixer gate capacitances, buffer topology similar to that was used between LNA1 and LNA2 is utilized as the second stage of the LO buffer.

An LO signal with a constant level over a wide frequency band is required to achieve a flat conversion gain response from the mixers. However, only one differential inductor is wanted per buffer for a compact layout. Therefore, the differential inductor of the modified shunt-peak load presented in Section 3.5.3.2 is re-used in both stages. According to the simulations, the presented method increases the -1-dB band by 2 GHz at the band of interest when compared to a buffer where a single load inductor was used only for one or other of the stages. In addition,

the achieved bandwidth is wider even compared to a case when separate load inductors for both stages were used. Therefore, bandwidth extension is achieved with a single load inductor only.



Fig. 6.46. Schematic of the LO buffer with bandwidth extension load.

6.6.3 Experimental results

The chip was fabricated in a 0.13-µm CMOS process and the chips were directly bonded onto a printed circuit board (PCB). The gain, noise, and linearity performances of the front-end were verified from IF outputs with an external high-impedance buffer.

The gain and input matching of the receiver in BG1 are shown in Fig. 6.47 with solid and dashed lines, respectively. The gain is approximately 26 dB and the gain deviation is less than 2 dB over the 3.1 - 4.8 GHz band. The gain starts to degrade after 5.4 GHz. The S_{II} at the BG1 band is better than -9 dB. The notches seen in S_{II} response are due to the PCB. The center frequency (f_C) of the LC notch filter is located at 2.3 GHz. Due to the slight deviation from the targeted f_C of 2.45 GHz, some of the linearity tests are performed at 2.3 GHz as shown later. The 1/f noise corner frequency is approximately 2 MHz. Because the BW of the external buffer utilized in measurements is limited, the output noise is integrated from 2 MHz to 30 MHz. That will give a result comparable with the other reported designs, because the flicker noise averages along with increasing IF frequency. The measured noise figures of different BG1 sub-bands are 4.9 - 5.6 dB. When the additional CG path of the BG1-LNA1 was shut off, the gain increased approximately by 1 dB and the NF lowered to 3.6 - 4.7 dB. The S_{II} , however, degraded to -6 dB in the worst case, which shows the effectiveness of the CG transistors.



Fig. 6.47. Conversion gain and S_{II} of the BG1.

The gain and S_{II} in BG3 mode are shown in Fig. 6.48 with solid and dashed lines, respectively. The maximum gain is approximately 26 dB and the gain deviation over 6.3 – 7.9 GHz band is approximately 4 dB. The measured NF of BG3 sub-bands are 6.9 – 7.7 dB. Due to the high operational frequency, the optimization of the PCB input was challenging. Despite the PCB modeling, the notches of S_{II} are nearly at a 1 GHz lower frequency than targeted. Therefore, even more effort should be applied to PCB design and modeling to achieve adequate performance at the wanted frequency band.



Fig. 6.48. Conversion gain and S_{11} of the BG3.

The front-end IIP3 performance was measured by using typical two-tone test signals at 20-MHz and 39.8-MHz offsets from the LO frequency. The in-band IIP3 result is approximately –16 dBm and –20 dBm in BG1 and BG3, respectively. The IIP2 was measured with test tones having 19.8-MHz and 20-MHz offsets from the LO frequency. The IIP2 result was approximately +20 dBm among BG1 and BG3 sub-bands.

The intermodulation caused by the GSM900 transmitter (f_{RF1}) and 2.4-GHz interferer (f_{RF2}) was measured by applying test signals at 880 MHz and 2.25 – 2.33 GHz into BG1 input, such that the latter signal is 15 dB stronger than the former. The most harmful interferers locating at UWB BG1 band were downconverted from the frequencies of 2 x $f_{RF1} + f_{RF2}$ and 2 x $f_{RF2} - f_{RF1}$. The corresponding out-of-band IIP3 result is +1 dBm for both cases. The measured result at the latter frequency is presented in Fig. 6.49.



Fig. 6.49. Intermodulation caused by GSM900 transmitter and 2.4-GHz interferer.

The in-band input compression point (ICP) is better than -27 dBm and -29 dBm in BG1 and BG3, respectively. The desensitization of the front-end was characterized by supplying a strong interferer at frequencies below 3 GHz such that the wanted weak signal is at BG1. The level of the interferer, which causes the gain of the wanted signal at BG1 band to be decreased by 1 dB, is depicted in Fig. 6.50. For example, at 2.25 - 2.33 GHz frequencies, which cover the 80-MHz BW allocated for 2.4-GHz systems, the -1-dB desensitization is achieved when the level of the interferer at BG1 input was approximately -20 dBm. Compared to in-band results, the filter improves the performance at the notch by 6 - 7 dB.



Fig. 6.50. Desensitization of BG1.

The measured current consumptions of the front-end were 48.1 mA and 42.7 mA for BG1 and BG3 operation modes, respectively. The measured gain and linearity results match well with the simulations. However, the measured NF deviates moderately from the simulated one. The measured front-end performance is summarized in Table 6.4. The chip micrograph is shown in Fig. 6.51. The active chip area is 0.73 mm².

Parameter	Unit	BG1 (GHz)			BG3 (GHz)			
		3.432	3.960	4.488	6.600	7.128	7.656	
Supply voltage	V		1.2					
Current cons.	mA		48.1		42.7			
Gain	dB	26.1	26.0	25.0	23.4	25.8	24.6	
Noise figure ¹⁾	dB	4.9	5.0	5.6	7.3	6.9	7.7	
IIP3 ²⁾	dBm	-15.8	-15.8	-14.8	-17.4	-20.1	-18.8	
IIP2 ³⁾	dBm	+21	+19	+18	+19	+23	+19	
ICP	dBm	-26.3	-26.5	-25.8	-25.8	-28.8	-27.5	
Active area	mm ²	0.73						

Table 6.4. The measured front-end performance.

1) Integrated from 2 MHz to 30 MHz

2) Measured with 20-MHz and 39.8-MHz offsets from the LO

3) Measured with 19.8-MHz and 20-MHz offsets from the LO



Fig. 6.51. Chip micrograph.

6.7 Summary and comparison to other published designs

In Sections 6.1 and 6.2, two direct-conversion receivers for wireless sensor applications operating in the 2.4-GHz ISM band were presented. Both design examples use a modified Bluetooth system that has optimized radio parameters for low power applications. The first demonstrator receiver consists of a merged LNA and mixers, LO buffers, and one baseband channel. It achieves 47-dB voltage gain, 28-dB NF, -21-dBm IIP3, and +18-dBm IIP2. The second prototype receiver includes an LNA, downconversion mixers, a 90-degree phase shift circuit, analog filters, a 1-bit analog-to-digital converter, and a received signal strength indicator (RSSI). The receiver consumes 4.1 mA from a 1.2-V power supply and achieves 43-dB voltage gain, a 25-dB noise figure, -22-dBm IIP3, and +11-dBm IIP2. The active silicon area was not increased, although several blocks were added. In addition, improvement in the overall performance was achieved. Both ICs were processed in same 0.13-µm CMOS process.

In those designs, approximately 90 % of the total receiver power is consumed in the RF blocks, i.e. in the LNA, mixer, and the LO buffers. Thus, several circuit design techniques were developed to minimize the power consumption. For the first low-power receiver prototype, a topology, where the LNA and mixer were merged, was chosen. A current boosting was utilized to separately optimize the performance of the LNA input devices and mixer switch transistors and the following baseband interface. A method to connect the boost current to the center tap of the load inductor was proposed. The benefit of the presented current boosting method was that the noise from the boost current source is common-mode at the mixer outputs. Therefore, the contribution to the overall front-end noise is insignificant. In addition, the effect of the parasitic capacitance caused by the boost current transistor is alleviated. Furthermore, a current boosting alternative, where the constant current source was replaced with other active blocks used in the receiver, was presented in Section 6.3. A design example, where the constant boost current source was replaced with LO buffers, was shown. In that case, the presented current reuse method decreased the front-end current consumption by 45 % without performance degradation. For the second low-power receiver prototype, the LNA and mixer were on separate dc paths. Due to the extremely stringent power budget, the separate transconductor input stage was not beneficial to the design of the mixer. Therefore, the output signal of the LNA is directly fed to the switching stage and the quadrature mixer is a low-impedance load for the LNA. The quadrature LO signal was generated with a single-stage polyphase filter. Although it has poor amplitude balance performance, it was chosen to minimize the LO signal loss and hence the power consumption.

Table 6.5 contains recently published CMOS RF front-ends and receivers for 2.4-GHz wireless sensor systems found in the literature. All the designs use either direct-conversion or low-IF architecture. The power consumption comparison is not trivial because some of the design examples collected in Table 6.5 include RF front-end (i.e. LNA and mixer) only, but some of them include baseband filters and ADCs as well, such as [P2]. In addition, the other designs apply higher supply voltage due to larger channel length. In general, the power consumption of the presented circuits is equal or lower than that of the circuits developed for ZigBee.

The measured NF in RF front-ends presented in [P1] and [P2] is high and deviates from the simulated one. The discrepancy is briefly discussed and improvements are considered. Firstly, since the target was to minimize the die area, a custom-designed source inductor having small area was used in the CG-LNA input matching. It is possible that the Q-value of the source

inductor is lower than predicted. According to (3.88), a source inductor having a large value and high Q-value should be used to minimize the noise figure, although the penalty is the increased layout area. Secondly, since the current consumption budgeted for LO buffers was small, the amplitude of the LO signal driving the mixer was moderate and the signal waveform was sinusoidal rather than square wave. Thus, the overlapping on-state of the switch transistors may be longer than predicted, resulting in worsened mixer noise performance. Since the voltage gain before the mixers is low, the switch stage noise is not adequately suppressed and improper switching has a significant influence on the overall RF front-end noise. In addition, the LO buffers used in [P1] and [P2] did not include the capacitor C_2 shown in Fig. 6.32, which filtered the noise coming from the bias circuit. Since there is some imbalance in the mixers, the noise coming from the LO port is not perfectly rejected but it nevertheless enhances the mixer output noise as well. Thus, even more attention should be paid to LO buffer design and interface with the downconversion mixer.

Metric	Unit	[P1]	[P2]	[24]	[25]	[26]	[27]	[28]	[29]	[30]
V _{DD}	V	1.2	1.2	1.8	1.8	1.8	1.8	1.8	1.8	1.8
I _{DD}	mA	2.8	4.1	9 ¹⁾	n/a	n/a	14.7	6	n/a	11.5 ³⁾
P _{DD}	mW	3.4	4.9	31 ²⁾	6.3	9	n/a	n/a	4.8	31.7 ⁴⁾
Gain	dB	47	43	91	30	30	n/a	n/a	23	n/a
NF	dB	28	25	8	7.3	n/a	5.7	10	8.1	25
IIP3	dBm	-21	-22	-15	-8	-4	-16	-15	-15	-11
IIP2	dBm	+18	+11	n/a	+40	n/a	+6	n/a	n/a	n/a
ICP	dBm	n/a	-35	n/a	-18	n/a	n/a	n/a	n/a	-18
Techn	. [µm]	0.13	0.13	0.18	0.18	0.18	0.18	0.18	0.18	0.18
Торо	ology	DCR	DCR	Low-	DCR	Low-	Low-	Low-	DCR	Low-
				IF		IF	IF	IF		IF

Table 6.5. CMOS RF front-ends and receivers for 2.4-GHz wireless sensor systems.

1) current consumption of the RX chain only

2) power consumption of the whole receiver including the synthesizer

3) current consumption of analog parts only, digital part consume 7.4 mA from a 1.4-V supply

4) total power consumption (analog+digital)

A direct-conversion RF front-end designed in a 65-nm CMOS was presented in Section 6.4. Transistors having minimum channel length are utilized whenever possible, e.g. LNA input stage, LO divider, and LO buffer, for maximum device speed and minimum parasitics. The simulated performance of the presented LNA is comparable with other CMOS LNAs designed with longer channel lengths. In downconversion mixer design, however, transistors with non-minimum channel lengths were used to optimize the linearity, 1/f noise, and device matching. Due to limitations caused by low supply voltage, satisfactory mixer performance was achieved by using folded structure. To minimize the 1/f noise, PMOS transistors with small quiescent current were used as switches.

This RF front-end operates at 2.4-GHz center frequency. Performances of recently published RF front-ends realized in 90-nm to 150-nm CMOS are shown in Table 6.6. The comparison is nontrivial due to requirements of different applications and when comparing simulation results to measured ones. This design example has an excellent minimum noise performance and moderate linearity vs. dissipated power. To improve the 1/*f* noise corner frequency and linearity, methods presented in [17], for example, could be investigated.

	Tech.	f_C	V _{DD}	I _{DD}	P _{DD}	Gain	NF	IIP3	IIP2	ICP
Ref.	[nm]	[GHz]	[V]	[mA]	[mW]	[dB]	[dB]	[dBm]	[dBm]	[dBm]
[31]	150	2.4	1.8	65	n/a	96	4.4	-19	n/a	-4 *
[17]	130	1.9	1.2	n/a	105	50	3.9	-9	+30	n/a
[32]	130	2.4	1.2	20	n/a	23.4	5.8	-4.8	+54	n/a
[33]	90	2.1	1.5	12.7	n/a	9.0	9.4	+8.9	+55	-3.3
[34]	90	1.8	1.4	n/a	75	37	2.9	-11	+63	n/a
[35]	90	1.9	0.75	15	n/a	31.5	3.5	-10.5	+51	-18
[P3]	65	2.4	1.2	29.3	35.2	39	3.0	-17	n/a	-29

Table 6.6. The comparison of recently published 1.8 – 2.4 GHz CMOS RF front-ends.

*) measured with 0 dB gain setting

An RF front-end targeted for a dual-band UWB receiver operating at BG1 and BG3 was presented in Section 6.6. The prototype chip was realized in a 0.13-µm CMOS. Because the target is a mobile environment, special attention was paid to interference issues. Therefore, separate inputs and on-chip filtering were designed for sufficient suppression against narrowband systems. The efficient component sharing and small die area were achieved by combining the input signals in the second LNA and by designing the LO buffers to uniformly cover both BGs. Furthermore, special emphasis was given to the optimization of interfaces between different blocks.

The performance of the presented front-end is compared to the other CMOS UWB receivers and front-ends found in the literature mentioned in Table 6.7. Most of the UWB receivers are targeted for BG1 and BG3. The overall performance comparison is nontrivial, because some of the receivers include more circuit blocks than the others. For example, both the current consumption and layout area are significantly increased if the on-chip synthesizer is included [38], [39].

The gain, which is given in Table 6.7, is the maximum gain achieved at the operational band. The gain itself is not a relevant figure-of-merit to be compared, since it is affected by the blocks included. More interesting merit would be the gain variation within a single bandgroup or subband. Since the gain responses are not always shown and the detailed information is seldom given for separate sub-bands, this gain deviation is omitted from the table. In the publications where the gain deviation is mentioned, the in-band gain ripple within a single BG is typically less than 2 dB. The NF results are typically separately mentioned for each sub-band. The minimum and maximum NFs within the whole operational band are collected in Table 6.7. For BG1 devices, the typical NF is 4 - 6 dB, while in the case of BG3 UWB ICs, the NF is approximately 2 dB higher. The NF of [P6] is comparable with the other published BG1 and BG3 designs.

All the linearity performances (IIP3, IIP2, and ICP) are given as in-band results if not otherwise mentioned. If the results are separately published for different sub-bands or BGs, the worst-case linearity results are included in Table 6.7. It should be noted that some of the linearity results are achieved with minimum gain setup. Because no specification for input test signals exists for UWB, the variety of test signal frequencies is found in the published designs, but inconveniently, they are not always mentioned. Although the linearity performance comparison is not straightforward, the in-band linearity performance of [P6] is in line with the other

designs. A more relevant measure of the UWB receiver linearity is the robustness against the interference caused by other active radios, i.e. GSM900, GSM1900, and systems operating at 2.4-GHz and 5-GHz ISM and in U-NII bands. Such intermodulation measurements are performed in, for example, [P6], [41], [43], and [45]. For example, [P6] has a measured IIP3 of +1 dBm when the interference is caused by a GSM900 transmitter and 2.4-GHz system.

Metric	Unit	[P6]	[36]	[37]	[38]	[39]	[40]	[41]
f_C	GHz	3.1 - 7.9	3 – 5	3.1 - 9.5	3.1 - 8	3 – 5	3 – 5	3 – 8
V _{DD}	V	1.2	1.5	1.1	1.5	1.8	1.5	1.2
I _{DD}	mA	48	n/a	n/a	81.5	n/a	34 ¹⁾	43 ¹⁾
P _{DD}	mW	57	105	224	n/a	412	237 ²⁾	114 ²⁾
Gain	dB	26	73	64	32.3	63.8	37	24
NF	dB	4.9 7.7	6.5 8.4	6.3 7.8	6.5 7.9	4.04.7	3.64.1	5 5.5
IIP3	dBm	-20	n/a	-17 ³⁾	-12.6 ³⁾	-0.8 ³⁾	-22	+5 4)
IIP2	dBm	+18	n/a	n/a	$+21.6^{3)}$	+22 3)	n/a	+24 4)
ICP	dBm	-29	-27.5	n/a	-24.5	-9 ³⁾	n/a	n/a
Area	mm ²	0.76	1	3.5	7.25	15.96	6.6	0.4 5)
Techn	ology	0.13-µm	0.13-µm	90-nm	0.18-µm	0.18-µm	0.13-µm	65-nm

Table 6.7. Recently published CMOS UWB receivers and RF front-ends.

1) receiver chain without LO generation

2) whole receiver

3) with minimum gain setup

4) intermodulation measured with out-of-band interferers

5) active area of the whole transceiver

Table 6.8 shows the UWB receivers implemented with SiGe process. Compared to CMOS designs, the NFs of SiGe UWB receivers are typically 1-2 dB better. The supply voltages of SiGe circuits are higher due to technology limitations, but the overall power consumption is typically comparable with CMOS designs. All published UWB receivers and front-ends utilize direct-conversion architecture except [38], which uses low-IF topology.

Metric	Unit	[42]	[43]	[44]	[45]	[46]	[47]
f_C	GHz	3.1 - 8.2	3.1 - 4.8	3.1 - 10.6	3 – 5	3.1 - 10.6	3 - 10
V _{DD}	V	2.7	2.5	2.7	2.7	1.8	2.5
I _{DD}	mA	88	78	31	47 ¹⁾	30	114
P _{DD}	mW	n/a	n/a	n/a	n/a	53.7	n/a
Gain	dB	52	59	21.8	59	39	78
NF	dB	3.3 4.1	4.5	4.1 6.2	4.5	3.3 5	5 10
IIP3	dBm	-4.5 ²⁾	$-6^{3)}$	-12.7	$-6^{3)}$	n/a	-9
IIP2	dBm	$+37.2^{2}$	$+25^{3)}$	+29	+25 3)	n/a	n/a
ICP	dBm	-10.2 ²⁾	n/a	-23.5	n/a	-46	-27
Area	mm ²	7.0	4	2.7	4	2.3	5.6

Table 6.8. Recently published SiGe UWB receivers and RF front-ends.

1) receiver chain without LO generation

2) with minimum gain setup

Technology

3) intermodulation measured with out-of-band interferers

0.18-µm

0.25-µm

0.25-µm

0.25-µm

0.18-µm

0.25-µm
References

- M. Honkanen, A. Lappeteläinen, and K. Kivekäs, "Low end extension for Bluetooth," *IEEE Radio and Wireless Conf. (RAWCON'04)*, Atlanta, GA, 19-22 Sept. 2004, pp. 199-202.
- [2] T. Melly, A.-S. Porret, C. C. Enz, and E. A. Vittoz, "An analysis of flicker noise rejection in low-power and low-voltage CMOS mixers," *IEEE Journal of Solid-State Circuits*, vol. 36, no. 1, Jan. 2001, pp. 102-109.
- [3] H. Darabi and A. A. Abidi, "Noise in RF-CMOS mixers: A simple physical model," *IEEE J. of Solid-State Circuits*, vol. 35, no. 1, Jan. 2000, pp. 15-25.
- [4] S. Mahdavi and A. A. Abidi, "Fully integrated 2.2-mW CMOS front end for a 900-MHz wireless receiver," *IEEE J. of Solid-State Circuits*, vol. 37, no. 5, May 2002, pp. 662-669.
- [5] H. Sjöland, A. Karimi-Sanjaani, and A. A. Abidi, "A merged CMOS LNA and mixer for a WCDMA receiver," *IEEE J. of Solid-State Circuits*, vol. 38, no. 6, June 2003, pp. 1045-1050.
- [6] W. Sansen and R. Meyer, "An integrated wide-band variable-gain amplifier with maximum dynamic range," *IEEE J. of Solid-State Circuits*, vol. SC-9, no. 8, Aug. 1974, pp. 159-166.
- [7] J. Ryynänen, K. Kivekäs, J. Jussila, A. Pärssinen, and K. A. I. Halonen, "A dual-band RF front-end for WCDMA and GSM applications," *IEEE J. of Solid-State Circuits*, vol. 36, no. 8, Aug. 2001, pp. 1198-1204.
- [8] D. Python, A.-S. Porret, and C. Enz, "A 1V 5th-order Bessel filter dedicated to digital standard process," *IEEE Custom Integrated Circuits Conference (CICC'99)*, San Diego, CA, 16-19 May 1999, pp. 505-508.
- [9] A. Baschirotto, U. Baschirotto, R. Castello, "High-frequency CMOS low-power single-branch continuous-time filters," *IEEE Int. Symp. Circuits and Systems (ISCAS'00)*, Geneva, Switzerland, 28-31 May 2000, pp. II-577-580.
- [10] T. Hollman, S. Lindfors, T. Salo, M. Länsirinne, and K. Halonen, "A 2.7V CMOS dual-mode baseband filter for GSM and WCDMA," *IEEE Int. Symp. Circuits and Systems (ISCAS'01)*, Sydney, Australia, 6-9 May 2001, pp. I-316-319.
- [11] A. Behr, M. Schneider, S. Fihlo, and C. Montoro, "Harmonic distortion caused by capacitors implemented with MOSFET gates," *IEEE J. of Solid-State Circuits*, vol. 27, no. 10, Oct. 1992, pp. 1470-1475.
- [12] R. P. Jindal, "Gigahertz-band high-gain low-noise AGC amplifiers in fine-line NMOS," *IEEE J. of Solid-State Circuits*, vol. SC-22, no. 4, Aug. 1987, pp. 512-521.
- [13] A. Shoval, D. A. Johns, and W. M. Snelgrove, "Median-based offset cancellation circuit technique," *IEEE Int. Symposium on Circuits and Systems (ISCAS'92)*, San Diego, CA, 10-13 May 1992, pp. IV-2033-2036.
- [14] N. Tan and S. Eriksson, "Low-voltage fully differential class-AB Si circuits with common-mode feedforward," *Electronics Letters*, vol. 30, no. 25, 8th Dec. 1994, pp. 2090-2091.

- [15] A. Zolfaghari and B. Razavi, "A low-power 2.4-GHz transmitter/receiver CMOS IC," *IEEE J. of Solid-State Circuits*, vol. 38, no. 2, Feb. 2003, pp. 176-183.
- [16] S. G. Lee and J.-K. Choi, "Current-reuse bleeding mixer," *Electronics Letters*, vol. 36, no. 8, 13th April 2000, pp. 696-697.
- [17] P. Sivonen, J. Tervaluoto, N. Mikkola, and A. Pärssinen, "A 1.2-V RF front-end with on-chip VCO for PCS 1900 direct conversion receiver in 0.13-µm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 2, Feb. 2006, pp. 384-394.
- [18] T. W. Kim, B. Kim, and K. Lee, "Highly linear receiver front-end adopting MOSFET transconductance linearization by multiple gated transistors," *IEEE J. of Solid-State Circuits*, vol. 39, no. 1, Jan. 2004, pp. 223-229.
- [19] B. Razavi, K. F. Lee, and R. H. Yan, "Design of high-speed, low-power frequency dividers and phase-locked loops in deep submicron CMOS," *IEEE J. of Solid-State Circuits*, vol. 30, no. 2, Feb. 1995, pp. 101-109.
- [20] V. Saari, M. Kaltiokallio, S. Lindfors, J. Ryynänen, and K. Halonen, "A 1.2V 240MHz CMOS continuous-time low-pass filter for a UWB receiver," *IEEE Int. Solid-State Circuits Conference (ISSCC'07)*, San Francisco, CA, 11-15 Feb. 2007, pp. 122-123.
- [21] H. Samavati, H. R. Rategh, and T. H. Lee, "A 5-GHz CMOS wireless LNA receiver front-end," *IEEE J. of Solid-State Circuits*, vol. 35, no. 5, May 2000, pp. 767-772.
- [22] J. Lee and J. D. Cressler, "Analysis and design of an ultra-wideband low-noise amplifier using resistive feedback in SiGe technology," *IEEE Trans. on Microwave Theory and Techniques*, vol. 54, no. 3, March 2006, pp. 1262-1268.
- [23] M.-C. Chiang, S.-S Lu, C.-C. Meng, S.-A. Yu, S.-C. Yang, and Y.-J. Chan, "Analysis, design, and optimization of InGaP-GaAs HBT matched-impedance wide-band amplifiers with multiple feedback loops," *IEEE J. of Solid-State Circuits*, vol. 37, no. 6, June 2002, pp. 694-701.
- [24] I. Kwon, Y. Eo, S.-S. Song, K. Choi, H. Lee, and K. Lee, "A fully integrated 2.4-GHz CMOS RF transceiver for IEEE 802.15.4," *IEEE Radio Frequency Integrated Circuit Symposium (RFIC'06)*, San Francisco, CA, 11-13 June 2006, pp. 275-278.
- [25] T.-K. Nguyen, V. Krizhanovskii, J. Lee, S.-K. Han, S.-G. Lee, N.-S. Kim, and C.-S. Pyo, "A low-power RF direct-conversion receiver/transmitter for 2.4-GHz-band IEEE 802.15.4 standard in 0.18-μm CMOS technology," *IEEE Trans. on Microwave Theory* and Techniques, vol. 54, no. 12, Dec. 2006, pp. 4062-4071.
- [26] P. Choi, H. C. Park, S. Kim, S. Park, I. Nam, T. W. Kim, S. Park, S. Shin, M. S. Kim, K. Kang, Y. Ku, H. Choi, S. M. Park, and K. Lee, "An experimental coin-sized radio for extremely low-power WPAN (IEEE 802.15.4) application at 2.4 GHz," *IEEE J. of Solid-State Circuits*, vol. 38, no. 12, Dec. 2003, pp. 2258-2268.
- [27] W. Kluge, F. Poegel, H. Roller, M. Lange, T. Ferchland, L. Dathe, and D. Eggert, "A fully integrated 2.4-GHz 802.15.4-compliant transceiver for ZigBee applications," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2767-2775.
- [28] I. Nam, K. Choi, J. Lee, H.-K. Cha, B.-I. Seo, K. Kwon, and K. Lee, "A 2.4-GHz low-power low-IF receiver and direct-conversion transmitter in 0.18-μm CMOS for IEEE 802.15.4 WPAN applications," *IEEE Trans. on Microwave Theory and Techniques*, vol. 55, no. 4, April 2007, pp. 682-689.

- [29] M. Annamalai Arasu, H. K. F. Ong, Y. B. Choi, and W. G. Yeoh, "A 2.4-GHz CMOS RF front-end for wireless sensor network applications," *IEEE Radio Frequency Integrated Circuit Symposium (RFIC'06)*, San Francisco, CA, 11-13 June 2006, pp. 455-458.
- [30] H. J. Bergveld, K. M. M. van Kaam, D. M. W. Leenaerts, K. J. P. Philips, A. W. P. Vaassen, and G. Wetkzer, "A low-power highly digitized receiver for 2.4-GHz-band GFSK applications," *IEEE Trans. on Microwave Theory and Techniques*, vol. 53, no. 2, Feb. 2005, pp. 453-461.
- [31] S. C. Yen, *et al.*, "A low-power full-band 802.11abg CMOS transceiver with on-chip PA," *IEEE Radio Frequency Integrated Circuit Symposium (RFIC'06)*, San Francisco, CA, 11-13 June 2006, pp. 103-106.
- [32] A. Lischidini, M. Brandolini, D. Sanzogni, and R. Castello, "A 0.13 μm CMOS frontend for DCS1800/UMTS/802.11b-g with multiband positive feedback low-noise amplifier," *IEEE J. of Solid-State Circuits*, vol. 41, no. 4, April 2006, pp. 981-989.
- [33] B. Bakkaloglu, P. Fontaine, A. N. Mohieldin, S. Peng, S. J. Fang, and F. Dülger, "A 1.5-V multi-mode quad-band RF receiver for GSM/EDGE/CDMA2K in 90-nm digital CMOS process," *IEEE J. of Solid-State Circuits*, vol. 41, no. 5, May 2006, pp. 1149-1159.
- [34] N. K. Yanduru, D. Griffith, S. Bhagavatheeswaran, C.-C. Chen, F. Dulger, S.-J. Fang, Y.-C. Ho, and K. M. Low, "A WCDMA, GSM/GPRS/EDGE receiver front end without interstage SAW filter," *IEEE Radio Frequency Integrated Circuit Symposium* (*RFIC'06*), San Francisco, CA, 11-13 June 2006, pp. 19-22.
- [35] M. Brandolini, M. Sosio, and F. Svelto, "A 750-mV 15kHz 1/f noise corner 51dBm IIP2 direct-conversion front-end for GSM in 90nm CMOS," *IEEE Int. Solid-State Circuits Conference (ISSCC'06)*, San Francisco, CA, 5-9 Feb. 2006, pp. 470-471.
- [36] B. Razavi, T. Aytur, C. Lam, F.-R. Yang, K.-Y. Li, R.-H. Yan, H.-C. Kang, C.-C. Hsu, and C.-C. Lee, "A UWB CMOS transceiver," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2555-2562.
- [37] A. Tanaka, H. Okada, H. Kodama, and H. Ishikawa, "A 1.1V 3.1-to-9.5 GHz MB-OFDM UWB transceiver in 90nm CMOS," *IEEE Int. Solid-State Circuits Conference* (ISSCC'06), San Francisco, CA, 5-9 Feb. 2006, pp. 120-121.
- [38] S. Lou, H. Zheng, and H. C. Luong, "A 1.5-V CMOS receiver front-end for 9-band MB-OFDM UWB system," *IEEE Custom Integrated Circuits Conference (CICC'06)*, San Jose, CA, 10-13 Sept. 2006, pp. 801-804.
- [39] S. Lo, I. Sever, S.-P. Ma, P. Jang, A. Zou, C. Arnott, K. Ghatak, A. Schwartz, L. Huynh, V. T. Phan, and T. Nguyen, "A dual-antenna phased-array UWB transceiver in 0.18-µm CMOS," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2776-2786.
- [40] C. Sandner, S. Derksen, D. Draxelmayr, S. Ek, V. Filimon, G. Leach, S. Marsili, D. Matveev, K. L. R. Mertens, F. Michl, H. Paule, M. Punzenberger, C. Reindl, R. Salerno, M. Tiebout, A. Wiesbauer, I. Winter, and Z. Zhang, "A WiMedia/MBOA-compliant CMOS RF transceiver for UWB," *IEEE J. of Solid-State Circuits*, vol. 41, no. 12, Dec. 2006, pp. 2787-2794.
- [41] J. Bergervoet, K. S. Harish, S. Lee, D. Leenaerts, R. van de Beek, G. van der Weide, and R. Roovers, "A WiMedia-compliant UWB transceiver in 65nm CMOS," *IEEE Int.*

Solid-State Circuits Conference (ISSCC'07), San Francisco, CA, 11-15 Feb. 2007, pp. 112-113.

- [42] A. Ismail and A. A. Abidi, "A 3.1–8.2-GHz zero-IF receiver and direct frequency synthesizer in 0.18µm SiGe BiCMOS for mode-2 MB-OFDM UWB communication," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2573-2582.
- [43] R. Roovers, D. M. W. Leenaerts, J. Bergervoet, K. S. Harish, R. C. H. van de Beek, G. van der Weide, H. Waite, Y. Zhang, S. Aggarwal, and C. Razzell, "An interference-robust receiver for ultra-wideband radio in SiGe BiCMOS technology," *IEEE J. of Solid-State Circuits*, vol. 40, no. 12, Dec. 2005, pp. 2563-2572.
- [44] B. Shi and Y. W. Shia, "A 3.1–10.6 GHz RF front-end for multiband UWB wireless receivers," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'05)*, Long Beach, CA, 12-14 June 2005, pp. 343-346.
- [45] J. Bergervoet, H. Kundur, D. M. W. Leenaerts, R. C. H. van de Beek, R. Roovers, G. van der Weide, H. Weite, and S. Aggarwwal, "A fully integrated 3-band OFDM UWB transceiver in 0.25 μm SiGe BiCMOS," *IEEE Radio Frequency Integrated Circuits Symposium (RFIC'06)*, San Francisco, CA, 11-13 June 2006, pp. 301-304.
- [46] F. S. Lee and A. P. Chandrakasan, "A BiCMOS ultra-wideband 3.1–10.6-GHz frontend," *IEEE J. of Solid-State Circuits*, vol. 41, no. 8, Aug. 2006, pp. 1784-1791.
- [47] A. Valdes-Garcia, C. Mishra, F. Bahmani, J. Silva-Martinez, and E. Sánchez-Sinencio, "An 11-band 3–10 GHz receiver in SiGe BiCMOS for multiband OFDM UWB communication," *IEEE J. of Solid-State Circuits*, vol. 42, no. 4, April 2007, pp. 935-948.

7 Conclusions

In this thesis, RF circuit techniques for short-range direct-conversion receivers were presented. The two main target applications were low-power sensor radio and WiMedia UWB BG1 and BG3. The targeted operational frequencies were 2.4 GHz and 3.1 - 4.7 GHz and 6.3 - 7.9 GHz, respectively. Two prototype circuits for sensor radios and one IC realization for UWB systems were presented. The demonstrated direct-conversion front-ends included LNAs, down-conversion mixers, and LO generation circuits. The front-ends presented in papers [P1] and [P2] have been a part of a whole receiver including baseband amplifiers and filters and 1-bit analog-to-digital converters.

Several LNA topologies were first studied and analyzed in this thesis. Typically, the input matching circuit provides additional voltage gain, which is measured with Q-value. It affects the noise performance, linearity, and current consumption of the LNA. The input matching circuit also has an impact on the achievable input matching and output current bandwidths. The input matching circuit design parameters were derived for different LNA topologies. The presented calculations were adapted to several wideband LNA design examples and good agreement with the theory was observed.

The IDCS LNA input is typically sensitive to the parasitic capacitance associated with the input node. In contrast to the common-gate LNA, the parasitic capacitance at the IDCS stage cannot be absorbed into the input matching network. Therefore, the parasitic capacitance associated with the typical bonding pad can limit the achievable input matching performance. To mitigate that problem, a simple modification to the pad usage was proposed and analyzed. The presented pad structure was used at UWB BG3 LNA input, where benefit was gained due to high operational frequency.

The different amplifier load structures were discussed. A typical RLC resonator load is applicable for narrowband applications. In addition, shunt-peak load and its alternatives were briefly presented and compared. Such load structures are needed, when the target is to cover wide bandwidths simultaneously with high center frequency, in the case of UWB, for example. The bandwidth shrinkage of two cascaded shunt-peak loads was briefly described. An alternative wideband load, which was targeted for two-stage amplifiers but contained only a single on-chip inductor, was proposed and analyzed.

The presented work also includes a theoretical analysis of passive polyphase filters. A general method with which to calculate the transfer functions of cascaded PPF filter stages is presented. According to the analysis, design guidelines for two- and three-stage PPFs, which are the most typical PPF structures encountered in practical RF IC design, are provided. Whenever it was reasonable, formulas for general *n*-stage PPFs are given. Based on the input signal feeding technique, two variants of PPFs can be separated. The analysis of both variants was carried out and a performance comparison between the PPF types was made.

The main targets of the presented application examples were quite separate from each other due to power consumption, operational frequency, and bandwidth points of view. From the sensor applications, the primary goal was to design a complete receiver with current consumption of only a few mWs. The challenge of the low power consumption is to achieve sufficient performance for RF blocks, which dominate the total receiver current consumption. On the other hand, the challenge of the UWB design example was to design LNAs having sufficient

input matching and gain response over the 1.6-GHz band and to optimize the interface between different blocks, still bearing mind the power consumption. For both applications, one relevant criterion was to keep the number of on-chip inductors as low as possible. Although the focus of the design examples was on these systems, the circuit presented and analyzed in this thesis can be utilized in other kinds of applications as well.

Three experimental circuits presented in this thesis were fabricated in a 0.13-µm CMOS. Part of the theory here was validated with design examples where actual IC component models were used. In addition, circuit design examples in 0.13-µm and 65-nm CMOS processes were included. Technology scaling allows an extensive use of digital signal processing and there is ever increasing pressure to replace analog parts with digital ones. Nevertheless, radio communication is still based on transmitting and receiving signals in the analog domain and some analog blocks will remain in forthcoming transceivers as well. It is feasible to design RF front-end in the analog domain, even with deep-submicron process, as was presented in this thesis.

Appendix A: General common-source LNA

A general common-source (CS) LNA is presented in Fig. A.1. The CS-LNA includes gate and source impedances Z_g and Z_s , gate-to-drain and gate-to-source impedances Z_{gd} and Z_{gs} , and a load impedance Z_L . The general CS LNA circuit is used to analyze the input matching and gain of the several LNA topologies shown later in this thesis. It should be noted that, when using the following equations, the intrinsic gate-source and gate-drain capacitors C_{gs} and C_{gd} of M_1 are included in Z_{gs} and Z_{gd} , respectively.



Fig. A.1. General common-source stage.

The input impedance of the common-source LNA of Fig. A.1 is

$$Z_{in} = Z_g + \frac{\left(Z_{gd} + Z_L\right) \left[Z_{gs} + Z_S \left(1 + g_m Z_{gs}\right)\right]}{Z_{gd} + Z_{gs} + \left(1 + g_m Z_{gs}\right) \left(Z_L + Z_S\right)},$$
(A.1)

where the components can be identified from Fig. A.1. An equation similar to (A.1) is also given in [1], where bulk conductance and source-bulk and drain-bulk impedances are also taken into account. If Z_{gd} is infinite, i.e. there is no feedback or a unilateral transistor model is assumed, (A.1) simplifies to

$$Z_{in} = Z_g + Z_{gs} + Z_S \left(1 + g_m Z_{gs} \right).$$
(A.2)

The insertion gain of a CS LNA shown in Fig. A.1 is

$$A_{i} = \frac{2Z_{L} \left(1 - \frac{g_{m}Z_{gd}Z_{gs}}{Z_{S} + Z_{gs} + g_{m}Z_{S}Z_{gs}}\right)}{Z_{gd} + Z_{L} + \left(R_{S} + Z_{g}\right) \left(1 + \frac{Z_{L} + Z_{gd} + g_{m}Z_{L}Z_{gs}}{Z_{S} + Z_{gs} + g_{m}Z_{s}Z_{gs}}\right)}.$$
(A.3)

If Z_{gd} is infinite, (A.3) simplifies to

$$A_{i,Z_{gd}\to\infty} = -\frac{2g_m Z_{gs} Z_L}{R_s + Z_g + Z_{gs} + Z_s + g_m Z_{gs} Z_s}.$$
 (A.4)

The voltage gain, neglecting bulk conductance and source-bulk and drain-bulk impedances, is given by [1]

$$A_{v} = \frac{Z_{in} - Z_{g}}{Z_{in}} \cdot \left(Z_{L} \| Z_{gd}\right) \cdot \left[\frac{1}{Z_{gd}} - \frac{g_{m}Z_{gs}}{Z_{gs} + Z_{S}\left(1 + g_{m}Z_{gs}\right)}\right].$$
 (A.5)

Reference

 H. Hashemi and A. Hajimiri, "Concurrent multiband low-noise amplifiers – Theory, design, and applications," *IEEE Trans. on Microwave Theory and Techniques*, vol. 50, no. 1, Jan. 2002, pp. 288-301.

Appendix B: Analysis and optimization of 4-stage PPF

The optimum pole splitting is calculated for a four-stage PPF. First, the pole-splitting factor k_4 is defined as the ratio of the pole frequencies ω_1 and ω_4 . Due to symmetry reasons, the pole ratio between ω_3 and ω_4 is the same as the pole ratio between ω_1 and ω_2 . Thus, k_4 is determined by k_2 and k_3

$$k_4 = \frac{\omega_1}{\omega_4} = \frac{\omega_1}{\omega_3} \frac{\omega_3}{\omega_4} = k_3 k_2 . \tag{B.1}$$

The minimum IRRs are achieved at three frequencies given by

$$\omega_{IRR,\min 2} = \frac{\omega_1}{\sqrt{k_2 k_3}}, \qquad (B.2)$$

$$\omega_{IRR,\min 1,3} = \frac{\omega_1}{\sqrt{2}k_2k_3} \sqrt{F_3(k_2,k_3) \pm \sqrt{F_3(k_2,k_3)^2 - 4k_2^2k_3^2}}, \qquad (B.3)$$

where

$$F_{3}(k_{2},k_{3}) = k_{2}(1-k_{3})^{2} + k_{3}(1-k_{2})^{2} + 2k_{2}k_{3}.$$
 (B.4)

The IRR minimums at $\omega_{IRR,min1}$ and $\omega_{IRR,min3}$ are always equal. To achieve equal IRR with $\omega_{IRR,min2}$, too, k_3 is calculated as

$$k_{3} = \frac{1}{3k_{2}} \left[k_{2} \left(k_{2}^{2} + k_{2} - 1 \right) + F_{4} \left(k_{2} \right) \cos \left(\frac{1}{3} \arccos \left(\frac{F_{5} \left(k_{2} \right)}{F_{4} \left(k_{2} \right)^{3}} \right) \right) \right],$$
(B.5)

where

$$F_4(k_2) = 2\sqrt{k_2(k_2^5 + 2k_2^4 - k_2^3 - 5k_2^2 + 4k_2 + 3)},$$
(B.6)

$$F_{5}(k_{2}) = 4k_{2}^{2} \left[-9 + k_{2} \left(-9k_{2}^{3} + 2\left(-1 + k_{2} + k_{2}^{2} \right)^{3} \right) \right].$$
(B.7)

Equation (B.5) is quite tedious for hand calculations. Therefore, an approximate formula for k_3 calculation given as

$$k_3 \approx 1.2k_2^2 + 0.17k_2 - 0.37 \tag{B.8}$$

will predict (B.5) with an maximum error of 1.1 % when $k_2 < 3$. At $\omega_{IRR,min,2}$, the amplitude balance becomes

$$A_{bal,4-stg} = \frac{k_2 \left(1+k_3\right)^2 + k_3 \left(1+k_2\right)^2}{2\sqrt{k_2 k_3} \left(k_2+1\right) \left(k_3+1\right)},$$
(B.9)

from where the IRR can be calculated with (5.21). The corner frequencies $\omega_{c,min}$ and $\omega_{c,max}$, where the minimum IRR is achieved, can be calculated with

$$\omega_{c,\min\&\max} = \frac{\omega_1}{\left(4k_2k_3\right)^{3/2}} \left[F_6^+(k_2,k_3) + F_6^-(k_2,k_3) \pm 2\sqrt{F_6^+(k_2,k_3)F_6^-(k_2,k_3)} \right], \quad (B.10)$$

where the '+' or '-' -sign of F_6 is chosen according to

$$F_6^{\pm}(k_2,k_3) = k_2(k_3\pm 1)^2 + k_3(k_2\pm 1)^2.$$
(B.11)

The relative bandwidth, where the IRR_{min} is achieved, is

$$BW_{rel,4-stg} = \frac{F_6^+(k_2,k_3) + F_6^-(k_2,k_3) + 2\sqrt{F_6^+(k_2,k_3)F_6^-(k_2,k_3)}}{F_6^+(k_2,k_3) + F_6^-(k_2,k_3) - 2\sqrt{F_6^+(k_2,k_3)F_6^-(k_2,k_3)}}.$$
 (B.12)

Equation (B.12) can be approximated with better than 1.0-% accuracy with

$$BW_{rel,4-stg} \approx 2.73k_2^3 - 3.63k_2^2 + 2.9k_2 - 1, \qquad (B.13)$$

when $k_2 < 3$. The minimum IRR and relative bandwidth with optimum pole splitting are shown in Fig. 5.17.

If the IRR is calculated for the 4-stage PPF by using (B.5), quite complicated equations are achieved. To get a simple equation for 4th-order PPF IRR, the generic formula $k_{n+1}/k_n = k_2$ is used to calculate pole-splitting factors k_3 and k_4 . As a result, the IRR at the frequency given by (B.2) becomes

$$IRR_{\min,4-stg} = \left(\frac{\sqrt{k_2}+1}{\sqrt{k_2}-1}\right)^4 \left(\frac{k_2 - \sqrt{k_2}+1}{k_2 + \sqrt{k_2}+1}\right)^2.$$
 (B.14)

Compared to the other two IRR minimums at frequencies given by (B.3), the IRR of (B.14) is approximately 5 dB higher when k_2 is close to unity. This is shown in Fig. B.1, where the IRR of a 4th-order PPF is shown. A BW-optimized PPF is shown with a solid line. In that case, k_3 is calculated with (B.5). The dashed line presents a PPF where k_3 is calculated as k_2^2 . In both cases $k_2 = 1.6$ and k_4 is calculated as in (B.1).



Fig. B.1. IRR of a 4th-order PPF. The solid line presents a BW-optimized PPF. In that case, k_3 is evaluated with (B.5). The dashed line presents a PPF, where k_3 is evaluated as k_2^2 . In both cases, $k_2 = 1.6$.

ISBN 978-951-22-9264-6 ISBN 978-951-22-9265-3 (PDF) ISSN 1795-2239 ISSN 1795-4584 (PDF)